Duquesne University

Duquesne Scholarship Collection

Electronic Theses and Dissertations

Spring 5-5-2023

The Ethics of Governance of Data Analytics in Healthcare

Andrew Harrington

Follow this and additional works at: https://dsc.duq.edu/etd

Part of the Applied Ethics Commons, Bioethics and Medical Ethics Commons, Biomedical Informatics Commons, Ethics and Political Philosophy Commons, Genomics Commons, Health Communication Commons, Health Information Technology Commons, Legal Ethics and Professional Responsibility Commons, and the Medical Humanities Commons

Recommended Citation

Harrington, A. (2023). The Ethics of Governance of Data Analytics in Healthcare (Doctoral dissertation, Duquesne University). Retrieved from https://dsc.duq.edu/etd/2256

This One-year Embargo is brought to you for free and open access by Duquesne Scholarship Collection. It has been accepted for inclusion in Electronic Theses and Dissertations by an authorized administrator of Duquesne Scholarship Collection. For more information, please contact beharyr@duq.edu.

THE ETHICS OF GOVERNANCE OF DATA ANALYTICS IN HEALTHCARE

A Dissertation

Submitted to the McAnulty College and Graduate School of Liberal Arts

Duquesne University

In partial fulfillment of the requirements for

the degree of Doctor of Philosophy

By

Andrew Harrington

May 2023

Copyright by

Andrew Harrington

THE ETHICS OF GOVERNANCE OF DATA ANALYTICS IN HEALTHCARE

By

Andrew Harrington

Approved December 15, 2022

Joris Gielen, Ph.D. Director, Center for Global Health Ethics Associate Professor, Center for Global Health Ethics (Dissertation Director) Gerard Magill, Ph.D. The Vernon F. Gallagher Chair Professor, Center for Global Health Ethics (Committee Member)

Kristine Blair, Ph.D. Dean, McAnulty College and Graduate School of Liberal Arts Peter Ikechukwu Osuji, C.S.Sp., Ph.D. Assistant Professor, Center for Global Health Ethics (Committee Member)

ABSTRACT

THE ETHICS OF GOVERNANCE OF DATA ANALYTICS IN HEALTHCARE

By

Andrew Harrington

May 2023

Dissertation supervised by Joris Gielen, Ph.D.

Recent literature and studies on data governance usually focus on data ethics and governance from a specific country or industry, which has been mostly business, without considering the larger global impact that is affected. Data is a worldwide asset; thus, its implications and considerations should be viewed as such. Hence, this dissertation attempts to incorporate this view by weaving together topics in healthcare, such as the blending of finance and delivery, data governance at the micro and macro levels, data analytics in healthcare, ethics of AI and information management, and technology's impact on end-of-life practices. Through highlighting these areas of healthcare, the dissertation explores characteristics of the healthcare landscape nationally and globally. As the use of data analytics in healthcare increases, the outcomes and their derivatives from these technologies increase proportionally.

Ethical governance of data analytics is focused on applying ethics to all aspects of data analytics, from beginning concepts to ending enforcement stages. This approach aims to take data governance throughout the "three P's" of an organization: policies, processes, and procedures. This type of governance is crucial since it ensures that data analytics is safe and secure to utilize but also takes ethical principles into the foundation of the whole cycle of data. By reprioritizing ethics at the forefront of governance, those involved with these technologies will be able to produce and maintain ethical data analytics. Thus, each chapter will attempt to look at the past, present, and future issues surrounding the governance of data analytics. Understanding the ethical concerns within these issues will provide a forum for those issues that have yet to be addressed by data experts. Further, by locating the commonalities between these ethical implications, it becomes more feasible to formulate and recommend solutions. This dissertation identifies many ethical concerns and provides recommended solutions through frameworks for data governance. While these global issues are often significant, usually a concrete solution will not always work practically in all countries. Therefore, a framework for ethical data governance for its stakeholders.

DEDICATION

To my family ~ thank you for all your unconditional love, support, prayers, and encouragement. Your actions and words have given me the strength to persevere.

To the memory of my father ~ Timothy J. Harrington ~ thank you for your heavenly support, I know you saw this process through to its completion. Love and miss you every day.

ACKNOWLEDGEMENT

First and foremost, I must thank God for granting me the wisdom and knowledge to finish this dissertation – many prayers have been said by numerous people that this day would come. Praise

God!

This dissertation would not have been possible without the guidance and education provided by my professors:

First, thank you Dr. Joris Gielen, my dissertation director, for challenging me to become a better writer and giving me the support and mentorship that I needed.

Thank you, Dr. Gerard Magill, for taking an interest in me and helping me unlock my potential and assisting me in molding my dissertation into what it has become today.

Thank you, Dr./Fr. Peter Osuji, for being on my committee and assisting in supervising my dissertation and providing valuable feedback. Your contributions have been invaluable.

I must thank the woman who helped bring me into the program and was the shining star of the

Center who made so many students feel welcome and at home over the years, Glory Smith.

Thank you for all the lovely memories and helping me learn to love the Center just as much as

you did.

I would like to acknowledge all my friends, colleagues, and classmates whom I have spent time researching, presenting, and discussing bioethics issues with – we are the future, let's show them

what we can do!

Finally, I want to acknowledge with abundant gratitude the love, support, and prayers of my family.

Abstract	iv
Dedication	vi
Acknowledgement	vii
Chapter 1: Introduction	1
Chapter 2: Data Analytics Blending Finance & Delivery in Healthcare	
2.A. The Ethics Behind Healthcare Reform: Saving Private America	11
2.A.i. Background of Healthcare Reform in America	
2.A.i.a. Brief History of Healthcare Reform in the United States	
2.A.i.b. Social Movements	17
2.A.i.c. Catholic Efforts	19
2A.ii. Distributive Justice	
2.A.ii.a. Background of Foundational Theories	
2.A.ii.b. The Basic Concerns	
2.A.ii.c. Allocation	
2A.iii. Implementation Considerations	
2.A.iii.a. Problems in the Present System	
2.A.iii.b. Considerations of Implementing a New System	
2.A.iii.c. Universal Access to Basic Care	
2.B. The Ethical Impact of Data Analytics in Healthcare	
2.B.i. Challenge & Opportunity in Data Analytics	
2.B.i.a. Challenge	
2.B.i.b. Opportunity: Real-Time Analytics	
2.B.ii. Personalized Care	
2.B.ii.a. Predictive Modeling (Algorithms)	

2.B.ii.b. Integrative Sequencing	51
2.B.iii. Organizational Implications for Health Systems	54
2.B.iii.a. Integrated Health Systems: Finance & Delivery	55
2.B.iii.b. Integrated Health Systems' Obligation for Population Data	58
Chapter 3: The Global Health Data Battle: Data Governance at the Micro & Macro Levels	63
3.A. Individual Autonomy & Rights vs. Population Health & the Common Good	64
3.A.i. Individual Autonomy vs. Population Health	64
3.A.i.a. Personal Autonomy & Human Rights	65
3.A.i.b. Population Health & Societal Obligations	68
3.A.ii. The Big Data Problem	71
3.A.ii.a. What is Big Data & Why Does It Matter?	72
3.A.ii.b. Ethical Challenges	75
3.A.iii. Individual Data Privacy vs. Public Health Surveillance	79
3.A.iii.a. Challenges in Data Use	79
3.A.iii.b. Ethical Justification of Public Health Surveillance	82
3.B. Ethical Challenges of Current Biometric Practices	89
3.B.i. What Are Biometrics & Why They Matter	90
3.B.i.a. Brief Biometrics History	90
3.B.i.b. Applications	94
3.B.ii. Ethical Challenges	98
3.B.ii.a. Fundamental Issues	98
3.B.ii.b. Specific Issues	102
3.B.iii. Creating a Biometrics Governance Framework	106
3.B.iii.a. Regulating Biometrics	106
3.B.iii.b. Global Examples	110

Chapter 4: Ethical Challenges of Data Analytics in Healthcare	. 117
4.A. The Ethical Challenge of Data Analytics in Genomics	. 118
4.A.i. Social Policy of Genomic Data	. 118
4.A.i.a Genetic Discrimination in Health Insurance	. 119
4.A.i.b. Common Heritage	. 122
4.A.i.c. Distributive Justice in Genomics	. 125
4.A.ii. Creating a Genomics Governance Framework	. 130
4.A.ii.a. Regulating Biobanks and Privacy	. 131
4.A.ii.b. The Need for Privacy & Security Law Harmonization	. 133
4.B. A Cautionary Approach to Digital Technologies in Healthcare	. 137
4.B.i. The Digital Revolution	. 138
4.B.i.a. Technology & Society Relationship	. 138
4.B.i.b. An Information Society	. 140
4.B.i.c. Forward to a Social-Digital Future	. 142
4.B.ii. Philosophy of Technology	. 145
4.B.ii.a. Technological Determinism	. 145
4.B.ii.b. A Medical Sociologist's Perspective	. 148
4.B.ii.c. The Precautionary Principle & Technology Assessment	. 150
4.B.ii.d. Repositioning the Medical Ethos for the Future	. 153
4.B.iii. Applied Ethics Analysis to Human Enhancement	. 156
4.B.iii.a. Benefits Beyond the Biological	. 156
4.B.iii.b. Radical & Moderate Enhancement	. 159
4.B.iii.c. Ethical Concerns	. 162
4.B.iii.d. Advancing the Ethical Debate	. 165
Chapter 5: Ethics of Governance of Artificial Intelligence and Information Management in Healthcare	. 168

5.A. An Applied Analysis of the Ethics & Effect of Artificial Intelligence on the D Healthcare	elivery of
5.A.i. Setting the Stage for Artificial Intelligence	
5.A.i.a. The Exponential Potential of AI	
5.A.i.b. Machine Learning & The Human Brain	
5.A.i.c. The Turing Test & Other New Versions	
5.A.ii. Philosophy & Ethics of Artificial Intelligence	
5.A.ii.a. Conflict with Free Will & Moral Disagreement	
5.A.ii.b. Artificial General Intelligence	
5.A.ii.c. Building Ethics into AI and the Idealization of Moral Agency	
5.A.iii. Challenges & Opportunities for Artificial Intelligence	
5.A.iii.a. Challenges and Opportunities in Digital Health Research	
5.A.iii.b. Static or Changing Capabilities & Dynamics in Healthcare	190
5.A.iii.c. Artificial Intelligence Safety Engineering	
5.B. The Impact of Information Management on Data Analytics	
5.B.i. Foundations of Health Information Management	
5.B.i.a. Acquisition, Storage, & Use	
5.B.i.b. Health Information Infrastructure	
5.B.ii. Legal & Ethical Aspects of Health Information Management	
5.B.ii.a. Privacy, Confidentiality, & Data Sharing	
5.B.ii.b. Rethinking the Ethical Approach to HIM	
5.B.iii. Health Information Technology Governance	
5.B.iii.a. Defining HIT Governance	
5.B.iii.b. Normative HIT Governance	
Chapter 6: Technology's Impact in End-of-Life Practices	

6.A. Technological Opportunity & Ethical Challenge in Hospice & Palliative Care: He Hinderance in the Art of Dying Well	lp or 219
6.A.i. Historical & Cultural Background of Death & Dying	220
6.A.i.a. Systematic Changes Towards Death Over Time	220
6.A.i.b. Toward a New Ethical Framework for Dying Well	223
6.A.ii. Opportunities of Technology in Hospice & Palliative Care	227
6.A.ii.a. Improvements	227
6.A.ii.b. Electronic Learning	231
6.A.iii. Ethical Challenges of Technology in Hospice & Palliative Care	234
6.A.iii.a. Fundamental Issues	234
6.A.iii.b. Specific Issues	238
6.B. Ethical Considerations of Using Care Robots in Aging Populations	243
6.B.i. The Mechanization of Care Work	244
6.B.i.a. Scope of the Aging Population	244
6.B.i.b. Elderly Care Needs	246
6.B.i.c. Rise of Care Robots	248
6.B.ii. Designing Robots Using Care Ethics	250
6.B.ii.a. Defining Care Ethics & The Value of Care	250
6.B.ii.b. Relational Autonomy in Elder Care	252
6.B.ii.c. Human Dignity and Vulnerability of the Elderly	255
6.B.iii. Ethical Challenges of a Care-Centered Framework	263
6.B.iii.a. Human Capabilities	264
6.B.iii.b. Context-Sensitive Process	266
6.B.iii.c. Deception	267
Chapter 7: Conclusion	271
Endnotes	279

322 Bibliography

CHAPTER 1: INTRODUCTION

The success of practically all enterprises, big and small, depends so heavily on data now that it is treated as an asset, and this is where the phrase "data-driven" has resulted. The literature shows that the healthcare industry is no stranger to this fact as the technologies it uses, from data analytics, artificial intelligence/machine learning, and electronic information management, all use data to bring about many benefits, such as faster and more accurate results for all involved in healthcare including patients, professionals, and managers. Data must be controlled since it is a vital resource. Without proper management, data often becomes duplicated, of low quality, and is unable to support the insightful conclusions that come from excellent data. Moreover, improper management and governance of data can cause numerous ethical concerns regarding privacy, confidentiality, and trust. Adverse outcomes will ensue without proper management and governance, causing patient harm, financial woes, and endless legal paperwork. Therefore, ethical governance of data analytics in healthcare is essential. This dissertation aims to explain the ethics of data analytics governance in healthcare and the challenges and opportunities accompanying this management.

What makes this research different from other studies is that this dissertation focuses on multiple areas of healthcare while demonstrating how all these areas come together to reveal the need for ethical data governance across the global healthcare industry. Recent literature and studies on data governance only focus on data ethics and governance from a specific industry or country, mostly business, without considering the more significant global impact. Data is a worldwide asset, and its implications and considerations should be viewed as such. In the literature and studies centered around data governance, the authors tend to keep narrow views on the effect of data governance, or lack thereof, within the specific country – never implicating the

raging global battle of health data. While both perspectives are necessary and feasible, one cannot ignore the global, interwoven nature of healthcare and society. Hence, this dissertation will attempt to reconcile these approaches by weaving together topics in healthcare, such as the blending of finance and delivery, data governance at the micro and macro levels, data analytics in healthcare, and technology's impact on end-of-life practices. By highlighting these healthcare areas, the dissertation explores characteristics of the healthcare landscape both nationally and globally. Whether these areas affect patients, professionals, or managers directly or indirectly – each of these areas has a significant impact on the healthcare landscape. As the use of data analytics in healthcare increases, the outcomes and their derivatives from these technologies increase proportionally.

Ethical governance of data analytics focuses on applying ethics to all aspects of data analytics, from beginning concepts to ending enforcement stages. This approach will aim to take data governance throughout an organization's "three P's:" policies, processes, and procedures. This kind of governance is crucial since it ensures that data analytics is safe and secure to utilize and takes ethical principles into the foundation of the whole data cycle. By reprioritizing ethics at the forefront of governance, instead of having it as merely a component of data analytics governance, all those involved with these technologies will be able to produce and maintain ethical data analytics. Since the field of data analytics is not groundbreakingly new, this dissertation does both a retrospective and prospective view on the ethics of governance of data analytics in healthcare. This dynamic view will allow for a thorough analysis of the ethics of data analytics governance in healthcare. One of the first steps to ethical data governance is identifying current and future issues in the field. Thus, each chapter in the dissertation will examine the past, present, and future issues surrounding data analytics governance. Understanding the ethical

concerns within these issues will provide a forum for those issues that have yet to be addressed within the field by data experts. Further, it becomes much more feasible to formulate and recommend solutions by locating the commonalities between these ethical implications.

Ethical data governance is not a single instance fix but an involved and ongoing process. As previously noted, ethical data governance covers a wide array of the data analytics cycle and hence involves multiple stakeholders. These stakeholders include but are not limited to developers, government agencies, healthcare professionals, patients, and researchers. All of these stakeholders play some part in the ethical data governance process. For example, developers have a moral responsibility to create and design artificial intelligence and information systems that will not harm their users via improper use, storage, or acquisition of private healthcare data. This can be achieved through fostering an ethical workforce and culture in the office. By encouraging and promoting ethics in these environments, the data analytics products and services in a research and development phase will consider ethical principles and outcomes throughout the entire product or service design. This goal can be achieved in various ways, such as consulting an ethicist specialized in data analytics governance to provide oversight or to teach ethics and its practical applications to developers – who often lack proper training and experience in that area.

Government agencies are also significant stakeholders because they have the authority to pass and enact legislation that can directly or indirectly affect the artificial intelligence/machine learning algorithms being developed or released, in addition to regulating healthcare data that affects nearly all its consumers. The responsibilities of the healthcare professionals and patients' stakeholders differ from that of the developers and government agencies. They usually advocate for change in the technology itself as they spend the most time utilizing the product or service

due to data analytics. This point can be further highlighted by the fact that healthcare professionals are becoming more legally liable for the consequences of unethical use of healthcare data products and services – most notably a privacy breach – which negatively impacts the patient. The researcher is a unique stakeholder because the role of the researcher perseveres throughout the previous stakeholder categories as well – developers, government agencies, healthcare professionals, and patients – all of whom contain and utilize the services of researchers. Acknowledging the significant stakeholders and their roles will raise awareness of the need for ethics of data analytics governance in healthcare.

Numerous ethical principles addressed in this dissertation align with much of the current research and literature surrounding the topic. Ethical issues in data governance, such as privacy and confidentiality, accountability, transparency, justice/fairness, autonomy, and more, are highlighted throughout this dissertation. Where the literature fails in comparison is taking the application of these ethical issues into various areas of healthcare, nor does it attempt to offer any sector-specific solutions to these issues. Fortunately, this dissertation identifies these ethical concerns and provides several recommendations for answers in the form of frameworks for data governance. While these issues are often significant in scope, globally even, usually a concrete solution will not always work practically in all countries. Therefore, a framework for an ethical data governance system will help organizations and systems worldwide to build and utilize a more ethical data governance for all its stakeholders.

The technological revolution has been around for a long time. However, the technology of data analytics is much more recent, over the past decade bringing tremendous changes in science, society, culture, and daily life, including health care. The technologies related to data analytics cover a large terrain, including artificial intelligence, big data, and electronic

information management, each with accompanying ethical challenges. Data ethics refers to the accumulation of these ethical challenges in a shorthand manner.

Clinical and health leaders and organizations must engage these challenges of data ethics by fostering more effective governance oversight of ongoing reform in health care. This governance oversight includes a vast array of ever-expanding medical capabilities, constrained resources and staff shortages, fast-emerging new diseases (including pandemics), and an increasingly diverse mix of patients and consumers whose needs only grow. This dissertation addresses these challenges of data ethics in this fast-developing environment as governance oversight develops effective policies organizationally, nationally, and globally. Hence, the thesis focuses explicitly on the ethics of data analytics governance in health care.

After an introduction in the first chapter, chapters two and three establish the context for the dissertation. Chapter two explains the significance of the blending of finance and delivery in health care today. Chapter three specifies this context further by discussing the widespread tension between individual data privacy and public health surveillance from a global health perspective. Chapter four presents my approach to data ethics, focusing on the key ethical challenges of data analytics from a technological perspective that guides the applied analysis in the subsequent chapters. The applied chapters engage widespread and illustrative topics dealing with data ethics. Chapter five discusses the ethical challenges of artificial intelligence and data science in health care. Chapter 6 discusses technological opportunities and ethical challenges in end-of-life practice. Finally, chapter 7 presents a brief conclusion to the dissertation.

Chapter 2 will describe how data analytics effectively blends the finance and delivery of healthcare and can be used to assist in healthcare reform within the United States. This analysis will begin by looking back to the past in order to understand what worked and what failed in

healthcare reform and why. U.S. healthcare reform began in the late 1800s and stretches to the present day. This section will detail how during this reform period, important social movements have spurred certain meaningful changes in legislation, while also critically reflecting on the Catholic Church's viewpoints of these changes due to the fact that the Catholic Church is a leader in hospitals around the country. The analysis will then focus on one of the ethical principles behind healthcare reform, distributive justice, and its foundational theories, basic concerns, and resource allocation issues. In the last section, this analysis will address some implementation considerations of a new healthcare system such as a popular reform effort of universal access to basic care.

The second half of Chapter 2 concerns the ethics of data analytics within the healthcare landscape. A brief overview of what big data is and why it matters including ethical challenges of security, privacy, and informed consent will be explored. As with any invention, data analytics poses both opportunities and challenges that must be met as society progresses. Furthermore, data analytics provides an expansion in personalized care through predictive modeling and integrative sequencing opening new realms to the possibilities of medicine. This chapter will show how data analytics has organizational implications for health systems such that it has altered and will continue to drastically alter the delivery and finance of healthcare at the same time as needing to respect an obligation for security, privacy, and confidentiality with patient population data. This turn towards patient population data sets the stage for Chapter 3 that focuses on the global health data battle – with data governance on the micro and macro levels.

The extent to which policymakers are able to strike a balance between the need to protect genuine patient and clinician preferences on the one hand and the significance of healthcare data to general public on the other will determine whether additional data access is permitted to

produce more robust data about how the health care system operates.¹ The ability to manage research-related data, such as that from the Federal Coordinating Council for Comparative Effectiveness Research, securely and effectively will play a role in determining how to strike a balance that will allow for the production of evidence without jeopardizing these ethical obligations.²

The third chapter moves on to the public health data battle in which individual autonomy and rights are pitted against population health and the 'Common Good.' Concluding the section at the heart of the argument will be the debate between individual data privacy and public health surveillance – with ethical arguments made on both sides. This section will detail the imperative to influence policy as technology rapidly changes the norms of everyday life. Changes in policy in the form of ethical governance frameworks that enable regulation and monitoring of the technologies that are a double-edged sword are crucial. Policy at not only a national, but global level will be conveyed, using examples such as the European Union and UNESCO.

The second half of Chapter 3 will focus on biometrics. Biometrics have passed the prototype phase and are now being used and implemented in all technology and computer science areas. Biometrics are being used for personal recognition, authentication, health systems, government security, and border control - just to name a few. The potential ethical ramifications of every program using novel technology must be continuously investigated. At the municipal, national, and international levels, biometrics may be essential to guaranteeing more dependable identification systems. Humanity must, however, carefully weigh the advantages against the moral and societal concerns. A number of concerns, such as the preservation of privacy, secrecy, individual freedoms, and the interaction between personal rights and the common good are related to individual identities. Biometrics and their associated data (fingerprints, retina/facial

scans, voice recognition) very closely relate to data analytics governance because biometrics are the unique personal identifiers that help data analytics and artificial intelligence/machine learning to differentiate one individual from another. Biometric ethics place a premium on the identification and subsequent de-identification steps that safeguard an individual's personal data's security and privacy.

The central question to be addressed in this section is: Are current biometric practices – such as a lack of privacy and data protection, over-surveillance, and large-scale applications – harmful towards human dignity? This question will be investigated by first explaining what biometrics are and why they matter, then the section will investigate ethical challenges associated with biometrics, and finally describe the need for creating a biometrics governance framework. While data analytics can assist in creating a biometrics governance framework, there are some concerns regarding the use and misuse of data analytics itself that must be addressed. The next chapter describes some of those concerns in two senses: one practical, within genomics, and another theoretical, within the future of digital technologies themselves.

Chapter 4 takes a different look at data analytics and focuses on the ethical concerns and challenges that this technology can display. The social policy of genomic data such as genetic discrimination, common heritage, and more concerns with distributive justice will be detailed. The section goes on to detail challenges in data analytics with the autonomy paradigm and cloud computing respectively. These issues pose a challenge to data analytics due to the cloud computing's de-centralization of data storage. A key factor in respecting the privacy and autonomy of the individual is having a secured, centralized data storage that can easily be maintained and traced back to in the event of a data breach. The rest of the chapter will lay a philosophical/theoretical foundation discussing the impact of technology in healthcare via the

digital revolution and its relationship to society while looking at the future of this dynamic relationship and how data analytics could vastly alter this relationship down a slippery slope of a dehumanized world and human enhancement. A cautionary approach is taken in which concepts like technological determinism and the Precautionary Principle will be explored. An applied analysis of these concepts will be used within the topic of human enhancement citing ethical concerns and providing a foundation to advance the debate.

The fifth chapter begins by looking at a few capabilities of artificial intelligence (AI) and its potential for revolutionizing healthcare. Since AI acts as the interpreter of all the data it receives using machine learning and algorithms, AI helps scientists, researchers, and healthcare data stewards among many others to understand and adapt to the clinical outcomes that the data forecasts. Philosophical concepts such as free will, moral disagreement, and moral agency will be explored in light of AI. The chapter will conclude with an analysis of the challenges and opportunities AI has on the delivery of healthcare. The second half of the chapter continues describing the foundations of health information management and its legal and ethical aspects. An applied analysis will be detailed within systematic reviews and risk management. Finally, the chapter concludes with a call for health information technology governance – defining its need and core components highlighting concepts such as information stewardship and data quality.

The end of Chapter 4 looked at an example of how data analytics could be applied in a negative or dystopian way if humanity is not careful of the slippery slope that human enhancement can lead down. In contrast, the sixth chapter focuses on an overall positive or utopian application of various data analytics-driven technologies within the specific context of the elderly population and end of life practices such as palliative care and hospice. A historical and cultural background of death and dying will be detailed outlining systematic changes

towards death over time. Opportunities of technologies in end-of-life practices will be explored such as telemedicine and electronic learning. Challenges will be described as well such as the digital-divide, professional-patient relationship via technology and fundamental issues such as the technological imperative vs humanizing. The second half of the chapter continues with ethical considerations of using care robots in aging populations using care ethics. Concepts like artificial morality, relational autonomy, and dignity will be described within this applied chapter of data analytics in healthcare.

CHAPTER 2: DATA ANALYTICS BLENDING FINANCE & DELIVERY IN HEALTHCARE 2. Introduction

The second chapter describes how data analytics can be used in a broader system context to initiate healthcare reform within the United States. Invoking its authority to tax, spend, and regulate trade, the federal government of the United States has regularly approved the gathering and use of health information to inform its activities since the 1950s.³ However, somewhat recently, in 2009, the government created a Federal Coordinating Council for Comparative Effectiveness Research, which established a data steward under health reform.⁴ The enormous volumes of healthcare data submitted to government agencies under Medicare, Medicaid, and countless other federal health programs are collected by this data steward. This council may "enable the research and investigation on a reformed health system, while in common purpose to enable advances in population health" by operating under open policy guidelines and having the power to gather, aggregate, manage, and safeguard data.⁵

Beginning with a background of healthcare reform in America, in the first section of the chapter, the chapter highlights one of the critical ethical concepts that data analytics assists in exercising – distributive justice. In this section, the foundational theories of distributive justice

are explored through an array of literature indicating the basic concerns of need and contribution within allocation problems. Implementation considerations of a new healthcare system are described by listing criteria for a better system, four conditions that must be met, and universal access to basic care that technology will help achieve. The second half of the chapter concerns the ethics of data analytics within the healthcare landscape. A brief overview of what big data is and why it matters, including ethical challenges of security, privacy, and informed consent, is explored. Social policies of genomic information are discussed in length, including issues of genetic discrimination, common heritage, and distributive justice. In the chapter, possibilities and difficulties in data analytics related to the autonomy paradigm and cloud computing are discussed. Personalized care using predictive modeling and integrative sequencing are described in their effect on the finance and delivery of healthcare.

2.A. The Ethics Behind Healthcare Reform: Saving Private America

2.A. Introduction

For the last half-century, healthcare reform in America has been and will continue to be an essential topic that arises among family dinner tables and the rooms of Capitol Hill. It is no secret that the United States healthcare system is far from being a world leader and, to this day, contains numerous disadvantages, nuances, and funding issues that keep it from becoming the most robust healthcare system in the world. This analysis begins by looking back to the past to understand what worked and failed in healthcare reform and why. US healthcare reform started in the late 1800s and stretches to the present day. To understand this history of reform in the US, it is essential to mention the critical social movements that have spurred certain parts of reform with an emphasis on the Catholic Church's viewpoints since they are a leader in hospitals around the country. The analysis then focuses on one of the ethical principles behind healthcare reform,

distributive justice, and its foundational theories, basic concerns, and resource allocation issues. The last section addresses some implementation considerations of a new healthcare system, such as a widespread reform effort for universal access to basic care. Competition exists among the world's leading nations to have the most robust healthcare system. However, it is only by ethically reforming the healthcare system that the heroic act of 'Saving Private America' can be accomplished.

2.A.i. Background of Healthcare Reform in America

While healthcare reform is a complicated and essential topic, to properly understand where the future of healthcare reform in America may lie, it is imperative to know how it started and became what the current healthcare landscape is today. Highlighting a synopsis of American healthcare reform's history from the late 1800s to today helps us understand what has worked and why or why not. Then, by examining the social and grassroots movements that have helped spur certain pieces of legislation, future movements can be predicted and their outcomes forecasted. Finally, by reviewing the history of healthcare reform in light of the efforts and responses by the Catholic Church, it becomes apparent that there is an observable moral difference between what has been proposed by the most recent reform legislation in the Affordable Care Act (ACA) and what the United States Conference of Catholic Bishops (USCCB) stand for.

This is crucial to the dissertation because the USCCB fights to preserve immigrants' historical countrywide availability of health care and eliminate any potential future hurdles to access that the ACA could construct. With the U.S. immigrant population of roughly 46.2 million people as of late 2021⁶, this accounts for a vast population of healthcare data that would be unusable in the government's Federal Coordinating Council for Comparative Effectiveness

Research if the immigrant population lost access to healthcare. Yet, the matter becomes more complicated as it is not as simple as Catholic social justice principles teaching the importance of the right to healthcare for the individual, as long as it is equally distributed.

2.A.i.a. Brief History of Healthcare Reform in the United States

The evolution of healthcare in America is unique compared to other developed nations worldwide. Because of the country's deep-seated commitment to capitalism, healthcare reform in America has historically taken a different course than in the UK, France, or Canada, countries that have established some universal healthcare.⁷ While America's system has diverged from the rest, it has changed dramatically over the past century to fit the needs of its citizens better. However, the healthcare landscape still requires a significant amount of amending.

Late 1800's to Medicare/Medicaid

Healthcare originally consisted of patients receiving treatment and then working out some form of payment to doctors directly. This worked during simpler times, but as the population and industry grew, this was no longer feasible. The industrial revolution brought many steel mill jobs, economic booms, and dangerous work environments, leading to more workplace injuries. A big drive for organized medicine under the direction of the American Medical Association (AMA) began as the 20th century arrived.⁸ As progressivism advanced across Europe, several countries approved their first social welfare legislation and created the foundation for required government-run healthcare.⁹ Kant Patel explains, "A unique American history of decentralization in government, limited government, and a tradition of classical liberalism are all possible explanations for the suspicion around the idea of compulsory government-run insurance."¹⁰ A draft bill requiring all workers to have health insurance was issued by the American Association for Labor Legislation (AALL), and some states saw

increased support for the initiative. As the United States entered World War I, a few states expressed interest but did not pass it. The AMA first endorsed the concept, but by 1920 it had changed its mind.¹¹

Healthcare became a pressing problem that required attention after World War I as a result of the expense of treatment continuing to climb and becoming more expensive than the typical person could bear. As a result, a cohort of educators established a system through Baylor University Hospital where they would "agree to pre-pay for future medical services."¹² This non-profit corporation, which solely offered coverage for hospital services, served as a precursor to what would eventually develop into the well-known Blue Cross insurance. The problems of unemployment and retirement or "old age" benefits captured the public's attention after the Great Depression wreaked havoc on society in the 1930s. Thus, the Social Security Act was passed into law by President Franklin Delano Roosevelt (FDR) in 1935. This legislation created a system of "old age" benefits and "allowed states to create provisions for people who were either unemployed or disabled (or both)."¹³

After American involvement in World War II, national attention shifted to the war and away from healthcare reform. In order to retain existing workers and recruit new ones at this period, employer-sponsored health insurance evolved. This excluded many Americans at the time, including those who were jobless, had impairments, or worked for a company that did not provide such insurance. The Wagner-Murray-Dingell Bill, which advocated for universal healthcare supported by a payroll tax, was launched in 1943 as a response to this gap. However, the bill faced strong opposition from the AMA and others; thus, it eventually died in committee review.¹⁴ Healthcare reform was put on hold in the 1950s while the American administration concentrated on the Korean Conflict, which was third conflict in 40 years, and as expenses

continued to climb as a result of new medical advancements like the first organ transplant and the polio vaccination.¹⁵

Medicare to Present Day

The American government started keeping track of National Health Expenditures (NHE) in the 1960s and calculating them as a proportion of Gross Domestic Product (GDP). During that period, "NHE accounted for 5% of GDP."¹⁶ President John F. Kennedy pushed for healthcare reform, especially for senior citizens, but failed again due to fear tactics of an overwhelming government-run universal healthcare program spread by the AMA.¹⁷ Following Kennedy's death in 1963, Lyndon B. Johnson gained office and expanded the Hill-Burton Program and the Social Security Act of 1935. With the help of doctors and hospitals, this initiative sought to ensure that older residents and those with disabilities had access to cheap healthcare. The Social Security Act of 1965 was approved with this concept, establishing the framework for Medicare and Medicaid. President Richard Nixon made a less comprehensive health insurance reform in the 1970s than previous single-payer system plans.¹⁸ He proposed federalizing Medicaid for lowincome people with small children in their care, requiring employers to provide private health insurance if their workers agreed to pay 25% of the expenses, and supporting health maintenance organizations (HMOs).¹⁹

NHE made for 8.9% of GDP in the 1980s, and healthcare privatization was widely practiced under the Reagan Administration. With this privatization, health insurance was a requirement for all companies with at least 5,000 workers.²⁰ States would create health insurance alliances to buy individual health insurance policies for people unable to receive health insurance via their place of employment. Subsidies would be made available for those who could not afford insurance. Flint Wainess notes, "In 1986, the Consolidated Omnibus Budget Reconciliation Act

(COBRA) was signed into law by President Ronald Reagan, enabling former employees to continue to be covered by their previous employer's group health plan as long as they agreed to pay the entire premium (employer portion plus employee contribution)."²¹ In the 1990s, NHE now represented 12.1% of GDP, and during his administration, President Bill Clinton immediately took steps to address this. He suggested the Health Security Act of 1993, which combined universal protection with consideration for the market for private insurance. In 1993, after considerable debate, the legislation was quietly killed as Congress adjourned for the holiday break.

Later, in 1996, Clinton enacted the Health Insurance Portability and Accountability Act (HIPAA), establishing stricter guidelines for individual privacy. Wainess continues, "Furthermore, it set restrictions on how pre-existing conditions were handled by group health insurance and made sure that a person's health records will be available if needed."²² The Clinton Administration also launched the Children's Health Insurance Program (CHIP), which extended Medicaid coverage to adolescents up to age 19 who lack health insurance. The intended market for CHIP was those with earnings that were too high to qualify for Medicaid coverage. Every state still manages CHIP, which is still in operation. According to 2022 data from the Centers for Medicare & Medicaid, CHIP provides low-cost health insurance to approximately 7 million children and adolescents.²³

In order to include prescription pharmaceuticals, George W. Bush wanted to update Medicare after he was elected since, by 2000, NHE amounted for 13.3 percent of GDP. Millions of Americans still participate in the Medicare Prescription Drug, Improvement, and Modernization Act of 2003 (often known as Medicare Part D). The next several years saw little healthcare reform as the U.S. focused on terrorism and the Iraq War. By 2010, the Patient

Protection and Affordable Care Act had been enacted, and NHE accounted for 17.4% of GDP. According to the legislation, "Americans earning less than 400 percent of the poverty level would qualify for subsidies to help cover the cost and more," and insurance providers may not deny coverage due to pre-existing conditions.²⁴ While this brief history of healthcare reform highlights some of the significant changes and shows the direction for US healthcare policy going forward, one must note the social movements that helped to drive all these changes, whose effects are still seen to the present day.

2.A.i.b. Social Movements

The 20th century was a prime example of the strength of social movements in reforming healthcare policy. Several different "cohorts" of people, such as women, workers, seniors, physicians, and welfare recipients, all organized themselves to induce some form of change in the several modifications to America's healthcare policy over the century. The major social movements that will be discussed are the Civil Rights Movement and grassroots movements, specifically women's and the AIDS movement. Each movement, despite having leaders, was composed of regular people who demanded change.

Civil Rights Movement

Racial prejudice kept African Americans from receiving essential medical treatment for a large portion of the 20th century. They were compelled to focus on creating their institutions, such as fraternal clubs, life insurance firms, and neighborhood public health initiatives.²⁵ Rightfully, civil rights campaigners disregarded reform plans that either upheld segregation or indirectly assured inequity, such as granting states authority over health care. Because the Wagner-Murray-Dingell plan lacked "safeguards to ensure equitable distribution of funds in the states where Negroes and Whites [were] forced to use separate hospitals, clinics, and other health

services," the NAACP, for example, only grudgingly backed the legislation.²⁶ After some time, civil rights organizations expressed concern that several features of Clinton's Health Security proposal, comprising of its stress on employer-sponsored insurance and the existence of individual private insurers with a longstanding record of discrimination and predatory lending practices, may be harmful to African Americans.²⁷

Desegregation has not been enough by itself to establish racial equality in healthcare, according to civil rights activists. The Medical Committee on Human Rights was established in 1964 by activist physicians to provide medical aid to civil rights campaigners in the southern United States. Nevertheless, they quickly became engaged in the northern American campaign against "inadequacies in health care."²⁸ However, they persisted in tackling the major barriers that low-income African Americans still had to overcome in order to get medical treatment. In the middle of the 1960s, civil rights activists succeeded in their tough campaign for hospital integration. Hospital policies on treating the poor and many medical facilities' and doctors' refusal to accept Medicaid made the connection between racial and financial obstacles to access very evident. Organizations for civil rights requested, via many class-action lawsuits, hospitals with government funding take additional residents who were struggling financially and remain in inner-city neighborhoods instead of relocating to the suburban areas.²⁹

Grassroots Movements

Reforms in the medical field has been demanded by and inspired by a variety of grassroots groups, including those for women's rights, civil rights, and those with particular diseases like AIDS. However, their healthcare demands were for particular reforms on behalf of their groups, such as the availability of experimental AIDS medications and the racial desegregation of hospitals.³⁰ Despite these modifications, a movement for universal healthcare

has not developed from these grassroots demands. Small-scale advancements in the delivery of healthcare have been viewed as obstacles to or deterrents to more thorough transformation by certain academics and reformers.³¹

Grassroots calls for healthcare have also been discovered to represent the start of a more thorough criticism of the American healthcare system. Advocates for quick and specific reforms discovered that they had a new and urgent awareness of the need for fundamental transformation as a result of their experiences with health care.³² Eventually, requests for universal access were incorporated in the platforms of various social movements. We may be able to better understand how these two types of movements have indeed been kept separate and how they can come together by fusing the history of healthcare reform attempts with grassroots action for healthcare system improvements.³³ Perhaps the modern-day link between national and grassroots campaigns comes from the power of data analytics. Since data analytics has the scope and capability to reduce healthcare costs, financing a universal healthcare system in America could become one step closer to reality – giving these campaigns some much-needed rejuvenation and a more robust platform to advocate change.

2.A.i.c. Catholic Efforts

Catholic social theory advocates for an inclusive society where everyone has access to basic healthcare and for seeing people as subjects rather than just objects.³⁴ Although understanding the "body as an object", distinct from other bodies, and composed of several parts or systems has greatly increased our capacity to treat illness, it has also negatively impacted patient care and human dignity.³⁵ Healthcare expenses in the US prevent easy access to care, which causes financial ruin for millions of people every year. This is essentially a generational phenomenon brought on by growing geriatric longevity aspirations (discussed in Chapter 6),

where practices are based on the assumption that health is a product of human inventiveness and care. Catholic social thinking recognizes God as the ultimate source of healing and believes love is necessary for good health care.³⁶ A viewpoint like this could show how to (re)humanize the hospital experience and address the social disparities in contemporary health care.³⁷

The United States Conference of Catholic Bishops (USCCB) has long emphasized the need of having access to the highest medical treatment in order to protect human life and advance human rights from conception to old age.³⁸ According to the USCCB, legislative and administrative efforts to achieve health care reform should be supported which "1) ensures access to quality, affordable, life giving health care for all; 2) retains longstanding requirements that federal funds not be used for elective abortions or plans that include them, and effectively protects conscience rights; and 3) protects the access to health care that immigrants currently have and removes current barriers to access."³⁹ In the USCCB article, *A Framework for Comprehensive Health Care Reform: Protecting Human Life, Promoting Human Dignity, Pursuing the Common Good*, The bishops support providing the economically disadvantaged and vulnerable with accessible health care, bringing our nation one step closer to the goal of universal health coverage. Similar clarity is provided by the USCCB when it emphasizes that this must be done in a manner that respects each person's dignity and their livelihood, wellbeing, and consciousness.⁴⁰

As Pope Benedict XVI stated, within the healthcare industry, "it is important to establish a real distributive justice which, on the basis of objective needs, guarantees adequate care to all."⁴¹ Moreover, "if it is not to become inhuman, the world of health care cannot disregard the moral rules that must govern it. We wholeheartedly commit ourselves to healthcare reform that achieves these worthy goals. We will advocate for addressing the current problems in the Patient

Protection and Affordable Care Act, as well as others that may become apparent in the course of its implementation."⁴²

The U.S. bishops claim in the introduction to Part One of the Ethical & Religious Directives (ERDs) that "the first right of the human person, the right to life, entails a right to the means for the proper development of life, such as adequate health care" citing Pope John XXIII's encyclical *Pacem in Terris*.⁴³ A core human right necessary for human flourishing is health care. People may be less able to thrive and experience spiritual, temporal, and financial deterioration if they are not in excellent health—their capacity to pursue fulfilling relationships declines. Life itself, together with human dignity, may be lost. Due to the fact that one's health (or lack thereof) directly affects their potential to grow, the Catholic social teaching tradition considers having access to quality healthcare a universal human right.⁴⁴ The tradition views fair access to basic healthcare as a right and sees denying that right as a breach of distributive justice.⁴⁵

As a result of our social nature, humanity is accountable for fostering social conditions that enable individuals, families, and communities to find fulfillment in a relatively thorough and quick manner.⁴⁶ Regardless of color or ethnicity, the Catholic health ministry respects every individual with the same unalienable dignity and respect. The Catholic health ministry stands apart because of this exhortation to adopt a Christian sense of justice and mercy. It is thus perfect for taking the lead in the fight against healthcare inequities. The Catholic health ministry is pushed into the public sphere, where healthcare discrepancies find their roots and continue to maintain ailing populations at an unfair level of disease and decline, in addition to being required to serve each individual in a just and equitable manner.⁴⁷

The history of healthcare reform and the social movements accompanying it are crucial to understanding the future of reform in America. The United States' long and difficult journey

toward healthcare reform has distinguished the country from the other industrialized nations in the globe, but not always for the better. There are still numerous problems that are in place in the current system; however, many of these issues can be remedied by looking at distributive justice principles & applying them in future legislation.

2A.ii. Distributive Justice

Should everyone have government-funded access to reasonably priced healthcare that can assist in maintaining life? In the US, society is beginning to endorse the concept that all citizens should have equal access to healthcare, including insurance coverage.⁴⁸ However, there needs to be more agreement on how the government should function, how to pay for insurance, how to provide healthcare, and what it means to have fair access.⁴⁹ It is still being determined if such a weak agreement can lead to a further agreement on implementing a fair access system. This problem centers on the concept of distributive justice. This section aims to break it down by examining its foundational theories, main concerns, and allocation problems on a micro and macro level.

2.A.ii.a. Background of Foundational Theories

To understand what distributive justice is and how it works, it is critical to know how it has logically evolved through history to apply it today. Therefore, it is essential to focus on a brief history of critical theories contributing to the concept of distributive justice. While not a comprehensive list, the traditional and recent theories of justice provide a starting point for an understanding. Utilitarian, Libertarian, Communitarian, and Egalitarian views of justice have traditionally prevailed, although Capability and Well-Being theories have emerged more recently.

Traditional Theories
Utilitarian theory was founded in the nineteenth century by John Stuart Mill and Jeremy Bentham in England.⁵⁰ The central part of Utilitarian theory concerning distributive justice is the Greatest Happiness Principle, which states: "do that which delivers up the most pleasure and least pain."⁵¹ To maximize utility wherever feasible, or to act in a way that benefits the largest number of people, is how this expression is often translated. This theory expressly focuses on maximizing (or the right distribution of) welfare when applied to distributive justice. This notion, however, lacks precision and has raised questions about whether particular welfare tasks need to be prioritized.⁵² A utilitarian with a realistic view of justice must be able to describe how welfare works in society and how to balance welfare circumstances within the framework.⁵³

Libertarian beliefs are often linked to John Locke's fair and natural rights to liberty philosophies. According to a libertarian, justice focuses on "the autonomous operation of fair procedures and transactions under conditions of law and order" rather than on utility or acting to satisfy individuals' health and welfare requirements.⁵⁴ According to this theory, there is no just distribution pattern independent of free-market processes for acquiring property, lawfully transferring that property, and offering reparations to people whose property was lawfully taken or who were otherwise illegally prevented from participating in the free market.⁵⁵ In this system, medical professionals and investors enjoy property rights, and society is not ethically obligated to provide healthcare. Beauchamp & Childress emphasize that in a libertarian sense of justice, "Health care is not a right, the ideal system of health insurance is privatized, and charitable care institutions are nonprofit and untaxed."⁵⁶ According to this view, a fair society upholds the rights to property and liberty, allowing each individual to take independent action to better their situation and safeguard their health.⁵⁷

Aristotle, Hegel, and many other influential philosophers were the foundation for various communitarian beliefs.⁵⁸ These ideas are founded on distributive principles from moral societies' notions of the good. An approach to justice that prioritizes the common good above individual rights rejects all conceptions of justice that are based on contracts and individual rights.⁵⁹ According to communitarians, there are many distinct perceptions of what is just or moral in the world, and so there are many different principles of justice.⁶⁰ These ideas contend that society should be allowed to apply a substantive definition of the good and aim to loosen liberal norms about the impartiality of the government.⁶¹

Compared to other core distributive justice theories, egalitarian views have the longest history. These beliefs are based on the theological idea that because all men are born equal, they all possess the same moral standing.⁶² Although these theories support equality in terms of treating individuals fairly in certain situations, none of them include a direct distributive principle requiring that all individuals get an equal share of all societal advantages.⁶³ These beliefs are founded on the idea that everyone should have equal freedom and access to the things in life that are important to every sensible person. Every individual would have equal access to sufficient, albeit not optimal, levels of healthcare in an egalitarian society; the precise amount of access would depend on the availability of social resources and public decision-making procedures.⁶⁴

Recent Theories

Amartya Sen and Martha Nussbaum are credited with giving rise to capabilities theory in the twenty-first century. They claim that the freedom to attain these states should be evaluated in terms of capabilities and that the chance to do so is of basic moral relevance.⁶⁵ According to the theory, a person's ability to attain goals determines the quality of their life, and a life well-lived is

one in which people maintain and use a set of fundamental talents. These talents include, but are not limited to, control over one's surroundings, emotions, physical well-being, and body integrity.⁶⁶ To maintain this justness, society must ensure that people's ability to develop their basic competencies is not hampered and that their ability to participate in politics is not hindered or damaged.⁶⁷

Despite their apparent similarities, the capabilities and well-being theories are essentially different. The primary distinction between the two theories is that capability theory emphasizes the capacities and possibilities for well-being, while well-being theory emphasizes well-being itself. The well-being theory's co-authors, Madison Powers and Ruth Faden claim that justice is concerned with achieving well-being rather than simply having the capacity to do so.⁶⁸ They list six fundamental components of well-being, including health, personal security, rationality, respect, attachment, and self-determination. According to these authors, justice ensures that each individual has a sufficient amount of each of these components.⁶⁹

2.A.ii.b. The Basic Concerns

Other ideas have been put out in addition to those already stated to address the distribution issue. Since they ignore one or more of distributive justice's fundamental concerns, most of these theories fall short in some manner. In reality, the significant elements that influence what is just in distribution are need and contribution.⁷⁰ A fair distribution of commodities is something that allocation, in and of itself, deals with. Distribution of these resources among the needy is seen as allocation since it leverages an individual's contribution to society and their resources to meet the needs of others. Therefore, a lack of anything deemed essential or obligatory may be used to define a need. You may have needs that are socially generated, subjective, or inherent.⁷¹ Regardless of a person's belief, there remains an inherent

need. For instance, everyone naturally needs a certain quantity of nutrients and hydration. An organism will weaken and finally perish if this inherent need is not met. In contrast to inherent needs, subjective needs arise from the thoughts and aspirations of the person.⁷² These wants may be unique to each person, as in the case of two star-crossed lovers who claim to "need" one another. Subjective requirements are often socially generated demands, where society decides what is required to establish a given social standing. For instance, when a teenager feels the urge to "need" the newest iPhone to fit in with their classmates.⁷³

People agree that certain goods are required for enjoyment, social interaction, or other objectives, which results in socially induced needs. This consensus is the product of the historical evolution of material and social circumstances.⁷⁴ It is evident that historically, socially induced needs have grown and have led to ever-increasing demands on people in society. Only under the real conditions of a specific society can the existence of these socially induced needs and their link to human dignity be recognized fully.⁷⁵ In other words, not everything that is required in one civilization is always required in another. As a result, there can never be a complete list of what is required since it is dynamic and ever-changing.⁷⁶

The next concept analyzed is contribution, which can be a bit confusing. However, the fundamental tenet of the concern for contribution is that, in the absence of contributions to society, there would be nothing to share since society cannot generate anything without the labor of its members.⁷⁷ This is why it has become vital for all communities to depend at least a portion of the distribution of commodities, including healthcare, on contributions from both organizations and individuals. Contribution is motivated by recognition, hence it is crucial to the operation of society.⁷⁸

It should be recognized that not all contributions are financial in nature, and not all rewards are monetary. People provide a wide range of political and social benefits that benefit society. For example, a family that raises good children makes a contribution of good citizens.⁷⁹ With its hospitals, health plans, educational perks, and other programs, the U.S. Department of Veterans Affairs honors people who have served in the armed forces. The continuing contributions in important societal sectors significantly affect a society's political and social health.⁸⁰ A fair and responsible distribution theory should consider all of these contributions.⁸¹

2.A.ii.c. Allocation

The allocation of scarce resources is an enormous and complex problem within the realm of distributive justice, and by no means does this analysis attempt to solve this huge issue. This section's main goal is to establish the idea of allocation before highlighting some of the major issues with doing so on both a micro and macro scale. David Kelly, Gerard Magill, and Henk ten Have summarized the idea in the following form: "it is morally wrong to spend X money on A while situation B exists, on which the X money ought to be spent instead."⁸² Analysis of this kind of argument reveals that it is mostly ineffective without supporting data about the advantages or futility of initially investing X dollars on A.⁸³

As with any allocation or economics discussion, concepts can be detailed on micro and macro levels (further detailed in Chapter 3), with each having its own set of nuanced problems that each present with the goal of just distribution. The consequence of society's efforts alone does not result in a balanced distribution.⁸⁴ The conversation must include the patient, healthcare professionals, and institutions. In other words, microallocation also impacts how health care is distributed and political and social considerations. Although society has authority over macro allocation, lay and professional people and organizations like nursing homes and hospitals

actively participate in micro allocation every day. For instance, the patient and the healthcare provider are doing triage or, at the very least, allocating their time and resources. Hospitals and other healthcare facilities must restrict treatment based on their resources and the patient's capacity to pay. Although the macroeconomic distribution of the society restricts their power, all of these groups substantially impact the health care options available to each person.⁸⁵

The allocation argument discussed earlier has a number of issues, one of which is that it is too loose and has to be more specific. To be clear, objective A (for which the funds should be redistributed) is open-ended. Everything is seen as subjectively less important than purpose B.⁸⁶ Americans, for instance, would agree to spend more on preventive care than we do today but cannot explain where those dollars should come from.⁸⁷

The shame associated with transferring money from purpose A to purpose B may lead to sentiments of desertion and guilt, which is another issue with the allocation argument.⁸⁸ Just like in palliative care, ceasing treatment does not mean stopping care – and similarly, just because some resources are reallocated away from A does not mean no care or attention should be paid to A. When resources are redistributed, the blame may recur like an open wound, creating an endless cycle of shame.⁸⁹

The moral significance of the divide between identifiable persons and ambiguous unidentified groupings is a third issue.⁹⁰ Is it ethically acceptable to divert funds from a specific identifiable group of orphan kids to a bigger, unnamed group that is in just as much need? The allocation argument has to be strengthened in order to make a stronger case that it may establish universal ethical rules of behavior. This is another complicated problem, like the ones before it, without a logical answer.⁹¹

While distributive justice has had a long history stretching back hundreds or even thousands of years, humanity has built a background for theories that provide a starting structure for understanding what distributive justice is and how it can work. Two main concerns are apparent in analyzing these theories: need and contribution. A large portion of distributive justice is concerned with allocating resources within a micro and macro level – while more problems continue to arise, fewer rational solutions exist.

2A.iii. Implementation Considerations

While the Affordable Care Act has significantly improved health care coverage in America, 31.1 million Americans remain uninsured as of mid-2021.⁹² This results from myriad problems with the current healthcare system, including inadequate access to healthcare, insufficient health insurance protection, rising costs, and an overall burdensome system for patients, their families, and physicians. This section explores alternative proposals for extending health insurance coverage while highlighting criteria that must be met for a better system and the four conditions that must be present. Following a European model, the probable answer to healthcare reform in the United States lies within a universal health insurance program. The advantages and disadvantages of this type of program are described while looking at other international healthcare program examples.

2.A.iii.a. Problems in the Present System

The United States' existing healthcare system has many issues; otherwise, the last several presidential administrations would not have made it a major part of their agendas. The Affordable Care Act, sometimes known as "Obamacare," has decreased the number of Americans without health insurance. However, it does not address many other healthcare issues that plague America. America still struggles with providing adequate access to healthcare to its

citizens, adequate insurance protection, suppressing rising costs of healthcare, and implementing a less burdensome system for all.

Access will be the first issue that has to be resolved since a lack of enough funding to provide almost universal access to necessary health care, along with a lack of resources like facilities and staff, leads to medical problems, unjustified suffering, and unnecessary agony.⁹³ If a diagnosis and treatment are not received in a timely manner, costs may be incurred.⁹⁴ As the expense of uncompensated treatment rises, this insufficient finance mechanism threatens a national system of providing healthcare. Lack of insurance has a large and direct impact on access to healthcare, with those who lack insurance far more likely than those who do to delay or forego medical treatments.⁹⁵ These unintended repercussions may have devastating effects, particularly when they include avoidable disorders or undiagnosed chronic diseases.⁹⁶

Studies show time and time again that people with insurance are more likely to get preventative treatment and assistance for serious health issues and chronic illnesses than those without insurance.⁹⁷ The Henry J. Kaiser Family Foundation states in their annual economic report "One in five (20%) nonelderly adults without coverage say that they went without care in the past year because of the cost compared to 3% of adults with private coverage and 8% of adults with public coverage."⁹⁸ Nearly half of those without insurance do not have a regular location to go when they are ill or in need of medical guidance, which is one of the causes of the poor access among the uninsured.⁹⁹

The next problem to be addressed is costs and quality, in which the US outpaces numerous countries in healthcare expenditures with no signs of slowing down. Unfortunately, these costs are not justified as the US population has poorer health than multiple countries that spend considerably less on healthcare.¹⁰⁰ After years of growth in life expectancy, certain

communities have seen a decline recently, a trend made worse by the opioid epidemic. Furthermore, as the baby-boomer generation ages, more individuals worldwide and in the United States are experiencing age-related diseases and disabilities, putting strain on healthcare infrastructure to deal with this growing population (further detailed in Chapter 6).¹⁰¹

The expenditures associated with documenting, invoicing, evaluating, processing, auditing, and justifying medical charges are among those of the health care costs that are growing the quickest.¹⁰² These costs include the price of insurance marketing, the earnings and cash on hand of public and private insurance providers, as well as additional overhead costs brought on by the way we now pay for healthcare.¹⁰³ The medical liability system and doctors' attempts to reduce their exposure to allegations of professional negligence result in significant extra expenditures. It is not unexpected that the quantity of professional liability claims, the magnitude of malpractice judgments, and the price of malpractice insurance are increasing.¹⁰⁴

The current systems for paying for healthcare, which include several public and commercial insurers as well as third-party payers, are difficult to understand, expensive, inefficient, and invasive.¹⁰⁵ A complicated system of documenting, invoicing, auditing, evaluating, and processing claims is required for health insurance coverage under commercial insurance plans and state programs like Medicare and Medicaid.¹⁰⁶ Most everyone engaged in US healthcare, including patients and physicians, are aware of the time- and money-consuming, tedious paperwork that goes along with it. People do not want to deal with stressful financial issues or paperwork during a private and delicate procedure.¹⁰⁷

2.A.iii.b. Considerations of Implementing a New System

It is crucial to take into account a number of factors while proposing or trying to adopt a new healthcare system in the United States in light of some of the issues with the country's

present healthcare system. It is unethical and unwise to implement any new system without having a set of standard criteria for evaluating proposals as well as the vote of the people. In addition to the criteria, the four conditions outlined by Norman Daniels must be present within the new system for accountability in distributive justice.

To evaluate proposals for a better health care system, it is imperative to identify and use a standard set of criteria to assess any proposals. The following criteria are not meant to be all-inclusive but serve as a beginning guideline for evaluative purposes. These criteria are broken down into four categories (benefits, financing, organization & delivery, satisfaction) and are not in any particular order of priority.

In terms of benefits, a method for figuring out the extent of benefits that would be gained by putting a new system into place should exist.¹⁰⁸ With this, there should be a standard minimum package of benefits for everyone — not just the really poor or wealthy. Clinical efficacy should be the basis for coverage choices; if a treatment is ineffective or just marginally beneficial, a more compelling argument and proportionality will be needed to justify coverage. Benefits and coverage need to be permanent and unaffected by where you live or work.¹⁰⁹

Financially, adequate money must be available to remove obstacles to receiving necessary treatment.¹¹⁰ Additionally, processes should be kept to a minimum, and measures for limiting costs, particularly administrative expenditures, should be in place. Any new finance structure should consider current income streams and aim to reduce professional liability expenses. Patients, doctors, and other healthcare workers should all feel satisfied regarding their work.¹¹¹

Facilities and staff must have the necessary infrastructure in terms of organization and delivery for optimal healthcare services to be provided quickly and effectively.¹¹² Additionally,

systems should be in place to guarantee quality in every element of healthcare. The system should be adaptable, and innovation and improvement should be encouraged rather than stifled. Finally, rewards should motivate people to take control of their health, seek preventive treatment, and engage in health promotion activities.¹¹³

Norman Daniels & James E. Sabin outline four critical conditions for implementing any new healthcare system in the US. They state, "Justice requires limits to care, and the lack of consensus on distribution principles means that society must develop a fair process for setting limits and learn how to apply that process in real-world situations."¹¹⁴ For such a method to successfully handle the legitimacy issue, they list four requirements. Decision-makers are responsible for the rationality of their judgments if these requirements are met.¹¹⁵

Limit-setting decisions must first be made public. Judgments must be made publicly, and so must the justifications for those decisions.¹¹⁶ Transparency is guaranteed by this publicity requirement. It is important to emphasize that it does not need that all decision-making factors be determined upon or openly agreed upon in advance. The proper institutions and authorities can create such measures as they encounter issues over time.¹¹⁷

Second, the justifications for the choices must be ones that realistic resource restrictions and fair-minded individuals think are relevant to equitably providing health care demands.¹¹⁸ This relevance requirement is essential because it centers the discussion of constraints on a common objective or benefit. In the American system and many governments in developing countries, a privatized, for-profit industry raises fundamental questions about what explanations shareholders may see as relevant and what data would be necessary to prove that relevance.¹¹⁹

Along with the first two conditions, limit-setting decisions must also be open to review and appeal.¹²⁰ Generally, judgments must be subject to revision in the face of new information, arguments, and debate. Lastly, there must be some regulation to fulfill the other conditions.¹²¹

2.A.iii.c. Universal Access to Basic Care

There are numerous proposals for extending health insurance protection among Americans, such as mandating employer coverage, creating health insurance risk pools, and expanding charity care. However, the most widely accepted proposal is establishing universal health insurance program access. Thus, the next section will be dedicated to showing how the rest of the world works in this type of program and some significant advantages and disadvantages of the system.

International Examples

Except for the United States, almost every industrialized country has national healthcare systems to provide universal access.¹²² These initiatives come in various formats, but they all have the same goal: enabling all individuals to access healthcare services without financial limitations. Coverage is generally consistent and universal (everyone is eligible regardless of health condition) (everyone is entitled to the same benefits). Costs may be covered entirely by tax revenues or by a combination of employer and individual premiums and public assistance.¹²³

A single, consistent program implemented nationwide, like in Sweden and France, is one format.¹²⁴ Another version, used in countries like Canada and Switzerland, establishes minimum benefits at the federal level while allowing programs to vary at the provincial or state level.¹²⁵ Allowing eligible, non-governmental programs to run locally, as is done in Germany, is another variant.¹²⁶ Another alternative is to have local councils manage a national program, as in Belgium and the Netherlands.¹²⁷

Advantages & Disadvantages

With any new healthcare system comes a series of advantages and disadvantages; thus, all things must be considered when evaluating whether to adopt a new approach. The way some of the main advantages and disadvantages will be presented are in a coupled form – as in the advantage (or pro) with its accompanying disadvantage (or con) will be listed together. In no way does this mean to be an exhaustive list, but some of the main points and positions to consider in a universal health insurance program.

All people would have access to specified health insurance benefits that, at the very least, cover basic medical services, which is the first major benefit. Private insurance might act as a safety net to augment coverage, giving people access to conventional medical treatment.¹²⁸ The associated drawback is that significant change, including reorganization of healthcare finance methods and programs, would be necessary. Due to increased access through universal coverage and benefit extension, healthcare expenses may grow, adding to the need to keep prices down.¹²⁹

The ability to receive consistent benefits at cheaper premium rates per dollar of coverage than is achievable under a variety of plans with different benefits and coverage is another major benefit.¹³⁰ Costs for invoicing, processing claims, marketing, reserves, and earnings might be decreased or eliminated to provide further savings. However, more governmental engagement in healthcare and tighter restrictions on volume, usage, expenditures, and quality assessments might lead to additional erosions of physician autonomy.¹³¹ Additionally, limiting medical decision-making, patient autonomy, and the accessibility of these treatments would be more directly hampered by government choices to restrict or forego funding for specific operations.¹³²

Ultimately, such a system's potential benefits outweigh the burdens, making the switch to universal access to basic care the key to bringing America back into the race for the best

healthcare system in the world. While the transition may be rough and certainly will be met with opposition, such a system must prevail if ethical, quality and equitable care are given to the exponentially increasing population of the United States. This reform will be more significant than the change in Medicare/Medicaid and the Affordable Care Act combined – it will be the most significant change in healthcare in the history of the United States. Universal access to basic care follows the basic principles of distributive justice and allocation while focusing on the needs and contributions of society. This system can meet the four conditions that Daniels mentioned earlier, follows international examples, and all the criteria for a better system. The citizens of the United States must band together to create such a system and urge our government representatives to do the same. This is the chance to save private America.

2.A. Conclusion

The current state of healthcare in America could be summed up as sub-optimal. Millions of American citizens remain uninsured, while others can't have the proper access to basic care that is needed. Suppose the history of healthcare reform in America hints at what is to come. In that case, a more financially thorough investment of the US government in healthcare will have a tough time passing through the legislative branch. However, this does not mean that universal access to basic healthcare is impossible – as this kind of reform is possible. Taking the critical principles of distributive justice, America can turn its ugly healthcare system into something truly remarkable. By following the criteria for a better system, the four conditions previously mentioned, and the basic concerns of distributive justice – a new healthcare system can be devised that benefits all American citizens. This is the only proper way America will be able to regain a place at the top of the worldwide healthcare leadership board and is the answer to saving private America.

The healthcare problem in America remains apparent as the Affordable Care Act has left a considerable problem to be solved – how to provide healthcare to everyone. Many complications exist within the current system, such as inadequate healthcare access, insurance protection, rising costs, and an unnecessarily burdensome system for all parties involved. Looking forward to proposals for a new healthcare system, specific criteria and conditions need to be addressed if not met by implementing a new system. A universal health insurance program may be the answer, but it comes with its own set of problems that must be considered. Many of these healthcare problems can be mitigated using data analytics, which can reduce costs, provide more personalized care access, lead to more accurate diagnostics, expedite care, and provide numerous other benefits for all—the following section details data analytics' current and future impacts on the healthcare landscape.

2.B. The Ethical Impact of Data Analytics in Healthcare

2.B. Introduction

Over the last approximately 20 years, healthcare data's digitization and electronic use have exponentially increased. With the emergence of the electronic medical record (EMR), health information systems (HIS), smart devices, wearables, and so much more, healthcare data has moved into the digital world. Data analytics is the pursuit of taking all of this data (healthcare or otherwise) and converting it to some form of meaning and interpretation on a large scale through advanced computer systems. These systems transform, organize, and model the data in numerous valuable and informative ways. As with any new invention, data analytics poses opportunities and challenges that must be met as society progresses. Furthermore, data analytics expands personalized care through predictive modeling and integrative sequencing opening new realms to the possibilities of medicine. Data analytics also introduces

organizational implications for health systems such that it has and will continue to drastically alter the delivery and finance of healthcare at the same time as respecting an obligation for security, privacy, and confidentiality with patient population data.

2.B.i. Challenge & Opportunity in Data Analytics

Numerous sectors have seen fast digitization during the last ten years. Mehta et al. state, "Healthcare has also undergone this digital transformation, with an increase in the use of electronic medical records (EMRs), healthcare information systems (HIS), and handheld, wearable, and smart devices."¹³³ As a consequence, a vast array of health-related data, including genetic data, sociodemographic data, insurance claims, and clinical data, to mention a few, is now available in digital form.¹³⁴ The immense potential for improving healthcare delivery presented by this high-quality healthcare data necessitates transforming the raw data into information that can be used. This is where big data analytics steps in – and with it come many opportunities and challenges that need to be discussed. Some of the significant challenges posed to data analytics are currently by the General Data Protection Regulation (GDPR) and Fair Information Practices, simultaneously posing a new challenge of an autonomy paradigm that must be addressed to progress this field further. On the other hand, an opportunity is significant in data analytics with the ability to provide real-time analytics via cloud computing methods and improve healthcare treatment methods.

2.B.i.a. Challenge

GDPR/EU Charter Fundamental Rights/Fair Information Practices

When it comes to big data analytics, the challenges and opportunities are endless – as technology and industry continue to evolve, so will the problems and possibilities that come with it. Some of the most prominent challenges to expanding data analytics came with the recent

implementation of the GDPR and the existing Fair Information Practices; while more control over privacy and consent isn't a bad thing, current constructors of the data analytics models and networks will have to abide by and adapt to these new protocols which in turn, slows progress.

First, a quick recap of what the European Union (EU) did when it passed the GDPR starting on May 25, 2018. It puts privacy and security requirements on all companies, regardless of location, so long as they target or gather information about individuals in the EU's 27 member nations, mostly in western Europe.¹³⁵ The project updates the Data Protection Directive 95/46/EC, a previous data privacy effort that was first established in 1995.¹³⁶ An express approval from the client is now required under GDPR before a business may use, keep, or analyze their personal information, as opposed to the previous opt-out approach. Nancy Davis Kho explains that GDPR added many types of information previously outside the scope of personal information (PI), such as genetic data, race, religion, and more, to be protected under the law.¹³⁷ According to the GDPR, businesses are only allowed to collect the information necessary to function. In this case, the concept of data minimization is at play, pushing corporations to stop gathering additional personal data for other conceivable future projects.¹³⁸ The GDPR has given consumers far more control over their data. They now have the right to request information about the specific data that businesses have gathered and saved about them. Additionally, customers have the "right to be forgotten" to ask that the corporation delete their personal information.¹³⁹

Contrary to what many people seem to think, the GDPR does not just apply to European enterprises; it casts a broad net. As previously noted, no matter where a company is physically located, it must adhere to the GDPR if it utilizes the personal information of EU people in any manner.¹⁴⁰ Companies have been scrambling to catch up; a rather bleak 2015 survey from Ovum

found that "63% of U.S. businesses expected the GDPR to make it harder for them to compete, while 70% felt that the GDPR would throw the competitive advantage to EU businesses."¹⁴¹ Businesses have seen greater expenses, process changes, and risk since its deployment.¹⁴² Every employee's day-to-day responsibilities have been affected, from contact center agents who now start every interaction with a privacy warning to software developers who must now build in privacy controls.¹⁴³ The consumer is the final group to discuss. While it may seem like today's online customer journeys, involve endless website clicking in addition to the customary cookies notice, the impact on consumers has also meant that they have significantly been empowered with a better understanding of their data and rights.¹⁴⁴ The GDPR has consequences, which is the final significant truth about it; in accordance with the regulation, businesses that do not fully comply "can be fined up to 4% of annual global turnover or €20 million (about \$24 million)— whichever is higher."¹⁴⁵

The Fair Information Practice Principles (FIPPs), which were initially published in 1973 as the Code of Fair Information Practices and served as the foundation for the United States federal Privacy Act of 1974, were an earlier US effort to develop anything like GDPR.¹⁴⁶ The *Nationwide Privacy and Security Framework for Electronic Exchange of Individually Identifiable Health Information* was recently established by the U.S. Department of Health and Human Services (HHS) and outlined these guidelines.¹⁴⁷ These guidelines have been approved by the White House as part of a Consumer Privacy Bill of Rights that will serve as a standard of conduct for companies doing online commerce.¹⁴⁸ These guidelines provide a strong foundation for data exchange while upholding trust. The HIPAA Privacy Rule or other relevant legislation, which has prompted the establishment of governance processes to apply the concept, is part of the FIPPs included in the HHS advice.¹⁴⁹ Each of these guidelines has been accepted by the HHS

as direction for healthcare entities sharing personally identifiable health information electronically. Each principle is followed by an explanation, a set of examples, and, if appropriate, examples of how the principle has been applied to other areas of law.¹⁵⁰

The United States Fair Trade Commission issued rules for fair information practices more than 40 years ago, noting, "Computers linked together through high-speed telecommunications networks are destined to become the primary medium for storing and using records about people. Innovations now being discussed throughout government and private industry recognize that the computer-based record-keeping system if properly used, can be a powerful management tool."¹⁵¹ Fair information practices include clauses requiring notification or knowledge so that data collectors, uses, and users may be pinpointed.¹⁵² The ability to opt-in or opt-out of certain applications should thus, in theory, be available to data providers. Establishing and maintaining confidence in the individuals whose data is being gathered and utilized is the primary driver for this law and any follow-up legislation, even if it is being done for their benefit.¹⁵³ Besides, the new paradigm of autonomy within data collection and aggregation poses a severe threat – people must have explicit and informed consent to make a self-determining choice. Any attempt to confuse or overload that choice is a limit to autonomy.¹⁵⁴

Autonomy Paradigm

Respecting people's right to self-determination is a critical component of autonomy. Autonomy is the indicator of how much the results of your process respect or obstruct people's right to self-determination or the capacity to make their own decisions concerning ethical information management.¹⁵⁵ This measure is determined by the degree to which individuals express their preferences for processing and how transparently choices are allowed or suppressed by the system or process design.¹⁵⁶ An information imbalance when someone "agrees" to

something they are not aware they are doing limits their autonomy, invalidates their decision, and might also be seen as a flaw in the verity/non-deception dimension.¹⁵⁷ Similar to information overload, which makes it difficult for individuals to grasp their options, obfuscation may likewise limit autonomy.¹⁵⁸

In a broader sense, processing that limits an individual's ability to choose—for instance, by withholding information about the product, service, or another benefit from them based on the processing of data about them—or that restricts their ability to exercise other rights or freedoms, would also have an effect on autonomy.¹⁵⁹ This quality is crucial for providing clear, transparent communication that can help individuals make decisions and comprehend the decisions that may be made on their behalf due to your information processing.¹⁶⁰ The conflict between this duty to preserve data (and respect the autonomy of persons to whom the data relate) and the obligation to contribute to the common good, particularly when doing so involves no effort at all, is at the very heart of our efforts to strike a balance (and so respect the positive human inclination to solidarity, to help others).¹⁶¹ We all value and even rely on our autonomy, and maintaining it fosters trust. However, we also depend on public health services, which rely on cooperation. Therefore, we must exercise caution should the forceful use of a solidarity trump card undermine that confidence.¹⁶²

Fortunately, most public health organizations in the West have the confidence of the people they serve. How else can the computational upkeep of vital statistics, illness registries, vaccination records, poison-control data sets, many health monitoring initiatives, etc., be justified without real opposition? Most of us rely on these services and think they will be there, often without moral concern about losing any of our autonomy.¹⁶³ We have no preference for creating and researching these archives utilizing information technology (as if any of them could be

maintained on paper). The enormous success of the public health effort, which has enhanced, if not saved, millions, if not billions, of lives, offers a strong utilitarian justification for the kind of "presumed solidarity" being proposed here.¹⁶⁴ Nevertheless, an opportunity is abundant in data analytics with the more recent implementation of cloud computing services, providing efficient and manageable storage and retrieval mechanism for large amounts of data. This has to do with better treatment strategies that might profoundly change how healthcare is delivered. These techniques provide breakthroughs on all biological dimensions, including sequencing tools, health monitoring devices, imaging collections, and increasing electronic health record systems, all while bringing down the cost of data capture.¹⁶⁵

2.B.i.b. Opportunity: Real-Time Analytics

Cloud Computing

Insight acquired via data analytics through the analysis of vast and complicated data sets is becoming a driving force behind advancements in biomedical research and healthcare delivery. Real-time analytics is possible in healthcare and other fields as more and more hypotheses may be created and tested utilizing high-throughput technologies. Cloud computing has recently evolved as a robust, adaptable, and highly efficient responses to a variety of computational and information-laden problems, made possible by the widespread use of the internet and its expanding capabilities and market demand.¹⁶⁶ The National Institute of Standards and Technology (NIST) describes cloud computing as "a model for enabling convenient, on-demand network access to a shared pool of configurable computing resources (e.g., networks, servers, storage applications, and services) that can be rapidly provisioned and released with minimal management effort or service provider interaction."¹⁶⁷

Due to varying degrees of security and other operational requirements, cloud adoption, regardless of technology, has been uneven across the sector. Public and private clouds have historically been used less often by regulated businesses and more commonly in unrestrained environments, although this is starting to change.¹⁶⁸ Public clouds are being used to open scientific data sets financed by the government. An example provided by Navale & Bourne, "Human Microbiome Project (HMP) data, funded by the National Institutes of Health (NIH), is available on Amazon Web Services(AWS) simple storage service, and more biomedical data sets are becoming available in the cloud."¹⁶⁹ Research scientists may currently seek NIH clearance to retain and study restricted "genomic and associated phenotypic data obtained from NIH-designated data repositories."¹⁷⁰ The NIH is dedicated to requiring public accessibility to digital content for all NIH-funded studies, with the appropriate access limitations for data pertaining to people.¹⁷¹

Chaowei Yang, Yan Xu, and Douglas Nebert note that "Cloud computing has emerged as a new paradigm to provide computing as a utility service with five advantageous characteristics: a) rapid and elastic provisioning computing power; b) pooled computing power to better utilize and share resources; c) broadband access for fast communication; d) on-demand access for computing as utility services and e) pay-as-you-go for the parts used without a significant upfront cost like that of traditional computing resources."¹⁷² In cloud computing, a serviceoriented architecture is used. It makes everything as a service, possible, including "Platform as a Service (PaaS), Software as a Service (SaaS), and Infrastructure as a Service (IaaS)."¹⁷³ Cloud computing with big data opens up new opportunities for geospatial research and Digital Earth while illuminating possible answers to problems with large amounts of geographical data in several geosciences and related domains.¹⁷⁴

Complicated studies of healthcare, such as predictions of protein-ligand docking and assessments of human activity, may be completed more quickly using a cloud infrastructure.¹⁷⁵ In order to reduce the costs of storing, accessing, and transmission, personal health data may be centrally housed in a cloud computing system. However, this also creates some new security threats for healthcare information. Typically, cloud computing-based medical systems use task scheduling and data monitoring as techniques to increase their efficiency.¹⁷⁶ Reducing the amount of analysis done on a portable device is a fascinating research topic that involves "offloading" the complex computing to a cloud, from mobile devices like PDAs.¹⁷⁷ A cloud infrastructure provides a helpful answer to preserving the energy of mobile devices, portable equipment, and smart wearable technologies, for instance, by communicating the implementation binary code of a cellphone or tablet to a cloud service at system startup.¹⁷⁸

Improved Treatment Methods

One example of improved treatment methods is real-time health tracking for chronic illness patients who reside distant from their healthcare providers. Due to distance, real-time health monitoring solves the issue of physicians finding it challenging to keep track of their patient's health state. One striking example is patients with heart arrhythmias that need ongoing episode identification and surveillance.¹⁷⁹ Wearable sensors provide real-time electrocardiogram (ECG) monitoring, as well as the categorization and diagnosis of arrhythmia episodes.¹⁸⁰ Navale & Bourne provide another example, "using AWS EC2, mobile computing technologies were integrated, and ECG monitoring capabilities were demonstrated for recording, analyzing, and visually displaying data from patients at remote locations."¹⁸¹

Additionally, cloud SaaS made software applications that tracked and analyzed ECG data accessible to the general public. Using the Microsoft Azure platform, a 12-lead ECG

telemedicine service is now available.¹⁸² To store and retrieve medical pictures, components of the Picture Archive and Communication System were put up on a public cloud. An examination of articles about cloud computing in the medical field revealed that, in contrast to the definitions of cloud computing, platforms, and applications, the terms "cloud" and "virtual machines" or "web-based tools" are often used interchangeably.¹⁸³ Through cloud computing, corporate vendors work with healthcare institutions and suppliers to provide healthcare services.¹⁸⁴ Another example is a system developed for ICU patients that combines static and real-time data and might alert caregivers to critical occurrences using cloud computing.¹⁸⁵ Similarly, a neonatal ICU-specific architecture incorporates information from several technological support systems to help with caregiver assistance in real time.¹⁸⁶

When it comes to big data analytics, the challenges and opportunities are endless – as technology and industry continue to evolve, so will the problems and possibilities that come with it. Some of the most prominent challenges to expanding data analytics went with the recent implementation of the GDPR and the existing Fair Information Practices; while more control over privacy and consent isn't a bad thing, current constructors of the data analytics models and networks will have to abide by and adapt to these new protocols which in turn, slows progress. Besides, the new paradigm of autonomy within data collection and aggregation poses a severe threat – people must have explicit and informed consent to make a self-determining choice, and any attempt to confuse or overload that choice is a limit to autonomy. Nevertheless, an opportunity is abundant in data analytics with the more recent implementation of cloud computing services, providing efficient and manageable storage and retrieval mechanism for large amounts of data.

2.B.ii. Personalized Care

Data analytics opens numerous opportunities and challenges in personalized care – from predictive modeling to integrative sequencing techniques, the ability to tailor care and treatment to each individual quickly becomes a reality. Advances in predictive modeling through algorithms and machine learning will help cut medical costs, evaluate health risks, and improve the speed and delivery of healthcare. This can be implemented in clinical decision support systems that use data analytics in real-time to assist healthcare professionals and patients in making the most informed decisions possible. Integrative sequencing has taken genomic medicine to a new level with the ability to analyze the patient's genotype data, determine how the disease affects this patient specifically, and then select the therapy and/or medication based on the data. The P4 initiative takes these goals and devises ways to apply them within genomic medicine in the real world. With the full availability of direct-to-consumer genomic testing, several positive and negative occurrences are discussed in this section– everything from empowering individuals to make lifestyle changes based on their genomic profile to privacy and security concerns governing the handling and storage of genetic data.

2.B.ii.a. Predictive Modeling (Algorithms)

One particular data analytics method that forecasts future medical expenditures is predictive modeling. It functions by forecasting future health risks and issues and projecting future medical use for the patient and the hospital using a patient's current medical information.¹⁸⁷ Patients are given a particular risk level or score using one of the many available predictive modeling techniques.¹⁸⁸ Risk markers are later used to record these values for each patient in certain situations. Predictive modeling estimates projected patient expenses using personal health data from the EHR, including demographics, prior diagnoses, and other pertinent

data. This information is important to physicians and insurers. As a consequence, patients who need special care and treatment are given priority.¹⁸⁹ A fraction of a given population may use the majority of healthcare expenses, according to certain forecasting models. These groups often need more time, resources, and attention, which raises the organization's total expenses.¹⁹⁰ *Algorithms & Machine Learning*

Although not a new issue, it is becoming more common to improve healthcare or health programs research initiatives by using prediction models and machine learning algorithms. There is an expanding amount of research on non-hypothesis-driven analytical techniques for finding patterns in the rising mass of clinical and healthcare data.¹⁹¹ Recent patient health studies have used these strategies in various diverse contexts, for example, "to identify high-cost patients, aid in the reconciliation of potentially erroneous entries in medication lists in patient's medical records, and identify patients at high risk for developing specific clinical conditions."¹⁹²

It would be preferable if these analytical techniques were developed and used to help population health research and monitoring "identify patient cohorts and enhance our understanding of the phenotypic representation of clinical concepts in data."¹⁹³ To more clearly comprehend and identify trends among potential determinants and response variables relating to clinical findings or health outcomes of interest (HOIs), machine learning and other data-driven predictive modeling methods are being created and used inside the distributed data network.¹⁹⁴ In reality, because distributed data networks are by definition made up of several distinct data sources, data-driven simulation techniques are essential in these systems.¹⁹⁵ At one location, an algorithm could be deemed ideal, but not necessarily at another. As a result, it is expected that academics will get a better understanding of the underlying data using predictive modeling

methodologies that draw on and modify the data upon which they are based in order to learn from it.¹⁹⁶

At one level, the physical infrastructure of a hospital, its personnel, its medical staff, its administrative staff, its diagnostic equipment, and its information systems are examples of relatively self-contained healthcare system components. At another level, perhaps the overall delivery of healthcare in a city or county, these components are parts of a more extensive healthcare system.¹⁹⁷ These things or organizations judge resource allocation and process redesign backed by simulation and predictive modeling. Due to significant improvements in computer power and other technical advancements, complex modeling is now viable for use in real-world contexts. The use of animation in modeling to enhance strategic decision-making has made it possible for decision-makers to comprehend better possibilities or less costly alternatives without having to deal with the intricacies and technical challenges of extensive calculations.¹⁹⁸ These predictive models and simulations are the go-to method for improving business, applications, and designs throughout the whole enterprise, and their usage and popularity are growing.¹⁹⁹

Clinical Decision Support Systems

A clinical decision support system (CDSS), which a clinical computer program, enables users to connect to a level of understanding, patient-specific data, and relevant details at the point of treatment to enhance healthcare administration as well as service.²⁰⁰ Clinical decision-makers, including pharmacists, medical technologists, and other health professionals and doctors and nurses, use CDSSs. A CDSS is useful because it delivers suggestions and alternatives for medicine, therapy, and diagnosis that are tailored to the particular patient. The effectiveness of care delivery, patient safety, and healthcare quality are all improved by this capacity.²⁰¹

Additionally, CDSS enables quick access to medical data for professionals. Furthermore, CDSS helps minimize hospital admission costs and drug-related emergencies caused by interactions between various drugs and allergies.²⁰² To fully grasp the range of advantages offered by CDSS, more investigation is required. It is important to emphasize that these systems do not engage in healthcare decisions; instead, they offer pertinent information assessments that help decision-makers, such as physicians, patients, and healthcare organizations, make better decisions.²⁰³

According to Musen, Middleton, and Greenes in *Biomedical Informatics*, in an ideal world, these five appropriate activities may be used to define CDSS systems: they "provide the right information, to the right person, in the right format, through the right channel, at the right point in the workflow to improve health and health care decisions and outcomes."²⁰⁴ They go on to say that there are three main types of CDSS systems: (1) "they may use information about the current clinical context to retrieve highly relevant online documents, as with so-called info buttons; (2) they may provide patient-specific, situation-specific alerts, reminders, physician order sets, or other recommendations for direct action; or (3) they may organize and present information in a way that facilitates problem-solving and decision-making, as in dashboards, graphical displays, documentation templates, structured reports, and order sets."²⁰⁵ An illustration of the latter is order sets, which may improve productivity by offering a rapid way to choose a group of relevant operations and assist decision-making by serving as mnemonic devices.²⁰⁶ Many observers see information sources that summarize the medical literature and allow people to choose information that is relevant to the situation as straightforward decision support systems.²⁰⁷

It is critical to remember the underlying truth that there will always be trade-offs when adopting any computer-based system while creating a system for clinical data. One side of this is

the desire to employ the greatest technology available, and the other is the need for the system to take the users' information requirements into account and know how the system functions. A universal framework that applies to all circumstances does not exist either.²⁰⁸ Systems must be customized to the organization's unique therapeutic goals and services and the requirements of doctors, employees, and patients. Clinical information systems are being implemented with a lot of effort and money. However, there are certain drawbacks. For many healthcare institutions, implementation costs might be a hurdle. Despite all the attention that privacy and security-related concerns are receiving right now, difficulties and challenges still exist.²⁰⁹ Additionally, many physicians claim they do not have enough time to use the computer-based system during patient interactions. In general, however, the systems are working well and improving healthcare in various ways.²¹⁰

2.B.ii.b. Integrative Sequencing

Utilizing DNA to identify genetic risk factors for illness and improve disease diagnosis is a key objective of genomic medicine.²¹¹ Gene therapy and the development of medications based on molecular data will result from this. To create specifically tailored drugs for each patient, researchers are looking into how to connect DNA variations with those reactions. The term for this is personalized medicine. Ana & Donald Cleveland explain, "The concept of personalized medicine is to analyze the patient's genotype data, determine how the disease affects specifically this patient, and then select the therapy and/or medication based on the data."²¹² The results of the DNA study do not determine the diagnosis or course of therapy. Nevertheless, the analysis is included in the standard clinical workup, including the medical history findings, blood tests, and imaging scans.²¹³ Personalized medicine seeks to deliver the greatest treatment possible in a very customized manner, improve the precision of the suggested diagnosis and treatment, and provide speedy and effective care.²¹⁴

P4 Initiative

With the capacity to examine the genotype data of the patient, ascertain how the illness specifically affects this patient, and then choose the treatment and/or medication based on the data, integrative sequencing has elevated genomic medicine to a new level. The P4 project takes these objectives and finds practical methods to use them in genomic medicine.²¹⁵ The cost of decoding the human genome, which contains 30–35,000 genes, has quickly decreased as better and more efficient sequencing technology has been developed.²¹⁶ Computational biology has a big issue in quickly producing meaningful suggestions from genomic skill data analysis, which has ramifications for existing public health policy and healthcare delivery.²¹⁷ When providing information and advice to patients and caregivers in the healthcare context, time and money are always essential. "The P4 paradigm, which is the predictive, preventative, participatory, and personalized health initiative, seeks to track these issues over 100,000 subjects in a 20–30-year time, creating personal genomics profiles for each subject."²¹⁸

The P4 initiative aims to accomplish a number of objectives, including analyzing genomic data to identify disease conditions, making blood-based monitoring tools the standard for subjects, developing new approaches to achieving drug targets, and improving and more effective modeling for each subject.²¹⁹ Based on physiological monitoring and genome sequencing, the personal genomics profile, also known as the integrative personal omics profile or iPOP, generates thorough health and illness data for a subject.²²⁰ Delivering practical advice in a hectic clinical atmosphere is still difficult. Big data analytics must fill the gap between the vast volumes of data and trustworthy suggestions and results.²²¹

Genomic Health

Direct-to-consumer genomic testing now has a new market thanks to the development of low-cost single nucleotide polymorphism (SNP) profiling tools. Customers may submit a saliva sample to a number of genomic testing businesses (including 23andMe, deCodeMe, Navigenics, etc.) to obtain an online report outlining their hereditary disease markers and risks.²²² Statistical methods are used to compare the patient's SNP profile with profiles of a broader cohort and then summarize the relative risk for various conditions, such as heart disease and restless leg syndrome.²²³ Some studies include recommendations for preventive actions, however, it is widely believed that most risk evaluations are not actionable.²²⁴ Few individuals in a genome analysis modified their lifestyle as a result of the testing, according to recent research.²²⁵

The personal health record's integration of these findings will provide a substantial informatics opportunity. Personal health record providers will have to choose whether to keep the raw genetic data in addition to the interpretations, which may be updated over time (storage intensive).²²⁶ It will also provide difficult system design and policy concerns when deciding whether the information in the PHR should be disclosed to providers and under what conditions via interoperability efforts supported by the Meaningful Use part of the 2009 American Recovery and Reinvestment Act.²²⁷ Patients' access to genetic information may encourage them to participate in the clinical process as co-producers by alerting them to potential health hazards that may affect them and their families.²²⁸

Health databases owned by the public, commercial, and governments increasingly include personal genetic data. The databases could be used for clinical work, epidemiological studies, pharmacological research, etc.²²⁹ Guidelines for using genetic data in these databases are

being developed by the World Medical Association (WMA).²³⁰ According to the WMA Chair Anders Milton,

The public is rightly concerned about whether their right to privacy and confidentiality is threatened by these databases and whether information about them as individuals could be misused. Centralized health databases can make a tremendous contribution to the improvement of health. But the public's right to privacy and consent are essential to the trust and integrity of the patient/physician relationship, and we must ensure that these rights are properly protected. Any guidelines must address the issues of privacy, consent, individual access, and accountability.²³¹

Personalized care and medicine are being enhanced in numerous ways by data analytics.

Predictive modeling through algorithms and machine learning has created new and dynamic connections in medicine that can improve healthcare delivery worldwide. The predictive models will become more advanced and able to tell healthcare more as time progress with technological advancement. The goal is that one day these models will be able to forecast and see future health needs and public healthcare trends and modify and adapt the resources to address those trends.

Clinical decision support systems are one product of predictive models and machine learning that can assist healthcare professionals and patients make real-time decisions. Integrative genome sequencing has revolutionized personal care – the cost, time of sequencing, and availability of sequencing techniques have opened new avenues for care that is tailor-made for individuals and how their genetics play a role in their diagnosis, treatment, and effectiveness of medications. Overall, data analytics may transform the forefront of personalized care and medicine for all by reducing costs making healthcare more accessible for the poor & immigrants, providing more accurate diagnostics, personalizing results, increasing service, and expediting the decision-making process.

2.B.iii. Organizational Implications for Health Systems

Organizations continue to be significantly impacted by data analytics and its unique implications for their health systems. Information systems have a long-term effect on the future

of healthcare organizations, influencing everything from delivery and financing to data management and organizational changes. By lowering costs overall by decreasing labor, waste, and inefficient supply utilization, data analytics has a significant positive impact on cost and quality. Enhancing documentation accuracy and thoroughness, shortening the time spent recording, and many other factors improve the quality of healthcare delivery. Health plan membership, eligibility, benefits, claims and reporting, and payment structure have all become standardized because health information systems have developed. From a different angle, good data management and the organizational changes that follow from such control make data analytics necessary for population data. Good data management is vital to stop the slippery slope of unethical data and information storage, usage, and retrieval. The confidentiality and privacy of patients may be seriously jeopardized in the absence of a standard or code of information ethics.

2.B.iii.a. Integrated Health Systems: Finance & Delivery

Cost & Quality

Adoption of healthcare information systems has been significantly influenced by their ability to reduce data management expenses in hospitals and other businesses. To enhance departmental and administrative processes, businesses in the healthcare industry continue to make strategic partnerships in information technology.²³² Reduced labor expenses, less waste (such as out-of-date surgical supplies bought but never used or food trays delivered to the incorrect location and discarded), and more effective administration of supplies and other inventory are the main advantages that offset certain information systems costs.²³³ By effectively allocating priceless resources like operating rooms and imaging equipment, significant savings may be realized. Additionally, health information systems may aid in preventing accidental duplication of tests and treatments.²³⁴ Information technology may lower the expenses of

maintaining, obtaining, and transferring charts in the medical records department if critical patient data are made accessible online.²³⁵

The capacity to raise the standard of care has emerged as an additional advantage as the reach of healthcare information systems has expanded to include clinical process support.²³⁶ Clinical documentation is more accurate and thorough, doctors spend less time documenting (and therefore spend more time with patients), fewer medication mistakes and a speedier reaction to unfavorable incidents and management communications.²³⁷ Healthcare companies may increase the delivery of specialized treatment to distant and rural locations by using telemedicine and remote links to broaden their geographic reach.²³⁸ After converting patient data from a simple transactional model to one more appropriate for data analysis, clinical decision support systems and a health information system with a clinical focus may provide significant benefits, such as boosting treatment quality while cutting costs.²³⁹

Economic analyses may help healthcare decision-makers allocate limited resources more effectively to enhance health outcomes. To assess the value proposition of ground-breaking health technology, decision-makers often utilize cost-effectiveness (CE) and cost-utility (CU) analyses.²⁴⁰ Health outcomes are used in CE research to allow for cost-burden analyses, while quality-adjusted life years are used in CU studies.²⁴¹ In addition to the incremental cost-effectiveness ratio (ICER), which uses a math ratio to help determine the cost-effectiveness of an intervention - very similar to risk-benefit ratios used in bioethics arguments - these are crucial when determining whether and how different HITs can be used to allocate scarce resources.²⁴² The ICER is often used by healthcare decision-makers to support expenses associated with adopting and implementing innovative health technologies.²⁴³

Health information technology (HIT) can increase the efficiency of the healthcare industry as it continues to influence medical expenses and quality. Many politicians are turning to HIT as a crucial instrument to enhance the efficiency of the healthcare sector by eliminating medical mistakes, reducing needless testing, and improving health outcomes in the face of a continually high increase in health expenditure.²⁴⁴ According to the RAND Institute, HIT will lead to a \$142-\$371 billion annual decrease in health costs.²⁴⁵ The HIT implementation may affect the level and scope of care given.²⁴⁶

First, using electronic medical records (EMRs) may reduce the time and energy needed for a doctor to advise a comprehensive check, which might lead to a more intensive course of treatment.²⁴⁷ Second, EMRs may enhance provider-to-provider communication, boosting reliance on experts and decreasing the need for duplicate testing.²⁴⁸ Finally, clinical decision support systems (CDSS) may improve routine care by prompting clinicians and reducing medical mistakes.²⁴⁹ However, it is unclear how these three channels will ultimately affect overall medical costs, health outcomes, and the standard of treatment.²⁵⁰

Payment Reform

Healthcare funding has suffered from a lack of uniform terminology that standardizes health plan membership, eligibility, benefits, claims and reporting, and payment structure, similar to how clinical services have.²⁵¹ As a competitive tactic, the insurance sector has maintained stark discrepancies. The federal government actively promotes the health insurance sector's transition from paper to electronic transactions and has established guidelines for this.²⁵² For hospitals, healthcare organizations, clinics, and individual practitioners, managing reimbursement and the revenue process has proven challenging and expensive (in terms of

processing claims and days in accounts receivable).²⁵³ Patients and families have experienced a great deal of uncertainty and frustration as a result.²⁵⁴

The adoption of standardized terminology and the conversion of data to digital format provides the chance to solve a crucial financial problem that has not been well addressed: the disconnect between payers, patients, and healthcare providers.²⁵⁵ The financial performance will probably be enhanced by this link, which will also align it with the clinical function. The existing business and clinical activities must be disrupted by such revolutionary change, and there is always the risk of just automating the system. This revolution will be ushered in by new business models.²⁵⁶

Medical banking has just made its way onto the scene. It is an electronic system for processing and monitoring financial activities, including payments, reconciliation of accounts, investment account balances, invoicing and collections, and making payments.²⁵⁷ Medical banking aims to handle billing and receivables, ensure that billable services were successfully done, and connect financial transactions with data on patient status and treatment results. By integrating financial data, handling collections, and offering consumers a gateway for managing transactions, banks have a significant potential benefit to the healthcare system.²⁵⁸

2.B.iii.b. Integrated Health Systems' Obligation for Population Data

The obligation to protect population patient data must be tackled at the organizational level, and subsequent changes must be made and met to ensure ethical compliance. Organizational structures of industrial and service businesses have been impacted by information technology. Automating back-office tasks like payroll was a common early application of computers. This kind of computer use necessitated no organizational adjustments on the part of the business. Later, businesses started using computers in production facilities. Companies could
tailor their goods and provide their clients with greater service thanks to computers. By encouraging the decentralization of sales and support services and distributing greater responsibility to line employees, this usage of computers helped organizations become less bureaucratic.²⁵⁹ With the development of computer networks connecting various areas of the company, information technology inside businesses entered its third stage. For instance, businesses can now purchase replacements instantly thanks to the integration of cash registers and inventory systems.²⁶⁰

The introduction of information technology has a flattening impact on organizational hierarchies. Most information flows followed the lines on organizational charts when memos with carbon copies were the main source of information sharing. A broad range of modern technologies enables easy and inexpensive communication between any two members of an organization. As a consequence, fresh chances present themselves. Many businesses put together "tiger teams" comprising knowledgeable employees from different levels of the organizational structure.²⁶¹ To handle an urgent problem, a team will come together for a short while before disbanding. Companies may implement "just-in-time" manufacturing and distribution techniques thanks to flexible information flow, which lowers the cost of inventory.²⁶²

Information technology also streamlines businesses by getting rid of middlemen in transactions. Take supply chain automation as an example. Let us say firm A purchases widgets from business B. At one time, a business A employee would contact a company B employee to place an order for the widgets. Supply chain automation is becoming a common practice across many businesses. Michael J. Quinn notes, "A computer at company A is linked to a computer at company B. The computers are responsible for ordering the widgets, eliminating the need for the middlemen."²⁶³ The number of personnel needed to write checks, make sales orders, invoicing,

pay utilities, etc., may be decreased by automating these paperwork-related tasks.²⁶⁴ Information technology will most certainly have a favorable effect on businesses by driving up demand for some job categories while driving down demand for others.²⁶⁵

A leader in supply-chain automation is Dell Computer, as an actual example. Dell accepts direct phone or online orders from customers for its PCs. Large enterprises account for 70% of Dell's sales.²⁶⁶ These businesses have unique websites with pre-configured systems catered to the buyer's requirements. Dell keeps its inventory low, just enough for a few days' worth of manufacturing, since it only builds computers when customers purchase them.²⁶⁷

Data Management

Data quality management is described by the DAMA DMBOK as "the planning, implementation, and control activities that apply quality management techniques to data to assure it is fit for consumption and business purposes."²⁶⁸ Data quality management is the management of information using tried-and-true quality management methods. An identifiable quality or aspect of data is called a data quality dimension. These parameters might be subjective (such as usability, transactability, or reputation), objective (such as completeness), or highly contextual (accuracy, accuracy against surrogate source).²⁶⁹ Organizations must determine which attributes are crucial to company operations and are also measurable. It is essential that whatever is being measured be connected to a business risk or problem. For instance, if a patient's blood type is incorrectly documented at a hospital, this might lead to treatment issues. As a result, a hospital could need data with a 99.999 percent accuracy level and a population of all blood groups.²⁷⁰

Data quality business rules include characteristics of data quality as inputs. These guidelines specify how that data must be for it to be useful. They formalize the requirements for

data quality and take action to stop, catch, or fix problems when they arise.²⁷¹ Numerous data quality business rules may be found in the DAMA DMBOK and other publications.²⁷² This section will concentrate on what occurs once a quality measurement has been made. Here, the problem is more with the organization's quality ethics. Organizational leaders were urged by W Edwards Deming to "adopt the new philosophy" of quality.²⁷³ That is a moral precept. Organizations that think about quality systems and methods often do three things. First, they prioritize the demands of their consumers; they ensure that they accurately identify and comprehend both the customers and their wants.²⁷⁴ Second, they eliminate anxiety and promote pride in a job well done; holding individuals accountable for the system's flaws and blunders is unproductive.²⁷⁵ While it may be easy to blame the individual closest to the symptom, the main cause often resides elsewhere in the organization. Third, they concentrate on continual development. The company may achieve a sustainable quality level by gradually raising the bar without taking its success for granted.²⁷⁶

Research with customers has shown the inadequacy of the "hero ethic," which involves throwing personnel at issues and trying to repair them personally.²⁷⁷ Applying a quality culture across the company and emphasizing defect prevention more would be a far better strategy than just rushing to discard and redo data. This data manipulation is unethical and can potentially create structural and organizational issues. In order to prevent this, ethical personnel must concentrate on the stakeholders' requirements, desires, and worries in the data—which might include internal and external customers—so that quality is built in.²⁷⁸ The organization's method of measurement and how that impacts how quality is perceived are additional ethical considerations in information quality management. When referring to quality management, this might entail gauging the effectiveness of a process rather than keeping tabs on individuals based

on the number of records that are compliant with the standards set.²⁷⁹ As a result, the ethics of quality improvement shift from penalizing departure from the norm to rewarding and recognizing high-quality results.²⁸⁰ This relates to the more general ethical ideas of protecting and advancing human dignity. Imagine you often call attention to the company's shortcomings, those of teams within the organization, or single out specific people as being below average. Integrating ethics into the information quality process from the start would help avoid this problem, maintain strong work morale, and better promote human dignity.

2.B. Conclusion

Data analytics has had a profound effect on the current and future states of healthcare in numerous ways – from the individual level with personalized care and the ability to tailor a patient's care based on an integrative sequencing of their genome to an organizational level impacting the way healthcare is financed, delivered, and systematically changed throughout the entire organization. While the impact of data analytics is substantial, challenges in the legislation provide suitable barriers that this raw information is kept safe, secure, and private. As long as these challenges are met, the opportunity for data analytics to revolutionize healthcare is infinite.

2. Conclusion

New advanced data management tools are rapidly becoming the norm in healthcare facilities. These technological advancements raise some intriguing concerns about how patient data is used. In the digital era, the age-old dilemma of combining patient confidentiality protection with accessibility to patient records takes on new dimensions. Information is much more readily available, and one system security breach can violate many patients' privacy. Data integrity concerns are also heightened as more data become available to more people and agencies. Finding adequate means to guarantee that electronically stored patient care information

is accessible to healthcare practitioners and other parties with a valid need to know about the patient is a problem for health information management experts. The next chapter highlights this issue by describing the health data battle on the micro and macro levels and how that plays out when individual privacy and confidentiality are pitted against public health and the common good.

CHAPTER 3: THE GLOBAL HEALTH DATA BATTLE: DATA GOVERNANCE AT THE MICRO & MACRO LEVELS

3. Introduction

The extent to which policymakers are able to strike a balance between the need to protect genuine patient and clinician preferences on the one hand and the significance of healthcare data to general public on the other will determine whether additional data access is permitted to produce more robust data about how the health care system operates.²⁸¹ The ability to manage research-related data, such as that from the Federal Coordinating Council for Comparative Effectiveness Research, securely and effectively will play a role in determining how to strike a balance that will allow for the production of evidence without jeopardizing these ethical obligations.²⁸² The third chapter moves on to the public health data battle in which individual autonomy and rights are pitted against population health and the 'Common Good.' Concluding the section at the heart of the argument is the debate between personal data privacy and public health surveillance – with ethical arguments on both sides. The section continues by detailing the imperative to influence policy as technology rapidly changes the norms of everyday life. Changes in policy in the form of ethical governance frameworks that enable regulation and monitoring of technologies that are a double-edged sword are crucial. Policy at not only a national but global level is conveyed, using examples such as the European Union and UNESCO.

3.A. Individual Autonomy & Rights vs. Population Health & the Common Good 3.A. Introduction

Healthcare electronic data is everywhere – on computers, phones, tablets, and virtually any device that can access the internet. Patient electronic health records, medication information, genetic testing results, and so much more are out there. What is being done with all this information, and did patients consent to its use in public health interests? This section's focus will address whether it is ethically justifiable for big data banks to use personal health information without consent for public health surveillance. The answer to this question requires a deeper look at personal autonomy, human rights and limitations, population health and an individual's societal obligations, and the concept of the common good. This section will then address how these concepts are applied in the real world by describing big data, what it is, and how it can help healthcare, in addition to its ethical considerations and limitations. The section will next discuss some of the issues with data usage and provide an explanation of the ethical basis for public health surveillance, including the hazards and obligations related to this profession, before addressing the issue of personal privacy vs public health surveillance.

3.A.i. Individual Autonomy vs. Population Health

Bioethicists nowadays are often classified based on how they see the interaction between the person and society. Ethicists have been characterized by terms like "communitarian" and "liberal," among many others, in their critiques and refutations of opposing viewpoints.²⁸³ For a long time, bioethics was dominated by liberal individualism, which was built on the philosophical pillars of liberty, dignity, and privacy. However, ideologies that promote giving the concepts of collaboration, unity, and common good a more substantial role have begun to pose an increasing threat to it.²⁸⁴ It has been argued that privacy laws, informed consent laws, and

individual autonomy laws significantly inhibit medical research that aims to advance society. Although acquiring and analyzing information technology has advanced quickly, stringent laws defending individual rights have made it difficult to use such information.²⁸⁵

Even while many opponents of communitarianism do not overtly reject individual rights, it is difficult to create a compromise between such rights and concerns for the greater good.²⁸⁶ The autonomy of individuals is often weighed against the interest of the public or collective good when considering the recommended ethical frameworks for genetic and clinical database systems. It ultimately turns into an either-or situation. This split may have resulted from the inability to see the intimate connection between autonomy and the common good. One possibility is to draw influence from Aristotle's political theory, which seeks to link autonomy with the general good in a clear, alluring, and persuasive manner.²⁸⁷ This statement alone should help to bridge the gap between communitarian and liberal bioethics since Aristotle neither supports nor reflects either position.²⁸⁸

3.A.i.a. Personal Autonomy & Human Rights

Bioethics has often been built in the West on concepts and language connected to human rights. This idea has often been centered on the rates of individual patients vs some kind of authority, such as medical paternalism, constraints on stopping treatment under the law, or the dull operations of hospital practice.²⁸⁹ There is an increasing push for bioethics to move beyond discourse centered on rights. However, this term often refers to a shift toward a patient's or family's deeper narrative or values framework. Cultural values are sometimes included, but they are oriented in a way that is more unique than multiple and universal.²⁹⁰ The move from a solitary mindset to a plural one is much more inclusive when bioethics and human rights are discussed.²⁹¹

Human Rights

The quest for a bioethics concept that encompasses human rights that is broader and fuller than one that is not already acknowledged in all civilizations, older cultures that value community above individuals, would likely be the biggest barrier to this universal approach. It is clear how much cultural values influence how we see morality in biomedicine from the emphasis on independence and self-expression in Western bioethics, especially from American bioethicists.²⁹² Our beliefs about uniqueness and autonomy affect all medical transactions, such as consent and privacy in the patient-clinician relationship. The patient is seen as the center of decision-making competence and complete consent in the United States and other Western countries, although many other people accept the community or familial decision-making model, notably in Eastern-Asian cultures and nations.²⁹³ In civilizations where family and community bonds are prioritized above all else, informed consent is essentially meaningless. Decisionmaking ability in these situations is only possible on a social level.²⁹⁴ In other words, medical decisions are relational or made by numerous family members or close friends instead of just the individual receiving medical care. Populations in patriarchal or matriarchal societies may also have medical decisions made by the family's leader.

In western civilizations, it is also believed that one's confidentiality of information should be preserved. However, this idea does not necessarily imply a universal application. The importance of privacy varies widely, according to social research on attitudes and customs related to cultural conceptions of the self.²⁹⁵ To further understand the ethical complexity associated with personhood, more empirical study is needed. Therefore, the importance of autonomy is not a solution for bioethics on a global scale but rather a symptom of a deeper issue for most societies.²⁹⁶

Limitations of Autonomy

However, the autonomy concept that guides bioethics and bio law has real challenges. Autonomy is an ideal term since it alludes to a person's complete self-control. However, an individual's capacity to regulate behaviors, desires, or both may have flaws.²⁹⁷ Jacob Rendtorff states, "The individual's capacity of reasoning may be limited or nonexistent, and this is not only the case for children, senile, insane persons, etc., but also for normal, intelligent people who feel themselves weak and dependent on others or who simply do not understand the scientific project in which they are asked to participate."²⁹⁸ People may also base their conclusions on unreliable or incorrect information they have learned from various sources. Additionally, the person's wants or wishes could be unclear. The individual's sense of self is not always consistent, and there are instances when they are unsure about their desires.²⁹⁹

This image of the free and independent person with the power to make their own decisions about their own life has drawn substantial criticism for being so far from the realities of the clinical environment and the specifics of bioethics and law. Reciprocal self - determination is allegedly insufficient as a conceptual framework for comprehending the relationship between patients and physicians. According to the objection, autonomy assumes a person with no social connections. Additionally, it makes an overly strong assumption about a person's ability to make their own judgments. Rendtorff continues, "In extreme situations of medical treatment, it is not likely that people can make independent and autonomous decisions. They are vulnerable, weak, and not in control of the situation."³⁰⁰ In addition, it is unlikely that individuals make independent and logical judgments during their daily lives without the assistance of others. In actuality, the need for individual autonomy is seldom met when making judgments in severe circumstances.³⁰¹

Furthermore, others claim that diversity and freedom are not the sole unalterable principles. Sometimes having a strong sense of community based on shared values is a need for self-determination. In this setting, autonomy assumes an institutional and cultural basis built around the shared values of tolerance for variety and individual freedom. Because of this, an explanation of autonomy cannot be wholly libertarian; instead, it must acknowledge that each person is embedded in various social customs, ties, and relationships with others.³⁰² One loses sight of the delicate and sensitive aspects of the human situation that call for respect and care if one just concentrates on autonomy.³⁰³

3.A.i.b. Population Health & Societal Obligations

As Dr. Henk ten Have states in his book *Global Bioethics: An Introduction*, "Focus on the concept of individual autonomy implies a relative neglect of notions such as the common good, public interest, and community. The individualism of bioethics discourse makes it harder to examine issues such as resource allocation, technology assessment, the aims of healthcare, and justice."³⁰⁴ Renée Fox, a sociology professor, argues in favor of this criticism, saying that "bioethics is centered on value complex of individual rights, self-determination, and privacy, at the expense of social responsibility and social justice."³⁰⁵ The flimsy ethics of individual autonomy have historically focused on the patient issues brought on by clinical medicine and medical technology. Still, they cannot effectively handle the social and institutional framework of medical decisions.³⁰⁶ Perhaps it is time to adopt a different principle and change how the West interacts with the concept of autonomy in bioethics and bio-law.

Changing Ethical Frameworks

The classic Hippocratic medical ethics were created within a framework of professional ethics based on virtue, emphasizing the obligations of beneficence and secrecy on the part of

physicians.³⁰⁷ The foundation of the doctor-patient relationship was the patient's confidence in and respect for the doctor's authority. However, the emancipation that was taking place in many other spheres of life led to criticism of this paternalistic attitude in the late 1960s. According to Alfred Tauber, "Medical ethics was caught up in rights movements and took the position that patients needed to be empowered against paternalistic doctors and an intimidating technology that threatened to dehumanize patients."³⁰⁸ The movement in favor of the individualism approach may also be seen as a response to the repressive past of eugenics and the forced involvement of people in activities under the guise of the greater good, namely the Tuskegee case in the United States and Nazi physicians' experiments on POWs. Thus, bioethics pledged to treat each person's liberty, privacy, and equality with the utmost regard.³⁰⁹

It has been claimed more and more often since the 1990s that bioethics committed a grave error by elevating human autonomy to the status of a core ethical concept at the expense of other crucial values. A few arguments have proposed new ethical frameworks for greater communal values like reciprocity, mutuality, solidarity, citizenship, and universality.³¹⁰ As communitarian ethicists constantly point out, humans are social creatures, and "our nature is distorted if we think of ourselves only as co-existing social atoms."³¹¹

These objections and issues suggest that an ethics that is more contextualized is necessary. The gathering, storing, and use of personal data, including genomic and health records, cannot be controlled by the same ethical principles as clinical testing, treatment-focused clinical care, public health, or other areas.³¹² The distinctions between these settings and their internal disparities are attracting increasing attention in the modern day. Invasive biomedical research involving people is where the so-called guidelines for informed consent were developed. In other circumstances, such as demographically based genomic repositories or

population health, they may not be appropriate. However, some detractors are concerned about creating a risky precedent. They fear that if ethical guidelines are loosened to allow for a study on human genetic databases, this might lead to modifications in other research areas and, ultimately, result in research subjects losing protection.³¹³ One concept related to the push to focus new ethical frameworks on more collective values is the common good.

Common Good

The welfare of society was a priority for the ancient Greek polis. The reciprocity of such benefits is essential to the survival of the shared benefits that belong to everyone in society, including peace, stability, success, fairness, and solidarity. It was believed that the goals of the society and the individual would be compatible under normal conditions. In a dispute, the common good would be seen as superior. The core of Aristotle's political theory is his claim of, although neither is sufficient on its own, "the state is by nature clearly prior to the family and the individual since the whole is of necessity prior to the part."³¹⁴ Every citizen's well-being is intertwined with the republic, and individuals are exposed to the republic's virtue daily. The common good, for Aristotle, relates to "a good proper to, and attainable only by the community, yet individually shared by its members."³¹⁵ For Aristotle, the common good must be both collective and individual. It spans the objectives and reasons individuals seek in their social and interpersonal interactions instead of correlating with the whole of individual commodities.³¹⁶

Both the virtue of the people and the institutional framework of the republic are recognized by Aristotle as requirements for the common good. A shared existence oriented to the common good is necessary to have collaborative pursuits, thus citizens must act fairly.³¹⁷ Unfortunately, many people lack justice and are always in conflict for things like power, money, or honor. Since there is always a shortage of outside resources, their lousy behavior causes fierce

rivalry and ongoing disputes. As a result, reorienting people's moral priorities toward virtue via education is a crucial need for achieving the common good. For example, overcoming human disparities in interests, abilities, and jobs is one of the challenges to achieving the common good.³¹⁸

According to Margit Sutrop, society is more than just a collection of different people vying for the same few resources.³¹⁹ We all have a number of common values as members of the community. Society's role extends beyond just giving individuals the resources to live fulfilling lives. It is challenging to formulate a definition of the common good that fully encapsulates and takes into account our shared interests.³²⁰

Suppose one views the common good in Aristotle's definition as equivalent to the public interest. People would then realize that what is required of them is careful consideration of its constraints and circumstances in addition to commitment to the common good.³²¹ The interesting question is, how do such limitations apply to the growing big data problem and the genetic databases that could be used for the public interest of health and wellbeing? This question will be further investigated as to whether it is ethically justified to conduct public health surveillance (via biobanks, genetic data repositories, or other means) without patient consent. This will be done by first explaining what big data is, how it can help healthcare and the ethical challenges of privacy and security that go along with it in a modern-day application of the battle between individual autonomy and rights versus public health and the common good.

3.A.ii. The Big Data Problem

Since the beginning of recorded history until 2003, there has accumulated around 5 billion gigabytes of information, making now the most data-rich time ever.³²² For comparison, in 2011, 5 billion gigabytes of data were generated every two *days*.³²³ In 2013, 5 billion gigabytes

were generated every ten *minutes*.³²⁴ In 2015, 5 billion gigabytes of data were generated every ten *seconds*.³²⁵ That is a lot of data, more than the human mind can comprehend, and it will only continue to exponentially increase over time, hence 'big data.' This is only the beginning when it comes to the big data problem – while it offers exciting and new avenues to healthcare by reducing costs, personalizing medicine, and predicting future health problems – many ethical challenges need to be addressed in these new developments. These obstacles include, but are not exclusive to, concerns with informed consent, privacy and security, and epistemological questions. These challenges must be addressed as the healthcare world moves forward with big data; otherwise, the system will fail to respect some of the most basic bioethical concepts of autonomy, consent, and justice.

3.A.ii.a. What is Big Data & Why Does It Matter?

The healthcare sector has generated large volumes of data for many years. They will continue to do so due to patient records, compliance, regulatory obligations, and patient care, to mention a few.³²⁶ Although much of this data is still in paper form, digitizing this vast quantity of data has become more prevalent in recent years. This transition is driven by a few variables, the most noteworthy of which is its potential to lower costs while improving the quality of healthcare service. These enormous data volumes, or "big data," have the potential to serve a range of healthcare tasks that might be advantageous to the majority, if not all, healthcare sectors.³²⁷

The Four V's of Big Data

According to academics, there is not a single, generally agreed definition of big data. Four essential characteristics—volume, variety, velocity, and veracity—are often used to characterize big data.³²⁸ Regularly created and collected health-related data will amass to a

reasonable volume of data. Raghupathi et al. describe, "The already daunting volume of existing healthcare data includes personal medical records, radiology images, clinical trial data, FDA submissions, human genetics and population data genomic sequences, etc. Newer forms of big data such as 3-D imaging, genomics, and biometric sensor readings are also fueling this exponential growth."³²⁹ Huge quantities of data exceed the capacities and retrieval capabilities of traditional data processing, organization, and storage and retrieval, including data warehouses, making volume a crucial component of big data.³³⁰ Big data hence necessitates scalable and efficient information management and storage strategies.³³¹

Healthcare data may be found in many different places, including electronic health records (EHRs), medical tests, smartphone apps, and social media. The application of this information to improve healthcare is, however, quite limited. Big data methodologies allow for the effective connecting and analysis of data in many formats to address specific operational, commercial, or research problems.³³² Raghupathi et al. continue, "Future applications of real-time data, such as detecting infections as early as possible, identifying them swiftly and applying the right treatments (not just broad spectrum antibiotics) could reduce patient morbidity and mortality and even prevent hospital outbreaks."³³³ For newborns in the NICU, real-time processing is already being examined to discover potentially fatal infections earlier.³³⁴ If real-time data analysis could be applied to vast volumes of healthcare data from all specialties, adding the second "v" for variety, healthcare would experience a renaissance.³³⁵

Velocity is another defining characteristic of big data. Massive volumes of continuously updated, variously structured data cannot be processed and analyzed in real-time by conventional IT health infrastructures.³³⁶ Data management may be done faster and with greater flexibility thanks to big data infrastructure. It is a blessing that improvements in data management,

especially in virtualization and cloud technologies, have made it simpler to design systems for more efficient data collection, storing, and analyzing.³³⁷ Real-time and fast data accumulation is taking place. Unprecedented issues arise from the steady influx of new data accumulated. The amount, variety, and collection of data needed for output retrieval, evaluation, comparison, and decision-making have changed along with the speed at which data is generated.³³⁸

According to some professionals and academics, the fourth element is veracity, sometimes referred to as data assurance. In other words, the large data, analysis, and results are accurate and reliable. Veracity is the desired outcome, but it has yet to become a reality. Data quality issues are of highest relevance in the sector because correct information is essential for making survival decisions and because the quality of medical files, particularly unstructured information, is extremely inconsistent and sometimes wrong.³³⁹ The most notorious instance is probably incorrect translations of shoddy handwriting on prescriptions.³⁴⁰

Advantages to Healthcare

According to Roski et al., "Big data may have the potential to create approximately \$300 billion annually in value in the health care sector, two-thirds of which would be generated by lowering health care expenditures."³⁴¹ Big data has so far proven useful in the therapeutic field in a number of circumstances, such as with the advent of tailored medicine (particular diagnostics and therapies predicated on a patient's thorough risk assessment) in the treatment and care of cancer patients.³⁴² Another example is to use decision support systems to match medications to the risk categories of specific patients. These systems are improved by automated detection of diagnostic imaging pictures, and literature mining.³⁴³ Another example of how patient-generated data is utilized is to personalize screening and therapeutic options along with educational communications to support desired patient behaviors.³⁴⁴ A fourth, though not the last, example of

how big data has already established its utility is the fact that population health assessments powered by big data have shown patterns that could not have been identified had fewer quantities of regularly structured data been examined instead.³⁴⁵ The Centers for Medicare and Medicaid Services' use of big data fraud prevention and detection methodologies has superseded older manual documentation techniques. Over \$4 billion in expenses have been recovered thanks to these methods in only 2011 alone.³⁴⁶

Big data is helpful in healthcare research and development, which is another field. The research and development pipeline for medications and devices may be made leaner, quicker, and more focused using big data and predictive modeling.³⁴⁷ Clinical research designers and participant recruiters may more effectively match medications to specific patients with the use of statistical approaches and algorithms, which lower trial failure rates and hasten the development of cutting-edge cures. Finally, before pharmaceuticals are put up for sale, big data analysis must be used to go through health records and clinical studies to uncover follow-on implications and harmful impacts.³⁴⁸ Raghupathi et al. conclude, "These advantages also help areas of public health in which big data can analyze disease patterns and track disease outbreaks in transmission to improve public health surveillance and speed response. Faster development of more accurately targeted vaccines, such as choosing the annual influenza strains and turning large amounts of data into actionable information that can be used to identify needs, provide services, and protect and prevent crises, especially for the benefit of populations."³⁴⁹

3.A.ii.b. Ethical Challenges

It should be no surprise that with all this patient information and data streaming around, this infrastructure has some ethical concerns and challenges. Many ethical issues are covered in the literature on big data, including informed consent, security (encryption and data protection),

stewardship, and epistemological questions (such as those of power or control and the digital divide).³⁵⁰ The scope of this section will address the two most popular themes in the literature of privacy and security and informed consent concerns with big data.

Privacy & Security

Our present and past are becoming obvious to individuals with the right access and expertise as our actions are captured and replayed in digital reality. For instance, the Rapid Information Overlay Technology (RIOT) software, created by US defense contractor Raytheon, reviews a single person and tries to make their daily activities perfectly clear using data that is freely available from social network and information systems affiliated with an IP address, among other sources.³⁵¹ This information was made public by the British newspaper *The Guardian*.³⁵² Data analysts use big data to assess our internet consumption habits, friendships, sleep patterns, and shopping preferences. This information is personalized only in a few circumstances, usually in intelligence circles.³⁵³

However, de-individualization is just one facet of anonymization, which involves taking away components that enable data to be related to a particular person.³⁵⁴ Andrej Zwitter notes, "Location, gender, age, and other information relevant for the belongingness to a group and thus valuable for statistical analysis relate to the issue of group privacy."³⁵⁵ Whether or not a dataset has been normalized depends on how many and which group characteristics are still there. This would entail stripping the data of its substance if it were to be stripped of all aspects about any group. As a result, groups are continuously getting more open even if the data is anonymous because it is de-individualized.³⁵⁶ Privacy and security will continue to pose an ethical challenge to big data until the anonymization problem can be tackled. Another major ethical challenge that is part of the big data problem is informed consent, as explained in the next section.

Informed Consent

The concept of consent denotes the ability and right of an individual to exert control over information about them, to limit how it is used and reused by organizations like research institutes and healthcare providers, and to revoke such use or reuse.³⁵⁷ According to existing consent regulations, it is better for society to place an emphasis on personal data ownership above the social advantages that might be gained by disclosing sensitive data for a variety of reasons.³⁵⁸ In contrast to rigid rules for every single conceivable use, an equilibrium between control of the person and informed disclosure in the context of public safety, environmental preservation, or other aims may be necessary for approval in order to effectively exploit the possibilities of big data.³⁵⁹

For instance, the Human Genome Project has put in place permission procedures that allow data to be used and reused for a variety of scientific purposes.³⁶⁰ The project has tight guidelines for data protection. Even while the details of that use are unclear when the study asks for their consent, it also warns individuals as to how their volunteered personal data may be used and repurposed for their benefit and that of others. In order to maximize participants' chances of benefiting from the study, its permission procedures explicitly specify how participants' data may be shared with other scientific research projects and to what degree data can be safeguarded.³⁶¹ According to Tene & Polonetsky, "In examining such an approach and the potential for its wider adoption, policymakers should balance the value of data uses and potential privacy risks, assess the practicability of obtaining true and informed consent, and understand the enforceability of restrictions on future data sharing and use."³⁶²

Sometimes, gene sequencing research is restricted to "information altruists," or people eager to offer their data (and occasionally their identities) in the hope that their social standing or

financial resources will sufficiently protect them from future prejudice or negative effects.³⁶³ 'Radical honesty' approaches, in which people volunteer to share their de-identified genetic information with the public, are comparable. Establishing "honest broker" and "stewardship" consent models is another strategy to safeguard the interests of data subjects by having unbiased third parties negotiate broad consent agreements.³⁶⁴ By focusing on professionalism or enacting sanctions for data misuse, it is feasible to delegate a duty of researchers who have access to participants' personal information to instill emotions of accountability in them. Prohibiting unethical research practices like re-identifying anonymized data would lessen any possible harm to data subjects.³⁶⁵

By serving as a jumping-off point and a point of reference for future study and debate on the ethics of big data, the quick analysis outlined in this section aims to change widespread and excessively optimistic views. An overview of big data through the four v's (volume, variety, velocity, and veracity) and why it matters in its advantages to the healthcare system was reviewed. The ethical challenges that must be addressed with this new avenue of processing data were mentioned, precisely privacy and security issues and informed consent problems. However, these issues are just the tip of the iceberg – many other ethical and technological issues must be addressed before big data comes into full swing in healthcare, which is a rapidly approaching deadline. One of the key topics at the center of the big data discussion is the conflict between upholding individual data privacy versus deanonymizing personal data to assist in public health surveillance. The challenges in big data use and the ethical justifications for public health surveillance are detailed in the next section.

3.A.iii. Individual Data Privacy vs. Public Health Surveillance

By necessity, public health monitoring takes place without express patient agreement. The continued use of personally identifiable information for reporting communicable illnesses and other forms of public health monitoring has legitimate legal and scholarly backing. The ethical justification for monitoring without a patient's express permission is examined in this section using current medical and population health ethics topics. Lee et al. stress, "Overriding individual autonomy must be justified in terms of the obligation of public health to improve population health, reduce inequities, attend to the health of vulnerable and systematically disadvantaged persons, and prevent harm."³⁶⁶ Additionally, data components obtained without permission must only cause the absolute minimum disruption, support efficient public health action, and be kept in a secure environment.³⁶⁷

3.A.iii.a. Challenges in Data Use

Big data presents several challenges with its use – some were highlighted briefly in the previous section, such as privacy, security, and informed consent. This section will take a closer look at electronic healthcare (e-healthcare), specifically at privacy and confidentiality issues that physicians face when debating whether to disclose information such as an infectious disease to the public. In addition, additional data utilization issues will be addressed, and how those should be handled within the legal and professional ethics worlds – subject to future change as time and data changes.

Privacy & Confidentiality

The quick development of e-healthcare has significantly impacted information security for patients. Furthermore, breaches of patient confidentiality and privacy may occur quickly, in large numbers, and without being noticed in the e-healthcare setting.³⁶⁸ State and professional

organizations have acknowledged that safeguarding the information entrusted to health care professionals is a fundamental ethical obligation for all clinical staff in light of these e-healthcare possibilities and the populace's concerns about security and confidentiality. A key element that gives information management actions by those in charge of regulating personal health information ethical and moral dimensions is the fact that patients provide information to healthcare professionals in confidence and out of need.³⁶⁹ Preserving the confidentiality of patients' individually identifiable health data is essential for e-healthcare to be successful and to live up to its promises.³⁷⁰

To get therapy and enhance their health, people must provide information to healthcare professionals. This information is provided on a confidential basis. The patients are aware that their present and future medical treatment is the primary reason for revealing, collecting, and retaining personal healthcare information. It raises serious privacy and confidentiality concerns when such private medical information is exploited for other reasons unrelated to their treatment.³⁷¹ The government claimed the common good to support additional usage of individual e-healthcare data in socially beneficial initiatives. Nevertheless, it is debatable whether financial considerations and further uses of patient data are acceptable.³⁷²

Calculating crucial data is necessary when planning and allocating resources over the country.³⁷³ The sharing of sensitive medical information outside of patient care is also closely scrutinized in the national control of contagious and epidemic illnesses. To advance democracy at the national level, personal privacy and secrecy must be legally protected.³⁷⁴ How is e-healthcare information used? Does one use have purpose over another? What happens if the use is ambiguous? These questions and more are answered in the next section involving e-healthcare utilization.

Utilization

Chronic difficulties with safeguarding e-healthcare information have arisen because of its many uses. Personal e-healthcare data's primary goal is to help clinical care choices for a specific identifiable individual.³⁷⁵ Therefore, it is very important to identify each individual subject of healthcare information correctly and accurately for main use objectives. Additionally, the law and professional medical ethics preserve the patient's privacy and confidentiality.³⁷⁶ The term "secondary uses" refers to further applications of e-healthcare data. The veil of protection effectively prohibits the secondary purposes and uses of health data that support controlling disease outbreaks.³⁷⁷

The secondary applications of clinical information might be considered as a compromise among the privacy rights of a person and society's desire to lower healthcare expenses while enhancing the quality of treatment and effectiveness within the delivery system. To evaluate the quality of treatment, plan and manage healthcare services, and conduct clinical or epidemiological research, it is essential to employ EHRs.³⁷⁸ A well-managed healthcare system has thereby benefited patients more because of the secondary applications of electronic healthcare data.³⁷⁹

Every secondary utilization of health information, whether it benefits the patient or the public or not, such as utilizing it to deny a job or health insurance application, raises concerns about privacy and confidentiality as well as legal and ethical difficulties.³⁸⁰ Through the many healthcare professions, ethical issues are handled. In the US, legislation like HIPAA that safeguards personal information are handled within the legal departments of hospitals or, if escalated, the courtroom.³⁸¹ The complete erasure of the subject's personally identifying information is thus required for secondary use of e-healthcare data, as is informed consent. An

essential issue for e-healthcare is the elimination of personally identifying information, which is being addressed through anonymizing and pseudonymizing the data.³⁸²

3.A.iii.b. Ethical Justification of Public Health Surveillance

Lee & Thacker define public health surveillance as "the ongoing, systematic collection, analysis, and interpretation of health-related data with the a priori purpose of preventing or controlling disease or injury, or of identifying unusual events of public health importance, followed by the dissemination and use of information for public health action."³⁸³ Public health monitoring differs from other forms of monitoring, such as security, in that its primary goals are the prevention or control of sickness or harm as well as the improvement of public health.³⁸⁴ In certain circumstances, public health monitoring is used only to assess the severity of health issues, characterize the course of a disease, identify epidemics and outbreaks of well-known or newly discovered pathogens, and track the dispersion and progression of health events.³⁸⁵ In addition to supporting epidemiological and clinical research, it also creates and tests theories, evaluates prevention and management strategies, monitors isolation activities and infectious agent changes, pinpoints alterations in medical procedures, plans population health interventions including via means of assets, and appropriates and distributes funds for treatment and control.³⁸⁶ *Considerations*

Public health monitoring techniques may be regarded as morally admissible and systems as ethically justifiable if they comply with the positive operational criteria and abstain from transgressing the negative ones. Building a system for monitoring public health is the first action in current best practices, which involves having explicit knowledge of the system's public health goals to guarantee that the relevant data are gathered to address the critical issues.³⁸⁷ Prioritizing health outcomes depending on the scale of the impact, assessed by frequency, severity, cost, or

preventability, is often necessary for determining what public health concern to address.³⁸⁸ The operational principles of gathering accurate information and responding quickly to the healthcare requirements of a population are met by this prioritizing.

The requirement for adequate data to guide both the management of public health and the ability to protect the privacy of individual patients' medical records must be balanced when deciding which data categories to collect.³⁸⁹ Lee et al. emphasizes the "Current recommended approach is to collect the minimum number and simplest data elements necessary to meet the goals of the system to minimize risk to individuals, thus meeting the operating principle of imposing the least possible infringement. Early engagement of partners and affected communities is recommended in developing public health surveillance systems, especially when the data are sensitive, or populations are particularly vulnerable."³⁹⁰ Choosing what kind of information should be gathered in a surveillance system often involves consulting the impacted groups. In the 1990s, community involvement was successfully employed to win over objections to name-based HIV reporting, even in places like New York City and San Francisco, California, where there had been vocal early opposition.³⁹¹

The best practices that are currently being used for the data stages, such as data gathering, aggregation, processing, and evaluation, cascade straight into a well-planned system.³⁹² Given the system's emphasis on public health, gathering the fewest and least sensitive data pieces is required. This is consistent with the tenets of the system, which include rationality, minimal intrusion, and attention to root causes of illness. Lee et al. states "Collecting information from all affected cases (or a representative sample) is the hallmark of population-based public health surveillance, leaving no subgroup excluded from data collection or, perhaps more importantly, from identifying the need for intervention."³⁹³ The lack of restriction due to demographical,

behavioral, or social factors operationalizes the goals of removing injustices, correcting structural inequities, and inclusion.³⁹⁴

Data gathering is crucial to the fair and equitable distribution of treatment and preventative measures in a sector like public health that is scientifically based. Consider the case when a demographic is often left out of public health monitoring, perhaps as a consequence of receiving treatment from a certain kind of practitioner who has less reporting responsibility. If so, structural discrepancies in the benefits gained for system participation will be revealed to this group, which is underrepresented in the statistics.³⁹⁵ Groseclose et al. continues "Acquiring accurate data during collection and collation is a minimal requirement of a public health surveillance system. To ensure data accuracy, ongoing monitoring and periodic evaluation are recommended for all systems."³⁹⁶ By guaranteeing the use of the data gathered and giving proof that the advantages of data collection exceed the drawbacks, this approach helps satisfy the requirements of two operational principles. Complete understanding of the surveillance system is necessary for using suitable analytical methods and correctly interpreting results.³⁹⁷

The operational pillars of responsible interpretation ensure the effectiveness of the intervention, under this case, surveillance, so that advantages outweigh violations. The last component of best practices for data security is defined processes for how data are collected, transported, accessed, kept, utilized, or shared, as well as specifications for technical and physical security safeguards of data.³⁹⁸ Lee et al. says "Limiting the number of printed and digital copies of case data, anonymizing and encrypting data during transport, limiting the number of users to the smallest possible, annual security training, and incorporating specific sanctions into annual confidentiality agreements signed by personnel working with surveillance data are some of the recommended best practices for maintaining confidentiality and data

security."³⁹⁹ These processes demonstrate how to protect confidentiality and prevent personal injury by protecting data and reducing the danger of damage. Additionally, they support little intervention.⁴⁰⁰

The best available procedures for information use are deduced from the core principle that data gathering must not be an end in itself but rather a means to improve the condition of the area where the information was gathered. These practices include information exchange and conversation to those who need to know it as well as implementation of the data to health promotion strategies and initiatives to achieve the best outcomes.⁴⁰¹ It need excellent communication techniques and understanding about the people and organizations that can utilize the data from a monitoring system to take action for public health. Furthermore Lee et al. explains "Effective communication models require assessing the quality of data, the definition of the purpose of the communication and the audience, the development of the message, selection of the channel, marketing of the information, implementation of the plan, and evaluation of the process and outcome."⁴⁰²

During this stage of the public health surveillance action, operating principles such as intervening on the data to benefit society, proving that advantages outweigh violations, and minimizing disparities by highlighting inequities and recommending initiatives to counteract them—are all applied.⁴⁰³ When spreading unfavorable information about socially underdeveloped minorities, caution must be taken to guarantee that the least amount of damage is done. By engaging affected groups and community leaders on the best communication strategies and populations, the harm and heightened marginalization that may come from direct mainstream media messaging may be minimized.⁴⁰⁴

Risks and Responsibilities of Public Health Ethics

The public health authority is now responsible for the reported data after a reportable condition has been identified and submitted in accordance with the legal duties imposed on the healthcare practitioner.⁴⁰⁵ It is a significant obligation to collect and store personally identifiable information. Since public trust is the cornerstone of public health monitoring, treating patients with respect and dignity is essential for the continuous operation of public health management.⁴⁰⁶

The public health system is aware of the consequences that individuals and communities face as a result of public health practitioners publishing name-based data about private medical matters. Potential damages have been recommended to be protected by ethical guidelines.⁴⁰⁷ Lee et al. suggest "These include the following: a collection of individual identifiers only when essential to achieve the public health goals of the system, acquisition of the minimum amount of information needed to meet the public health objective, and engagement of affected communities and stakeholders to consult regarding the most effective way to disclose and disseminate the findings of public health surveillance data, especially when they might substantially add to the stigmatization of an already marginalized group."⁴⁰⁸ For classification initiatives that pose the greatest social concerns, such as HIV and other STDs, the acquisition, preservation, and use of monitoring data must adhere to strict programming criteria.⁴⁰⁹

The agreement between the public health sector and the general community must include the utilization of monitoring data to improve health. The fundamental ethical condition that the risk involved in obtaining and preserving the data is justified by the possibility that it will be utilized to accomplish the intended goals continues to apply to every public health dataset. In the nineteenth century, Herman Biggs, a physician from New York who helped establish surveillance, was certain that "data were collected to be used, not to keep adding machines busy."⁴¹⁰ This attitude remained into the twenty-first century when William Foege, the director

of the Centers for Disease Control and Prevention from 1977 to 1983, indicated that gathering data was crucial for limiting sickness and that spending resources should not be permitted if no mission is completed.⁴¹¹

In fact, it is still morally required that we utilize the data we acquire to advance public health; any exceptions to this rule need to be supported. Ethics was often used as a defense for switching to personal identification tracking in public health monitoring programs that had previously used anonymized systems in order to best use the data for health optimization. As an example, the United States ended its national anonymous HIV monitoring program in 1995 among expectant mothers in pursuit of routinely checking expectant mothers and infants and sending the results to their pre-existing identity surveillance system.⁴¹² This made it possible to share the findings with expectant moms and new mothers so they could decide whether to get chemoprophylaxis or early therapy.⁴¹³

Overall, the United States public health sector has a proven history of handling private information responsibly. The need for rules and codes for the ethical gathering, preservation, and use of health care data is growing as more and more electronic storage makes it possible for easy accessibility, verification, and distribution.⁴¹⁴ Due to the diversity of authorities, public health programs confront difficulties at the municipal, state, and federal levels, but these difficulties are surmountable. Scientific and legal justifications for public health monitoring in the absence of explicit and specific patient permission, contemporary public health ethical frameworks and their guiding principles, and, at least in part, a thorough monitoring system involving disadvantaged populations, gathers the minimal amount of data necessary, stores information safely, and utilizes information for public health promotion.⁴¹⁵

3.A. Conclusion

The dominant concept and practice of bioethics have been called for to change from a more individual-centered to a more global-centered emphasis throughout the last several decades. With this shift comes a move from less stress on personal autonomy and privacy to more emphasizing the community, solidarity, and the common good. Applying this shift in thought to today in America, the problem of data sharing for public surveillance comes to the forefront. The mainly Western notion of anti-paternalism and advocacy for the concept of respect for autonomy takes precedence in the medical field within the physician-patient relationship. However, there is more to bioethics than this simple relationship. Humans are interconnected and social beings and therefore are connected to a community, so the individual good should be the same and connected to the community good. This way of being connected to the community good can come through the analysis and use of electronic health records for public surveillance.

Combined with the possibilities of algorithmic computing and data sorting, much public health information is now available to be used in numerous ways. Of course, with these data use and sharing come ethical considerations such as privacy, security, informed consent, and confidentiality. These issues must be addressed to the best extent possible to use this data for research – and anonymization and pseudonymization allow these possibilities to be addressed. However, with some public health data, this is not possible. As discussed in the last section, it can be ethically and legally permissible to use such data without the individual's informed consent. Several federal statutes that regulate entry to and the potential utilization of specific categories of information, like genetic information or data on other biometrics, offer challenges since they need explicit, informed consent for data use. Some of these challenges, such as privacy, confidentiality, surveillance, and large-scale biometrics application, are detailed in this chapter's next section.

3.B. Ethical Challenges of Current Biometric Practices

3.B. Introduction

Biometrics has passed the prototype phase and is now being used and implemented in all technology and computer science areas. Biometrics are being used for personal recognition, authentication, health systems, government security, and border control - just to name a few. The potential ethical ramifications of every program using novel technology must be continuously investigated. At the municipal, national, and international levels, biometrics may be essential to guaranteeing more dependable identification systems. Humanity must, however, carefully weigh the advantages against the moral and societal concerns. A number of concerns, such as the preservation of privacy, secrecy, individual freedoms, and the interaction between personal rights and the common good are related to individual identities. Biometrics and their associated data (fingerprints, retina/facial scans, voice recognition) very closely relate to data analytics governance because biometrics are the unique personal identifiers that help data analytics and artificial intelligence/machine learning to differentiate one individual from another. Biometric ethics place a premium on the identification and subsequent de-identification steps that safeguard an individual's personal data's security and privacy. The central question to be addressed in this section is: Are current biometric practices – such as a lack of privacy and data protection, oversurveillance, and large-scale applications – harmful to human dignity? This question will be investigated by explaining what biometrics are and why they matter, exploring ethical challenges associated with biometrics, and finally, describing the need for creating a biometrics governance framework.

3.B.i. What Are Biometrics & Why They Matter

3.B.i.a. Brief Biometrics History

Biometrics is a technique that identifies and reliably authenticates an individual's identity by using physiological or behavioral attributes that are unique to that person.⁴¹⁶ Sutrop & Laas-Mikko explain, "Such characteristics are facial image, fingerprints, hand geometry, the structure of the retina or iris, DNA, gait, heart pulse, and voice, among others. Biometric recognition offers many advantages over the traditional personal identification number (PIN) or password and token-based (such as ID cards) approaches."⁴¹⁷ Since biometrics are tied to a specific human being and are difficult to replicate, they may help avoid identity theft and restrict the usage of multiple credentials by one individual.⁴¹⁸ Biometrics are constantly accessible and cannot be forgotten, which gives them an advantage over conventional passwords and PINs.⁴¹⁹ These factors have led to the adoption of biometric systems by both public and commercial entities in a variety of situations, including social security benefits, border protection, healthcare systems, banking, collection management, insurance, and business, to mention a few.⁴²⁰

Origins

When civilization had already advanced to the point that it necessitated frequent interactions between strangers, the need for biometric identification systems could be traced back to the earliest urban communities throughout the Middle East and China.⁴²¹ Military personnel, sailors, and merchants are examples of people who often went outside of their hometowns and needed to be identified. The earliest means to identify someone else and be recognized was presumably via a written description of physical characteristics. Due to the increasing frequency and complexity of human interactions, descriptions of bodily appearances alone become

insufficient. Mordini & Massari say that "The first recognition schemes were probably based on artificial body modifications (such as branding, tattooing, scars, etc.) and tokens."⁴²²

The first genuinely global civilization in the west was the Roman Empire. In addition, it was the first instance of a system that mainly relied on badges and written documentation for identifying individuals.⁴²³ Individuals in medieval Europe were identifiable by permits and safe conduct granted by religious and civic authorities since most people never ventured outside their homes or communities. Substantial civilizations' growth and urbanization's ease of movement required the use of new identification procedures. The first passports were granted by King Louis XIV of France in 1669, but by the end of the 1700s, passports as well as other types of identification were widely used.⁴²⁴ However, a widely used passport system did not start to control interstate travel through till turn of the nineteenth century. Around the beginning of the 20th century, passports and identity cards—which, in some cases, also included facial photographs and fingerprinting—were the major means of identifying individuals.⁴²⁵

Auto-IDs were developed by the late 1960s. However, it took some time since it was recognized that biometrics were different from previous Auto-IDs in specific important ways. For perhaps the first point in the history of the humankind, biometrics has improved the body's inborn physiological identification system order to evaluate the effectiveness of outer physiological and behavioral features, allowing individuals to identify one another more easily.⁴²⁶ Mordini & Massari conclude that "Complex personal recognition schemes, tattoos, seals, passports, badges, save conducts, passes, passwords, PINs: biometrics make obsolete all these traditional identification paraphernalia and – at least in the long run – promise to replace all of them."⁴²⁷

Present Day

Since the turn of the 20th century, biometric characteristics have been routinely employed to identify specific groups, including convicts, employees, political refugees, etc.⁴²⁸ The use of fingerprints in forensics for criminal investigations became well-established in the 20th century. Personal identification, or the link of identity with a particular person, has become an urgent problem due to population increase and rising mobility, as well as the emergence of new risks, including terrorism, international crime, illegal immigration, and identity theft.⁴²⁹ Airports now have body scanners, scanning software, using fingerprints and face recognition technology, biometric passports and identification cards that will help fight terrorism.⁴³⁰ Simultaneously, more people realize that even these precautions cannot keep us safe from malicious intent. "We live in a globalized world of increasingly desperate and dangerous people whom we can no longer trust based on their identification documents which may have been compromised."⁴³¹ Today, there is more interest in biometrics because governmental agencies seek to control and manage inappropriate activity and criminal intent to stop terrorism and other illegal activity.⁴³²

The tremendous advancement of biometric technology over the last several decades is mainly attributable to digitization. Non-automatic biometrics for identification or verification, such as conventional fingerprinting, are dependent on analog models in contrast to automated biometric technologies, which rely on digital representations.⁴³³ Mordini & Massari explain, "Digital biometrics differ from traditional biometrics both quantitatively (the digit format allows us to collect, store and process a vast amount of data electronically in a short period) and qualitatively (being numeric strings instead of icons, digital representations have different qualities from analogical representations). Current biometrics include fingerprints, ultrasound fingerprinting, iris scans, hand geometry, facial recognition, ear shape, signature dynamics, voice

recognition, computer keystroke dynamics, skin patterns, and foot dynamics."⁴³⁴ Upcoming (2nd gen) biometrics can probably employ methods like "improved face recognition, remote iris scanning, skin luminescence, brain wave analysis, body odor, etc."⁴³⁵

Today, the primary purposes of using biometrics are to differentiate one person from another via identification or verification processes, as well as to anticipate someone is behavior or intents.⁴³⁶ According to Sutrop & Laas-Mikko, "the first aim is related to the practical identity and accountability of the person that makes them trustworthy. What must be ensured is that the person identified is the person they claim to be, thus proving their rights, entitlements, and responsibilities."⁴³⁷ To get entry to a facility, for instance, the individual may need to have their fingerprints verified. Prior to this procedure, the person's fingerprint is scanned in order to provide them access rights. The second objective is more challenging; it entails automatically categorizing a person based on an automated assessment or judgment about that individual. A preset attribute that justifies include the individual in a group of people who might represent a danger or risk to society is the basis for the judgment.⁴³⁸

The knowledge of data collection is what distinguishes first gen biometrics from second gen biometrics. First-gen biometrics use physical contact for data collection and management, giving users the chance to become aware of what is going on, learn why the data is being acquired, and respond appropriately to protect their rights.⁴³⁹ Second-generation biometric surveillance involves remote and covert data collection and processing that is carried out without the subject's awareness.⁴⁴⁰ As a concluding concept for this part, what if society included a person's biometric profile with information about their beliefs, principles, and ideologies? Leavitt suggests "This, in essence, would make possible a measure (metric) of the mental aspect of one's

life (bio)."⁴⁴¹ This could be a broader definition of the term "biometrics," but does not language development include words taking on new meanings?

3.B.i.b. Applications

The applications of biometric use are infinite – as technology, science, and society grow, there will only continue to be opportunities to use and apply biometrics within many aspects of life. From using it to log in to personal devices to medical and big data analytics, biometrics has a massive reservoir of untapped potential. Seals, tokens, documents, and other conventional forms of identity may be forged or used fraudulently.⁴⁴² Biometrics will be susceptible to attack but are much harder to commit fraud with.

Medical

Medicine will soon enter the big data age. Clinical use of genomic analysis is advancing the development of targeted molecular treatments and pathogenesis. A biobank is one way to successfully use genomic data; depending on the project design, this kind of study participant pool might include patients from the general community.⁴⁴³ As a result, ethical concerns including privacy protection have grown. Due to the widespread usage of electronic medical records, a vast quantity of electronic health information has amassed in the clinical context. It is now feasible to collect health data from smart gadgets as they have spread. Its treatment becomes increasingly difficult as data from the clinical context and associated medical studies increase. Without reliable data management, it will be hard to defend the rights of study participants.⁴⁴⁴ Additionally ongoing are patient-focused information delivery and interactive research utilizing digital platforms. Big data gathered from various sources will be examined by artificial intelligence to provide new information.⁴⁴⁵
Future medical care and research have been suggested to be based on precision medicine, which treats and prevents illness by taking individual patient variances in genes, environment, and lifestyle into account.⁴⁴⁶ Any application of information and communication technology (ICT) for healthcare will be crucial in this age of customized medicine, sometimes known as "eHealth." Only a few of the issues with eHealth include security, express authorization for data exchange and future usage, standards, organization, and deidentification.⁴⁴⁷ The development of a trusting connection between study participants and researchers will inevitably lead to such new ethical difficulties.⁴⁴⁸

To provide patient-centered information in research, some efforts have adopted a new methodology. Teare et al. state that "One of these is RUDY, a study in rare diseases of the bones, joints, and blood vessels organized by the research team at the University of Oxford, which uses a patient portal that allows patients to enter their information online."⁴⁴⁹ Furthermore, it detects and gathers sporadic advanced illness or health information which, while not yet well studied, will soon be crucial data. The march toward consumer healthcare management is compatible with the transmission of information that is motivated by the requirements of the patient. This idea has already gained a lot of support in the US because of the development of technology that makes it simpler for patients to get and communicate vital health information with someone on their healthcare staff. Blue Button is an illustration of this endeavor.⁴⁵⁰ A website with a blue button indicates that it allows users to obtain their medical records. By making the information easily accessible for viewing, this aids patients, families, and caregivers in taking greater control of their health information and assisting in the making of better-informed choices.⁴⁵¹

This shift is crucial for putting any future healthcare system into place. Online patient submission of medical data, including genetic data and medical records, is predicted to increase.⁴⁵² As a result, it is anticipated that electronic techniques, such as online permission processes and health information exchanges, will alter the current landscape of medical research. Information and communication technologies may also make it easier for study participants to participate, feel empowered, and communicate with the researchers.⁴⁵³ Electronic technologies may, however, raise a few ethical concerns. Some of these ethical difficulties in both fundamental and specific areas will be covered in the following portion of this chapter.

Types of Personal Recognition

When utilizing electronic healthcare systems, it is thought that several types of authentication are required during registration and login. When a registered user returns to the network, they must log in. Since others might mimic the user when they log in, there is a greater risk of information leakage consequently. As a registered participant, you must authenticate yourself upon login. This may be accomplished in one of three ways: (1) using what you know; (2) using what you have; or (3) using who you are.⁴⁵⁴ Users are authenticated by "What you know" based on information that is personally known to them, such as a secret phrase or password. Other individuals may be able to impersonate the user if they can find out or guess this information. The risk increases if customers use a basic passcode that is easy to deduce and one that is the same between another website. However, it gets increasingly challenging to remember passwords as they grow more complex and are used more often.⁴⁵⁵ Even a complicated password may be retrieved through keylogging by a hacker or malicious computer program, making this method of authentication less secure.

"What you have" uses a user's possession of something, such a gadget, to authenticate them. To prevent unauthorized usage, users register their gadgets in advance. Then, the person using that device is verified. The usage of a token that creates a unique password is an option. Users may only utilize preregistered devices using this approach, which poses a security risk in the event of loss or theft of the registered device.⁴⁵⁶ Biometric authentication, sometimes known as "what you are" systems, is finally made possible by modern technologies. Users may be verified by the systems using their voice, iris, ear, face, or fingerprints. Biometrics was included as a means of authentication in the US Food and Drug Administration's guidelines for employing electronic consent.⁴⁵⁷ Biometric data, however, is delicate and has a built-in flaw: if disclosed, it cannot be modified. As a result, it must be protected carefully. Sutrop & Laas-Mikko explain "In identification functionality 1:N, one-to-many, or N:1, a many-to-one match is performed."458 In the case of 1:N, a single unique identity is compared to and matched with its corresponding identity within the infinite repository of identities stored in the server. In the case of N:1, the infinite number of identities in the database are compared and matched with a single unique identity. More commonly, a 1:N match is performed when a user logs into a database. The verification process's conclusion determines whether the identification claim is accepted or rejected. In contrast, the outcome of identification is a list of identities that have a chance of being the individual. Utilizing the identifying feature, a person may be uniquely linked to any further identifying data that has been gathered or kept about them.⁴⁵⁹

According to the most recent recommendations made by the US National Institute of Standards and Technology, the most durable technique now is two-factor authentication.⁴⁶⁰ Kogetsu et al. emphasize, "Sufficient security cannot be secured by single authentication, and the problems of authentication by username and password alone cannot be ignored. Combining two

or more factors is more secure. Two-factor authentication combining 'what you know' and 'what you have' is currently the most viable.³⁴⁶¹ This procedure might be used to get people's permission to share their genetic and other health information online. However, in order to improve security or reduce the likelihood of key loss, it could be necessary to utilize graphical passwords or other methods.⁴⁶² The kind of data kept or shared in the system affects how acceptable two-factor authentication is. For arbitrary medical or health information, such as blood pressure, gathered from home or using a smartphone, ordinary two-factor verification could be suitable.⁴⁶³ If data or information that might be used to identify a specific person is shared and maintained, stricter security procedures could be required. Consequently, biometric authentication will probably be highly beneficial in the future, but the process needs further debate. We believe "what you know" to have a significant role in this purpose; thus, the combination of "what you have" and "what you are" cannot be used to prove intent.⁴⁶⁴ It is thus vital to establish an authentication method that can attain higher levels of security.

3.B.ii. Ethical Challenges

3.B.ii.a. Fundamental Issues

Several ethical challenges concern the use and implementation of biometrics. These challenges have different levels of detail, consequences, and solutions and should be separated as such. Hence, I have divided the ethical challenges into fundamental issues, or those problems that lie at the core of biometrics dealing more in the theoretical and philosophical discussions such as human dignity and liberty, and specific topics, or those that deal with real-world application and have a more practical and direct effect on the world. This separation between fundamental and specific issues should help to understand that the ethical challenges posed by

biometrics are two-fold and, therefore, should have some form of a governance framework that will address the issues in both realms as the future of this technology develops.

Human Dignity & Individual Liberty

Physical traits and behavioral traits are transformed into biometric data via measurability. Only quantifiable anatomical and functional characteristics may be regarded as biometric characteristics. A human characteristic may be viewed as a physical, quantitative phenomena if it can be quantified.⁴⁶⁵ Physical parameters including length, mass, temperature, and electric current may be used to measure this. Although medical research has long quantified human biological characteristics for therapeutic uses, the question is now whether it is ethically acceptable to use this method to identify persons.⁴⁶⁶ As Mordini asks, "If identity and identification concern the essence of the individual, would biometric identification run the risk of reducing the richness of human identity to a sum of mere physical quantities?"⁴⁶⁷

The phrase "informatization of the body" relates to this idea, which describes the dispersion of a person's physical and behavioral characteristics throughout the global information network via digitization.⁴⁶⁸ The fundamental worry with the "informatization of the body" is how the digitization of bodily characteristics may impact how we portray ourselves and lead to "disembodiment" processes.⁴⁶⁹ While privacy experts and civil and human rights groups are concerned about the possibility of function creep, which is "defined as identification systems incorporating biometric scanning that would gradually spread to additional purposes not announced or even intended when the identification systems were initially implemented" philosophers frequently worry about "informatization of the body because it would touch our inner nature, the human essence."⁴⁷⁰

The ability of biometrics to extract out of any biographical details, and to concentrate primarily on "bare life" in the opinion of some academics, carries the potential of a future in which people will transform from their biological bodies into "cyborgs," where they will be free to build or merely recreate themselves and alter their individual identity as though they were changing apparel.⁴⁷¹ Biometrics and information technologies' self-descriptions as "technologies of immateriality" have drawn criticism from other, more critical viewpoints.⁴⁷² Dematerialization is a process that some academics have described as moving from item, commodity, symbol, to simple information.⁴⁷³ An item inhabited by "us" would be created by the body via this disembodiment process, similar to those used by biometric technology.⁴⁷⁴ Mordini & Massari state "Disembodiment occurs when a person's identity is separated from their physical presence. This dematerialization leaves the body and its personal identifiers as objects separate from the identity of the person attached to them.⁹⁴⁷⁵ This has prompted some sincere inquiries into the possibility that biometrics endangers the body's humanity and violates human dignity.

Any personal identification procedure does in fact imply that people are recognized as having rights and duties, which may be seen as restricting individual freedom.⁴⁷⁶ Furthermore, biometric applications cannot avoid providing additional information that might be exploited and is unrelated to recognition, making them far from being a "clean" identifying method. ⁴⁷⁷ The following section will examine a few ethical and privacy problems that specific applications and systems brought up.

Universal Declaration on Bioethics and Human Rights

Biometrics are considered to be "borderland" technologies since they combine knowledge from several fields, including information, security, the life sciences, and medicine.⁴⁷⁸ The Universal Declaration on Bioethics and Human Rights, which expressly tackles "ethical issues

related to medicine, life sciences, and associated technologies as applied to human beings," is a suitable source to refer to when considering ethical issues with biometrics.⁴⁷⁹ Mordini points out "The main principles of the Declaration that are relevant to biometrics are likely to be Article 3 (about human dignity and human rights), Article 6 (about consent), Article 8 (about respect for human vulnerability and personal integrity), Article 9 (about privacy and confidentiality), Article 10 (about equality, justice, and equity), Article 11 (about non-discrimination and non-stigmatization), and Article 12 (about respect for cultural diversity and pluralism)."⁴⁸⁰

Article 3 of the Universal Declaration on Bioethics and Human Rights says, "Human dignity, human rights, and fundamental freedoms are to be fully respected. The interests and welfare of the individual should have priority over the sole interest of science or society."⁴⁸¹ When the interests of a person and those of society collide, the individual must win ultimately. Undoubtedly, it is often viable to approach security issues from a win-win perspective. Increasing both individual rights and society security might resolve many perceived tensions between those two requirements. However, the Declaration emphasizes that when a genuine dispute arises, the only viable and moral solution is to put individual rights first.⁴⁸²

In accordance with Article 2 (Consent), "scientific research should only be carried out with the prior, free, expressed and informed consent of the person concerned."⁴⁸³ Informed consent is a need for all types of scientific study, not only biomedical research, as this article demonstrates. This implies that when human subjects are involved, biometric studies should adhere to the same strict guidelines that apply to the area of full consent in healthcare. Human fragility and individual dignity are preserved according to Article 8, which states that "individuals and groups of special vulnerability should be protected, and the personal integrity of such individuals respected."⁴⁸⁴ The article's significance for biometrics centers on the concern

that these systems would not be built adequately to provide complete accessibility to poor and handicapped groups and individuals. Engineers must use universal design concepts, methodologies, and tools to create biometric devices that are usable by all people, including those with impairments, especially when it comes to extensive applications.⁴⁸⁵

Article 9, which is concerned with privacy and confidentiality, definitely has a place for biometrics. This clause explicitly demands that collecting personal information be compliant with "international human rights law."⁴⁸⁶ It suggests that when adopting biometric applications, a more extensive human rights effect assessment should be necessary, including but not exclusive to a privacy impact analysis.⁴⁸⁷ Mordini summarizes that "Article 10 (equality, justice, and equity), Article 11 (non-discrimination and non-stigmatization), and Article 12 (respect for cultural diversity and pluralism) collectively draw attention to the risk that biometric identification could favor discrimination, stigmatization, and racial and ethnic harassment."⁴⁸⁸ This is a genuine issue, especially for big apps and biometric databases for profiling individuals. But this does not signal the inevitable demise of biometrics. On the other hand, by facilitating civic identification and consequently increasing equity, justice, and fairness, biometrics may turn out to be an essential component of development projects in low-income countries.⁴⁸⁹

3.B.ii.b. Specific Issues

Privacy & Data Protection

A digital version of the biometric might be lost, stolen, or otherwise compromised. Numerous problems, not the least of which is privacy, might arise from illegal access to biometric storage systems. Because the biometric cannot be altered and, if compromised, poses a risk to the donor's life, abuse of biometric data is a severe problem. John R. Vacca notes "Even when used as intended, the biometric control captures personal information, such as fingerprints,

iris scans, palm geometry, and more."⁴⁹⁰ Even if users are informed that their biometric information is being utilized, policy restrictions prevent them from always having the option to refuse. They may need to employ biometrics to do their job or to access connected systems or services. Some people may not want to utilize it because of the "Big Brother" idea. Others, however, could really fear that the data might be used against them to monitor their activities, fake exchanges, or for other nefarious reasons.⁴⁹¹

Biometric technology is expanding and will continue to pick up steam. Vacca continues, "Organizations must accept biometrics and determine the best approach to ensure that they are used appropriately, that the information stored is adequately secured, and that data collected on the user remain private."⁴⁹² Essential factors including data gathering, storage practices, and the consent of the persons from whom the information are being gathered must be carefully taken into account during an audit or review.⁴⁹³ In order to decide, it is also necessary to take into account the relevant rules if the data collection, retention, and usage were lawful as well as any hazards connected with using biometrics within a business.⁴⁹⁴

Historically, biometric-based technology has been seen as dehumanizing and a danger to people's right to privacy. A person's location, for example, leaves a trail of unmistakable private information even as identifying technologies get more and more secure.⁴⁹⁵ Jain et al. stress "In the case of biometric-based identification, this problem is even more severe because the biometric features may inform others about an individual's medical history or susceptibilities, e.g., retinal vasculature may divulge information about diabetes or hypertension."⁴⁹⁶ As a result, there is a valid worry about privacy concerns related to biometric identification. It is not acceptable to use a biometric measurement's data for purposes other than those for which it was created.⁴⁹⁷ It is challenging to guarantee that the biometric evaluations gathered will just be

employed for the intended goal in just about any linked information processing model. It is challenging to assure that the biometric evaluations collected will only be utilized for the intended purpose in almost any linked information processing architecture.⁴⁹⁸

Surveillance

Governments and organizations, like airports or railroad associations, have a propensity to be more interested in the kind of person you are than your identity, including if you have malicious or "hostile" intents.⁴⁹⁹ While this is happening, it is becoming more clear that even more advanced identity documents, such biometric passports that include a face picture and fingerprints, will not completely protect us all from malicious intent.⁵⁰⁰ It is now quite evident that just knowing someone's identification will not be enough to stop a danger. These factors have led to a continual hunt for fresh surveillance techniques.⁵⁰¹

Second-generation biometrics allow for the observation of every person in public areas. Since the act of monitoring suggests a lack of confidence in society as a whole and the identification of people who should not be trusted based on some trait that is assumed to indicate a specific goal, surveillance may include broad mistrust against everyone.⁵⁰² Continuous and ongoing monitoring might result in a double morality where everyone could be acting out. Some researchers worry that individuals will change how they behave as monitoring awareness increases.⁵⁰³ As Ploeg et al. say, "Society is at risk of losing its innocence. Before you know it, we will all behave similarly so that we are not seen as suspicious."⁵⁰⁴ Those who have experienced living in a totalitarian regime with its extensive monitoring system are acquainted with these effects.⁵⁰⁵

In addition to offering tremendous potential for conventional forms of monitoring, biometric identification technologies seem to be central to the vision of the emerging "society of

control."⁵⁰⁶ One of the critical components of globalization is the constant movement of people, which biometrics may identify, track, and monitor.⁵⁰⁷ In theory, biometric technology could even give each person on the planet a singular, unmistakable identity by enabling identification procedures on a worldwide scale. If the world embraces a more thorough use of biometrics – the end of anonymity could be at hand. It would be feasible to compile a comprehensive record of a human's life in this imagined new world, including their whereabouts, spending patterns, interpersonal ties, connections, etc. This end to anonymity would also mean that any individual could be tracked, even without their knowledge creating an asymmetry in information.⁵⁰⁸ Compared to other freedoms like the right to search for, get, and share information, which this kind of system aims to restrict, this amount of monitoring and accompanying breach of privacy does not stand well.⁵⁰⁹

Large Scale Applications

Two broad categories of issues are anticipated for a fingerprint matcher. The first category of problems deals with circumstances in which it is vital to confirm or authenticate a person's identity. As a conceptual foundation for one-to-many matching, this one-to-one matching issue is of particular importance.⁵¹⁰ Jain et al. describe, "The second, more challenging problem occurs when it is essential to ensure that a particular database contains only a single entry for any given individual. This happens in the case of social services wherein one wishes to prevent individuals from collecting welfare under multiple aliases or in the case of identity card issuance."⁵¹¹ For this identification issue, it is necessary to scan an extensive database of people to see whether a particular person is already present.

Criminal justice applications have received most of the attention in previous work on automated fingerprint recognition systems. It is reasonable to demand the hiring of skilled

fingerprint inspectors to review a significant set of possible matches in the criminal justice application space since there is a relatively substantial cost associated with missing a possible match, such as the release of a wanted criminal.⁵¹² Additionally, a large degree of filtering is almost always required for felony fingerprint verification searches, which reduces the size of the data that is checked. This filtering may have an impact on categorization based on the overall ridge pattern, as well as demographic screening based on factors like age, race, and region.⁵¹³ Last but not least, criminal justice fingerprinting systems save pictures of each of the 10 fingers.

Vast social service or national identity registration requests, which will need inquiries and archives holding imprints from a significant fraction of the state's or country's whole population, require the ability to examine records of millions worldwide.⁵¹⁴ Jain et al. continue, "An understanding of the systematic changes in the error rates of a fingerprint identification system with database size is also needed as part of a framework for extrapolating measurements from small benchmarking or sample databases. Criminal justice fingerprint systems are not characterized as allowing extrapolation of measure performance data to large database sizes."⁵¹⁵ Characterizing outcomes in the form of how often they occur in the top or top 10 places is given special importance. These metrics help compare results obtained on different databases but are unsuitable for estimating identification error rates on more extensive databases.⁵¹⁶

3.B.iii. Creating a Biometrics Governance Framework

3.B.iii.a. Regulating Biometrics

The concept of limited conservation of data pertains to how personally identifiable information about data subjects is stored.⁵¹⁷ Data should only be retained for as long as is required to fulfill the reasons for which it was obtained. Additionally, the subsequent processing of personal data is subject to this regulation. If the administrator wishes to maintain the data after

it has expired and lost its usefulness, it should be anonymized.⁵¹⁸ If no identifying components remain in a single collection of personal data, the data are considered anonymous. Any information that may be used to re-identify a specific person should not be included in the data. Successful data anonymization removes any traces of personal information.⁵¹⁹

National Level

There appears to be a legitimate conflict between personal and communal values when it comes to biometrics. The problems that remain unresolved primarily concern who will determine that biometric data will now be gathered to safeguard domestic security, the process used to make this choice, if appropriate discussion is allowed, and whether key individuals in society are to be considered, and whether or not the utilization of biometric technology as a protective measure is appropriate despite the dangers involved.⁵²⁰ The "risk society" concept's creator, Ulrich Beck, has noted that the debate on risk starts where our confidence in our safety and our faith in development stops.⁵²¹ The idea of risk thus alludes to a "peculiar, intermediate state between security and destruction, where the perception of threatening risks determines thought and action."⁵²² Whenever people talk about risk, people are worried about anything that could happen if they do not do anything to alter the way things turn out. Risk measures one's exposure to risk and the possibility of suffering loss.⁵²³ Risks are hence estimates of potential occurrences; they happen in the context of uncertainty. As a result, we have limited capacity to assess hazards, take precautions against them, and modify the proportionality of these precautions.⁵²⁴

Ethics training and applying relevant laws and regulations are crucial first steps in ensuring privacy. The scientific community has had to balance open data access with genetic privacy. Technology improvements are accompanied with worries, discussions and conflicts on a range of subjects, such as ethics, rules, and legislation pertaining to the safeguarding of genomic

privacy.⁵²⁵ Although many of these rules and regulations have been slow to take effect, they nonetheless significantly influence personal genomics research, teaching, and clinical practice. Shi & Wu point out, "As indicated by recent privacy and identity infringement work, protecting genomic anonymity becomes next to impossible because researchers increasingly combine patient data with many types of data from social media posts to entries in genealogy databases."⁵²⁶ In order to educate students, investigators, medical professionals, and society concerning these mechanisms for safeguarding genomic security, it is crucial to develop and extend educational activities.⁵²⁷

The Health Insurance Portability and Accountability Act of 1996's (HIPAA) Standards for Privacy of Individually Identifiable Health Information, or the "Privacy Rule," is a crucial piece of health information protection legislation.⁵²⁸ The Privacy Rule was created to control how covered firms used and disclosed customers' health information. It lays forth standards for people's rights to know how their health data is used and to exercise that control.⁵²⁹ It should be noted that business partners, insurers, data clearinghouses, and health providers are all protected organizations under the privacy regulation. HIPAA does not protect health and medical information from these listed organizations. As a result, recently founded, for-profit sequencing and genetic testing businesses (like 23 and Me) produce a lot of healthcare data that is sensitive and identifiable yet is exempt from HIPAA rules.⁵³⁰ The recently emerging user personal health information from social networks, activity trackers, digital healthcare apps, and home paternity testing is not protected by HIPAA.⁵³¹ Shi & Wu conclude, "Additionally, the privacy rule only protects identifiable health information, and there are no restrictions on the use or disclosure of de-identified health information. Metadata, such as age, race, and geographical regions, can be publicly accessible."⁵³² The potential risks to genetic privacy are one illustration of how such

non-identifiable data and documents may be utilized to reveal private information about particular people.⁵³³

Global Level

Currently, certain countries—like Finland, Sweden, and Norway—have established national biometric systems; other countries—like France and the United Kingdom—have not, but plan to.⁵³⁴ Germany deserves special attention because the parliament previously disallowed the creation of a national biometric database. The German parliament claims that a national biometric registry does not include one of the fundamental rights protected by German law, which is the ability to decide what information is publicly available.⁵³⁵ The Fourth Amendment's ban on unwarranted searches and seizures is the main impediment to collecting biometric data in the USA.⁵³⁶

Although there are several reasons to share biometric data with other parties and organizations, the most common one is to facilitate international law enforcement and security. The "Prüm Treaty" and the "Five Country Conference Protocol" are the international agreements pertaining to the exchange of biometric data that are better known.⁵³⁷ 13 EU members have signed the Prüm Treaty, which permits them to share internet access to their respective countries' databases for DNA and fingerprints.⁵³⁸ The Five Country Conference Protocol authorizes the sharing of fingerprints or other biometric information for the purposes of intelligence and law enforcement.⁵³⁹ The United States of America, Canada, the United Kingdom, and New Zealand are all parties to the protocol. The Association of Southeast Asian Nations (ASEAN) nations are also working on a comparable multinational endeavor.⁵⁴⁰ Additionally, several bilateral agreements exist enabling the exchange of biometric data between specific nations.

Data sharing raises substantial political and ethical challenges in addition to several technical issues, the main ones of which are related to standards and technological compatibility.⁵⁴¹ Reciprocity is most likely the fundamental problem. Mordini explains "Provided that all actors and principals agree with the need to ensure a basic level of data protection and respect for informational privacy in these databases, the international community has widely divergent views about acceptable levels of data protection. To date, there is no agreement on common international privacy principles that could provide a framework for using biometric data."⁵⁴² For instance, the EU forbids the transmission of peoples' personal information to an area or nation outside the EU unless that region or nation guarantees a comparable degree of data protection as the EU.⁵⁴³ As has been the case, for example, with the never-ending argument between the United States as well as the EU over the sharing of passenger data, such official views will always lead to deadlocks.⁵⁴⁴ In the worldwide and interconnected world of today, while it is difficult to restrict the flow of data to a select few nations, it is also unrealistic to think that all nations would agree with the EU's stance on the protection of personal data.⁵⁴⁵ This brings up the major, and arguably more troubling, ethical problem with the exchange of biometric data internationally.

3.B.iii.b. Global Examples

There are numerous global examples of biometrics governance frameworks – but few are more robust than the EU since the implementation of the GDPR. It is *the* standard, right now, for the most comprehensive biometric data privacy and security. Therefore, the analysis will begin with the European Union and then move to look at UNESCO's example to further what a global biometrics framework should aim for.

European Union

Since the passing and implementation of the GDPR, the EU has been one of the global leaders in data privacy, security, and surveillance – biometrics included. However, many essential legal documents were implemented before the GDPR that set the stage for its arrival, including the Data Protection Directive, the Charter of Fundamental Rights of the European Union, and the Treaty on the Functioning of the European Union – all of which will be touched on in this section. The Data Protection Directive governs how personal data is processed in Europe. "It affirms that the processing of personal data is subject to principles of purpose, specification, and proportionality: personal information may be collected only for specified and legitimate purposes and not further processed in a way incompatible with those purposes"; in addition, such information needs "adequate, relevant and not excessive and about the purposes for which they are collected and/or processed."546 According to the proportionality principle, there should be a proper balance seen between quantity of information gathered, documented, and given weight or worth of the goals being pursued.⁵⁴⁷ Data volume might relate to the amount of processing being done. However, because even a small quantity of sensitive material entails a great deal of danger, it may also relate to the type of data being handled. ⁵⁴⁸

Securing proportionality may be challenging since many variables need to be considered, many of which are context dependent.⁵⁴⁹ In theory, only extreme instances may be resolved in a way that is undisputed. In most real-life situations, no proportionality decision can be made without carefully considering contextual elements. These considerations encompass both the processing's purpose and its overall context, such as the national security, educational, healthcare, or workplace contexts; the need for the handling and the existence of workable or less threatening solutions; the type of data is being generated and its vulnerability, such as whether the data reveal personal health information.⁵⁵⁰

A basic human right that applies to the usage of biometric information is the right to privacy when processing personal information. The Charter of Fundamental Rights of the European Union and the Treaty on the Functioning of the European Union both include provisions that are fundamental to this right.⁵⁵¹ Deliversky & Deliverska state, "According to legal instruments, the right to personal data protection is a universal right provided to everyone, as this protection must comply with a person's fundamental rights and freedom. The European Convention on Human Rights proclaims the right to personal data protection. Through this mechanism, the right to respect private life has been guaranteed, as well as the right to home and correspondence, and it lays down the conditions under which restrictions of the right are permitted."552 The primary piece of law protecting persons against the release of information at the level of the European Union is Directive 95/46/EC. The free interchange of personal data is addressed by this piece of law, sometimes referred to as the data protection directive. The European Commission advocated for updating data protection laws at the EU level at the beginning of 2012 since ensuring security is the main goal of all laws governing the protection of personal data.553

Article 7 of the EU Data Protection Directive states that "no data collection can go unnoticed by the subject being monitored."⁵⁵⁴ The goal should be for the subject to be conscious of the many types of information being collected on them. However, precisely this is what integrated biometrics would stop. Article 7, paragraph 2 stipulates that paragraph 1 is not applicable in the event of "processing of data relating to offenses, criminal convictions, or security measures," provides the escape hatch from this legal predicament.⁵⁵⁵ However, is it proper to apply the idea of security precautions to every technology used in any situation? A never-ending expansion of an ambiguous security area cannot be used to justify the increasing

prevalence of embedded biometrics.⁵⁵⁶ Additionally, it does not appear that visitors' privacy rights will be honored even if they are informed that they will be inside a biometrically monitored location. Once individuals are accustomed to warning labels, they are less likely to be seen. Since the EU is part of the United Nations (UN), looking at a more significant part of the world through the lens of UNESCO – a specialized agency within the UN – seems to be a logical next place to look for how to create a biometrics governance framework.

UNESCO

Emerging technologies provide UNESCO and its partners with a wide range of new possibilities and channels for action, allowing them to actively participate with the rise of the information society and demonstrate the UNESCO guidelines below while still carrying out their responsibilities. An advisory board is created in the first suggestion.⁵⁵⁷ International research on the moral consequences of cutting-edge technology should regularly benefit from advice from an advisory board. Top technology specialists are sought after by policymakers and the public, and UNESCO's mission and certain university institutions' technology-focused programs are complementary since both organizations support principles such as information availability and observance of human rights. As a result, by establishing a specialist advisory committee on information ethics, UNESCO would be able to benefit from the expertise of leading technologists and win the recognition they deserve.⁵⁵⁸

One such group might help UNESCO maintain the kind of collaboration that were discussed during the World Summit on the Information Society while avoiding the issues that impede important, rising conferences.⁵⁵⁹ This advisory board should consist of youngsters and young people from all around the globe as well as technology and info ethics professionals to guarantee that the group's results are adaptable and forward-thinking. Additionally, UNESCO

might work with academic institutions to organize brainstorming sessions to address essential concerns in the future.⁵⁶⁰

The establishment of a technological community to safeguard personal data is the second suggestion.⁵⁶¹ Controlling the flow of personal data will be essential for ensuring that people can access information and exercise their human rights in the information age. The poll on the "Ethical Implications of Emerging Technologies" recognized this as one of the most significant factors since technology would likely be utilized more and more to exert control over people's lives.⁵⁶² Here, it is suggested that the issue of digital identity management be the focus of an uncommon cooperation since it will be the cornerstone for subsequent developments involving the transfer of personal data and because the market is preparing to introduce new products in the upcoming years.⁵⁶³

A hole in the global system would be filled by significant efforts in the security of personal data. Personal data protection is urgently needed right now. The Montreux Declaration, issued in September 2005 and calls for multilateral norms in this field, was endorsed by the International Group of Data Protection and Privacy Commissioners in recognition of this need.⁵⁶⁴ The need is also recognized by computer scientists, notably well-known members of the identity game business group and the World Wide Web Consortium (W3C), who believe their effort has immense potential, both for good and bad. Nevertheless, a lot of nations look for guidance on the moral principles their laws should uphold.⁵⁶⁵

Since the Council of Europe (COE) and the Organization for Economic Cooperation and Development (OECD) both enacted legislations well over 20 years ago to address the issue of incorrect treatment of personal information in the digital era, the idea of safeguarding personal data is not a new one.⁵⁶⁶ These measures make up a robust set of safeguards when taken

simultaneously. These OECD principles include transparency, accountability, openness, individual engagement, purpose specificity, usage restriction, and limited collection.⁵⁶⁷ An antidiscrimination clause is included in Article 6 of the COE Convention.⁵⁶⁸ The COE regulations, on the other hand, only apply to signatories, while the OECD guidelines are not legally obligatory. These instruments are purely legal in character, even if they were founded with international force. Without adequate worldwide enforcement, their contents are likely to be disregarded, particularly by non-signatory countries. They are unrealistic for ensuring the appropriate global handling of personal data by machines. UNESCO might assist the technology sector in creating instruments to enforce personal data safeguards when the legislation is deficient by collaborating with W3C and academic institutions.⁵⁶⁹

3.B. Conclusion

Individuals fall prey to the delusion that personal identification in and of itself endangers fundamental rights and invades the private realm. Any technique for determining a person's identity indicates that they are recognized with rights and duties, which may be seen as restricting their freedom. Nevertheless, without personal identities, there would be no rights and freedom. No one's civil, political, or social rights can be imposed on them since they are nameless. Only an identifiable person and someone with a public identity may assert their rights, including the right to remain anonymous. There is currently a lack of public knowledge of the types of data gathered and their intended uses. Since humanity's ability to cooperate with security protocols varies depending on their perception of risks as real and the effectiveness of security protocols in relation to those risks, it would be a serious mistake to ignore the public's behavior and physical attributes when conducting planned covert monitoring.

Many nations are working on second-generation biometrics systems, but there has not been much public debate about the benefits and drawbacks to employing this new technology. Imagine there is a risk of confidentiality being violated or data being misused after being gathered without permission or, in some instances, even with the subject's knowledge. In such events, individuals can feel duped and distrust governments more generally. The future reaction might be severe since it is hard to rebuild public confidence after misuse. Data should thus only be retained if it is required to fulfill the goals for which it was obtained to prevent this possible public reaction. Additionally, the subsequent processing of personal information is governed by this regulation. If the administrator wishes to maintain the data after it has expired and lost its usefulness, it should be anonymized. A collection of personal data becomes anonymous when all aspects that may be used to identify a person are taken out. The data should not include any elements that might lead to the re-identification of specific persons. Successful data anonymization removes any traces of personal information. This governance framework is then detailed at the national and global levels with specific examples of the European Union & UNESCO to show examples of existing early biometrics frameworks.

3. Conclusion

This chapter summarized the data battle between the individual's autonomy and rights versus population health and advancing the common good. Recall that the key challenge is whether decision-makers can strike a balance between the need to protect real patient and clinician concerns and the need to place value on health records for humanity. To provide proof without jeopardizing these legitimate interests, it will be necessary to consider whether data used for research, like that by the Federal Coordinating Council for Comparative Effectiveness Research, can be handled properly and securely. Regulation and control of biometrics are

essential since they are one of the main types of data utilized in public health statistics. While data analytics may help with the development of a framework for biometric governance, various issues with the use and abuse of data analytics itself need to be addressed. The next chapter describes some of those concerns in two senses: one practical, within genomics, and another theoretical, within the future of digital technologies themselves.

CHAPTER 4: ETHICAL CHALLENGES OF DATA ANALYTICS IN HEALTHCARE <u>4. Introduction</u>

The earlier chapters focused on the benefits and opportunities that data analytics governance can provide the healthcare landscape through its finance and delivery, as described in Chapter 2. In Chapter 3, data analytics governance assistance is detailed at the micro level, with the individual's autonomy and rights paralleled against the governance at the macro level with population health and the Common Good. The fourth chapter takes a different look at data analytics and focuses on the ethical concerns and challenges this technology can display. The social policy of genomic data, such as genetic discrimination, common heritage, and more concerns with distributive justice, are detailed. The section details challenges in data analytics with the autonomy paradigm and cloud computing, respectively. These issues pose a challenge to data analytics due to cloud computing's de-centralization of data storage. A critical factor in respecting the privacy and autonomy of the individual is having secured, centralized data storage that can easily be maintained and traced back in the case of a data hack.

The rest of the chapter will lay a philosophical/theoretical foundation discussing the impact of technology in healthcare via the digital revolution and its relationship to society while looking at the future of this dynamic relationship and how data analytics could vastly alter this relationship down a slippery slope of a dehumanized world and human enhancement. A

cautionary approach is taken in which concepts like technological determinism and the Precautionary Principle are explored while using these concepts to assist in repositioning a medical ethos for the future. An applied analysis of these concepts is used within the topic of human enhancement citing ethical concerns and providing a foundation to advance the debate.

4.A. The Ethical Challenge of Data Analytics in Genomics

4.A. Introduction

The twenty-first century has, without a doubt, thrown humanity into the informational technology age stemming from famous and influential inventions in the digital world. Innovations such as the smartphone, social networking, high definition, commercial GPS, and cloud computing –to name a few are some of the reasons why society lives in a digital world. Big data is one term used in describing all this digital information, and its potential uses for the world and the healthcare industry are limitless. However, with all this informational data comes a big data problem, and it, like many other technological ethical issues, must be addressed. Issues surrounding privacy, security, and consent concerning your personal data (especially genetic data, as discussed in this chapter) are of the utmost importance to keep the updated policy. The social policy of genomics highlights genetic discrimination, common heritage concept, and distributive justice in genomics and insurance companies. The section concludes by giving a global privacy governance framework that will help set the policies in the right direction to address the ethical loopholes in the big data dilemma.

4.A.i. Social Policy of Genomic Data

Big data, while providing numerous benefits to healthcare described in the previous section, obviously has several ethical implications that must be monitored and maintained as time progresses. Security, privacy, and informed consent are some of the most significant areas

of moral concern. However, there is reason to argue that the social policy of genomic data has the most potential cause for concern. This area involves all three previously mentioned concerns and adds new situations, such as genetic discrimination and problems with distributive insurance justice. Genetic discrimination can occur and affect life and health insurance as well as the workplace despite the current policies to fight it. Then, a different view is expressed by looking at genomic data as a common heritage of humanity. Any findings should benefit society, as declared by UNESCO. This section concludes with a look at genetic data regarding life insurance informational symmetry and how other nations handle genetic testing and insurance.

4.A.i.a Genetic Discrimination in Health Insurance

While it may seem evident that discrimination of any kind – including genetic as described here – is wrong, it is not always that simple to prevent. As technology and big data continue to advance rapidly, so do the availability and accuracy of genetic testing. While no one can choose their genetic makeup or predispositions, insurance companies still use this information to charge higher premiums to those who are genetically disadvantaged. While policies have been passed to reduce this discrimination against insurers as well as employers, these current policies are not complete and provide loopholes that insurers and employers can exploit. The ethical course of action would be to continue passing legislation that will patch these loopholes while maintaining a free and stable market.

Genetic Testing Concerns

Genetic testing has numerous benefits. For example, "Ashkenazi Jewish women who test positive for a mutation in the BRCA1 or BRCA2 tumor suppressor gene have a lifetime breast cancer incidence of 82 percent, compared to less than 20 percent for the female population at large."⁵⁷⁰ Even though a negative test result cannot guarantee that an individual will remain

cancer-free, partly because many instances of breast cancer have causes other than genetics, having a positive test result can lead to better proactive outcomes such as routine screening, more effective treatments, and lives saved.⁵⁷¹ In the same sense, for example, a young child without symptoms of Huntington's disease but with a family history of the condition, may think that only a genetic test will give them the capacity to make future-oriented responsible decisions such as whether to raise a family or the type of career to achieve. The knowledge that a genetic test could give this individual, especially in the case of Huntington's disease, which does not have a treatment or cure, may not be seen as a blessing to some, who would rather have the peace of mind that ignorance can bring them.⁵⁷² However, for numerous people, the availability and accessibility of genetic testing have been a blessing of priceless information.

At the same time, the widespread use and availability of genetic testing have revealed fears of discrimination in various areas: employment, health, life, and disability insurance, with the underlying motive of creating a "genetic underclass."⁵⁷³ Further, it is precisely these kinds of fears that are cause for public alarm and concern in the genomic revolution.⁵⁷⁴ While many may disagree about the extent of genetic discrimination that takes place, a general consensus exists that opposes genetic discrimination of any kind. As with many public opinions, views range from "no amount of such discrimination is acceptable" to more detail-laden ideas.⁵⁷⁵ Regardless, the consensus against genetic discrimination is remarkable, as governments in both North America and Europe have implemented actions to outlaw various parts of it.⁵⁷⁶ The following subsection will focus on some legislation enacted in the United States to prohibit various forms of genetic discrimination.

GINA & ADA

The Genetic Information Nondiscrimination Act (GINA), which had been the subject of a lengthy 13-year congressional debate, was finally approved on May 21, 2008, and it went into effect.⁵⁷⁷ Adding to the federal and state protections already in place for genetic discrimination, such as employer-sponsored group plans in HIPAA, GINA acts to outlaw employment and health insurance genetic discrimination⁵⁷⁸

The Americans with Disabilities Act (ADA), like GINA, prohibits discrimination in the workplace based on a person's impairment. The ADA's scope of which individuals considered to have a disability is somewhat limited. Therefore, it does not outlaw all forms of discrimination in employment. Disability is described under the ADA as "a physical or mental impairment that substantially limits one or more of the major life activities of an individual, a record of such an impairment, or being regarded as having such an impairment."⁵⁷⁹

The ADA and GINA cover individuals in an opposite parallel manner in the most severe cases. When it comes to genetic discrimination in employment, GINA covers the individual, even when asymptomatic; however, the ADA does not. Conversely, the ADA will cover the symptomatic individual, which is not specifically covered in GINA. The lack of coverage and problem for most lies in the middle of these extremes. Within the ADA, people with "mild, temporary, or pre-symptomatic conditions" do not meet the criteria for a disability as previously mentioned.⁵⁸⁰ Likewise, under GINA, people with a genetically-based anomaly but not physically symptomatic are not going to be covered.⁵⁸¹ This is one narrow way to view how genetic data should be seen and handled, resulting in confusion, discrimination, and misunderstandings. A different way to look at genetic data without prejudice is the view that it is the common heritage of humanity. Any breakthrough findings regarding genetic data should benefit humanity as a whole, as declared by UNESCO.

4.A.i.b. Common Heritage

The idea of a 'common heritage of humankind' was used in international law as far back as the 1960s to manage universal resources like outer space and the ocean floor.⁵⁸² The argument was that these "things" are a common heritage of mankind, they need to be utilized for goals that advance the interests of all people.⁵⁸³ Interestingly, this concept dates back even further to traditional Roman law, which distinguishes a "category of *res extra commercium*, or things outside of commerce; property that cannot be exchanged."⁵⁸⁴ A few examples of this common property were considered to be the seas and rivers. This common property category made a law that these things could not be traded or owned by anyone or any nation.⁵⁸⁵

Initially, this concept only applied to those resources like the moon, the ocean floor, and the Antarctic. Still, this notion quickly added vital resources that were of benefit to humanity, even if they resided within a specific country, such as the rainforests. As Henk ten Have explained, "labeling areas as common heritage means that they are vital for the survival of humankind. It also means defining global responsibilities."⁵⁸⁶ The concept of common heritage was first labeled regarding the human genome by the HUGO Ethics Committee. It was followed by UNESCO in its Universal Declaration on the Human Genome and Human Rights (UDHGHR) in 1997, declaring that "the human genome is the heritage of humanity."⁵⁸⁷ Universal Declaration on the Human Genome and Human Rights (UDHGHR)

The UDHGHR, which was approved by the UN General Assembly and UNESCO in 1998, is recognized as the pioneering effort for international bioethics governance.⁵⁸⁸ This legislation sparked an international moral debate regarding the human body and life with financial considerations. The discussion produced two notable aspects of global bioethics governance. The first aspect regards that the field of science is seen as a global commons

requiring shared principles. Starting in 1990, The Human Genome Project created knowledge that should be available to all globally, even though specific applications of the project can be privatized for various reasons. The new understanding should benefit humanity. It should be widely disseminated and available to countries not involved in the research endeavor. There is, therefore, a need to go beyond a strictly economic perspective since the "genome is the heritage of humanity."⁵⁸⁹

The other aspect of global bioethics governance is a greater emphasis on human rights. Genetic technologies can limit reproductive freedom and promote sex selection, thus introducing new forms of discrimination and stigmatization. Access to genetic services and potential new treatments can also be limited for various reasons in different countries. The new knowledge will only benefit humanity if special attention is paid to ethical concerns, particularly the human rights implications. Article 1 of the UDHGHR states, "The human genome underlies the fundamental unity of all members of the human family, as well as the recognition of their inherent dignity and diversity. In a symbolic sense, it is the heritage of humanity."⁵⁹⁰ This introductory statement has several practical implications: the genome cannot be appropriated as someone's property and finically exploited (Art 4); benefits from scientific advances should be available to all (Art 12); research results should be used for peaceful purposes (Art 15); scientific knowledge and information should be freely exchanged and publicly accessible (Art 19).⁵⁹¹

The UDHGHR was intended as an incentive for national legislation. Since important federal legislation is lacking, and global human rights regulations are too vague and unspecific in this area, there is an opportunity for a global initiative, not to impose a framework on all countries but to start a global dialogue on how human genetics can develop for the benefit of

everyone. But even this modest objective requires subtle processes and practices of deliberation and interaction to defuse controversies that otherwise result in ineffective statements.⁵⁹²

Global Public Goods

Global public goods are a notion in international law that resembles common heritage in certain ways. In the 18th century, the concept was conceived by David Hume and Adam Smith. However, it lost much popularity until the widespread debate by economically challenged countries for global public access to AIDS drugs in the twentieth century.⁵⁹³ Benefits from an item must not be exclusive for it to qualify as a global public good, and its use should not deplete the available supply for others or cause its use to be restricted.⁵⁹⁴ Additionally, a global public good should provide advantages that are accessible to individuals from various backgrounds, ages, and cultures.⁵⁹⁵ For example, knowledge about the human genome could count, given that it satisfies each of the aforementioned requirements, as a global public good.⁵⁹⁶

In designing and implementing genomic databases, a key idea has been the idea of global public goods. To illustrate further, a HUGO declaration from 2002 about human genetic databases said: "population genomic databases and the primary sequence data contained therein to be global public goods."⁵⁹⁷ HUGO declares that these goods [human genomic databases] are to be shared by all with no restrictions. Their statement continues that "Human genomic databases are global public goods. (a) Knowledge useful to human health belongs to humanity. (b) Human genomic databases are a public resource. (c) All humans should share and have access to the benefits of databases."⁵⁹⁸ The terminology surrounding the idea of global public goods is useful for ushering international agreement in genomic analysis and the dissemination of its findings specifically to contribute to health equity in the socioeconomically challenged nations.⁵⁹⁹

4.A.i.c. Distributive Justice in Genomics

Everyone has an equal right to the products made possible by the results of the human genome's study and testing since they are a global public good and a part of humanity's common heritage. Distributive justice aims to ensure this is precisely what happens. The most obvious area that this can be monitored is in the insurance arena, specifically life insurance, as will be discussed. To manage this justice, it is essential to look at other countries worldwide and how they have handled this matter.

Allocation of Life Insurance with Genomic Data

Currently, it is legal for life insurance firms to inquire about an insured person's medical background, way of life, and family history, all of which are relevant to the person's genetic information.⁶⁰⁰ The insurance company can use this information for underwriting the insurance contract; notable examples include a family history of a disease, pre-existing conditions, and high-risk hobbies (such as mountain rock climbing or parachuting).⁶⁰¹ A paradigm international life insurance example is the U.K., which uses a universal voluntary mutual insurance model instead of a solidaristic insurance model.⁶⁰² These are the two main types of insurance models, and the subsequent analysis comparing these two models follows. A mutual insurance market functions by pooling the collective risk of all the insured together and then requires payment based on an estimate of how much risk the insured brings to the collective pool.⁶⁰³ The quantity of coverage supplied is closely associated with the amount of insurance bought, and these transactions are often confidential.⁶⁰⁴ In addition, these transactions are voluntary, which means U.K. citizens have a choice as to the level of coverage they prefer.

The other insurance model is solidaristic insurance or sometimes called social insurance. This model is similar to mutual insurance as the risks are also pooled, and payment is based on a prior agreement (like an individual's ability to pay or requiring an equal contribution from everyone in the market). However, the solidaristic model differs as the payment required is unaffected by the degree of risk added to the pool, which means that insurance coverage is separate from the payment amount and is more correlated with the level of need of the individual.⁶⁰⁵ These transactions are usually government-run and required by all government citizens. This model limits the amount of coverage an individual can choose. International examples of this model are health insurance services in the majority of EU nations and the UK, supplying coverage to all citizens and funded by general taxation.⁶⁰⁶

Given the compulsory nature of solidaristic insurance, it can still operate with vastly differing degrees of knowledge regarding a person's risk.⁶⁰⁷ Given that the cost of insurance is unrelated to the recognized risk that was added to the pool, as well as the lack of control the person has over the extent of coverage, The provision of more information will not be beneficial to a high-risk person for more coverage just as low-risk individuals cannot skip out on contributing to the payment pool in a true solidaristic fashion.⁶⁰⁸ On the other hand, mutual insurance requires informational symmetry to function correctly.⁶⁰⁹ Martin O'Neill highlights this importance in mutual markets: "systems of mutual insurance markets are stable only under conditions of informational symmetry whose need can be explained by virtue of an instability that mutual insurance markets undergo in the face of informational asymmetry."⁶¹⁰ This informational asymmetry is why insurance companies inquire about personal medical history, high-risk hobbies, and family medical history and precisely why the insurance agreements are only validly enforced when all relevant information is fully disclosed.⁶¹¹ Thus, informational symmetry is a fundamental part of mutual insurance contracts. However, our understanding and awareness of genetic information that could be risk-relevant are increasing with no signs of

slowing down. With contemporary biomedical advances in genetics happening every day, a look into how these advances change the nature of the life insurance marketplace is critical in shaping the future of the legal and ethical fields.

Most medical conditions have a genetic component that can assess the risk of developing the disease, thanks to advances in genetic technologies. The availability of genetic testing and subsequent knowledge of one's genetic risk can allow a person to mold their future to reduce that risk (via behavioral changes or preventative medicine) and come to peace with what may lie ahead. ⁶¹² Due to the significance, overall usefulness, and especially sensitivity surrounding genetic testing and information, it is reasonable to assume that people should be able to keep this information private if they choose. One powerful way of achieving this privacy would be dismantling the mutual life insurance market to be replaced with a solidaristic model. This model allows for preserving people's interests in maintaining privacy while accessing genetic information and avoids genetic discrimination by the life insurance company. ⁶¹³ This "one-sided" form of information gives the individual full access to their genetic data without the requirement to convey that to insurers. This choice has the drawback that individuals will be forced to sacrifice their choice in the level of coverage that the life insurance can provide as customary with solidaristic insurance models.⁶¹⁴

While this is a rather extreme option for solving the problem, social justice demands will help create a consensus to forward this cause. This can be explained if you think about which issues the theories of social justice part from and what is similar about these theories. Martin O'Neill gives this example: "many disputes among theories of distributive justice involve thinking about the degree to which *choices* made by individual agents out to affect their economic prospects."⁶¹⁵ However, one glaring problem with that is genetic endowment is not

something you can choose, nor is it related to individual effort and is wholly outside one's agency to control. Given that genetic endowment has no regard for personal preference, there is little room for argument over responsibility, choice, or desert issues.⁶¹⁶ In a similar spirit, ideologies of justice often vary on how much outcome equality should be valued versus economic sufficiency, giving precedence to the weak and vulnerable, or any other distributive goal.⁶¹⁷ However, if life insurance is a "gateway social good," as O'Neill describes, which means it is otherwise "essential for delivering basic economic opportunities, securing fundamental rights, and for avoiding social exclusion, then all these sorts of theories should agree that it is an essential matter of justice to ensure access to life insurance."⁶¹⁸

Insurance and Genetic Testing in the International Context

In the Netherlands, France, Switzerland, Austria, and the United Kingdom, policies do not exist that monitor the use of genetic tests as a precondition for the end of a life and/or health insurance contract.⁶¹⁹ In countries like the Netherlands and the UK, insurance companies differentiate the agreements with their users by the sum insured.⁶²⁰ Judit Sándor notes, "If a specific limit is exceeded, there is at least the possibility of requiring the insurance applicant to disclose the results of previous genetic tests. In Austria, Switzerland, and France, the disclosure of test results is neither a right nor a duty."⁶²¹

In most European countries, laws require the insurance applicant to divulge any and all information concerning the insured risk, such as diseases. The insurer may end the contract if the insurance applicant does not reveal any of this information. However, it is only in the United Kingdom that this disclosure duty applies to life insurance contracts.⁶²² Another example in the Netherlands is a disclosure obligation if the applicant has any information referring to diseases that have already presented symptoms. In this sense, genetic information is not a predictor of

future health events but rather a diagnostic tool similar to the results of a typical medical examination.⁶²³ Regarding procedural law, there are two sides. On the one side, there is Austria, with a very robust and widespread legal basis, while on the other side is the United Kingdom, which does not have a robust legal basis but numerous reports and codes that guide practices.⁶²⁴ To add to the differences, Switzerland takes more of an Austrian approach, while France maintains a strict outlaw of any kind of genetic testing for procedural purposes.⁶²⁵ Despite these differences, most countries display some sort of effort to protect private genetic information in health insurance.

Many, if not all, other European nations have passed legislation regulating the use of genetic screening in insurance agreements and other legal contexts. Further examples include the unlimited utilization of results from genetic testing in Spain, Japan, and Canada.⁶²⁶ Also, in Canada, the insurance applicant must disclose genetic testing information; to the applicant's detriment, results can be used to terminate a contract. Japan may lag a bit behind as it allows genetic testing without many ordinances to regulate the practice. However, due to a few aspects of the Japanese healthcare system, genetic test results can only be used in life insurance contracts and nothing else.⁶²⁷ In addition, any negative results of previous testing or investigations of the applicant must be disclosed.

In Denmark, disclosure is obligatory if the genetic results concern the applicant's current or former health condition, prohibiting any discrimination based on potential future health conditions when negotiating insurance contracts.⁶²⁸ In Sweden, there is an underlying agreement between the state and national insurance associations that control the country's usage of genetic testing. The insurance applicant must agree to add to the genetic database and add any information related to themselves.⁶²⁹ The harmonization of legal regulations on an international

level is crucial if you were to consider the potential of massive multi-national insurance companies competing with each other in smaller domestic markets and could bypass other national laws and regulations by contracting individuals in other countries where the law may not apply in the same way. The European Union, however, has thought of this potential and has enacted laws to respect other European regulations, even if they are not signed into law yet.⁶³⁰

This is but a brief introduction to the social policy of genetic big data and focuses on highlighting the importance of genetic discrimination in insurance, the flaws of the current policies in place to combat discrimination, a presentation of the view that mankind shares a genome as a common heritage and is a public good that all should benefit, and finally the problems that genomic data faces with distributive justice and some international examples of how it is handled. With this in mind, it is imperative to create a global privacy governance framework that will help address all of the above issues ethically. Tighter regulation of these genomic data banks, or biobanks, is necessary as a part of this framework, in addition to the global harmonization of laws regarding genetic privacy.

4.A.ii. Creating a Genomics Governance Framework

With an understanding of the big data problem and its ethical issues, including the social policy of related genetic data, the need for data governance and regulation is clear. Genetic testing companies, biobanks, and other data infrastructure facilities must be regulated on what they can and cannot do regarding personal privacy and security of our personal, genetic, and other forms of data. Therefore, this section aims to give a starting point for the need and framework for such governance to be enacted into future policy.
4.A.ii.a. Regulating Biobanks and Privacy

A well-rounded idea of privacy as a foundational principle can help sort between the multiple stakeholders and get to the root of the problem with biobanks. Most places, especially in bioethics, hold the right to privacy as a long-standing and widespread ideal to be achieved.⁶³¹ Numerous texts from laws, ethical manuscripts, and regulations worldwide have all pointed to the essential nature of the private life or the right to privacy and maintaining confidentiality in professional relationships, contracts, and most laboratory testing.⁶³²

Graeme Laurie et al. describe four dimensions of privacy issues in biobanking:

(1) physical privacy (e.g., gathering and storing biospecimens and testing them without consent);
(2) informational privacy (e.g., possible misuse of information);
(3) decisional privacy (e.g., control or influence over what is done with data and biospecimens); and
(4) proprietary privacy
(e.g., ownership of biospecimens and the control of identity as it relates to one's genes).⁶³³

A critical question arising from these four dimensions of privacy related to biobanking is how can scientific advancement and research promotion effectively take place with so much privacy around the biobanking industry. This question and more will be answered in the following subsection.

Scientific Advancement & Research Promotion

The risks of a privacy breach, especially data misuse in biobanks, can be prevented or mitigated by protecting and promoting privacy at all costs. Unfortunately, even with an active stance on security, breaches of confidentiality and data misuse can still happen. Violations in privacy breaches can span a wide range of harms, such as "group discrimination following a scientific publication containing group associations (e.g., cancer and Ashkenazi Jews), individual insurance or employment discrimination, and reusing DNA collected for research for criminal profiling."⁶³⁴ The risks to these examples are real. Still, fortunately, there is a low incidence rate, and there are laws and regulations in place (or soon to be) to even further reduce the risk and

manage any harm resulting from a privacy violation or misuse of data.⁶³⁵ The danger still exists, which begs the issue of how the biobanking sector can continue to provide adequate privacy protection, support for research, and scientific development.⁶³⁶

One response suggests that an assessment should occur of the restrictions on biobank clients' privacy rights under different legal systems. This should be done with a mindset that privacy is seldom an absolute in law or ethics while simultaneously clarifying the research improves public interest in biomedicine.⁶³⁷ Edward Dove states that "indeed, most legal systems recognize that privacy must yield in certain circumstances or that the level of protection might be calibrated relative to other social values and interests."⁶³⁸ This next paragraph is only a short description of the extent of biobank customers' privacy with a critical look at the decisional and informational privacy dimensions.

Protecting biobank users' genomic and related data, including biospecimens, is an obligation of researchers and healthcare professionals. There are numerous ways that this objective can be achieved, but most commonly is accomplished by the biobank's data stewards who are "(1) seeking the explicit (although not necessarily specific) consent of participants to share their data with other researchers, (2) replacing personal identifiers in a dataset with at least one code (also known as reversible de-identification or key-coding), or (3) anonymizing a dataset (i.e., permanently removing direct identifiers, hence the synonymous term irreversible de-identification) before making it available to other researchers."⁶³⁹ Laurie et al. add that "each method has value in the protection of privacy interests but also inherent limitations."⁶⁴⁰

A Fallacy of Sufficiency

There has been an ongoing discussion in the literature for quite some time that the traditional concept of consent is particularly questioned in biobanking. In the biobanking

industry, many institutions are designed to focus on the future of research. Within these institutions, however, a research subject cannot be fully disclosed the entire range of possibilities of projects that will make use of their information and samples for the duration of the study and institution– certainly not if the subject's data is transferred to another biobank.⁶⁴¹ Therefore, leaning on the traditional sense of consent in a specific context lends itself to a fallacy of sufficiency: "no participant can be sufficiently informed at the initial stage about the range of unknown actors and uncertain events to follow, and therefore to set up specific consent as the requisite criterion for participation means that most biobank initiatives will lack a sufficient ethical or legal basis on which to recruit and operate the resource."⁶⁴²

In addition, the traditional sense of consent usually involves an all-or-nothing approach, with it rarely, if ever, contributing to any control over privacy. Many people mistakenly believe that more control means more privacy. Biobank users have their consent summed up into two options: to participate or not to participate, with the ability to withdraw later but to a variable/unknown extent. For those users who value privacy as a significant dimension of autonomy, unfortunately, the ability to negotiate over the terms of data access is near impossible.⁶⁴³ Further, participants' consent does not release data stewards from their ethical and legal obligations to use data securely and responsibly.⁶⁴⁴

4.A.ii.b. The Need for Privacy & Security Law Harmonization

Bottlenecks of Global Data Sharing

This subsection will highlight two bottlenecks of global data sharing caused by the lack of a worldwide privacy governance framework. The first bottleneck occurs when ethics committees and similar regulatory bodies (such as IRBs) are out of sync with the current data privacy policies and laws – which becomes a difficult challenge to overcome since new laws and

policies are being frequently implemented worldwide.⁶⁴⁵ However, ethics committees often hold a stricter level of privacy protection than the current laws require, eliminating any ethical-legal equilibrium. These entities are generally meant to act as safeguards, so this creates a bottleneck when ethics and the law are not in-sync. Many research ethics committees have a severe deficit of any privacy-related training or certification among the members. ⁶⁴⁶ The types of consent that many of these research ethics committees reviews are traditional, with the research involving a direct action on a human body. In the case of biobanks, an influential research committee would have expertise in data involving "the nature of the population and longitudinal studies and the security and changing nature of data collected and linked over time."⁶⁴⁷

Instead of using actual people, research in biobanks uses big databases and samples of human tissue. Within the various kinds of biobank research, more nuanced considerations must occur, creating the second bottleneck. Unfortunately, while different institutions have various requirements for their research ethics committees, most are comprised of people with expertise in bioethics and law but not data privacy or governance law which can be seen in the current practices of biobanks.⁶⁴⁸ These data privacy and governance laws require any entity that collects, uses, or discloses health information to uphold high levels of security to prevent any breaches or hacks. While current practices require research ethics committees to determine whether or not sufficient safeguards exist to maintain the data's confidentiality, in order to achieve this goal, someone must be on the committee who has unique training and wisdom with relation to information systems and de-identification methods that accompany the data.⁶⁴⁹

Establishing Privacy Laws for Biobanking

While, currently, a global data privacy standard is yet to be developed, the European Union, for example, has produced some legislation regarding data privacy harmonization with plans to make it more robust over time.⁶⁵⁰ Regardless, progress is being made on the data privacy front. Another small step was made with the passing of the Madrid Resolution of 2009. However, it failed to include any privacy stipulations regarding biomedical research.⁶⁵¹ In the same vein, a call for harmonization of data privacy in the biobanking field has been requested several times.⁶⁵²

What has ultimately been missing between the domains of privacy law and biomedical practice is a thorough international and comparative legal review of biobanking procedures throughout the globe and how each country's privacy laws impact biobanking activities. These fields are in a hybrid nature of inquiry, drawing professional advice from numerous areas such as life and social sciences, policy, and law.⁶⁵³ Due to this fact, more thorough and widespread research is needed in the privacy law and biobanking fields that will help identify problem areas and overlapping areas in the respective fields. The goal of this research would be to lead to deeper discussions and subsequent analyses that could give direction to a future policy that could pave the way for much more accessible and ethical biobanking.⁶⁵⁴

Moving forward, it is no surprise that it will be a tremendous challenge to write a robust worldwide law or regulation concerning research on biobanking and privacy. Still, nonetheless, it does provide a starting point. While creating any sort of global framework will be a difficult and time-consuming task, it should be achievable if created with principle-based law.⁶⁵⁵ Edward Dove notes that "principles are compatible with harmonization, providing an embodiment of the core values and interests at stake within a common language and framework for action that clearly and determinedly promotes data sharing and use."⁶⁵⁶ For some countries and peoples, principles allow for too much flexibility while sacrificing certainty, however, it remains crucial to retain some flexibility at the local level. Regardless of one's view of principles, they are able

to give a common starting point to allow for discussion on how they can be utilized in regulation.⁶⁵⁷ It should be understood that the primary use of principles is to create a kind of harmonization that is useful, easy to apply, and supportive of the diverse cultures and peoples around the world.⁶⁵⁸

One could say it is in the realm of possibilities and desirable to pursue stronger and more robust harmonization of the currently very segregated and siloed data privacy laws around the world, as this goal would ultimately permit more ethical and efficient sharing and use of data that will most certainly contribute to advances in healthcare and biomedical research for the benefit of humanity.⁶⁵⁹ On the other hand, one must be aware that any notable breakthrough in data privacy harmonization will be a monumental challenge to implement due to various cultural, social, or technological differences amongst nations.⁶⁶⁰ Thus, the way forward should not aim to achieve a common data privacy framework with particular rules and regulations but rather seek to create a set of universal data-sharing principles that can be set to a larger governance framework.⁶⁶¹

4.A. Conclusion

The effects of big data on the technological world are apparent – it affects humanity in innumerable ways, most notably in healthcare. These effects have numerous benefits and can significantly reduce expenditures. However, several ethical implications come with this data, specifically genetic data. The social policy aspects highlight the current genetic discrimination in GINA and the ADA and the concept of the common heritage. The UNESCO declarations show that any research and testing should benefit all of humanity, not factor in discrimination. Therefore, a global privacy governance framework is suggested and should be used to enact

future legislation to protect society against the current forms of genetic discrimination that it faces.

4.B. A Cautionary Approach to Digital Technologies in Healthcare

4.B. Introduction

Globally important social and cultural transformations have always been led by technologies. The latest societal change technology has caused has been termed the digital revolution. With the ever-growing intertwining relationship between technology and society, it becomes increasingly important to take a cautionary approach to these digital technologies. Hence, ethics and philosophy have a chance to question and analyze technological developments that could save humanity from a dystopian future run more by machines than by man. This is argued by first describing the digital revolution, its close relationship to society, and how information communication technology is explicitly a large part of that relationship.

The analysis then proceeds with a philosophy of technology and an explanation of technological determinism and strategies to avoid it. A medical sociologist's perspective is used while the Precautionary Principle's close relationship to Technology Assessment is defined, which bases the reasoning for this section on why taking a cautionary approach to digital technologies is crucial. This section concludes with repositioning the medical ethos for the future, one that values ethics education in light of the numerous technological developments. The analysis applies this reasoning to human enhancement – one of the largest areas of contention with digital technologies – and a detailed account of what it is and what progress has been made thus far. The distinction between radical and moderate enhancement is described, and the analysis concludes with ethical concerns about enhancement. Overall, the ethical challenges

displayed by digital technologies must be dealt with to avoid a dystopian future of technological determinism.

4.B.i. The Digital Revolution

4.B.i.a. Technology & Society Relationship

It should be of little surprise, but the digital revolution is transforming human lives. As technology continues to impact various aspects of life, such as communication, healthcare, and entertainment – to name a few – society is becoming exponentially more integrated with these technologies and their impact on daily life.⁶⁶² The digital revolution can be defined in many ways. Still, one such definition by philosopher Nicholas Agar can be described as "the widespread and rapid replacement of mechanical and analog electronic technologies by digital technologies...Digital technologies are radically changing the ways that we share information, travel, treat disease, and party."⁶⁶³

Notable technology advancements have routinely been used to mark points in history before and after the upgrade. For example, in 1958, Hannah Arendt used specific technological objects to categorize civilizations (such as the bronze age and now the digital age).⁶⁶⁴ More recently, Sally Wyatt advocated that cultural factors influence these divisions.⁶⁶⁵ Further, Lewis Mumford proposed that we take a clue from anthropology and archeology fields and link specific artifacts to periods of time as we attempt to learn about the society and culture of human beings through their use of particular objects in certain time periods.⁶⁶⁶ In these different ways, Information and Communication Technologies (ICTs) have clearly designated the kind of change brought about by notable technological advancements, specifically how information is transmitted and processed.⁶⁶⁷

Newer and more advanced developments in the manufacturing world have started a worldwide change in the world of work, such as the Fourth Industrial Revolution.⁶⁶⁸ Shou-Yan Chou notes, "The fourth stage in any industry has become synonymous with state-of-the-art status, digitization, and smart automation."⁶⁶⁹ While a consensus is absent regarding the optimal way to use these developments, numerous industries and countries have ideas and plans to gain a competitive advantage in the global race to become the most technologically advanced society.⁶⁷⁰ Many of these countries' plans involve transforming industries to be "smart" or utilizing more digital software technologies instead of physical hardware.⁶⁷¹

The emergence of the Internet of Things (IoT), which may be characterized as an expansion of the Internet to incorporate physical items and systems, served as the impetus for the numerous changes that the Fourth Industrial Revolution brought about.⁶⁷² Chou notes, "IoT comprises a collection of digital technologies, including sensors, communication modules, and various software applications that can digitally integrate analog physical systems with the digital world, providing constant and readily available information about those systems."⁶⁷³ IoT enables quick, accurate, coordinated, and collaborative choices and actions, which makes the physical system intelligent.⁶⁷⁴ This combination of the digital realm with physical systems produces a synergistic effect that can be applied beyond the world of manufacturing to society.

It is no surprise that ethical problems can arise out of the use of technologies. A general consensus exists that early identification of the issues could provide a way to mitigate them with appropriate legal and regulatory action.⁶⁷⁵ One of the main ethical issues that arises is the struggle between technology's normative and empirical aspects. The Collingridge Dilemma, which claims that, among other things, "it is impossible to know with certainty the consequences of an emerging technology at an early stage when it would be comparatively simple to change

the technology's trajectory."⁶⁷⁶ When the technology has been around for some time, and it becomes more apparent what the ethical and social implications are, it also becomes much harder to change its outcomes and social contexts. Even though the Collingridge dilemma poses some severe problems for the future of technology, it attempts to anticipate the social and ethical problems that might arise through design choices and legal regulations.⁶⁷⁷

4.B.i.b. An Information Society

The relationship between society and technology is furthered with the use of Information and Communication Technologies (ICTs), specifically with ICTs have impacted how information is transmitted and processed. Oxford University's Luciano Floridi, a professor of philosophy and information ethics, "reconstructs these changes by identifying three macroperiods: pre-history, history, and hyper-history."⁶⁷⁸ He notes that the information cycle, defined as the "occurrence, transmission, process and management, use of information," has significantly changed over time.⁶⁷⁹ In the time before ICTs, known as pre-history, society had no way of recording information and thus was only processed orally.⁶⁸⁰ Then, with the invention of writing, however primitive it may have been, society entered the information age or history. Now, with the use of ICTs, society has entered the period of hyper-history. The notable difference between these periods "is not only in the quantity and speed of information transmitted but especially in *how* it is transmitted and processed."⁶⁸¹

Therefore, Pierre Lévy, a philosopher and professor of communications at the University of Ottawa, touches on "a speed of evolution of knowledge and of a collective intelligence formed by the social existences of the subjects who participate in it, which can continually reshape itself."⁶⁸² Federica Russo attempts to take this notion and relate it to the idea of connective intelligence, which describes how things are based upon how they connect to each other instead

of what we have been told they are.⁶⁸³ With ICTs making connections like this, they are able to push boundaries and change the world around them by creating a different way of viewing the field of ontology.⁶⁸⁴Spaces like this allow for technologies to interact amongst themselves, often times even without human input. ICTs such as e-health technologies, social media, and data analytics are just a few examples of technologies that have the power to cause new ethical and ontological problems in this period of hyper-history that we are living in.⁶⁸⁵

This ICT-powered hyper-history period can also be called the Information Society and has turned out to be extremely transformative to the contemporary way of living. Increasing every day, the role of ICTs in society's development is constantly changing and in need of assessment. Previously there have been numerous ideas to test and attempt to understand ICT's role in development better. The various ideas have been disseminated from a wide variety of fields, including organizations, corporations, government agencies as well as non-governmental organizations (NGOs).⁶⁸⁶

These various fields have deduced that "ICT is a core catalyst or key strategic resource for economic and social development."⁶⁸⁷ This notion of ICT is generally positive. However, it fails to keep in mind the various interactions of ICT in the subsequent social and material spaces it occupies. Philosopher Natascia Boeri describes the problem that "development projects frame ICT as a tool to create a more equitable place but ignore that it is within an unequal society that these technologies are created, and it is through an unequal society that these technologies will be used."⁶⁸⁸ It is no surprise that society has different, yet no less important, roles in the ICT world. Therefore, it is not safe to assume technology is the sole source of significant social change, but rather a 'socio-technical product.'⁶⁸⁹ In a related notion, Saskia Sassen, a professor of sociology at Columbia University, gives us a framework that precisely details what

technology and society have in common, "allowing humanity to recognize the embeddedness and the variable outcomes of these technologies for different social orders."⁶⁹⁰

Viewing ICT in a different frame like this has a meaningful impact on how someone would comprehend the ICT's function in society's growth. For example, Boeri attempts to better apply Sassen's technology and society's integration framework to comprehend this reinterpretation of ICTs function in development, a case study involving ICT centers in rural India was used.⁶⁹¹ Boeri strives to avoid making the same assumptions about technology that are made in many other evaluations of ICT centers: "a technological determinism that camouflages real social divisions."⁶⁹² Her assessment aims to question the assumption that development projects offer only positive results without noting the underlying or unrealized changes that happen from technological developments that can be summarized in the digital divide problem.⁶⁹³

4.B.i.c. Forward to a Social-Digital Future

This connection between society and digital technologies has been a popular topic amongst contemporary philosophers as the digital revolution significantly influences human affairs. Nicholas Agar writes, "the digital revolution poses a threat to humans as doers, as authors of our own destinies. We make significant choices about ourselves and the world, in large part because we reason in certain ways."⁶⁹⁴ Agar claims that one can predict that the digital revolution will radically remake work and redirect human agency.⁶⁹⁵ This is something that society cannot change. However, humanity will nonetheless fundamentally alter the world of work and where human agency can be sourced to. This will require careful consideration of the domains of human activity that society surrenders to the machines.⁶⁹⁶

In Agar's view, societies that advance to the digital revolution ought to be disseminated into "social-digital economies."⁶⁹⁷ These kinds of economies emphasize two completely divergent types of activity, each focusing on two completely divergent types of value. The digital economy focuses its activity on maintaining the value of efficiency. This kind of economy values outcomes above all else and the means only when they are impacting the results in a negative way. It is completely possible that one manufacturing process may be valued over another due to factors such as efficiency, costs, or using fewer materials. On the other hand, the social economy prioritizes humanness. This kind of economy prefers entities with minds akin to humans, with preferences to subsequently interact with entities that espouse feelings akin to humans. Society enjoys the company of other members of "the mind club," a phrase Agar takes from psychologists Daniel Wegner and Kurt Gray.⁶⁹⁸ Wegner and Gray this exclusive club as "that special collection of entities who can think and feel."⁶⁹⁹

Humanity's preference for alike members of the mind club takes particular interest in the personal aspects of our lives by leading our selection of friends, lovers, and even in the domain of work. Agar describes, "if we think about it, we want our baristas and nurses to have minds like ours too. We will rightly reject the inefficiencies of humans when they stray into parts of the economy that emphasize the skills of the computer. But we should have the courage to reject digital technologies when they trespass on distinctively human activities."⁷⁰⁰ Society should be asking the more important questions that progress of AI may lead to machines that contain human-like thoughts and emotions. Society's contract with the machines should be one in which people can complete the tasks where feelings are important and play a role, and machines can complete the numerous data-driven jobs where feelings do not matter.⁷⁰¹

As time progresses, it will become more apparent that humans will be substituted from areas of work that focus on pure efficiency, and humans will become more available to do types of work that fulfill social interaction needs. In today's technologically advanced societies, "social worker" is the name of a job that addresses the most extreme harms caused by social isolation and indifference.⁷⁰² Social work should be vastly diverse in a social-digital economy that resides within a mostly now digital age. With that being said, human social needs are quite dynamic and often complex. Unfortunately, "jobs placing humans into direct contact with other humans are foremost amongst those that our current emphasis on efficiency is causing us to seek to do without."⁷⁰³ For example, in automated checkouts and customer service, AIs are taking the place of workers who deal directly with other human beings.⁷⁰⁴

Agar notes that "machines will always be poor substitutes for humans in roles that involve direct contact with other humans."⁷⁰⁵ In this case, humans, prefer connections, no matter how small or insignificant, to other humans. People often want to know what is happening in the thought processes of those in the service industry in this kind of economy. While efficiency remains important, it is not the only factor to consider in these interactions. Of course, people would like the barista with whom they placed an order for a coffee not to forget about it. At the same time, people value, even subconsciously, the human interaction as the coffee gets delivered to the customer. "When a drive for increased efficiency causes us to do without human workers, we leave ourselves ill-prepared for the kinds of digital future that we should be seeking."⁷⁰⁶ If humanity would like to continue down the path of the social-digital economy, then these are the kinds of jobs it should maintain and preserve for humans. It should aim for a future in which machines do much of the heavy lifting and complicated calculating, but humans find work meeting the many social needs of other humans.⁷⁰⁷

4.B.ii. Philosophy of Technology

4.B.ii.a. Technological Determinism

Even in 2022, it is impossible to know which innovations will be remembered long after they were invented, and which ones won't. Regardless, the custom of using technologies to associate places and periods of history remains commonplace, even though technologies' causeand-effect relationship may not always be apparent. Considering how often people ponder about how technology and society interact, it never required a formal name.⁷⁰⁸ Nonetheless, researchers and scholars have labeled it technological determinism, which comprises two main aspects. According to Robert Scharff and Val Dusek in their book *Philosophy of Technology: The Technological Condition: An Anthology*, "The first part is that technological developments take place outside society, independently of social, economic, and political forces."⁷⁰⁹ As history has shown us, advanced manufacturing processes spark from people with a knack for inventing or designing and are absent of anything to do in respect to interpersonal relationships. The second, more crucial claim made by technological determinism asserts "that technological change causes or determines social change,"⁷¹⁰

A strong connection exists within technological determinism that social and technological progress are equal. Scharff and Dusek note that Vladimir Lenin shared this view he said "Communism is Soviet power plus the electrification of the whole country," and interestingly, this view stays popular among a wide variety of politicians today.⁷¹¹ In another example of a politician holding this view, In his 2006 State of the Union speech, George W. Bush said that he believed technology to be the answer to the impending energy issue in the United States.⁷¹² In the past, technological determinism has had the understanding that every generation comprises a handful of inventors and inventions that actually turn out to forward the progress of human

development. On the other hand, unsuccessful inventions are quickly forgotten and swept up in history. Noteworthy inventions are able to quickly and easily prove their usefulness and speedily make their way into the everyday lives of individuals and society, thus transforming them. Through this route, technological breakthroughs can have meaningful social impacts.⁷¹³

It is often the case that the invention of new technologies invokes completely opposite reviews. Some view certain technological advances as groundbreaking improvements, while others view them as a potential slippery slope to a disaster. Regardless of the side one takes, the fact cannot be ignored that information and communication technologies have put humanity on a "predefined path, which we cannot move away from."⁷¹⁴ This causes digital technologies to become either positively or negatively normative. Russo explains that "this normative dimension of technology, and a fortiori of ICTs, makes the link between digital technologies and ethics explicit: technology is closely related to the sphere of action and of decision."⁷¹⁵ In a similar sense, Hans Radder notes that technologies are tied to the place in which they operate, which has social and cultural aspects in addition to the material ones.⁷¹⁶ It would seem that technologies are somehow related to the norms observing how humans interact, such as which people can interact with certain people and what the technologies ought to do or ought not to do in those situations. Imagining how social interactions could be impacted by digital technologies, how they might be advantageously employed in politics, or how they might be the focal point of important political platforms, might be useful.717

There are two notable tactics to remove oneself from the psychological trap of technological determinism, the first tactic involves thinking about possibilities. This tactic originally comes from M.I.T. mathematician Norbert Wiener in his book, *Cybernetics: Or, Control and Communication in the Animal and the Machine,* where he states that cybernetics

gives humanity these possibilities to escape the mindset of technological determinism – and it's up to us which possibilities we choose to advance or not.⁷¹⁸ What exactly is cybernetics, though? Heather Love details, "Cybernetics emerged around World War II as a techno-scientific field of study that employed statistics and probability to develop machines capable of high-speed, highvolume data processing."⁷¹⁹ She continues that "The basic goal of the machines was to enable the transmission of useful information or to facilitate communication and the ability to incorporate feedback, at rates that exceeded the scope of human cognitive processing."⁷²⁰ In other words, cybernetics uses positive and negative feedback loops to further the ethical inquiry - similar to that of the Socratic method which continually asks the "why" question in order to get to the root of the problem. Wiener went further to say that technology is more like applied social and moral philosophy than applied science.⁷²¹ He also talked of hesitations regarding the use of cybernetics in military operations. He argued that cybernetics should be a service to humanity, not used as potential weapons or for exploitation. On the other hand, technologies such as cybernetics can also provide massive opportunities such that the digitization of society has the opportunity to control and regulate the free market.⁷²² Society need not fall to the potentially harmful effects of digitization but can proactively turn them into opportunities to improve its current condition. Similarly, computerization can pave the way to developing the discussion of certain norms.⁷²³ Russo concludes that "instead of being passive victims of (digital) technology, we *create* technology and the material, conceptual, or ethical environments, possibilities, or affordances for its production or use; this makes us also responsible for the space of possibilities that we create."⁷²⁴

The link between knowledge, or science, and its applications, or technology, might be reformulated as a second strategy to escape the technological determinism trap.⁷²⁵ Technologies,

specifically those of the digital variety as been discussed, cause ethical questions to be raised for how it is applied in certain situations. For example, in science, you study physics and nuclear reactions, whereas, in technology applications, you learn to build a nuclear bomb.⁷²⁶ Hence, science itself is not intrinsically moral or immoral. These conclusions are only applicable to the use of technology or science. It is crucial to remember that the border separating science from technology is often hazy and imprecise. Heather Douglas adds, "This distinction, as well as the one between pure and applied science, is a product of our historical, philosophical, sociological reflection on techno-scientific practices."⁷²⁷

These two strategies lead us down the same road: either the issue of normativity arises because digital technologies are inherently moral or immoral, or because they are a branch of hard science.⁷²⁸ This question forms because science and technology are related to action. People take action almost every moment, from the smallest decisions, like drinking a cup of coffee in the morning, to larger ones, such as testing a nuclear fusion reactor or giving consent to a life-threatening procedure. Humanity, specifically in groups such as institutions, and digital technologies, are constantly a part of the decision-making process.⁷²⁹

4.B.ii.b. A Medical Sociologist's Perspective

This section takes a bit of a break from the philosophical views regarding technological determinism and the like and looks at the topic through the lens of a couple of medical sociologists. Sociologists have always been fascinated with information communication technologies (ICTs) and their impacts on society's relationships with each other and towards technology itself. Medical technologies are of particular note here as they are simply intrinsic to the healthcare landscape today, whether it has direct or indirect effects on patient care. Sociologists Stefan Timmermans and Marc Berg have provided a detailed look at both medical

and sociological research-related technologies. According to Timmermans and Berg, there are three primary areas of sociological study on medical technologies, each with a specific emphasis on a distinct technology.⁷³⁰ The first area, of course, is technical determinism, which holds that technology is indeed a "driving political force in late modern societies, or as a political tool in itself."⁷³¹ Additionally, they provide an illustration of how radical feminist discourse that labels every new reproductive technology as a means for maintaining patriarchal control over women would fall under this classification.⁷³²

Any kind of literature on "new genetics" and a large portion of political economics study on the pharmaceutical sector might be said to fall under this first group. New genetics here refers to individuals who undergo the consequences of geneticization.⁷³³ Geneticization here refers to "the increasing tendency to define differences between individuals as largely or entirely due to genetics."⁷³⁴ Dr. Henk ten Have defined geneticization as "a heuristic tool that can help to refocus the moral debate on the implication of new genetic knowledge towards interpersonal relations, the power of medicine, the cultural context and social constraints, rather than emphasizing issues as personal autonomy and individual rights."⁷³⁵ Timmermans and Berg conclude this category by noting that this more theoretical part of medical sociology only looks at the decisive and generally newer technologies that take medicine to a new level by revealing more robust social and ethical problems to be solved.⁷³⁶

They continue to denote the second category as social essentialism. This category details that technologies are "blank slates to be interpreted and rendered meaningful by culture."⁷³⁷ This area of sociological study focuses on medical technology, such as radiology equipment, which may be very beneficial to the sociological research community. For instance, the investigative research by Strauss et al. outlines the structure of the work in technology-heavy hospitals, how

the staff works with complex technology and patients simultaneously, and how patients experience technologically invasive treatments.⁷³⁸

Timmerman and Berg's third and most current classification of sociological studies of medical technology combines ideas and strategies from both science and medical sociology.⁷³⁹ This category is based on the conviction that "technologies are not regarded as isolated inanimate objects with fixed characteristics, but as, at least metaphorically, actors."⁷⁴⁰ This does not imply that any technology is conscious of itself or has its own thinking. This concept should focus your attention on how these technologies can assume the norm of new practices due to their particular characteristics. Simultaneously, technologies can adapt to their interactions with society and their environment. According to Bruno Latour, "technologies are regarded as members of networks in the sense that they can be involved, when in use, in relationships with a whole range of persons, other technologies, practices, and so on."741 Therefore, instead of assuming that technologies are great influencers of society or carriers of various indirect cultural and social meanings, technologies, at least in practice, must deal with these questions in an empirical manner.⁷⁴² However, in attempting to answer these questions empirically, it is essential not to forget to use the Precautionary Principle and Technology Assessment, which will prevent any harm technology may cause in the empirical investigation. These notions are explained in further detail in the following section.

4.B.ii.c. The Precautionary Principle & Technology Assessment

In 1992, with the ratification of the Rio Declaration, the Precautionary Principle (PP) came to light despite disagreement on a standardized definition of what exactly it is.⁷⁴³ Nonetheless, the majority of the meaning of the PP can be found in the Wingspread Conference Statement: "When an activity raises threats of harm to human health or the environment,

precautionary measures should be taken even if some cause and effect relationships are not fully established scientifically."⁷⁴⁴ The PP's philosophical base has been somewhat flimsy from its founding, but numerous attempts have been made to change that fact. The main point of the principle comes from what has been termed the "paradox of precaution," which states that according to the PP, mankind must refrain developing dangerous but possibly useful technology in order to avoid harm.⁷⁴⁵ Jonathan Hughes explains that in this view, "harm occurs since the benefits of adopting the technologies are foregone with, for example...lost opportunities to prevent disease and death, resulting in a precautionary principle that would instruct society to refrain from implementing itself."⁷⁴⁶ Further, even supporters of this view have admitted that the principle is not well-defined and does not translate well into rules of behavior regarding technology.⁷⁴⁷

Progress has been made since 2006 to create more clear and sound philosophical renditions of the PP, whose goal is to provide actionable guidelines regarding technology in unique instances. Political philosopher Stephen Gardiner's products have helped establish specifications for the PP to be used realistically.⁷⁴⁸ A significant advancement according to Gardiner is that the PP "applies in conditions of uncertainty, but not when ignorance exists of the range of outcomes over which there is probabilistic uncertainty."⁷⁴⁹ At that time in 2006, the PP did not have a way to find and normatively assess any possible results of developing technology. An avenue to give the PP a way to make these assessments can be supported through empirical research that can be used to find various uncertain outcomes of the technologies and then use that information to provide a basis to make a normative claim.⁷⁵⁰ Different ways to ethically assess emerging technologies need a route that uses cross-referencing technology with various moral values and ethical principles.⁷⁵¹ By locating the results inside previously established moral

theories and notions, these identification procedures do more than just identify a possible range of ethical outcomes; they also give the outcomes a normative character. The blending of facts and standards is compatible with one another in this manner. Hence, without this identification process, the PP is "of little use in evidence-based policymaking and does not explore the relationship between facts and norms in future-oriented ethical assessment."⁷⁵²

Now it is necessary that, relating to the outcomes of technology, both prescriptive and descriptive kinds of information are necessary to use the PP in the decision-making process. As previously mentioned, this information should be collected in empirical ethics research to identify the outcomes and then use those outcomes to give insight into normative claims on emerging technologies. The most popular approach is called Technology Assessment (TA), which is used to gather this kind of information in an uncertain environment.⁷⁵³ "TA is a field that studies and evaluates the interaction between new technologies and the environment, industry, and society."⁷⁵⁴ There are numerous similarities between TA and other approaches that can be used to assess emerging technologies, such as value-sensitive design, participative design, and various socio-technical methods.⁷⁵⁵ Research and studies in TA typically focus on forecasting studies and/or foreseen applications of emerging technology.⁷⁵⁶ Most importantly, TA has a built-in form of identification of outcomes, a relatively normative perspective in emerging tech, and a wide array of empirical tactics to research this perspective.⁷⁵⁷

It is no surprise that TA has been born out of many of these emerging technologies' adverse or undesirable side effects. Technology Assessment is composed of "a family of approaches that combine empirical research on likely consequences of technologies with normative insights."⁷⁵⁸ TA has several parts, such as participative TA and constructive TA, both of which use a wide variety of methods, notably the participation of numerous stakeholders,

specifically people who could be seen as future users of the emerging technology. By carefully guiding technological development and oversight, TA research aims to anticipate and address the myriad problems that evolving technologies present. Armin Grunwald, a professor of philosophy of technology, recognizes a few obstacles that TA replies:

that of integrating at an early stage in decision-making processes any available knowledge on the side effects, that of supporting the evaluation of the value of technologies and their impact, that of elaborating strategies to deal with the knowledge uncertainties that inevitably arise, and that of contributing to the constructive solving of societal conflicts on technology and problems concerning technological legitimization.⁷⁵⁹

Therefore, TA becomes a stronger untied field of study that focuses on disseminating and evaluating knowledge regarding environmental, social, and economic outcomes to give a framework to navigate the subsequent social response to the future of emerging technologies.⁷⁶⁰ These future emerging technologies can be assessed with the use of the Precautionary Principle and Technology Assessment as described in this subsection, however, if the moral ideals and ethical standards we associate them with have not advanced in line with technological advancement, the evaluation may not be accurate. Society's ethos must be repositioned to be ready for the future that emerging technologies will most certainly bring.

4.B.ii.d. Repositioning the Medical Ethos for the Future

Recently, there has been a lack of ethical discussion and questioning in this more modern culture. Little public ethical discourse is ever seriously entertained philosophically; morality is strictly privatized, individualized, and compartmentalized in personalized spaces and other venues such as private residences, political offices, and in breakout-rooms at conferences.⁷⁶¹ The hostility of political culture supports the idea that any lack of passion in public debate can show disinterest in the cause no matter which side of the spectrum the candidate resides on. It is difficult to see how society can address the challenges of organizing medical care in this epoch

without improving the level of discourse on social values and ethical matters. This conceivably could be done by enhancing the education of all citizens about philosophical concerns and methods in debating issues of great importance. However, education alone may not accomplish this goal. As author Thomas McCollough notes, "in contemporary America, the emphasis is on efficiency, organization, technology, and profit, leaves little room for serious moral deliberation on difficult subjects."⁷⁶² The general public is overwhelmed by advertisements, distractions, entertainment, ruthless politics, and celebrity rumors that are absent of any legitimate, ethical discourse. Author Henry Giroux has noted that schools in the twenty-first century have become venues for selling products instead of creating deliberative citizens.⁷⁶³ "Society has ceded the *polis* to advertising in various electronic formats."⁷⁶⁴ This means that society has sacrificed what is good for its citizens to exploit them for the lucrative business of personalized advertising across multiple electronics such as mobile device ads (via apps or web browsers), tv, radio, and many other e-formats. It is precisely this slippery slope of tech capitalism exploitation that must be avoided with the use of the digital age tools that big tech companies provide.

How is humanity going to rationally discuss complex healthcare issues concerning resource allocations, the privacy of medical information, genomics, the impact of technology, and cost-effectiveness if it does not learn the underlying philosophical and mathematical principles and the structural facts that impact these issues? There is a need to seek a higher level of public discourse and a greater understanding of the implications of ethical and statistical analysis in solving societal healthcare problems. Roy Poses, in his article, *A Cautionary Tale: The Dysfunction of American Healthcare*, noted that large bureaucratic organizations, including the regulatory agencies, "act in accordance with their own agendas and threaten to undermine the

core values of medical practice, as the physician turns into a technocrat controlled by various interlocking bureaucracies."⁷⁶⁵

The growth of medical technology and the corporate control of medical practice create a further challenge to the practice of medicine as a holistic healing art. However, these technologies uphold a standard of efficiency that relies heavily on clinical data and regulation by bureaucracies at the internal and external levels. The assessment of quality, safety, computerization, and mechanization of processes have accelerated affecting the character of medical practice in both good and bad ways.⁷⁶⁶ Despite rising healthcare costs, federal and private policies mandate that computerization and mechanization must be better than the status quo and worth pursuing despite the substantial added costs and relatively little evidence of benefit. Serious deliberation on the cost of computerization of healthcare institutions continues with data, healthcare will sacrifice humanness and further separate patients and doctors who need more than just another statistic to the institution.

Professor of philosophy at the University of Montana and expert in the philosophy of technology, Albert Borgmann encourages consideration of the nature of technology interactions instead of focusing solely on the power of technology. He notes: "Technology...promises to bring the forces of nature and culture under control to liberate us from misery and toil...there is a promise that this approach to reality will yield liberation and enrichment by way of domination of nature."⁷⁶⁸ Borgmann further posits that this vision led contemporary society to believe technology is the key to the good life.⁷⁶⁹ His ideas assist in shedding light on the myth that unregulated technological growth is the solution to the problems that plague healthcare. The healthcare bureaucracy appears to subscribe to the dominant myth of the twenty-first century that

more advanced technology inevitably leads to better outcomes as it implements "meaningful use of electronic medical records and orders."⁷⁷⁰ The Leapfrog Group, a consortium of large employers, has taken a more cautious approach, concerned that the electronic health record (EHR) may introduce new types of medical errors and views the matter with greater discernment.⁷⁷¹

Could raising the knowledge of ethical discourse, starting with more ethics education in colleges and high schools, help? Several authors contend that good public deliberation is possible, but structure and process must be developed and followed diligently.⁷⁷² Today's culture has made such discussions more challenging to have, let alone try. Robert Sternberg, an American psychologist and psychometrician, has studied and advocated for the role of wisdom development in education.⁷⁷³ His theories of wisdom education that balance self-interest with other-directed concerns and tacit and formal knowledge were applied in a New Jersey school district with very positive results for these students.⁷⁷⁴ The question is, why isn't this catching on and becoming more popular? The reason most likely relates to "the pervasive consumerist ethos that eschews contemplating the selfless, the deliberate, and the unexciting."⁷⁷⁵ The chapter will now turn towards an applied analysis of how taking a cautionary approach to technologies may be of real ethical consideration when it comes to human enhancement. If society is not careful, the perceived benefits of human enhancement can lead down a slippery slope to creating a new host of issues described in the final section of this chapter.

4.B.iii. Applied Ethics Analysis to Human Enhancement

4.B.iii.a. Benefits Beyond the Biological

It is not hard to imagine a future in which science fiction cyborgs take another step closer to becoming a reality, with human beings more often becoming implanted with bioelectronic

devices. Advancements in numerous technologies such as "semiconductor devices, cognitive science, bioelectronics, nanotechnology, and applied neural control technologies are facilitating breakthroughs in the hybrids of humans and machines."⁷⁷⁶ For example, every day, we see increased instances of computing power, breakthroughs in prosthetics, artificial implants, and the like that facilitate the combination of man and machine. This fusion of the organic and inorganic is becoming much more normal and available as bioelectronic and mechanical parts are replacing or enhancing current body parts. It is crucial to understand that the improvements mentioned in this section do not have curative functionality, giving those with disabilities such as amputees, the paralyzed, or the blind a chance to overcome their condition. The curative devices that help these people with sensory, motor, cognitive, or other disabilities are generally not provocative of a debate of the ethical nature of these devices. The ethical issues lie when these technologies are used to enhance human capabilities beyond that threshold to where it could be considered a step in the evolution of humankind.⁷⁷⁷

Gifting humans with abilities such as x-ray vision, smelling sensitivity for carbon monoxide and mercury, night vision, extended range vision capabilities, and glow-in-the-dark pigmentation without a doubt change the natural human ability.⁷⁷⁸ Even more noteworthy and radical are the enhancements that could become possible through brain-machine interfaces. Within the realm of possibilities, brain-machine interfaces replace the capacity for thought and memory.⁷⁷⁹ These technologies will make it possible for people to maintain a continuous internet connection and instantly retrieve encyclopedic information levels. Ellen Mc Gee notes that "Building in these interfaces, surgically implanting them in the brain, will allow for greater energy and efficiency and will eventually enable humans to operate without radios, TVs, printed

newspapers, cameras, GPS units, credit cards, computer workstations, ATMs, wireless, corded or mobile phones, and other separate devices."⁷⁸⁰

Brain-machine interfaces use technology that can externalize data already in the brain and give the user the ability to access data from external sources.⁷⁸¹ This should raise huge ethical red flags concerning privacy and autonomy to have this data input-output flow outside of one's control – in a way, stripping ourselves of human dignity. Hypothetically, in the future, if human cloning ever does reach reality, this brain-machine interface would be able to take the memories, emotions, and essentially life and personality of its source and upload it into the clone – thus creating a sense of immortality.⁷⁸² Another way that could be achieved would be uploading one's consciousness into a computer or the cloud. This would enable ageless minds that could travel at lightspeed and communicate telepathically.⁷⁸³ In this hypothetical future, every person may face the choice of what kind of reality they want to live in: "a natural, an immortal with a body, or an entity that exists in virtual reality."⁷⁸⁴ Due to the possibility that these technologies could one day make this choice a reality, ethical and legal guidance should be written and enforced to usher humanity along the way.⁷⁸⁵

Unlike the enhancements talked about with genetic technologies or with drugs, brainmachine interfaces have had little to no ethical limelight from the scientific, government, or international communities. McGee argues, "Neither of these future enhancements – genetic or pharmaceutical – will alter future humanity, as will bioelectronic systems that mix electronic and biological components."⁷⁸⁶ She reasons that biology limits how far genetic enhancements can go – whereas human-machine cyborgs are not restricted in the same way. In addition, drug enhancements merely increase standard human memory and performance by just a bit, these enhancements don't even come close to the abilities of a brain-machine interface to share and

retrieve data at lightspeeds. Editing genes can only enhance the optimum for a biologically based human.⁷⁸⁷ Splicing a gene from another biological system into the human genome would surely augment human abilities. McGee gives an example of inserting the gene that gives dogs their superior smelling ability to humans. However, since the human intellect is already superior to animals, these genetic enhancements would only be able to go as far as the greatest degree of intelligence determined by genetics.⁷⁸⁸ The human being may undergo significant change thanks to bioelectronic devices, implants, and prostheses, augmenting benefits that are simply beyond the biological.⁷⁸⁹

4.B.iii.b. Radical & Moderate Enhancement

Numerous technologies have promised human enhancements beyond the imagination that could be provided by the more conventional means such as via injection, pill, or genetic modification currently available now. These are exciting prospects – but are they genuinely desirable? In his book, *Truly Human Enhancement: A Philosophical Defense of Limits*, Nicholas Agar argues that "some ways of enhancing cognitive powers or extending life spans are undesirable specifically because they enhance these attributes to a great degree."⁷⁹⁰ This happens to be the case even when more minor variations of cognitive enhancing or life-extending developments are considered good. Hence, it becomes necessary to differentiate between moderate and radical types of enhancement. Agar claims that "radical enhancement improves significant attributes and abilities to levels that greatly exceed what is currently possible for human beings."⁷⁹¹ On the other hand, Agar continues, "moderate enhancement improves significant attributes and abilities to levels within or close to what is currently possible for human beings."⁷⁹² Agar's overall view endorses some moderate enhancements but rejects radical

enhancement, which he calls the "ideal of truly human enhancement."⁷⁹³ This view takes humanity's desires and uses them as directors to give way to advances that improve humans. ⁷⁹⁴

The differentiation between radical and moderate enhancement is quite vague. Consider enhancing our cognitive powers - the futurist and inventor Ray Kurzweil envisages "adorning human brains with increasingly powerful electronic implants, soon generating an intelligence about one billion times more powerful than all human intelligence today."⁷⁹⁵ This is radical. This kind of power in a mind would have intellectual abilities greater than any human in history. It also becomes possible to imagine cognitive enhancements that are more vague. The world's highest IQ has been measured at an intimidating 210.796 Suppose that IQ is an accurate measure of intelligence. Would an education program that boosted students' IQs to 250 moderately or radically enhance intelligence? Should a human being with an IQ of 250, on the assumption that IQ is an accurate measure of human intelligence, be recognized with intellectual abilities that far transcend what humans are now capable of?⁷⁹⁷ Maybe the best way to classify it would be in the realm of vagueness between that of radical and moderate enhancement. It sometimes seems that vagueness poses a hurdle to settling into one camp or the other. Yet, it does not by any means diminish the separation of radical and moderate enhancement. For example, there are hairy and bald people on earth who still exist despite a formal boundary between the two. Understanding what constitutes a radical enhancement and what constitutes a moderate enhancement may be done despite some degree of ambiguity.⁷⁹⁸ Once society knows how to determine the clear cases of radical or moderate enhancement, it should have the intellectual tools needed to tackle vague topics.799

So, how could more robust enhancements be of less value than weaker enhancements? Agar makes significant points in a few rebuttals regarding the radical enhancement of human capabilities. Agar offers "moral criticisms of some radical enhancements – they impose significant, unjustified costs on others."⁸⁰⁰ The arguments tend to support legal bans on "technologies or uses of technologies that lead to certain varieties of radical enhancement."⁸⁰¹ Agar is principally concerned about enhancements that alter our sense of moral being to different benefits and our ethical safeguards against various potential harm.⁸⁰² Any kind of modifications that would jeopardize our morals should be prohibited since they reveal unenhanced humans to potential harm.⁸⁰³

Agar criticizes various radical enhancements in an efficiently rational manner. The advocates of radical enhancement display it as being cost-effective and valuable in the long term. Hence, to the proponents, it vastly improves the overall well-being and interests of people who have the procedure completed. Agar argues that radical enhancement will disappoint. Humans tend to value memories or experiences that are just outside the normal range of events due to the fact that they can properly engage them.⁸⁰⁴ Any memories or experiences that are extremely distant to the normal range of human events are less impactful because we engage with them significantly less. These times and memories are of less worth than the memories they would instead replace. Humans do not prioritize wisdom gained through radical cognitive enhancement because they view knowledge inefficiently without radical cognitive enhancement, which extends to a more existential understanding of humanity's place in the universe.⁸⁰⁵ These barriers either are non-existent in regard to lesser cognitive enhancements or are critically reduced. Agar claims that both his points about radical cognitive enhancement need an account of what it means to engage with the experiences and knowledge brought by enhancement and a theory about why successes or failures of engagement should be significant.⁸⁰⁶

4.B.iii.c. Ethical Concerns

The general ethical consensus as stated previously in the benefits beyond the biological section is that enhancements, whether bioelectronics or implanted chips, when used for therapeutic purposes, is not ethically concerning. However, concerns that are present with curative use of implants such as those of safety, access, and costs must be attended to. As with any procedure long-term and short-term risks exist. Short-term risks focus on the procedure itself such as bleeding, infections, and adverse drug-related reactions.⁸⁰⁷ Long term risks are harder to define and evaluate, but many of them revolve around immune rejection of the implantation. For implantation manufacturers, the challenge will remain to develop a non-toxic device that will not cause an immune system rejection in the body.⁸⁰⁸ McGee relates that "these issues are warranty availability, liability responsibilities of manufacturers, industry-wide standards for devices, methods of facilitating upgrades, and procedures for training users in implementing the systems."809 Moreover, it is crucial to research on the practicality of implants and if certain people benefit from the implants more than others. Unfortunately, some people, such as the mentally disabled – may never be able to learn to use an implant. While still, other people will need a great deal of counseling and advice about proceeding with an implantation.⁸¹⁰

It seems highly unlikely that a struggling health system that has difficulty in providing basic access and coverage for its participants would be able to include this kind of cutting-edge procedure in its coverage or government subsidy.⁸¹¹ Subsequently, these concerns will become more complicated as technology has an innate ability to change the normal standard constantly. This ability of technology will only grow and become more difficult to stop or slow progress and add to the already overburdened and understaffed healthcare system in America.⁸¹² Strategies and programs must be developed to limit the inequalities that are sure to arise from technological

developments in this field. In the name of fairness and equality, efforts must be made to avoid the failure of the HIV/AIDS epidemic that has caused shortages of the relief drugs that have caused and continues to cause many people to suffer needlessly. Perhaps it is not unreasonable to ask for a commitment to adjust or subsidize pricing for the socio-economically challenged area and countries so that they may have access to the technology for therapeutic purposes at a reduced rate.⁸¹³

Enhancement technology raises numerous ethical, technical, and existential questions regarding its ability even to control humans. Unsurprisingly, brain implants that can provide vision to the visually impaired are highly coveted. What is a very different issue is extending that implant to enable x-ray, night, or long-range vision to the average person. In the same vein, "Enhancement in and of itself is not necessarily objectionable; vaccines, in vitro fertilization, and breast enhancement surgery, are all instances of readily accepted and widely sought enhancement technologies."⁸¹⁴ The boundaries of autonomy, privacy, justice, and what it means to be a person will undoubtedly be pushed by brain-machine interfaces.⁸¹⁵ These interfaces will be able to assist in memory, help in language fluency, recognize unknown individuals, and enable sharing of information without speaking.⁸¹⁶

It can be assumed that these devices will have three main categories of users. The first is the most obvious, which are therapeutic users with legitimate disabilities. The next category will be the military, which will be used in communication and weaponry systems. The third category will comprise users in data-intensive fields and businesses such as data analytics. The Defense Advanced Research Projects Agency supports funding for these developments: "DARPA is interested in creating new technologies for augmenting human performance through the ability to non-invasively access these codes in the brain in real-time and integrate them into peripheral

device or system operations."⁸¹⁷ These devices have the potential and capability to alter the capacities of humans so much that they will be fundamentally changed.⁸¹⁸

Naturally, many are opposed to this concept of fundamentally altering humanity. "Both essentialists, including those who argue for bodily integrity, and creationists, argue against the development of implantable chips."⁸¹⁹ This tinkering with human nature causes a great amount of fear in many people. Political scientist, economist, and author Francis Fukuyama claims that "human nature, which provides continuity to our species and defines our values and politics, should not be altered."820 Former President's Council on Bioethics director Leon Kass writes, "If there is a case to be made against these activities - for individuals - we sense that it may have something to do with what is natural."⁸²¹ Even though people have been and will continue to be the creators of methods to improve their lives, it is wrong to say that nature is good and technology is bad. Thus, technology is not intrinsically evil – however, its uses by evil people can be. In a more practical sense, numerous ethical, technical, or social questions, comments, and concerns should be addressed before continuing down the path with these implants. "The concerns include apprehensions about safety, risk, and informed consent, issues of manufacturing and scientific responsibility, anxieties about the psychological impacts of enhancing human nature, worries about possible usage in children, concerns about increasing the divide between the rich and the poor, and most troublesome, issues of privacy and autonomy."⁸²² Any technology serves as a case study for how hard it is to foresee all events and impacts with accuracy. Whatever the case, the next step-however small-must be done to guarantee the ethical use and use of these and other next developing technologies.⁸²³

4.B.iii.d. Advancing the Ethical Debate

Humanity's ethical discourse progress is substantially different from science and technological progress. Evaluating a new technological advancement ethically and adequately is an intensive and time-consuming activity. Expert mathematicians may adeptly and efficiently use their knowledge to solve a mathematical problem. On the other hand, ethical expertise differs in drawing on an extensive and diverse collection of sources of information.⁸²⁴ Thoughts, ideas, and questions pertaining to the morality of human enhancement source information concerning how the various technologies will operate, be tested, and what kind of materials will be required to create it and its final cost to the consumer. This kind of information must be taken into advisement despite the desires and interests of the individuals or organizations who, directly or indirectly, are being enhanced. There is no easy path or shortcut to a well-rounded ethical evaluation. As with any investigation, too quick or simple of an evaluation of a piece of evidence can lead to an incorrect and error-filled conclusion. A widely acknowledged tendency for technological progress to outpace ethical understanding reflects differences in the nature of the tasks.⁸²⁵ The individuals who are focused on improving technology can direct their time and energy on the details of its design, whereas moral and philosophical evaluation is broader. Every new advancement of technology, enhancement included, has the ability to alter forever what it means and how it affects us as humans. All this strongly suggests the need to begin an ethical evaluation of possible future enhancements as soon as possible.⁸²⁶

Where does society go from here? Human enhancement's future needs to progress on two fronts – in both downwards and upwards directions. The downwards direction narrows its focus on more minor problems, breaking them up by their contextual features and tackling head-on the choices that decision-makers must make for future enhancements.⁸²⁷ On the other front, the

upwards direction deals with the more significant ethical and practical challenges that arise when society takes a step back and ponders the part that enhancement plays in the longer overall journey and future of humanity as a whole.⁸²⁸ Nick Bostrom and Julian Savulescu describe the requirements for the future of the enhancement debate:

This would require discussing how enhancement – including the prospect of future more radical enhancement – might interact with other macro-trends and global problems and prospects such as economic growth and inequality, existential risks and global catastrophic risks, molecular nanotechnology, artificial intelligence, space colonization, virtual reality, surveillance technology, democracy, and global governance, along with the deep epistemological, methodological, and moral questions that arise when one attempts to think about these interlocking issues in a serious and critical manner.⁸²⁹

While at first glance, it would seem that these two fronts would be in opposition to each other, that is not the case as they merely describe two coexisting intellectual paths, each with very different but equally important issues requiring attention.⁸³⁰ A potential problem for the future of enhancement ethics is having it get trapped in the middle of these two frontiers. The potential is high for the future of this century to develop unparalleled advances in science, technology, communication, analytics, and so many more spaces. Undoubtedly, these advances have the power to fundamentally alter the human condition as we know it – displaying both vast benefits and risks. The fate of humanity, now more than ever, is in our own hands.

4.B. Conclusion

It is clear that digital technologies have had and will continue to impact society and the economy immensely. While this impact and technology and society's relationship continue to grow exponentially, the need to take a cautionary "step-back" approach becomes increasingly more relevant to avoid the impending dystopia led by machines and technology with little room for humanity as currently known. Philosophy and ethics must be given a reasonable opportunity to question and analyze digital technologies to prevent a slippery slope of technological
developments that could cause a catastrophe. Technological determinism can be avoided by thinking of the possibilities in which the digital world can evoke a robust discussion of norms. Another way is to reimagine the way science and technology relate to action. In these actions, humans, surrounded by the science and technology we create – whether in a digitized form or not – are partaking in the decision-making progress that is slowly shaping the world's future.

These decision-making processes must include the described Precautionary Principle and Technology Assessment concepts. Failure to include these concepts in a decision-making process is irresponsible and unethical as it positions the future ethos towards digital technologies in the wrong way, ignoring a cautionary way of thinking. The application of these concepts extends to human improvement, especially the radical enhancements that give humans the ability to perform actions that would be considered outside the natural boundaries of humanity. Ethical concerns such as safety, fairness, access, and cost are just a few of the questions that need to be addressed moving forward with a cautionary mindset. Without a cautionary approach to digital technologies in healthcare, society could be doomed to a dystopian future because it fell victim to technological determinism.

4. Conclusion

As previously mentioned, the fourth chapter takes a different look at data analytics and focuses on the ethical concerns and challenges this technology can display. The social policy of genomic data, such as genetic discrimination, common heritage, and more concerns with distributive justice, were detailed. The section continued to detail challenges in data analytics with the autonomy paradigm and cloud computing, respectively. These issues pose a challenge to data analytics due to cloud computing's de-centralization of data storage. A critical factor in respecting the privacy and autonomy of the individual is having secured, centralized data storage

that can easily be maintained and traced back in the event of a data breach. The rest of the chapter laid a philosophical/theoretical foundation discussing the impact of technology in healthcare via the digital revolution and its relationship to society while looking at the future of this dynamic relationship and how data analytics could vastly alter this relationship down a slippery slope of a dehumanized world and human enhancement. A cautionary approach is taken in which concepts like technological determinism and the Precautionary Principle are explored. An applied analysis of these concepts was used within the topic of human enhancement citing ethical concerns and providing a foundation to advance the debate.

CHAPTER 5: ETHICS OF GOVERNANCE OF ARTIFICIAL INTELLIGENCE AND INFORMATION MANAGEMENT IN HEALTHCARE <u>5. Introduction</u>

The fifth chapter begins by looking at a few capabilities of artificial intelligence (AI) and its potential for revolutionizing healthcare. Data is evolving into a new type of money in today's world of interconnected things, fueling process transformation and commercial success. Data analytics is vital to AI, feeds it, and helps it grow. Data from various sources, including clinical records and "digital exhaust" generated by users as they carry out online information, are significant resources utilized to build sophisticated medical services driven by AI.⁸³¹ The best part is that a wealth of data is now accessible and expanding exponentially in the healthcare industry that can be utilized for analytics and AI. Since AI acts as the interpreter of all the information it receives using machine learning and algorithms, AI helps scientists, researchers, and healthcare data stewards, among many others, to understand and adapt to the clinical outcomes that the data forecasts. Philosophical concepts such as free will, moral disagreement, and moral agency are explored in light of AI. The section's consideration of the difficulties and possibilities faced by AI in providing healthcare serves as its conclusion. The fundamentals of

health information management and its moral and legal considerations are described in the following sections. Applied analysis is detailed within systematic reviews and risk management. The chapter ends with a call for healthcare information governance, outlining its need and essential elements while emphasizing information stewardship and data quality.

5.A. An Applied Analysis of the Ethics & Effect of Artificial Intelligence on the Delivery of Healthcare

5.A. Introduction

It may be difficult for the human mind to comprehend, but artificial intelligence (AI) possibilities are limitless. The technology associated with it is advancing at a pace that is faster than the law, ethics, and society can keep up with. By the time this chapter is published in the dissertation, it will already be outdated. So how does one keep all this knowledge organized and applicable? We, as human beings, cannot, but artificial intelligence can. This section presents the exponential potential of artificial intelligence, strengthened by machine learning and neural networks, and patterned after the human mind. The current state of AI 'awareness' is visited by mentioning the Turing test and other newer versions that show how far AI has come and where it can go.

The narrative then shifts to talking about the philosophy and ethics of artificial intelligence by highlighting its inherent conflict with free will and that moral disagreement will cause problems when designing an AI. The concept of artificial general intelligence is brought up to show that while AI can master a specific task, game, or area – it is far from being a 'jack-of-all-trades' and cannot do more than a few particular tasks at the moment. The philosophical and ethical issues with incorporating ethics into AI and the concept of idealizing moral agency are discussed in this section's conclusion.

The analysis continues by providing an applied study of artificial intelligence in healthcare delivery. Many challenges and opportunities are visited within digital health research in this field, proving that AI's changes in healthcare may not be as dynamic as one would presume. Finally, the section concludes by describing artificial intelligence safety engineering that should be noted as society progresses with building a better and more efficient AI.

5.A.i. Setting the Stage for Artificial Intelligence

Artificial intelligence evokes strong emotions. For starters, our preoccupation with intellect makes humans appear to occupy a unique position among living things. There are inquiries like "what is intelligence?" "how does one measure intelligence?" and "how does the brain function?" All these inquiries have relevance while attempting to comprehend artificial intelligence. However, engineers and computer scientists are primarily interested in the problem of the autonomous robot that behaves intelligently and acts like a person.⁸³²

5.A.i.a. The Exponential Potential of AI

AI is now the hottest topic in technology. In the business and technology sections of newspapers and publications, AI is often mentioned. According to Hector Levesque, many more companies are following the example set by prominent computer companies, including Microsoft, IBM, Google, and Apple, which have all made significant investments in the development of AI.⁸³³ Toyota made a \$1 billion investment in AI in November 2015, and Elon Musk launched a new nonprofit organization named OpenAI that received \$1 billion in financing in December 2015.⁸³⁴

What exactly is the fuss about? Let us say you think about what those vast amounts of money are anticipated to accomplish. Compared to the AI previously depicted in science fiction literature and films, the technology in that scenario looks noticeably different. There is not much

discussion of powerful disembodied intelligence like the HAL 9000 computer in the film 2001: A Space Odyssey or even sentient humanlike robots like those in the Blade Runner movie. Adaptive machine learning, or AML, is the name for the AI that tech firms are now considering. AML's main objective is to use substantial data processing to instruct a computerized system to display human-like intelligence.⁸³⁵ In reality, a significant factor in the current hype around AI is the potential utilization of what is sometimes referred to as big data.⁸³⁶

For the sake of explanation, Levesque gives a hypothetical example where the goal is to create a software program that can identify cats. In other words, its mission will be to separate the photos it receives into those with cats and those without. Then the question becomes how to develop this kind of platform. Previously, AI engineers would have attempted to create software that searched through images with particular cat attributes. It could resemble a cat face, with features like "an inverted pinkish triangle for the nose, whiskers, and fur, as well as greenish or yellowish eyes."⁸³⁷ The form of a cat, which has four legs, a tail that frequently pitches up, a small head, and triangular cat ears, may also be sought after. Or it could search for the cat fur's particular hue. If a picture has enough of these characteristics, the software will identify it as a cat picture or otherwise discard it.⁸³⁸

AML offers an extremely distinct approach, nevertheless. Users begin by presenting the system with a large number of digital photos, some of which contain cats and others that do not. The algorithm is then instructed to hunt for a set of traits that are present in several photographs to compress all of this visual data. A region with a certain uniform hue and brightness might be a characteristic. It could be a region with a noticeable edge-related shift in brightness and hue. The goal is to identify a collection of attributes that may be used to recreate the original photos in a manner that is as close to the original as feasible. Then you instruct your system to detach from those aspects and search for universal traits. There are a couple more rounds after that. Beyond what AI experts may have imagined even a few years back, this unregulated strategy to AML has been discovered to function quite effectively.⁸³⁹ Levesque notes, "The success of AML is usually attributed to three things: truly massive amounts of data to learn from (online, in specialized repositories, or from sensors), powerful computational techniques for dealing with this data, and very fast computers. None of these were available even thirty years ago."⁸⁴⁰

Naturally, no one wants to spend a fortune on cat recognition. Imagine instead that the images are mammograms, some of which may include cancers that are difficult for medical professionals to detect. Or consider a scenario where the data is entirely auditory. It could feature audio recordings of the topic of interest speaking on some of the tapes. Consider an alternative scenario where the data relates to financial transactions, some of which include forgery or embezzlement. Alternatively, the information may relate to online buyers with similar purchasing and browsing trends. Alternatively, the information might be the motion of the car's pedals and steering wheel relaying information given by the windshield. Numerous sectors with substantial economic and social repercussions may employ technologies that automatically learn using vast amounts of data.⁸⁴¹

5.A.i.b. Machine Learning & The Human Brain

It was formerly believed that computer vision couldn't even equal the skills of a one-yearold with visual development. This seems not to be the case anymore since machines can now identify objects in images just as well as the majority of humans can, and computerized cars can drive more securely than a rookie 16-year-old could.⁸⁴² However, these machines still have room for development. Additionally, computers have learned to perceive and navigate via experience,

following nature's route millions of years ago. Terrence Sejnowski gives an illustrative example, "What is fueling these advances is the gushers of data. Data are the new oil. Learning algorithms are refineries that extract information from raw data; information can be used to create knowledge; knowledge leads to understanding; and understanding leads to wisdom. Welcome to the brave new world of deep learning."⁸⁴³

With origins in computer science, neurology, and mathematics, machine learning has a branch called deep learning.⁸⁴⁴ Deep networks learn from inputs similarly to infants' learning from their surroundings, beginning with a blank slate and eventually picking up the abilities required to move about unfamiliar situations. In the past, there existed two hypotheses on how to create an AI in the 1950s, one of which dominated the industry for years and was centered on reasoning and software applications, and the other of which took much longer to develop and was centered on having to learn directly from raw data.⁸⁴⁵

Machine learning is necessary for artificial intelligence, and this goes beyond simple collection or coding issues.⁸⁴⁶ If a machine could not learn in a dynamic context, we would not define it as intelligent. On the condition that the system is willing to learn from and adapt to such adjustments, the developer need not anticipate every situation and supply remedies. Evolution worked as the system architect for humans, creating our biological makeup, entrenched impulses, and reflexes over thousands of years. Throughout our lives, we also learn how to modify our behavior. This enables humans to adapt to environmental changes that evolution is unable to foresee. All of an organism's behavior could already be present in short-lived creatures in well-specified habitats. Ethem Alpaydin states, "Still, instead of hardwiring into us all sorts of behavior for any circumstance we might encounter in our life, evolution gave us a large brain and a mechanism to learn so that we could update ourselves with experience and adapt to

different environments."⁸⁴⁷ Because of this, people have thrived in places worldwide with quite varying climatic circumstances. When we discover the appropriate course of action to take in a certain circumstance, that information is retained in our brains. When the same scenario occurs again, we identify it, remember the best course of action, and take appropriate action.⁸⁴⁸

Artificial intelligence is modeled after the human brain. Neuroscientists and cognitive researchers work to better understand how the brain works. They develop neural network models for this purpose and carry out simulation trials. But much like any technical discipline, computer science also studies artificial intelligence with the aim of creating practical systems. Even though the brain serves as our inspiration, we are not very concerned with the algorithms we develop feasibility in terms of biology. We believe the brain is fascinating because it might improve our design of computer systems. Alpaydin continues, "The brain is an information-processing device with incredible abilities and surpasses current engineering products in many domains – for example, vision, speech recognition, and learning, to name three."⁸⁴⁹ If used on machinery, these applications have obvious economic value. If we can comprehend how the mind executes these tasks, we may specify results, rewrite them as algorithms, and then apply them to systems.⁸⁵⁰

Computers are not brains, despite the term "electronic brains" being used in the past. A computer typically contains one or a few processors, whereas the brain comprises several parallel-operating processing units called neurons. The processors are said to be far slower and more basic than a standard desktop CPU; however, the specifics are unknown.⁸⁵¹ The vast connection of the brain is another feature that distinguishes it and is thought to be responsible for its processing ability. Tens of thousands of additional neurons are connected to one another in the brain through connections known as synapses, and they all function simultaneously. In a laptop, the memory is autonomous and idle, whereas the CPU is always working. However, it is

believed that computation and memory are distributed across the brain's network; the neurons process information while the synapses between the neurons store it as memory.⁸⁵²

5.A.i.c. The Turing Test & Other New Versions

The 1950 issue of the journal *Mind* included an article titled "Computing Machinery and Intelligence" by codebreaker and computer technology trailblazer Alan Turing.⁸⁵³ It was the first comprehensive academic examination of the idea of artificial intelligence. By the year 2000, individuals will, in Turing's words, "be able to speak of machines thinking without expecting to be contradicted."⁸⁵⁴ He believed that machines will be capable of passing the Turing Test, which has since become well-known. There is a gaming component to the Turing Test. Through a keyboard and screen, a third person—the "judge"—conversates with two "players"—one human and the other artificial.⁸⁵⁵ In order to determine which player is a human and which is a computer, the judge converses with each participant in turn. The argument makes the case that the computer aims to persuade the judge that it is a human—an achievement that would need human intellect.⁸⁵⁶ The machine passes the test if the judge is unable to tell the difference between a human and it. In 1950, Turing foresaw a day when "thinking machines" would be pervasive in the home and office, and computers that could pass his test would be the norm.⁸⁵⁷

Despite Turing's forecast, human-level AI was neither developed nor shown to be on the horizon by the year 2000.⁸⁵⁸ The Turing Test could not be passed by any computer. However, one crucial step forward in the area of artificial intelligence has been achieved. Garry Kasparov was dethroned as the global chess champion in 1997 by IBM's Deep Blue computer.⁸⁵⁹ Kasparov supposedly said that when he played against Deep Blue, there was an "alien intelligence" on the opposite side of the table, in contrast to earlier chess algorithms he had defeated, which were

foreseeable and conventional.⁸⁶⁰ Computer chess has now been conquered. Even still, it appeared like we were still a long way from having AI that could compete with humans.

How is this possible? Deep Blue had an issue in that it was a specialist.⁸⁶¹ It was limited to playing chess. In contrast, an average adult completes various tasks employing a variety of sensorimotor abilities. A human being is essentially a generalist or jack of all crafts. A top human chess player can do much more than merely play the game. A human person is also adaptable. Chess is only one of many successful applications of AI research, which stands in striking contrast to the field's inability to create a computer with general-purpose, adaptive intelligence.⁸⁶² Today's AI has become much closer to passing the test since the Deep Blue days – however, are ethics and morality being used in these AML data sets? How are these AI being designed from a philosophical perspective?

5.A.ii. Philosophy & Ethics of Artificial Intelligence

Creating sentient robots presents many ethical issues, including how to prevent these machines from harming humans and other morally significant species and the morality of the technologies altogether.⁸⁶³ This section examines potential ethical issues that might develop when we develop different types and levels of artificial intelligence. As Roberto Simanowski said, "The future no longer rests with politics or even the philosophy of Plato's *Republic*. It rests with science, or more precisely, with computer science... It's people like Sundar Pichai and Mark Zuckerberg who, every day, with every new piece of data, increasingly determine the future we're rushing toward."⁸⁶⁴ Do these people have the right to decide how scientific advancements like atom fission and DNA sequencing will be applied in society? Do they adhere to societal pressures to avoid researching technologies whose effects cannot be predicted or managed? Or would they look for a community with different moral principles and governing laws?⁸⁶⁵

5.A.ii.a. Conflict with Free Will & Moral Disagreement

Theresa May said at the 2018 World Economic Forum in Davos: "Imagine a world in which self-driving cars radically reduce the number of deaths on our roads. Imagine a world where remote monitoring and inspection of critical infrastructure make dangerous jobs safer. Imagine a world where we can predict and prevent the spread of diseases around the globe."⁸⁶⁶ But to what degree do science and start-ups control the future? How should we interpret the fact that, concerning artificial intelligence, the risk of unanticipated effects now includes the possibility that the creation may retain its creators as pets rather than serving them as slaves as was originally intended?⁸⁶⁷

There are two ways to think about the relationship between AI and ethics: technology ethics in how it interacts with humans and human ethics in using technology.⁸⁶⁸ The first group includes, in addition to the usual subjects of security, equality, accountability, prejudice, and monitoring, the dangers of a "Black Box Society" and the "ethical framework for a Good AI Society."⁸⁶⁹ The second category asks what moral standards we should include in technology to treat humans as partners rather than as pets, even though humans are no longer in authority. At its foundation, the problem is that the technology we develop calls into question our free will in two ways: first, it hinders our ability to do those things entirely differently, even if we truly want to, and secondly because it causes us to lose the ability to want anything else.⁸⁷⁰

One way is that it is difficult to deviate from the rules due to the if-then logic's extreme dominance.⁸⁷¹ When an algorithm drives our cars, we are not authorized to travel faster than the stated speed limit. The situation could be welcomed by police enforcement; however, constitutional attorneys do not welcome such a situation. The existence of autonomous cars at the expense of free, regular human activity was criticized by a German ethical panel on artificial

intelligence and self-driving cars established in 2016. The panel, which included a constitutional judge, was critical that AI required drivers to obey traffic laws.⁸⁷²

Their reasoning resulted in the following: "There is no ethical rule that always places safety before freedom," which has a long philosophical history in Germany.⁸⁷³ It may be found in Kantian ethics, which maintains that doing what is right must be done (as a responsibility) and that evil must exist in order for good to prevail.⁸⁷⁴ When AI assumes control of cars and, therefore, civilization, there will not be such a notion as right behavior in the concept of individual choice. However, positive conduct in the sense of tangible effects will be present. What is novel in this situation is that artificial intelligence, which is more aware of what is beneficial to humans, would establish and enforce roles rather than human authority.⁸⁷⁵ The subtext of the German ethical commission's concerns is that the self-driving car represents a demonstration of the future cybernetic management of society.⁸⁷⁶

The absence of agreement among moral philosophers over the most appropriate theory of ethics has been pointed out as a challenge to the advancement of machine ethics.⁸⁷⁷ The best method to use in creating artificial moral beings may be obvious to someone who adheres to a specific philosophy of ethics. However, it is still uncertain how institutions and society will go ahead. Businesses, authorities, and scholars will have to make a tough decision between conflicting moral frameworks over which approach to use for intelligent machines.⁸⁷⁸ Moral conflict may be very problematic for the machine ethics endeavor for two potential reasons. As conflicts among engineers, legislators, and philosophers obstruct cooperative efforts, it might be a practical issue. Kyle Bogosian notes, "In a worst-case scenario, decisions over which research programs to fund will turn into bitter ideological battles, research agencies will become bogged down in disputes, and developers will split up their resources and devolve into a competitive

mindset which reduces information sharing and slows research progress, thus making it more difficult to construct more on machines."⁸⁷⁹

Additionally, there could be a moral dilemma. Given how uncertain we are about morality and how split we are among numerous moral notions, it is theoretically improbable that anyone's particular moral theory is entirely true.⁸⁸⁰ As a result, if we create AIs based on a certain moral framework, they will undoubtedly make many unethical choices.⁸⁸¹ Whether the computational architecture of a robot is essentially utilitarian or not is unrelated to a utilitarian engineer; as long as utility is being maximized, the engineer is motivated only by self-interest. This is true even when individuals agree on moral decisions while differing on moral theories.⁸⁸² In general, if we are certain that computers will behave morally, there is no cause to believe that the precise engineering of machine code is ethically meaningful.⁸⁸³ Therefore, moral differences across ethical theories would provide comparable practical and ethical issues for the advancement of bottom-up computer morals in the same way they may for the advancement of top-down computing ethics.⁸⁸⁴

5.A.ii.b. Artificial General Intelligence

Even while AI systems have surpassed humans in many specialized fields, such as chess, there is almost unanimous consensus among current AI specialists that artificial intelligence misses the mark of human abilities in some key aspects.⁸⁸⁵ Even the IBM researchers who created Deep Blue agree that something crucial is missing from contemporary AIs. Some have argued that once AI researchers discover how to do something, that ability fades away from being considered intelligent. Chess was once thought to be the pinnacle of intellectual prowess until Deep Blue took home the global title.⁸⁸⁶

"Artificial General Intelligence" (AGI) is the growing term of art used to define "real" AI, even though this branch of artificial intelligence is just recently beginning to take shape.⁸⁸⁷ As the name suggests, the widespread agreement is that generality is the missing attribute. Current AI algorithms are defined by a purposefully designed competence exclusively in a specific, constrained area, with human-equivalent or better accuracy.⁸⁸⁸ Deep Blue won the chess world championship, but it is unable to do anything else, let alone drive a vehicle or find anything useful. Except for Homo sapiens, these contemporary AI systems mimic every form of organic life. Bostrom & Yudkowsky give an example, "A bee exhibits competence at building hives; a beaver demonstrates competence at building dams, but a bee doesn't build dams, and a beaver can't learn to build a hive. A human watching can learn to do both, but this is a unique ability among biological life forms. It is debatable whether human intelligence is truly general – we are certainly better at some cognitive tasks than others – but human intelligence is significantly more generally applicable than non-hominid intelligence." [Emphasis not added]⁸⁸⁹ It is not difficult to imagine the scope of safety problems that could occur from an AI functioning within a specific context, like Deep Blue in chess. However, the scope of potential safety problems becomes entirely different when referring to AGI functioning across numerous contexts - all of which cannot possibly be predicted by engineers.⁸⁹⁰

In the development of Deep Blue, the human engineers did something revolutionary – they gave up their ability to predict Deep Blue's local, specific game behavior.⁸⁹¹ "Instead, Deep Blue's programmers had (justifiable) confidence that Deep Blue's chess moves would satisfy a non-local criterion of optimality: namely, that the moves would tend to steer the future of the game board into outcomes in the winning region as defined by the chess rules."⁸⁹² Although this prediction about future outcomes turned out to be satisfactory, it prevented the programmers

from foreseeing Deep Blue's local behavior or its reaction to a particular assault against its king. This resulted from Deep Blue computing the non-local game map—the relationship between a move and its potential aftereffects—more precisely than its developers could.⁸⁹³

Today, human beings can perform a nearly infinite number of actions to feed themselves.⁸⁹⁴ While many of these contemporary feeding actions may not be "envisioned by nature," it is no surprise that humanity has evolved and adapted to many of the obstacles to feeding that plagued our ancestors. Bostrom & Yudkowsky continue, "But our adapted brain has grown powerful enough to be *significantly more generally applicable*; to let us foresee the consequences of millions of different actions across domains and exert our preferences over outcomes."[Emphasis not added]⁸⁹⁵ Despite the fact that none of our predecessors faced a hardship comparable to a vacuum, humans have traveled across space and left footprints on the Moon. Designing a system that would function securely in thousands of situations, including circumstances not expressly imagined by developers or consumers and situations that no person has yet experienced, is a fundamentally different task from domain-specific AL.⁸⁹⁶ There may not be a concise local specification of all the methods that people get their daily bread in this situation, nor is there a local definition of good conduct or a clear specification of the actions themselves.⁸⁹⁷

In order to construct an AGI that behaves safely while operating in various contexts, one must define appropriate behavior using phrases like "X such that the consequence of X is not harmful to humans" (where X represents any action).⁸⁹⁸ This way of defining AGI acceptable behavior requires predicting the future outcomes of actions, making them non-local. These future outcomes of actions are non-local in the same sense that Deep Blue extrapolated a set of outcomes from chess moves and steered its actions in the winning direction. The actions of the

AGI will be based on how it interprets the consequences of its actions relative to whether or not it causes harm to humans. This specification is only useful if the AGI system clearly extrapolates the consequences of its actions and subsequently implements the output of how the action affects humans into a design characteristic. A toaster, for instance, cannot have this design characteristic since it cannot predict the results of toasting bread.⁸⁹⁹

Suppose AGI's behavior will be based on future outcomes of actions across a variety of contexts. How can developers promise that AGI will act in an ethical manner in all instances?⁹⁰⁰ To help illustrate how ridiculous it would be to predict an answer to that question from the developers' perspective, Bostrom and Yudkowsky present the analogy of a statement from an engineer, "Well, I have no idea how this airplane I built will fly safely – indeed, I have no idea how it will fly at all, whether it will flap its wings or inflate itself with helium or something else I haven't even imagined – but I assure you, the design is very, very safe."⁹⁰¹ Clearly, from the standpoint of public relations, this would be a bad situation. So, what can developers do to address the question? It should be noted that purely hopeful expectations have been and will continue to be a major problem in AGI research and development.⁹⁰² Building a trustworthy AGI will need new approaches and a fresh style of thought; it will demand that AGI has the mentality of a human engineer who is ethical, not merely a result of ethical engineering.⁹⁰³

5.A.ii.c. Building Ethics into AI and the Idealization of Moral Agency

There must be a higher degree of complexity in AI rules of ethics that addresses the conduct of machines. The issue of control may be solved in many ways. In addition to creating codes of ethics, one solution may be to include ethical behavior and decision-making in AI. A mandate for AI work or a suggestion to program ethical conduct into computers may even be included in such codes.⁹⁰⁴

Is designing ethical behavior into AI and computers the logical next step in developing AI? Paula Boddington summarizes the concept of building ethical behavior into AI and computers in her book *Towards a Code of Ethics for Artificial Intelligence*, where she attempts to reconcile the errors of programming ethical AI behavior in the literature that, in her view, idealize or over-simplify the idea of AI as a moral actor.⁹⁰⁵ There are compelling reasons to take every precaution in AI development and application to protect security and ensure that machine conduct complies with moral principles.⁹⁰⁶ Boddington goes further, "But the question about building ethical decision-making and action into AI goes further than this [precautions in AI development], for it concerns judgment in novel, perhaps unpredictable situations, where decisions and actions would be taken without any immediate human oversight; it goes further than the simple alignment of outcomes with our ethical values if it implies that it's the machine itself which is acting morally."⁹⁰⁷

Numerous antecedents exist of mechanistic actions taken without direct human oversight or intervention, like failsafes (which are safety mechanisms designed to cause a machine to revert to a "safe mode" in the event of a breakdown or malfunction to prevent harm or catastrophe) used in trains to deal with tragedies like driver collapse (in this case would involve an automatic stop of the train).⁹⁰⁸ However, the failsafes function in systems with constrained capabilities. Creating mechanisms to make machine judgments "ethical" could appear like an alluring option for networks AI where choices and actions might be made that could have wide-ranging and possibly difficult-to-find impacts.⁹⁰⁹ Discussions of difficult moral quandaries are not just a defining feature of ethics literature generally but also of AI ethics literature specifically. Therefore, without a thorough explanation of ethics, efforts to integrate morality

into technology may be effectively concentrated at a minimum in seeking to prevent horrible outcomes.⁹¹⁰

It is challenging to define what exactly are horrible outcomes when it comes to catastrophes. Which is worse: running over a newborn or six people who are 59 years old? Which scenario is worse for an accident victim—death or coma? Many of the potential consequences of AI now lie at the farthest reaches of our capacity for imagination; they swing from "wonderful" to "catastrophe" like an Olympic gymnast swinging from one end of the balancing beams toward the other.⁹¹¹ Without a clear understanding of our value objectives, designing a computer to answer ethical issues will be challenging. We lack such a clear vision, particularly for difficult-to-conceive, intricate alternatives. So, might we teach a computer to identify our "true" objectives? What criteria would the computer use to determine our genuine objectives? Maybe the machine or we can figure out what our "true" nature is, but do we even have one? Then, is our nature set? Can this be a topic for empirical research? This is a philosophical problem of the utmost complexity.⁹¹²

Furthermore, even if we were to imagine that a computer could be built to ascertain our moral objectives, this would only bootstrap the issue. We would constantly need to be able to confirm that the result met our standards for morality. Boddington gives a stark example, "Are we going to accept that, say, wife-beating was ethical after all, particularly if she's burnt the dinner and had sloppily applied makeup, just because we've got an app that told us it was okay? I hope not."⁹¹³

One of the main ideas is that ethics must always allow for discussion and collaboration with others who have justifiable interests since they could be impacted or have something to add. This is all opposed to outsourcing ethics to a computer that is not a part of a network of such

human interaction.⁹¹⁴ Should technology advance to the point where computers possess the same level of moral agency as people? Even if such a computer had intriguing things to say, outsourcing your ethics to another person would still be unethical.⁹¹⁵

Like outsourcing our ethical decisions and acts to another human, outsourcing them to a computer has major ethical implications. In consequentialism, just the results of our actions are morally significant; it does not matter how you came to a decision under this agent-neutral morality if it is the right conclusion.⁹¹⁶ Therefore, in theory, you may delegate your ultimate decision to a capable computer. However, note that this computer would develop a decision-making process to apply morality and do empirical calculations to determine the most effective way to accomplish a moral objective.⁹¹⁷

On the other hand, standing in opposition to consequentialism is deontology and virtue ethics, where intentionality behind the action is key to the morality of the decision. Therefore, if using Kantian and virtue ethics frameworks, you cannot outsource moral decisions to another individual or an AI. In other words, to be a moral agent, you need to take responsibility for your actions, motivations, and decision-making processes. Boddington concludes, "You have to do the right thing, for the right reasons, in the right manner. Even many consequentialists are troubled by this and try to work around it. Remember the Nuremberg trials? The quintessentially lousy excuse of the twentieth century was, I was only following orders."⁹¹⁸ The realization that individuals cannot delegate moral judgments is the essential moral discovery of the 20th century and the foundation from which all successive standards of ethical conduct and laws have been established. That is a determination of undeniable moral duty.⁹¹⁹ The next section will detail some of the various challenges and opportunities that artificial intelligence can have in the

healthcare landscape – from digital health research to ethical, social, and legal implications of AI in healthcare.

5.A.iii. Challenges & Opportunities for Artificial Intelligence

Healthcare is being altered because the digital revolution is disrupting health research. With the development of digital health technology, vast amounts of big data, both qualitative and quantitative, have been generated. These data are significant sources of knowledge about consumer transactions that may be useful to patients and their caretakers.⁹²⁰ Digital data "exhaust," or the bits of ordinary actions caught in daily digital interactions, is particularly interesting since it captures these behaviors in real time and contains our natural activities. The way these sociotechnical systems affect our daily lives and the way they affect society will undoubtedly depend on the vital social dialogues that take place.⁹²¹ Big data, enhanced by artificial intelligence, becomes a socio-technical phenomenon since these technologies can alter our lives in significant ways and change the direction of society for better or worse. It is through intellectual discussion, debate, and reflection that society can choose to let these technologies artificial intelligence will encounter in society and highlight certain points to keep in mind as the discussion moves forward.

5.A.iii.a. Challenges and Opportunities in Digital Health Research

The usage of artificial intelligence technologies in healthcare is growing as these AIs use data analytics to improve digital health research. Medical records offer training datasets for learning that educate AI systems that can discover abnormalities more efficiently than qualified staff in cancer diagnoses, internal medicine, and ophthalmology, for example.⁹²² AI assists in health research and development in order to supplement or provide a substitute for

standard medical treatments, for instance, digital therapeutics are making an effort to expand and get products into the health sector.⁹²³ Nebeker et al. state, "While the digital health revolution brings transformational promise for improving healthcare, we must acknowledge our collective responsibility to recognize and prevent unintended consequences introduced by biased and opaque algorithms that could exacerbate health disparities and jeopardize public trust."⁹²⁴ To prevent these negative consequences, it is essential to raise the minimum safety requirements for digital health technologies to be accessible to the general public. These technologies must endure significant safety examination and have been shown useful in genuine clinical research settings before they are released into the health sector.⁹²⁵

Despite the enormous potential of digital health technologies, those making decisions regarding the purchase, evaluation, application, and assessment of technology in healthcare face significant ethical challenges.⁹²⁶ The leaders in the health sector face ethical challenges due to the absence of policy regulations or standard requirements and a combination of new, quickly developing machinery, new relevant parties (such as tech behemoths, digital therapeutic start-ups, and civilian researchers), humungous stores of data, and continuous new computational techniques.⁹²⁷ These technologies should not enter clinical settings or research without proper verification. For instance, the adage "if the product is free, then you are the product" is often used concerning digital technologies today.⁹²⁸ This implies that the information used by businesses to guide product enhancement is generated by our search phrases, keyboard inputs, clicking, and taps. With the help of this "big data," computers are trained to produce, for instance, customized adverts. Customers acknowledge and agree to these terms of service, which are not often intended to be easily readable or comprehendible, by clicking "I Accept" to signify their understanding and agreement.⁹²⁹ Digital health technologies must demonstrate that they are

not using their consumers' personal health information for personalized ad revenues and have undergone rigorous testing to ensure that won't happen to users as this compromises personal health privacy and security.

At times, digital health technologies may be helpful in how they use your health data. For example, a smartwatch can remind you to exercise daily, drink more fluids, or schedule a yearly checkup with your doctor. The AI may be close to being accurate sometimes, yet it can also be completely off. Nebeker et al. give an example "if you were to write something on Facebook that its proprietary AI interprets as putting you at serious risk, it may send the police to your home! Is Facebook getting it right? We do not know; Facebook has claimed that, even though its algorithm is not perfect and makes mistakes, it does not consider its actions to be research."⁹³⁰ Facebook does not consider its algorithmic actions research because they do not publicize the data nor use any of its efforts to benefit anyone or anything outside the company. Facebook regards that these actions (such as sending the police) are done for the health and wellness of its users and have their best interests at heart, i.e., Facebook claims they are doing the "right" thing. This is most likely far from the case, as each of its users is just another revenue source from personalized ads, and it's in Facebook's best interest to keep as many as possible of those revenue streams open. We should examine the implications of informed consent concerning one's privacy, whether the risk of damage is objectively weighed against the possibility of benefit, and if those involved in the quality testing phase are those who will benefit from it at all.⁹³¹

Applying modern technology and AI systems to digital medicine presents new difficulties and requires various skills. It is possible for technology developers to lack patient-centeredness and produce technologies with minimal practical value.⁹³² AI may be trained by computational

scientists using datasets that are not typical of the general population, which would restrict its capacity to make accurate judgments or predictions.⁹³³ Clinicians can lack confidence in AI-generated choices or knowledge of handling complex data depth.⁹³⁴ It is crucial for research on digital health and the use of AI in the healthcare industry to understand this separation and develop tactics to close gaps and forge stronger ties across such groups.⁹³⁵

Numerous initiatives are being undertaken to address the digital age's moral, regulatory, and cultural ramifications in healthcare. Leading examples of these projects relate to AI. The areas of focus for AI range from autonomous driving to face recognition, the future of work, urban development, and occasionally even healthcare.⁹³⁶ Nebeker et al. highlight a few carefully chosen contemporary AI initiatives that seemed to be well-funded and cooperative projects whose goals are to explore fundamental issues of AI's relationship with society.⁹³⁷ These programs all include evaluating AI's possible ethical, legal, and societal implications (ELSI). Organizations examining AI via an ELSI perspective want to create guidelines that may be applied or changed globally, similar to the impact of the General Data Protection Regulation of the European Union on countries outside of the EU.938 In reality, however, most current attempts to link ELSI to AI are nebulous, might have overlapping focus, and lack precision. The healthcare technology revolution involves a broad spectrum of technologies, even though AI plays a part in it. Other projects focus more explicitly on ELSI in sensors worn for digital research, social network platforms, and mobile applications.⁹³⁹ These projects aim to inform governance and policy in a mostly uncontrolled environment. These programs serve as a few examples, however, it is essential to emphasize that several research facilities and institutions are engaged in ELSI for digital health.940

5.A.iii.b. Static or Changing Capabilities & Dynamics in Healthcare

In terms of variety and efficacy, it is easy to anticipate that AI systems for healthcare applications will keep expanding. Jim Warren comments, "Supercomputing is now readily available: the graphics processing units in the video cards of home computers turn out to be superb number-crunchers for neural network algorithms, or we can rent scalable computing power through cloud computing services by Amazon, Google or others."⁹⁴¹ Additionally, as computers are increasingly incorporated into healthcare systems, a growing library of electronic medical records that are ready for examination is a natural by-product. Even though this AI explosion is likely to revolutionize healthcare delivery, there are many grounds to believe that these changes will be gradual, controlled, and perhaps overall beneficial.⁹⁴² Here are three reasons why the shift to using AI in healthcare will not be far from the current technologies and practices.

First, deep learning-based AI algorithms are not all that dissimilar from the computer tools humankind has been using daily for years. For example, PREDICT combines decision assistance technology with ongoing, prospectively planned, open cohort research.⁹⁴³ Wells et al. continue, "The PREDICT software integrates with the practice management system to retrieve patient data, with any remaining required data entered interactively to provide an individualized estimate of the probability of a cardiovascular disease (CVD) event in the next five years, along with treatment recommendations."⁹⁴⁴ A recent study to enhance risk prediction was based on more than 400,000 patient contacts in New Zealand between 2002 and 2015, and participant risk variables were gathered by equipment often linked to government databases, including hospital admissions and mortality from CVD.⁹⁴⁵ A regression model, which assigns each risk factor a specific weight, is the foundation of the risk prediction. Despite having a much simpler structural

design than a deep learning model, the model offers the benefit of having a simple justification for each suggestion. Deep neural networks are now being researched for explanation capability.⁹⁴⁶ When adopting a deep learning AI, patients and healthcare professionals would likely have a somewhat different experience than they would with PREDICT, which has been seamlessly incorporated into the current healthcare system and professional responsibilities.⁹⁴⁷

Second, the relationship between a doctor and patient will be challenged by AI, but information technology (IT) has always posed similar problems. Access to knowledge has been democratized for more than 25 years thanks to the World Wide Web.⁹⁴⁸ Patients are free to bring printouts of the most recent study results and maybe dubious information slanted toward revenue-generating into their consultations (although it seems that today's patients are more inclined to wield their smartphones or tablets). Warren notes, "As the web has become more sophisticated and IT reaches ever more intimately into our lives, the diversity of ways patients may bring IT into their healthcare has grown, including mobile apps, fitness trackers, and blog posts. An interesting example is PatientsLikeMe, a web-based network where patients connect to others with the same disease and share experiences."949 Together with the planning of research studies, such as those on the efficiency of off-label medication usage, sharing statistical data is encouraged.⁹⁵⁰ Eric Topol writes in his book, The Patient Will See You Now, that medicine has reached a "Gutenberg moment" in which greater information freedom has allowed patients to take a revolutionary level of control over their treatment.⁹⁵¹ Topol lists various web-based and IT-enabled developments, such as big data sharing and commercial genetic test findings, as shown by 23andme.⁹⁵² Interestingly, evidence-based care is beginning to use smartphone message services more often. An example is a program that included behavior-change strategies and motivating messaging that dramatically increased smoking cessation rates after six

months.⁹⁵³ This service is a kind of AI since it combines several operationalized intervention approaches and interaction tactics. It may be downloaded from the internet or prescribed to a patient by a physician.⁹⁵⁴

Thirdly, by regularly investigating the source of information, healthcare practitioners may participate in, support, or restrain the development of AI. AI-based decision assistance could be offered to a patient or linked with the tools one uses at a district health board or primary healthcare organization. Warren concludes, "In any event, you can query where it comes from who is endorsing and distributing it, and what is their motivation (i.e., is it purely for profit through proliferation – licensing fees or banner-ad revenue – or is it publicly funded; does a medical body endorse it?). Is it part of big data's new wave of AI-based machine learning?"⁹⁵⁵ Or maybe, like in the case of quitting smoking above, the capacity is the result of "knowledge engineering," in which methods endorsed by a medical organization are utilized. If data were used to support these endorsed methods, when and where were they gathered? Do you think the data accurately represents the patient group, or are there clear gaps? Is it possible to retrain the system to utilize local data? Does the system have any evidence to support its suggestions, or is it simply a "black box" with no specific data to back up its analysis?⁹⁵⁶ Is there any proof of the system's efficiency? If so, how was its performance assessed? In what setting, with what population, for how long, and most importantly, against what? If the answers to these questions are elusive, one should be wary (or at the very least cautious); if the responses are inadequate, one should aggressively discuss the system's shortcomings.⁹⁵⁷

5.A.iii.c. Artificial Intelligence Safety Engineering

In spite of our accomplishments in creating computers that can pass the moral Turing test, performing like a human requires certain unethical activities that should not be tolerated by

the robots we build.⁹⁵⁸ Or, to put it differently, we need completely safe and law-abiding machinery, not robotic systems that are fully ethical individuals debating what is morally correct. Beautifully said by Robin Hanson: "In the long run, what matters most is that we all share a mutually acceptable law to keep the peace among us and allow mutually advantageous relations, not that we agree on the right values. Tolerate a wide range of values from capable law-abiding robots. It is a good law we should most strive to create and preserve. Law really matters."⁹⁵⁹

Yampolskiy thus suggests that in the framework of a new discipline he names artificial intelligence safety engineering, scientific activity targeted at constructing safe machines be added to primarily philosophical concerns of machine ethics.⁹⁶⁰ In this crucial area, some actual work has already started. An ongoing problem in a study on AI safety is the notion of enclosing a super-intelligent entity in equipment to inhibit it from harming people. These ideas were advocated by futurists in science like Eric Drexler, who proposed containing transhuman robots in order to investigate and use their impacts securely.⁹⁶¹ Similar to this, futurist Nick Bostrom has put forward the concept of an AI oracle that can only respond to questions.⁹⁶² Not to mention, in 2010, David Chalmers suggested the concept of a "leakproof" singularity.⁹⁶³ For safety reasons, he suggested that AI algorithms should first be restricted to virtual simulations until their propensities for conduct can be grasped in controlled settings.⁹⁶⁴

Yampolskiy published a defined idea for an AI containment method that characterizes AI-Boxing as a technology security issue.⁹⁶⁵ He adds, "The artificial intelligence confinement problem is defined as the challenge of restricting an artificially intelligent entity to a confined environment from which it cannot exchange information with the outside environment via legitimate or covert channels if the confinement authority does not authorize such information exchange. An AI system that succeeds in violating the confinement problem protocol is said to

have escaped."⁹⁶⁶ Asking "safe questions" and allowing just certain replies is the foundation of the suggested approach. A safe question is one that a person might respond to alone, without the aid of a superintelligence.⁹⁶⁷ How does this help? Yampolskiy gives a hypothetical example: consider a researcher looking at two treatment options for cancer. Without the AI's aid, each could be tested and verified, but it would take around three years of work. If the scientist thinks both have an equal probability of succeeding, which should be tested first? It will still take three years for humankind to find a cancer cure if the incorrect option is picked to be tried first. What if we could ask the AI to recommend the initial course of action?⁹⁶⁸

For a variety of reasons, this investigation is safe. First, the likelihood that each probable response will be accurate is equal. Second, requesting the AI for aid would just hasten the process without altering the conclusion since a human may respond to it without the AI's assistance. It is the same as getting fortunate while attempting to predict multiple-choice answers. Last but not least, the response to asking AI to recommend an initial course of action from the above example might be programmed in a single bit, making it difficult to conceal subsequent data. A panel of experts might be employed to assess a prospective question to ensure it is secure. All specialists should be qualified AI safety engineers, which means they are knowledgeable about the architecture of the AI, its nature of confinement, and the most recent advances in machine ethics.⁹⁶⁹ It may also be necessary for experts to get training in computer psychology, a field that does not exist but might do so in the future. The topic of analysis offered by Yampolskiy et al., which may be very useful in the field of AI development is "artimetrics," which "identifies, classifies, and authenticates AI agents, robots, and virtual reality avatars for security purposes."⁹⁷⁰

Yampolskiy suggests that creating safety procedures for self-improving systems is the main difficulty facing AI safety engineering.⁹⁷¹ He concludes, "If an artificially intelligent machine is as capable as a human engineer of designing the next generation of intelligent systems, it is essential to ensure that any safety mechanism incorporated in the initial design is still functional after thousands of generations of continuous self-improvement without human interference."972 With each successive generation, the self-improving system would hopefully be able to provide independent verification of its security for outside scrutiny. Allowing a safe, intelligent computer to create a fundamentally risky update, creating a more robust and deadly system, would be devastating. Some claim that either this problem cannot be solved or that, even if it can be solved, the accuracy of the answer cannot be established.⁹⁷³ The number of design flaws rises proportionally or sometimes even exponentially with system complexity, depending on the system. The most challenging system to debug, a self-improving system, will break all safety assurances with only one defect. Even worse, a problem could be created even after the design has been completed, either as a consequence of a natural event, such as a short circuit that alters a system component or as a result of a random mutation caused by defective hardware.⁹⁷⁴

5.A. Conclusion

This section has demonstrated the impact of artificial intelligence on the world with a specific focus on its application to healthcare delivery. The utilization of AI within healthcare can cause significant changes in the effectiveness, timeliness, and robustness of medical information. Fortunately, the changes to the doctor-patient relationship will not be significantly affected as small changes have been implemented over many years within healthcare, and it has not considerably impacted this dynamic. This analysis aims to bring knowledge and awareness of the exponential potential of artificial intelligence in its impact on society – specifically on

healthcare. This goal was achieved by setting the stage for artificial intelligence and providing information about machine learning and how much of how AI operates is modeled after the human brain via neural networks. The Turing test was mentioned, and it is one of many tests now to show the AI's overall general awareness of how it relates to the world around it.

The philosophy and ethics of artificial intelligence were introduced to provide a context for the many issues involved in designing and building an AI with a proper moral compass. The inherent nature of AI conflicts with humanity's free will, as detailed in the analysis of autonomous vehicles. In addition, the very nature of moral disagreement will lead to many conflicting views and opinions on which ethical theory or standards should be predominant when developing artificial intelligence. The concept of artificial general intelligence was brought to light in how it describes AI as having the ability to complete only a handful of particular tasks and not being able to handle numerous general tasks. This sets the AI apart from being 'real' and is one of the distinguishing abilities between humans and animals. This section described the thought process and necessary philosophical idealization of moral agency when building an AI.

The final section addressed an applied analysis of artificial intelligence in healthcare delivery. It described challenges and opportunities presented within the digital health research field, which AI prominently utilizes. The static or changing capabilities and dynamics in healthcare were addressed in light of AI. The field will not be overtly altered because many changes have slowly been implemented over the years and should not cause significant disparity. Finally, this section analyzed AI's ethics and effect on healthcare delivery in describing artificial intelligence safety engineering and how it must be utilized as society develops.

5.B. The Impact of Information Management on Data Analytics

5.B. Introduction

Information management dramatically impacts the development and organization of big data and data analytics. Health information management is a massive part of this development and has an exponentially growing influence on our society today. Starting from the foundations of health information management (HIM) in its acquisition, storage, and use of this data -one can begin to understand the infrastructure behind this massive management undertaking. Legal and ethical aspects of HIM are crucial as privacy, confidentiality, and data-sharing concepts must be addressed in this growing field. The current complexity of payment encoding systems exacerbates HIM's ethical problems. To remedy this, ethical decision-making must occur when coding HIM systems, and the workplace must foster a strong sense of staff justice. In an applied analysis of HIM, concepts of risk management and risk perception in the information management workplace are crucial to rethinking an ethical approach to HIM. Finally, society must progress towards a health information technology governance in which data quality characteristics are clearly defined. A proper application of normative health information technology governance and information stewardship can reshape the future of the healthcare landscape.

5.B.i. Foundations of Health Information Management

5.B.i.a. Acquisition, Storage, & Use

The concepts of illness and how to cure it have long been intertwined with those of data observation and analysis. Whether we look at the descriptions of diseases and recommendations for treatment in ancient Greek literature or the utilization of sophisticated laboratory and X-ray investigations by contemporary doctors, it is obvious that collecting data and understanding its significance are essential steps in the healthcare process. The sheer volume of data that may be utilized in patient care has increased dramatically with the shift toward genetic information in evaluating individual patients (their risks, prognosis, and anticipated responses to medication).⁹⁷⁵ As a consequence, data gathering, storing, and use issues are often brought up in the literature on health information management. This section will provide a starting point for summarizing these information management issues, covering numerous clinical applications.⁹⁷⁶

Evolution of Health Information Management

Data are essential to all aspects of healthcare since they are essential to the decisionmaking process.⁹⁷⁷ Any healthcare-related activity will include acquiring, analyzing, or utilizing data. Information may be used to identify subgroups within a patient population or to categorize any issues a patient may be experiencing. They also help a doctor decide what more studies are required and what actions should be taken to understand the patient's difficulty better or, more significantly, to handle the issue that has been detected.⁹⁷⁸

Health information management has developed into a specialty within the health professions. It combines methods, techniques, ideas, and concepts from several other health professions, making it a distinct specialist field of study. Health information management is a subject of research and a practice that combines elements of the social sciences, the medical sciences, and computing and data sciences.⁹⁷⁹ Similar to HIM methodologies, information technologies (IT) are used by health informatics specialists to gather, store, analyze, and disseminate patient data, documentation, understanding, and wisdom. IT and related gear and software are seen as tools that users, clinicians, employers, healthcare providers, and administration may use to attain the objectives of health informatics. Two goals of health informatics are supporting the delivery of care and enhancing everyone's state of health.⁹⁸⁰ *Health Informatics History*

Generally speaking, the 1950s saw the beginning of the use of computers in healthcare, which is when health informatics began to take shape. The characteristic of this early period in the history of informatics, which continued throughout the 1960s, was using this new technology to innovate in medical and nursing education.⁹⁸¹ The late 1960s work of graduate nursing student Connie Settlemeyer at the University of Pittsburgh School of Nursing offers a local illustration in the discipline of nursing.⁹⁸² She created a computer-assisted training application running on a mainframe to teach pupils how to use "the common problem-oriented format known as SOAPE or SOAP."⁹⁸³ Nelson & Staggers note, "The SOAP note stands for Subjective, Objective, Assessment, and Plan and is used to record the patient's condition and status by both doctors and nurses."⁹⁸⁴ The University of Pittsburgh utilized this approach to instruct undergraduate nursing students during the remainder of the 1970s.

At that point, filling up the paper documentation required to follow up on doctors' orders that had been scrawled on patients' notes was the responsibility of nurses and unit clerks working under their guidance. In addition to recording the hospital costs related to doctors' orders, SOAP-templated printed sheets were utilized to correspond the instructions to connected units. One of the earliest components of hospital information systems to affect real patient care was order entry, followed by outcomes reporting.⁹⁸⁵ The first group of healthcare professionals immediately impacted by using SOAP sheets in the medical field was nurses and staff members in specialized divisions like laboratories and radiology. The use of computers in specialized sectors began throughout this same decade. One instance of computers used in specialized sectors was hemodynamic monitoring devices in the cardiac lab. Computers were employed in these settings to do computations, providing precise answers in seconds.⁹⁸⁶ By the late 1970s, developments in computer science, library resources, and medicine, as well as academic

developments in these fields, had created an environment conducive to expanding and developing the recently established field of medical informatics.⁹⁸⁷

Due to its widespread usage across several sectors, the definition of knowledge management varies depending on the application. The context in this section pertains to management and clinical decision-making. Thus, it is appropriate to use Lee's definition of knowledge management: "A discipline that promotes an integrated approach to identifying, managing, and sharing all of an enterprise's information needs. These information assets may include databases, documents, policies, and procedures as well as previously unarticulated expertise and experience resident in individual workers."⁹⁸⁸ The use of knowledge resources inside a corporation is a key function of health informatics, which necessitates the adoption of cutting-edge IT. Data generated by biostatistics, diagnostics, medicinal, and public health informatics are all included in health informatics, and they are all integrated with open systems, organization theory, engineering, and behavioral sciences concepts.⁹⁸⁹

5.B.i.b. Health Information Infrastructure

Health information infrastructure (HII) is "community-level informatics systems designed to make comprehensive electronic patient records available when and where needed for the entire population."⁹⁹⁰ Privacy, stakeholder participation, securing all electronic information, and economic sustainability are challenging issues that HII systems must solve. Consequently, even though HII has long been sought using a variety of ways in several nations, progress has been gradual, and no established recipe for success has yet been discovered.

Although the creation of the HII in the United States is the focus of this section, many other nations are engaged in comparable operations and have made additional advancements in this direction. Australia, Canada, and European countries have invested a lot of effort and money

in their own national HIIs.⁹⁹¹ For instance, the UK has significantly invested several billion pounds in upgrading its health information system over the previous several years.⁹⁹² It should be mentioned that the government centrally manages the healthcare systems in these countries. The US's complex, mostly private healthcare system differs organizationally, leading to a unique set of challenges and concerns. The knowledge gained from successful HII development programs worldwide may be successfully shared to decrease the problems everyone pursuing these important goals will confront.⁹⁹³

Strategic Dimensions of IT Architecture

Knowing the notions of "reach, range, and richness" is necessary to comprehend how IT architecture and company strategy are related.⁹⁹⁴ The many strategic capabilities made possible by IT are described by their reach, range, and richness. Reach is "the degree to which an organization can manage its value chain activities to connect its customers to an accessible product or service."⁹⁹⁵ Simply said, reach is the number of individuals and households in the neighborhood eligible for medical care from a hospital or clinic. One might categorize the community as local, regional, national, or global. In the past, healthcare organizations have always focused locally and developed so that all health treatments should be available to the local community. Traditional definitions of reach in the healthcare industry focused on having physical access to clinical services, which by default constrained healthcare organizations' vision to the local community.⁹⁹⁶ Early on, telemedicine was used to make specific local medical treatments accessible in the area. E-health has developed from telemedicine and now includes various services transmitted across broad geographic areas, and treatments and medications can be given virtually to the patients.⁹⁹⁷

Range is "the degree to which an organization can offer its customers of value proposition containing a breadth of products or services."⁹⁹⁸ It illustrates how effectively a healthcare professional offers the patient various relevant services. The range may be widely defined as a complete health service for a person or family or, more specifically, all the services linked to the care and treatment of a patient with a specific ailment.⁹⁹⁹ IT may broaden the scope of services to include preventive and maintenance instead of only immediate and urgent needs. IT can expand the scope of healthcare by offering a wide range of wellness or health maintenance care to become available to a community that may only have health services that focus on acute or chronic conditions – connecting communities and specialists that otherwise would have never met due to geographic distances. However, doing so can only be possible if financial incentives are matched to IT's enormous potential and regulations hampering any access to the Internet or related IT technologies are relaxed accordingly.¹⁰⁰⁰ The range of health services that IT can bring can only be extended as far as the infrastructure will allow, recalling the topic of the digital divide addressed in Chapter 4.

Richness is "the degree to which an organization can facilitate the exchange of information to deliver products or services that match customers' exact wants and needs."¹⁰⁰¹ This indicates the ability of a healthcare supplier or clinician to provide patients with individualized or customized treatments, including the integration of services to address specific requirements. By getting over the institutional flaws of big, centralized, and bureaucratic organizations, IT implementation enables the richness of healthcare professionals to give quality recommendations during clinical interaction.¹⁰⁰² This refers to the ability of a doctor or medical institution to provide patients with specialized or personalized treatment, including the integration of resources to satisfy the unique requirements of each patient. Healthcare providers
and organizations who use IT infrastructure to customize their services to specific patient preferences are at a competitive advantage over those that do not.¹⁰⁰³

Future Prospective in HIM

The boundaries of a healthcare institution are no longer a barrier to the transmission and linking of data, thanks to health information management and information technologies. The migration of patient records to EHRs and EHR systems is made possible by the collaboration of HIM and IT. Successful migration to the EHR system requires well-defined, valid, and accurate data that can be easily transmitted, altered, and aggregated.¹⁰⁰⁴ Numerous new applications and users will surface as more patient data become accessible in electronic form. In order to secure proper uses of the EHR data and restrict unauthorized users, technological safeguards must be balanced with knowledge and implementation of HIM governance. The early HIM specialists employed the technology available at the time to collect, store, and sort paper patient records and charts. While today's HIM professionals have essentially the same obligations to patient data, their job is much more challenging due to the enormous volume, uses, and users of health information.¹⁰⁰⁵ To make modern-day HIM professionals' roles even more difficult, their duties and obligations must change to accommodate the demands of the healthcare industry and healthcare reform. The HIM field must continue to set the standard for managing and supervising patient records, information, and EHR systems. The future of healthcare and the intricate, integrated information system required to provide effective, high-quality treatment in a growing and changing healthcare system will continue to depend heavily on HIM specialists.¹⁰⁰⁶ The next section will discuss some of the legal and ethical aspects these HIM specialists endure in the profession.

5.B.ii. Legal & Ethical Aspects of Health Information Management

Human values should be used as a guide for the health profession's education and practice. Healthcare informatics must address moral dilemmas such as right and wrong, honorable and dishonorable conduct, and suitable and improper behavior, much like other health professions.¹⁰⁰⁷ It is important for students and professionals in the medical sciences—including informatics-to reflect on their profession's moral underpinnings and ethical conundrums. Although there are ever more ethical problems in research involving human subjects, psychiatry, social service, and allied fields of medicine and nursing, the main concerns are universally understood. Numerous academic, professional, and educational settings have addressed fundamental bioethical problems. Even though some of them have garnered attention for decades, ethical issues and health informatics are typically less well-known.¹⁰⁰⁸ Among all the health professions, informatics is today the subject of some of the most crucial and fascinating ethical discussions. The following subsections will address health information management's legal and ethical aspects. The first subsection will focus on the internal and external challenges that healthcare information managers might incur and the importance of ethical coding systems and staff justice in the workplace. The second subsection will describe ways to rethink an ethical approach to HIM through Corine Mouton Dorey's matrix and ethical decision-making process.

5.B.ii.a. Privacy, Confidentiality, & Data Sharing

Internal & External Challenges

Health information management specialists have seen increasing difficulty in gaining access to sensitive patient information due to the increased demand for patient data.¹⁰⁰⁹ A health information management professional has a long-standing ethical obligation to protect patient confidentiality and private communications. Think about a famous patient being admitted to a

hospital for acute treatment. It is unethical for hospital personnel or the press to request specific information regarding the celebrity patient's health state since none of these parties has an unalienable right to access such data without the patient's consent.¹⁰¹⁰ Dana McWay states, "With the advent of computers, the temptation is to satisfy curiosity by seemingly anonymously gaining access to the celebrity's health information. Such temptations should be tempered by audit trails, which allow the hospital to determine how frequently the celebrity patient's health information is accessed and by whom."¹⁰¹¹ The hospital may implement the necessary disciplinary measures if improper access is discovered.

Another example of unethical behavior in health information management is pressure placed on a healthcare institution to get full accreditation from an accrediting body by "fudging" data showing the department's chart completion rate is higher than the necessary percentage.¹⁰¹² Another approach is for a physician to write progress notes after visiting a client and request that the health management system record that the notes were written promptly. Other instances include failing to declare conflicts of interest and giving a false account of one's qualifications.¹⁰¹³

Health information managers may encounter external ethical concerns in addition to those that arise inside the healthcare company. Some third parties attempt to get unauthorized access to data about genetics, adoption, or mental health.¹⁰¹⁴ Others, including business suppliers, can try to be reimbursed for work that was not done. McWay gives an example of how a false invoice with different line counts than what was billed may be submitted by an external vendor that provides transcribing services.¹⁰¹⁵ The management may consider revising the transcribing contract to include penalty provisions for further mistakes and demand more specific

information about the invoice if it deems that the disparity is not an error but rather the consequence of an inflated line count.¹⁰¹⁶

Coding Systems & Staff Justice

The complexity of encoding payment systems exacerbates health information technology ethical problems. When health information technology specialists must code and secure payment for delicate patient areas like mental health and genetic information, problems with patient autonomy and confidentiality commonly arise.¹⁰¹⁷ Access to celebrity information, as was previously indicated, is of particular significance. Even though there may be firewalls, passwords, and other safeguards in place, they often do not guarantee patient confidentiality.¹⁰¹⁸

The importance of third-party payer payment has increased among experts who handle health information and issue clinical and procedure codes to ensure a healthcare institution's continued financial viability. Eileen Morrison notes, "Most, if not all, reimbursement mechanisms tie the coded diagnosis and procedure to the amount reimbursed to the healthcare facility. Along with the growing complexity of coding and reimbursement guidelines, some coders face the dilemma of focusing on the accuracy of coding versus coding to obtain a better reimbursement for the healthcare facility."¹⁰¹⁹ To get the highest compensation for the healthcare facility, the individual coder may feel pressure from healthcare institutions to depend on unreliable, incomplete, misleading information or lack a solid clinical foundation or supporting evidence.¹⁰²⁰ In situations with moral ambiguities, like this one, the American Health Information Management Association (AHIMA) offers principles for ethical coding to direct coding specialists.¹⁰²¹

Professionals in health data technology also need to handle staff justice problems in addition to patient confidentiality and integrity concerns. Ethics-based management should be

provided to employees who create health information technology systems, keep track of patient data, or code medical data.¹⁰²² Leaders in health information management must thus demonstrate moral awareness. Being morally conscious involves being aware of possible ethical concerns connected to fraud and exploitation of health information technology.¹⁰²³ Similar to this, executives in health information technology need to be able to appropriately enforce organizational regulations by comprehending their nature and how they should be used. Fair treatment is also necessary for promotions, overtime distribution, and other staff justice concerns.¹⁰²⁴ To strike a balance between the requirements of the workforce and those of the healthcare organization, administrators in the healthcare field will need to be realistic.

5.B.ii.b. Rethinking the Ethical Approach to HIM

Objectification of Individuals vs. New Scientific Advances with Big Data

Ethics takes precedence over law to use big data to manage health information. Big data eludes the standard structure of clinical research, the practice of medicine, and relationships among patients, donors, doctors, and researchers. Corine Mouton Dorey states, "The three rules of medical secrecy, the patient's right to the truth, informed consent, and the concept of individual indivisibility are challenged. Moreover, the high speed of big data development requires continuous normative adaptation to legitimize its use."¹⁰²⁵

Based on the three components of applied posterior ethics, moral standards, and antecedent ethics, Dorey's matrix suggested ethical assessment method for big data is structured around the idea of fairness.¹⁰²⁶ A sense of fairness is important because it promotes collaboration, such as making big data's sources, techniques, and results available to the public, along with making reparations when findings have hurt specific people.¹⁰²⁷ A bottom-up "democratic" engagement in big data governance will be favored by the appropriate social justice and equitable information dissemination principles. Citizens' engagement and education would shield patients and medical professionals from uncontrolled anxieties that may result in a cautious attitude and from potential covert population health coercion or the lack of caution in its use or exploitation of big data.¹⁰²⁸ This democratic approach to managing huge health data might improve ethical reflection among healthcare stakeholders, boost their feeling of agency and self-worth, and lessen the possibility of medical or public arbitrariness.¹⁰²⁹

Big data moral questions may reveal modern normative tendencies in ethical reasoning for handling health information. Dorey elaborates, "Finally, advances in data size and analytic methods in HIM could be seen as a resurgence of scientific positivism, adding new narrative paths in the research for truth, and challenging the moral references for medical and bioethical judgment. Depending on the queries, numerous different plots could be revealed, which not only enhance scientific research, but also carry new patterns in moral thinking for HIM itself."¹⁰³⁰ The present normative rules of justice may be in jeopardy as a result of the morally challenging tendencies of HIM.¹⁰³¹ Dorey continues, "The deliberation process starts at the level of posterior ethics with openness and prudence in founding the common good. Then, normative development proceeds by adjustment using anterior ethics, gradually revising public health legitimacy and distributive justice."¹⁰³² With the anterior ethics section specifying a specific ethical purpose, the matrix also creates limits in light of big data's potentially misleading moral guidance.¹⁰³³ The matrix, for example, would restrict excessive demands on one's health and distinguish between one's self-esteem and self-love.¹⁰³⁴

The matrix deduced from Ricoeur's "little ethics" is a viable approach to ethics for application in the management of health data due to its focus on patient agency, the trust relationship between healthcare institutions, and justice as inclusive and fair engagement in the

collective welfare.¹⁰³⁵ Although Kantian and Rawlsian theories have advantages, Ricoeur's ethics asserts that previous ethics supersede moral and legal norms.¹⁰³⁶ There may be ethical problems, but these may be resolved by turning to anterior teleological ethics. In managing health information for clinical practice, research, and public health, the remedy to anterior teleological ethics permits the potential moral conflicts to be overcome and facilitates informed and collaborative decision-making.¹⁰³⁷ This ongoing ethical reflection method may apply to various aspects of technology-assisted healthcare.

Ethical Decision-Making Process

How these ethical dilemmas might be resolved is as varied as how they can be presented. Decisions may be made individually, in a group, via discussion, or by reaching a consensus.¹⁰³⁸ These choices cannot be taken in a vacuum since they may impact other aspects of the same ethical issue. Regulations, ideals, technologies, liberties, society, norms, legislation, religion, and morality, are examples of ethical problems' constituent parts.¹⁰³⁹

Despite the fact that the literature on ethics mentions several different ways of problemsolving, these approaches may essentially be handled as a part of a multi-step process. Establishing the facts and properly describing the ethical problem are the first stages. The defining stage entails defining the issue, clarifying it, and looking at incongruent values. Data collection may be used to establish the facts. Speaking with people who may know anything about the problem or looking at pertinent documents are two ways to collect data.¹⁰⁴⁰ Finding out who has a stake in this choice must come after the issue has been precisely identified and facts have been acquired. Sometimes, the party or individual who must make the moral choice is not the same as the one who is the problem's victim. A wider approach to identifying stakeholders is necessary since there may be more parties with interests that need to be considered than are first evident. The connections between the stakeholders should also be investigated, as well as any possible biases or other incentives.¹⁰⁴¹

The knowledge gained at each stage helps the decision-maker identify the possibilities open to them. The pros and cons of each alternative and how they will affect the many individuals and groups involved should be carefully considered. To help in decision-making, each alternative may be weighed against an ethical framework. For instance, which choice supports completing one's duty (deontology), which option supports the rights and dignity of all people (social equality and justice), and which option gives the greatest benefit for the largest number of people (utilitarianism)? The decision-maker explains the rationale behind each choice by relating it to an ethical philosophy. In order to effectively support a HIT ethical decision-making process, a kind of HIT governance must be built to help with the enforcement and compliance with ethics throughout the organization—ensuring that data is kept safe and private and maintains integrity.¹⁰⁴²

5.B.iii. Health Information Technology Governance

A set of processes known as health information technology governance is in place to ensure that information assets are formally controlled across the entire business and that people liable for adverse consequences brought on by poor data quality may be held accountable. It involves managing the organization's data assets' availability, usefulness, integrity, and security. In the proceeding subsections, HIT governance will be more fully defined, including its need and core components and the essential characteristics of data quality.

5.B.iii.a. Defining HIT Governance

The following subsections will describe the need for HIT governance and the core components of that governance – what it entails and requires to be implemented within a healthcare system.

Need & Core Components

Healthcare organizations must ensure that their health IT initiatives align with their major objectives due to health IT's broad participation in all aspects of a company's purpose and activities.¹⁰⁴³ Additionally, health IT resources are often needed for a healthcare company's strategic and operational efforts to be as successful as possible. Health IT is often required for important healthcare initiatives, such as population health, service pathway adoption, and pharmaceutical safety.¹⁰⁴⁴ As a result, there will often be a considerable gap between the demand for and supply of health IT personnel (such as analysts and software developers). Nelson & Staggers state, "The critical nature of effective health IT governance then arises from the following fundamental interrelated needs: 1) the need to ensure alignment of health IT resources with institutional priorities and 2) the need to effectively prioritize the use of health IT resources in the face of numerous competing demands for these limited resources."¹⁰⁴⁵

In conclusion, there is a growing need for health IT resources and skills, and institutional interests, as well as demands from a wider range of constituents, are becoming less clear.¹⁰⁴⁶ Without efficient governance, important parties, such as healthcare providers, would experience delayed or insufficient IT assistance and a resource allocation rationale that they do not comprehend, such as how to distribute staff among units in the hospital or how to ration a medication shortage. Furthermore, because of resource allocation, similar organizations (like

hospitals) may be sponsoring projects that compete with one another and have opposing goals instead of working together to provide a wider range of health services to their communities.¹⁰⁴⁷

Determining goals, allocating resources, and, if required, approving funding to support the growth of immediately accessible health IT assets are all part of the function of health IT governance.¹⁰⁴⁸ The governance of health IT should be responsible for keeping track of and evaluating the results of these investments. The following elements must be included in effective health IT governance to achieve the funding investments needed to support its growth:

1) Organizational structures responsible for clearly defining institutional priorities. Typically, this function is primarily the responsibility of a healthcare organization's board of directors and senior leadership. 2) Organizational structures responsible for ensuring that health IT efforts are aligned with institutional priorities and used optimally. 3) Accompanying processes to operationalize the governance. Establishing such operational processes is much more straightforward than showing a governance structure that appropriately owns and uses these processes.¹⁰⁴⁹

As previously mentioned, defining HIT governance includes a description of data quality characteristics since so many of the governance concepts revolve around the quality of the organization's data assets. The next subsection details some of those qualities and why they are important.

Characteristics of Data Quality

The following definitions of "quality" are particularly relevant to the HIM professional, however, there are other connotations as well: "1) a degree of excellence of a thing and 2) a required character or property that belongs to a thing's essential nature."¹⁰⁵⁰ The accuracy of the data is referred to as data quality. When "qualities of data" is used, it often refers to the qualities or components that go into making up data or factual information. Developing health information

systems and tracking data are essential steps in ensuring the reliability and accuracy of the information generated.¹⁰⁵¹

Data quality management (DQM) leads to good data. HIM experts are familiar with the duties and talents of DQM. "Practice Brief: Data Quality Management Model," which the American Health Information Management Association (AHIMA) initially released in 1998 and then revised in 2012, serves as an example of data quality management.¹⁰⁵² There is also a checklist to evaluate DQM initiatives. Continuous data quality assurance is a part of all DQM operations, including data application, collecting, analysis, and warehousing. The following ten components characterize data quality:

Accessibility: Data items should be easily obtainable and legal to collect

Accuracy: Data are the correct values and are valid

Comprehensiveness: All required data items are included. Ensure that the entire scope of the data is collected, and document intentional limitations

Consistency: The value of the data should be reliable and the same across applications **Currency:** The data should be up to date. A datum value is up to date if it is current for a specific point in time. It is outdated if it was current at some preceding time yet incorrect at a later time **Definition:** Clear definitions should be provided so that current and future data users will know what the data mean. Each data element should have clear meaning and acceptable values **Granularity:** The attributes and values of data should be defined at the correct level of detail **Precision:** Data values should be just large enough to support the application or process **Relevancy:** The data are meaningful to the performance of the process or application for which they are collected

Timeliness: Timeliness is determined by how the data are being used and their context.¹⁰⁵³

The "Practice Brief: Data Quality Management Model" offers examples of each domain's characteristics, along with a graphic depiction of the DQM domains and how they pertain to the many components of data integrity. The model is universal and may be used in any care

environment. It serves as a tool or paradigm for HIM professionals transitioning into entity- or organization-wide DQM responsibilities.¹⁰⁵⁴ Now that HIT governance has been defined, it is appropriate to see how normative ethics play a role in its dissemination in healthcare. The following subsection discusses the ethics of quality and the concept of information stewardship to guide this description of HIT governance.

5.B.iii.b. Normative HIT Governance

The Ethics of Quality

Quality is crucial because it is a Kantian commitment, in addition to ensuring that a good or service satisfies the standards of regulatory agencies and financial institutions and that a business is profitable. The categorical imperative would be satisfied if the patient received safe, efficient, high-quality healthcare that is beneficial.¹⁰⁵⁵ Additionally, Kant's idea that all people have worth is consistent with delivering top-notch treatment to all patients regardless of their financial situation. Healthcare institutions often use the word "quality" in their mission statements. It makes no sense to have a goal statement encouraging subpar quality when people's lives and health are at stake. The problematic issue, however, does not lie in declaring that one's mission is to provide high-quality care in this period of significant change in the healthcare system and the financial constraints placed on caregivers; rather, it lies in having the moral courage to take actions to ensure that high-quality is present.¹⁰⁵⁶

Although for a different reason, utilitarians would concur with the Kantians in this case. The most damage can be prevented, and the most good can be done for the most impacted people by providing effective healthcare services. Utilitarians compare a decision's benefits or drawbacks to its effects. The repercussions of poor treatment might harm patients as well as the sustainability of the healthcare institution. For instance, poor care may worsen patients' suffering

and raise the price of their treatment.¹⁰⁵⁷ Bad patient satisfaction numbers or legal action might affect the company due to poor treatment. In the age of the Affordable Care Act, noncompliance with quality requirements may lead to lower payments, which might hurt the organization's financial line. Reduced income may result in staff cutbacks and a decreased ability to meet patient demands. The conclusion that quality assurance is just good business comes from considering the utilitarian implications of the practice.¹⁰⁵⁸

The ethics of quality assurance are subject to the same ethical rules. Think about the idea of nonmaleficence. Patients might suffer if they get subpar treatment, and caregivers' jobs can suffer if they make mistakes with medicine and patient identification.¹⁰⁵⁹ Although vital, the expenses associated with developing rules, and procedures, holding training sessions and monitoring to ensure excellent practices raise the cost of providing care.¹⁰⁶⁰ As a result, investing in quality assurance results also improves ethical behavior across the firm. When healthcare practitioners behave with patient-centered care and compassion, the sister principle of beneficence is put into practice. Healthcare managers also show their goodwill toward their staff by enforcing laws that protect their rights, respect their dignity, and promote their growth as individuals and professionals.¹⁰⁶¹ A business may show generosity to all workers and convey that they are appreciated by, for instance, investing in a competent employee support program.¹⁰⁶² A quality assurance program that is properly financed and maintained may encourage the use and implementation of the moral values of beneficence and nonmaleficence, which will improve normative governance.

Information Stewardship

In order to handle information ethically, it is essential to practice efficient information stewardship within an organized and well-defined information governance framework. To

summarize, information stewardship is often described as "An approach to information/data governance that formalizes accountability for managing information resources on behalf of others and for the best interests of the organization."¹⁰⁶³ One should think about broadening this notion from an information ethics standpoint. Since it places a heavy emphasis on the "best interests of the organization" and cannot specifically take stakeholders in larger society into account, as with other definitions, information stewardship is closely tied to organizational ethics.¹⁰⁶⁴

The risk is that the current ideas of information stewardship will be applied to the normative stakeholder theory paradigm of ethics. But this does not align with the underlying idea of stewardship, which is widely seen as "the acceptance or assignment of responsibility to shepherd and safeguard the valuables of others."¹⁰⁶⁵ Consideration of the notion of stewardship characterizes the idea as the "responsibility for sustainable development shared by all those whose actions affect environmental performance, economic activity and social progress, reflected as both a value and a practice by individuals, organizations, communities and competent authorities."¹⁰⁶⁶

This interpretation of stewardship expands the ethically acceptable range of norms and values. It also paints a complete picture of the participants in the data and operational results the organization aims to provide via its administration and management of data.¹⁰⁶⁷ Additionally, it underlines how crucial the stewardship method is. In this context, ethical information stewardship may be described as "an approach to information/data governance that formalizes responsibility and accountability for sustainable information management shared by all those whose actions affect the information and process outcomes experienced by stakeholders, both

inside and outside the organization, and that is expressed by individuals, organization communities, and competent authorities as both a value and a practice."¹⁰⁶⁸

5.B. Conclusion

Information management has dramatically impacted how we view, use, and store data – from the smallest to the most extensive files. Information management affects everyone –those who do and do not work directly with data and with multiple levels of influence. As a result, it is crucial to include a governance framework to ensure that this data is appropriately, ethically, and legally managed since, if not, it might lead to a major tragedy. This section has explained the foundations of health information management and how that can lead to an understanding of the infrastructure of the technology and the system behind it. Then, while striving to reconsider the ethical approach to appropriate health information management, some of the legal and ethical difficulties that may occur in this context were looked at. Next, applied analysis of health information management was described using systematic reviews and risk management principles to highlight the importance of having ethical information management. Finally, a health information technology governance was proposed in which its need and core components were emphasized within a normative health information technology governance and information stewardship.

5. Conclusion

As a reminder, AI and AML require massive amounts of data to learn and improve their computation and processing. Data analytics can assist with this information storage, acquisition, and use – which can be summarized in information management. The way this information is managed and the kinds of ethical and philosophical data input into the system are of the utmost importance. A proper governance system must be integrated at the base level of information

management so that the AI will have regulated, ethically robust data to learn from and can be designed with humans in mind. Quality control and information stewardship were just two ways of helping to guarantee that proper information regulation and governance can occur. The next chapter, Technology's Impact in End-of-Life Practices, examines how information management, AI, and data analytics through various technologies can be applied in a specific context in the care of the elderly and aging populations and help to revive the art of dying well in the modern world.

CHAPTER 6: TECHNOLOGY'S IMPACT IN END-OF-LIFE PRACTICES <u>6. Introduction</u>

The end of Chapter 4 looked at an example of how data analytics could be applied in a negative or dystopian way if humanity is not careful of the slippery slope that human enhancement can lead down. In contrast, the sixth chapter focuses on an overall optimistic or utopian application of various data analytics-driven technologies within the specific context of the elderly population and end-of-life practices such as palliative care and hospice. Why end-of-life practices? Due to the exponentially increasing scope of the aging population, an unmet need has arisen, now more than ever, to care for the elderly worldwide. Thanks to data-analytics-driven technology, machines have changed how the world deals with death and allow people to live much longer than they could a century ago—the historical and cultural background of death and dying outlines systematic changes toward death over time.

Opportunities for technologies in end-of-life practices are explored, such as telemedicine and electronic learning. Challenges such as the digital-divide, professional-patient relationship via technology, and fundamental issues such as the technological imperative vs. humanizing are also described. The chapter continues with ethical considerations of using care robots in aging populations using care ethics. In this narrative about the automation of care labor, the size of the aging population is calculated, the unmet care requirements of the older population are described and how carebots emerge to fill this expanding gap is noted. Care ethics and elements of care are detailed in how they can apply to technology, specifically care robots. This applied healthcare data analytics chapter describes concepts like artificial morality, relational autonomy, and dignity. It ends by noting the ethical challenges of a care-centered framework using a context-sensitive process and human capabilities in an artificial form.

6.A. Technological Opportunity & Ethical Challenge in Hospice & Palliative Care: Help or Hinderance in the Art of Dying Well

6.A. Introduction

Dying is a phenomenon that affects us all one day and therefore causes a universal interest in the experience. However, the general way of thinking in the modern secularized West has focused on how this phenomenon can be controlled, presenting options for assisted dying and mechanical intervention to delay dying as long as possible.¹⁰⁶⁹ The main idea of this section is to investigate the history and culture surrounding death and to revive the process of *Ars moriendi*, or the lost art of dying well, that can be applied to the modern world by showing how various opportunities and challenges of technology in the fields of hospice and palliative care attempt to achieve *Ars moriendi*.¹⁰⁷⁰ This will be defined by explaining the systematic changes to death over time through the perceived notion of a 'good death' and the advent of hospice and palliative medicine to revive the art of dying well. Subsequently, a new ethical framework is proposed for dying well via critical observations about the field and some contemporary issues that must be addressed for the ethical framework to be valid.

This section then continues by describing the opportunities modern technology affords hospice and palliative care, such as improvements to medical interventions and telemedicine, electronic learning, and the flexibility it provides patients and providers in healthcare. Finally, this section concludes with a description of some of the ethical challenges posed by this technology, including fundamental issues such as technological futility and dehumanizing the caring approach, in addition to more specific topics such as how technology creates a digital divide and stresses the professional-patient relationship within hospice and palliative care. The purpose of this section is to show that while modern technology provides many new and valuable opportunities in hospice and palliative care, it also produces ethical challenges that must be addressed if the future of healthcare and the revival of the art of dying is dependent on it.

6.A.i. Historical & Cultural Background of Death & Dying

6.A.i.a. Systematic Changes Towards Death Over Time

Systems alter as well, as was mentioned in the history of Western views on death. This section will try to summarize and conceptualize those systems changes that went with these significant historical moments that shifted the practices surrounding death. This will be carried out by describing how the localization of the death system changed from the home or community to the hospital, nursing home, or hospice unit. This subsection will explain how and why the hospice and palliative care fields attempt to bring back the art of dying in this contemporary world.

The Modern 'Lie' & The Good Death

Correlated with the localization of death to the hospital or institution, Philippe Ariès describes "the rise of the modern lie or shame of death: the sense that death signifies not spiritual completion but medical failure" in his book *Western Attitudes Towards Death: From the Middle*

Ages to the Present.¹⁰⁷¹ He describes that "one dies in the hospital because the hospital has become the place to receive care which can no longer be given at home."¹⁰⁷² Further, in the past, the hospital acted like more of a shelter for the homeless or refugees; later, it became more like the medical facility it is today where healing occurred, and people battled death.¹⁰⁷³ Even while it still has that healing capability today, many have started to conceive of hospitals as places where people go to die. Ariès notes, "One dies in the hospital because the doctor failed to heal. One no longer goes to or will go to the hospital to be healed, but for the specific purpose of dying."¹⁰⁷⁴

However, many Americans stated their preference to die away at home, close to their loved ones.¹⁰⁷⁵ This suggests that death will occur after some notice, giving people time to make arrangements. "Unfortunately, though the numbers have been slightly better in recent years, it is still the case that more than 70% of Americans dying hospitals or nursing homes, in the large percentage die after a disconcerting series of institutional admissions and readmissions."¹⁰⁷⁶ This type of 'medicalization' of death is stripping away people's chance to die in a way that allows them to experience some of the best and highest values in life, such as family, closure, religious enlightenment, etc.¹⁰⁷⁷ Even though it would be challenging for everyone involved, it would be beneficial for many patients and their families if medical professionals had the foresight and the guts to inform them that, while this might not be the end, it might be the beginning of the end.¹⁰⁷⁸

The notion of the good death or of dying well isn't a phenomenon that just passively happens to you. It is something you must actively do.¹⁰⁷⁹ The concept of "tame death" has been around since at least the eighth century, according to several social historians, including Ariès.¹⁰⁸⁰ Speaking mainly of Europe, he explains that a 'good death' meant the dying had the "liberty and presence of mind to set their physical, material, and spiritual affairs in order – unlike

in earlier times, particularly in our hunter-gatherer past, where death happened suddenly and often violently."¹⁰⁸¹ The pastoral age began when hunter-gatherers built farms and began living more stable lifestyles near to one another. The seasons served as a helpful metaphor for comprehending life and death as a process before the event of death happened, and they gave them the chance and time to make plans.¹⁰⁸² A repetitive, immobile, and largely unadventurous life gave time to anticipate death and, finally, to prepare for it. Brandy Schillace concludes, "Death as the process became something the dying could participate in *with* the living who remained – not just something that happened after they were gone."¹⁰⁸³

Hospice & Palliative Medicine

Hospice and palliative medicine, however, provides an alternative to this unartful way of dying, and it contributes to the project of recovering the practices of the *Ars moriendi* by helping to create the prerequisites for patients and their communities to engage in the activities of dying well (and caring well for those who are dying). In theologian/ethicist Paul Ramsey's terms, dying well is both "a gift and a task."¹⁰⁸⁴ For one to be conscious of the gift and partake in the task simultaneously requires patients to be aware of both their symptoms and their mortality.¹⁰⁸⁵

In 1996, the American Medical Association's (AMA) Council on Scientific Affairs published a report titled *Good Care of the Dying Patient*.¹⁰⁸⁶ The report states that "In the current system of care, many dying persons suffer needlessly, burden their families, and die isolated from family and community."¹⁰⁸⁷ The default pathway creates enormous obstacles to dying properly and regaining any art of dying by entangling patients in a technologically driven social infrastructure that causes unnecessary pain and isolates them from their communities.¹⁰⁸⁸ Expanding and improving palliative care and hospice have been the traditional approach to solving this issue and the only solution the Council on Scientific Affairs of the American

Medical Association could confidently identify.¹⁰⁸⁹ Numerous waves of outreach have urged and helped individuals create advance directives enabling them to refuse unwelcome life-sustaining treatments when they are terminally sick and deliver them to teams of palliative healthcare specialists who will support their death at home.¹⁰⁹⁰

Keeping one's "matter in motion" for as long as possible, away from one's community, and under the direction of machines and a system of processes that continue with unrelenting and uncaring logic is not what dying properly means, as both patients and doctors have realized.¹⁰⁹¹ As Atul Gawande notes in his book *Being Mortal*, "striving officiously to keep alive within the medicalized privileges of North America comes with severe costs."¹⁰⁹² Jeffrey Bishop's *The Anticipatory Corpse* builds on Gawande's pathology. Bishop argues that "a more profound medical objectification of the body in the modern West renders it lifeless long before it ceases to function as a living organism."¹⁰⁹³ In summary, the more we as a society attempt to control death, the more we become removed from moral realism, and the less our theological teachings may be used to address the contemporary death.¹⁰⁹⁴ Bishop concludes that "the religious traditions of today have a special job in challenging the fundamental assumptions of the medical establishment."¹⁰⁹⁵

6.A.i.b. Toward a New Ethical Framework for Dying Well

A solid and unambiguous understanding of human frailty and a realization that existence has limits are what Ars moriendi strives to achieve. If we do not initially acknowledge our limitations and impending mortality, how can we begin to establish a structure for the practice of dying well? We cannot discuss the art of dying without first acknowledging our own mortality. Careful consideration of the potential for beauty and growth is encouraged by art. We must handle death carefully; we must weigh its excesses against its lack and its beauty against its

decay. The art of dying can only be experienced by those who recognize their own death. Can the emergence of a contemporary Ars moriendi be facilitated by a bioethical framework that is sufficiently robust? The following subsection will describe how the overlap of virtue ethics and bioethics can come together to create a framework that includes five significant observations for a modern *Ars moriendi*. While the following five observations are a good start, they are by no means a complete and comprehensive list of all the ways that the field of bioethics can help to usher in a new framework for dying well in contemporary times.

Five Observations for a Modern Ars Moriendi

The first step in developing a framework for dying well is for bioethics to honestly assess the limitations of its own formal, procedural approach.¹⁰⁹⁶ Everyone who is dedicated to the mission of dying well agrees that it is a virtue-based undertaking. To support this goal, bioethics must fully reimagine its ethical framework to include a virtue methodology. Second, in order to achieve this, bioethics will need to develop the skills necessary to constructively and honestly support meaningful clinical and cultural discussions about goals and means.¹⁰⁹⁷ Serious considerations regarding the societal aims that such a protocol aspires to achieve would need to be raised if bioethics were to aid medicine in designing a procedure for dying well in the therapeutic environment.¹⁰⁹⁸

Third, bioethics will need to seriously consider economic issues rather than just adding up the high expenses of end-of-life medical treatment in a cost-benefit analysis or confounding the problems of who should foot the bill for what.¹⁰⁹⁹ A profound ethical framework will need to consider the many significant ways that modern economic systems and philosophical viewpoints influence those who enter the clinical setting, determine the framework within which patients

and healthcare are provided, and impact the biotechnology and healthcare industries.¹¹⁰⁰ These influences are often subtle and unnoticed.

Fourth, we will need to go beyond the clinical context and recreate the geography of dying in addition to going beyond the bioethics' constrained methodology and ethical framework for dying properly.¹¹⁰¹ Restoring death to its rightful, non-medical setting—the home or local community—is the most natural step if we are sincere about lessening the medicalization of dying.¹¹⁰² Beyond the intensive care unit, how can healthcare professionals and organizations assist communities in providing for the terminally ill? Lydia Dugdale asks the right questions, "might various religious congregations, for one example, reimagine how they care for the sick? Instead of sending communion ministers to healthcare institutions, might congregations propose vacant rectories as places to care for dying parishioners for whom home care is not possible?"¹¹⁰³

Fifth, focusing on the deathbed is missing the purpose, as ancient and modern Ars moriendi literature argues.¹¹⁰⁴ The art of dying is more than just a ritual performed once medical intervention has been exhausted and all associated tools and aids have been abandoned.¹¹⁰⁵ Those attempting to change the dying experience, from Ariès to Callahan and beyond, make it plain that actions at the deathbed must be connected to a more extensive set of practices acquired throughout life (and after death). "The art of dying – which really is the art of living – is a lifelong process cultivated in the home, in the congregation, and in the community."¹¹⁰⁶ The diverse facets of numerous traditions may thrive there, in which the advancement of the qualities required to die and live well could be accomplished by particular practices and traditions brought about by spiritual development.¹¹⁰⁷

Contemporary Challenges

Another significant obstacle in the art of dying well comes not from shifts in religious and cultural norms but changes brought about by human evolution.¹¹⁰⁸ Human evolution, whether biologically or technologically, has caused us to live longer despite the increase in adverse events such as unhealthy lifestyles, poverty, and suicide amongst the general population.¹¹⁰⁹ Sarah Coakley points out that "Death comes slowly to a generation serially saved by antibiotics, their lives often further elongated either by economic privilege or 'good genes.'"¹¹¹⁰ However, this increase in longevity isn't without a few adverse outcomes, like for those elderly adults who are plagued by dementia – confusing and isolating the patient from family members who belabor how to deal with the progressive disease.¹¹¹¹ In previous generations, the elderly patient would have been cared for in a wider social sphere, such as their religious congregation. In the Catholic tradition, elderly individuals are often given *viaticum* numerous times as their fragile lives transition from one life-threatening event to the next.

Coakley then asks where the Christian Ars moriendi comes into play in these instances.¹¹¹² Does even the holiness of the sacrament appear rather dull by its continued repetition (such as within the sacraments of Last Rites, Anointing of the Sick, etc.)? The unfortunate reality for all involved in this caring process is coming to accept the predictable path to death that these end-of-life sacraments often foreshadow.¹¹¹³ The challenge for religious leaders and autonomous individuals is to change death preparation from a suppressed subject forgotten about until some last-minute medical emergency intervenes to a lifelong duty that one continually prepares for. In short, "there is no possibility of a straightforward retrieval of a set of traditions and practices that have served the Christian tradition so richly for nearly two millennia, not at least without deep and realistic reflection on how the medical and cultural situation has now changed."¹¹¹⁴

Throughout time, major historical events have slowly stripped away the art and sense of dying well. From the Bubonic Plague to the AIDS crisis, there has been a shift away from dying with family members in the community to an unattended death in a hospital room with often tricky decisions about whether or not to turn off machinery that is keeping someone alive – machinery that didn't exist centuries ago and has now stretched the definition of death to something brand new. With this new definition of death, we need a new *Ars moriendi*, the art of dying, that is suitable for the current needs of medicine. Palliative and hospice care have been an excellent start to bringing back this tradition, but more must be done to achieve a good death. Hence, a new ethical framework, though still limited in scope, can help to work towards a contemporary art of dying that satisfies the needs of the misguided secularized West.

6.A.ii. Opportunities of Technology in Hospice & Palliative Care

It should be no surprise that the opportunities that technological advancements will provide to the growing needs of the rising elderly population will be profound. This next section will highlight some of the technological advancements that are being utilized in the fields of hospice and palliative care – two areas of healthcare that deal with the increasing ageing population the most outside of geriatrics. The hope is that in this contemporary and digital world, the art of dying well will be revisited and revived through the opportunities that technology now enables and was previously unavailable to past generations.

6.A.ii.a. Improvements

The fields of palliative care and hospice have given much room for technological opportunity to work in attempting to seek out new and innovative ways to revive the art of dying well. Some technological improvements used in palliative and hospice interventions are the digital pens and the videophone. These advances have improved patient access and

communication in the fields and helped patients have a higher quality of life when utilizing these interventions. These kinds of technologies can be broadly categorized under the umbrella term of telemedicine, which has brought improvements in the physician-patient relationship and caregivers and will be further detailed in the following subsections.

Medical Interventions

A journal article by Lia Willis et al. reviewed the interaction of palliative and hospice care patients and providers with the Internet and some of its technological interventions involved in care.¹¹¹⁵ The review highlighted several articles. However, two were prominent in using and explaining telemedicine in hospice and palliative care through digital pens and videophones.¹¹¹⁶ Willis et al. noted that "both articles described this equipment as being used to aid communication between patient/caregiver and professional."¹¹¹⁷ Summarizing the use of these telemedicine interventions, the digital pen's main function was to assist in pain assessment by the patient. In contrast, the videophone's main functionality surrounded enhanced communication and better physical evaluation.¹¹¹⁸ Regarding the digital pen, the patient (sometimes with the help of a caregiver) would rate their pain level on a scale on digital paper, which would send the information to a secure server via the Internet and then be seen by healthcare professionals.¹¹¹⁹

The results in the literature review of the digital pen were mostly positive and supported by the conclusions that patients' care improved along with the communication between them and their healthcare professionals.¹¹²⁰ The literature review results of the videophone were very similar in that videophones "offer the capability of physical assessment and communication that is more personal in nature due to adding the video component."¹¹²¹ Also noteworthy is the conferencing feature that videophones permit enhanced communication and conversation

between many different healthcare professionals, further augmenting the hospice and palliative experience for all involved.¹¹²² The reasons for hospice and palliative providers to use the Internet vary but top reasons include clinical information gathering, live consultation, email, and online journals with "a need for more online training opportunities to obtain continuing education credits."¹¹²³

On the other side of care, patients and families generally use the Internet for gathering information and searching for support groups for people who are going through similar illnesses or situations.¹¹²⁴ The review also indicated that patients enjoy the anonymity factor that is so easily obtained through the Internet support groups.¹¹²⁵ The literature review is clear that "patients and professionals are using the Internet to find answers to healthcare questions, communicate, and deliver healthcare interventions."¹¹²⁶ While the review aimed to focus on palliative and hospice technological interventions, many of the support groups were intended for people with terminal illnesses – mostly cancer related.¹¹²⁷ In addition, a shift in the mindset of patients and professionals to view these technologies as giving the opportunity to create a social network of sorts that can further connect patients and professionals and enhance the standard of care provided at the end of life.¹¹²⁸ Willis et al. explain that "the Internet can create peer-to-peer communities, allow more access to information, and address isolation rather than as a substitute for patient care."¹¹²⁹ While many people who are currently in hospice may have little to no access or experience with the Internet or telemedicine technologies, in the not so distant future it will be easier to assume that more people who will reach elderly ages will have access and comfortability using the Internet and its telemedicine technologies.¹¹³⁰

Telemedicine

Patients with terminal diseases predominantly experience varying levels of both physical and psychosocial issues throughout the course of the illness.¹¹³¹ Palliative and hospice care provide a way to manage these issues that prioritizes the quality of life of the patient and family members.¹¹³² Patients who are home-bound typically receive community-oriented palliative care services from their primary care physician(PCP). Since palliative care is a specialist service, the care it provides at times is not very accessible to patients for several potential reasons including PCPs disregarding the care networks, or the more likely scenario being a physical decline of the patient that would diminish their ability to travel to outpatient palliative campuses.¹¹³³ Yet, this kind of specialist care is much more needed and utilized when difficult problems present during the patient's care.¹¹³⁴ Hence, many healthcare professionals agree that a multi-team collaborative effort between PCPs, specialist services, and patients should be maintained to provide the best care continuum possible.¹¹³⁵ It remains unclear, however, how home-bound patients fit into this care continuum since their vulnerability prohibits them from travelling to these various care settings. For these patients, telemedicine may provide a promising opportunity to keep this care connection strong as it permits patient involvement in their care remotely.¹¹³⁶

Technologies for telemedicine have been developed to help with patient monitoring and communication who are confined to their homes. ¹¹³⁷ For example, "online monitoring enhances patients' incorporation of illness(es) into their daily routines and yields improvements in cancer-related pain and depression."¹¹³⁸ In addition, a "tele-hospice" report stated that both professionals and patients were excited about the ability to instantly communicate in a visual manner via videophones.¹¹³⁹ Another study reported that patient-physician webcam interactions lead to "honest conversations, revealed nonverbal cues, and opened personal spaces."¹¹⁴⁰

On the other hand, it is noteworthy that with telemedicine's potential to help in healthcare, also carries worries and fears that these technologies will dehumanize the relationships through establishing barriers that prevent compassionate care and disrupt the patient's home.¹¹⁴¹ These concerns have prevented telemedicine from being widely used in settings focused on palliative care. In order to better understand how telemedicine affects the patient-professional interaction, which is essential for providing good palliative care, various studies have been conducted.¹¹⁴²

6.A.ii.b. Electronic Learning

Reasons for Going Online

There seem to be multitude of reasons patients with life-threating illnesses or diseases as well as family members go online – to engage in telemedicine (via videophone) or the Internet for information (for example, to learn about the processes of palliative/hospice care) are just a few of the reasons.¹¹⁴³ Patients often use the internet to research their illnesses and discover if there are any available treatments.¹¹⁴⁴ This pattern supports the idea that, to varied degrees, patients do in fact want to be included in the therapeutic decision-making process.¹¹⁴⁵ With this in mind, the Internet shows its value as an accessible source of information to the general population. Other reasons for this trend include seeking a second opinion and for verification of information regarding their illness told to them by professionals already.¹¹⁴⁶

Rising levels of Internet learning very likely will parallel the rising interest rates in in alternative medicine.¹¹⁴⁷ Websites that promote alternative medicine and similar practices appear to be abundant on the Internet.¹¹⁴⁸ While there are numerous potential reasons as why a patient may seek alternative treatments, some of the most prevalent are dissatisfaction with the amount of information presented, lack of tangible results from current treatments, and general

dissatisfaction with the doctor.¹¹⁴⁹ These are just a few of the reasons that explain alternative treatment seeking behavior in addition to a patient's natural desire to control an event that is mostly out of their control. This can lead to experimental treatments, drugs, and therapies that may or may not in the human trials phase of research and development.¹¹⁵⁰

It also happens to be the case that many patients prefer the anonymity factor of the Internet desirable and would be more comfortable and have less anxiety than an in person encounter with a healthcare professional.¹¹⁵¹ Gustafson et al. note that "computer-based interfaces have been noted to increase participants' willingness to engage in frank discussions about health status, behavioral risks, fears, and uncertainties."¹¹⁵² Internet-based interactions, such as discussion groups, have facilitated patients' and families' capacity to locate the encouragement and guidance they deserve. Scolamiero suggests "that these virtual support communities may reflect people's tendency toward socialization and is one of the most common health-related uses of the Internet."1153 Klemm et al. noted some distinct advantages of these online discussion board communities over in-person support groups, such as around the clock availability and the lack of a peer-pressure to participate that a patient will not experience online.¹¹⁵⁴ The Internet also provides a way for home-bound individuals to stay connected to their families, providers, and the online support groups that give that can give them answers.¹¹⁵⁵ Not many other information resources can compare to the Internet: its speed and sheer volume of content outpaced many other sources decades ago. While lacking formal verification of research, one last reason to note of reasons patients may go online is the removal of an intermediate, such as a librarian, between the seeker and the information, provider faster access to a wealth of knowledge.

Promoting Participation

Recent academic writing has underlined the value of free and open debate. Democracy is seen as "government by discussion," and one should place more emphasis on "social realizations" rather than striving to imagine perfect fairness and ideal institutions.¹¹⁵⁶ A democratic society is distinguished more by political engagement, debate, public contact, and deliberation than by its institutions. These systems can express the voice of the weak, the oppressed, and the disadvantaged.¹¹⁵⁷ This perspective is refreshing since it is uncommon for marginalized communities' views and ideas to be heard in public conversation.

It is interesting that a number of initiatives concerning participation experiences have been published. One relates to monthly Swiss radio programs that provide voice to vulnerable populations so they are no longer ostracized in public discourse, such as critically sick individuals, people with disabilities, people with chronic conditions, and those who are addicted.¹¹⁵⁸ Being vulnerable became a shared experience because to the efforts, which gave those who were vulnerable equal power and responsibility and made others aware of their own vulnerability.¹¹⁵⁹ An additional study describes a phenomenological investigation of the hospital experiences of elderly patients. It demonstrates how prolonged observation (shadowing) may give voice to the perspectives of vulnerable individuals who are frequently left out of research studies because they find it difficult to express their feelings.¹¹⁶⁰ The case for participation is broadened to include different categories of study. Through participation in research, vulnerability can be decreased. Protecting vulnerable persons or taking extra security measures is insufficient. To prevent exploitation, they should take part in every stage of the study and supervise the trials.¹¹⁶¹

This claim is connected to the need to identify the autonomy of helpless individuals and communities in the institutional review process of investigation, especially in economically

disadvantaged countries.¹¹⁶² Vulnerable subjects must be involved so they may speak for themselves, rather than being represented by review boards who speak for them.¹¹⁶³ Review boards would thus serve as venues for vulnerable groups to gain power rather than just as procedural tools. By promoting and broadening involvement, the institutional review will become a tool for addressing inequality.¹¹⁶⁴ As a result, participation and inclusion are strategies to combat disempowerment. As with most technological advances in healthcare, ethical challenges accompany the practice in fundamental and specific areas which hospice and palliative care are no exception. The following subsection highlights a few of these challenges that potentially can limit the modern *Ars moriendi* and prevent the art of dying well from progressing.

6.A.iii. Ethical Challenges of Technology in Hospice & Palliative Care

The development of life-sustaining technology like the ventilator, artificial nourishment, artificial hydration, and dialysis, to mention a few, has profoundly altered how we see and cope with death. These technologies have enabled people to live considerably longer, but not always with quality of life. While the practice of advanced directives and living wills has greatly helped ensure patients' wishes are respected at the end of life, many patients end up without communicating their preferences after a medical crisis. This all-too-common situation puts loved ones in fundamental topical discussions of technological futility and the technological imperative, as described in the following subsections.

6.A.iii.a. Fundamental Issues

The issues of medical futility and the technological imperative are more fundamental in nature because they can affect anyone and everyone – no matter how advanced their country's healthcare is. Every hospital in the world must deal with futility cases caused by the change in

mindset of many individuals that now that the technology is available – it should always be used to do everything possible, regardless of the bigger picture or the goals of care for the patient. This subsection dives into more detail on this issue and how technology can pose just as much of a challenge as an opportunity in healthcare today.

Technological/Medical Futility

It is difficult to define what constitutes futile treatment since physicians, academics, and politicians each have a different subjective interpretation of the phrase.¹¹⁶⁵ According to one viewpoint, therapeutic procedures that do not result in positive patient outcomes constitute medical futility.¹¹⁶⁶ The term medical futility can also characterize "any effort to achieve a result that it is possible but that reasoning, or experience suggests is highly improbable, and that cannot be systematically produced."¹¹⁶⁷ In contrast, holistically prolonging medical procedures might not be pointless if they help a patient's family members accept the death of a loved one by reducing their uncertainty, enhancing their wellbeing, or simply by lengthening their life.¹¹⁶⁸ Lacking a universally accepted or legally binding definition of futility, debates over the end of life will continue to arise in medicine, politics, and ethics.¹¹⁶⁹ Having sincere conversations with patients and their families about their desires for end-of-life care is crucial to dispel mistrust about just continuing care that medical professionals may deem pointless and promote more peaceful death experiences in light of the predicted shifts and cultural demographics in the US population. Patients' and their families' choices to transition to hospice care may be influenced by educating them about the benefits of palliative care.¹¹⁷⁰

Critical illnesses frequently come on suddenly and without warning, leaving no time for thoughtful deliberation or dialogue.¹¹⁷¹ H. Montgomery et al. note "If futile intervention or unwanted suffering is to be avoided, patients must be made fully aware of the limitations of

intensive care, such that they can make decisions in advance. It is often too late when the patient is hospitalized, let alone when a sudden decline has begun: the full benefits of collaborative or shared decision-making are only realized when the conversation begins early."¹¹⁷² They can start before a crisis ever arises, when the citizen is still at home. However, humanism and compassion should be matched with options and choices.¹¹⁷³

The idea of futility has been used in the context of palliative care, even if research and pertinent literature are rare. For instance, it has been observed that relatives of patients who passed away noted futilities in 28% of the instances.¹¹⁷⁴ The most common was unnecessary medication (21%), followed by unnecessary diagnostic examinations (19%), treatments (18%), and caring processes (17%).¹¹⁷⁵ These unnecessary interventions caused suffering to dying patients, most frequently in hospitals (50%) and less often at home (8%).¹¹⁷⁶ These results, of course, show that palliative treatment has not been of high quality. What is troubling about this, though, is that even "caring processes" are seen to be futile; again, it is not made clear what is meant here, and no instances are given.¹¹⁷⁷ Insufficient communication and information are also linked to reports of futility. The idea of futility is also used in the policies and statements of organizations that provide palliative care, as when it is stated that there is no ethical requirement to discuss cardiopulmonary resuscitation with palliative care patients.¹¹⁷⁸

Technological Imperative vs. Humanizing

Many Americans believed they may be cured or that they could delay dying because of technological advancements and the rise of professional treatment in hospitals.¹¹⁷⁹ A new technological imperative emerged, which refers to doctors using more modern technology to "do

everything possible to extend the individual's life, prolonging the inevitability of death."¹¹⁸⁰ To put it another way, intensive therapies will be used to try to prolong life.

Clinical experts in critical care must balance the use of available technology with patient requirements and expectations, as well as opinions of family, friends, caregivers, and staff, given that these are sometimes difficult to ascertain.¹¹⁸¹ This balance is made much more difficult when considering the patients in the expensive and resource-intensive ICU who use a lot of resources that might be used more equally throughout the hospital.¹¹⁸² In many instances, this raises the question of whether or not to act.¹¹⁸³ It is crucial that the public may participate in the conversation so that society as a whole, not just healthcare experts, can decide what standards of compassion, affordability, and reason it will tolerate.¹¹⁸⁴ Intensive treatment centers have gradually become more and more familiar to ICUs.¹¹⁸⁵ This has been beneficial for many people since, for example, it can increase survival rates following major surgery. In other instances, though, the need for scientific solutions and the growing capability to give them may limit compassion.¹¹⁸⁶ The practice of such excessive therapy may, in certain circumstances, teeter on the unethical when carried out by professionals whose professional ethics instruct them to "first do no harm" to the individual patient.¹¹⁸⁷

Intensive care specialists' first objective should be to humanize the ICU experience.¹¹⁸⁸ We must reduce the misery we cause. H. Montgomery et al., in their article "critical care at the end of life: balancing technology with compassion and agreeing when to stop," state, "Circadian rhythm, sleep, and pleasant sight and smell should be maximized. Pain, the administration of disorienting drugs, and noise should be minimized."¹¹⁸⁹ In order for them to comprehend the limitations of such treatment and the potential misery it may bring about, healthcare workers must also start a discourse among themselves and later with other medical experts outside the

ICU. H. Montgomery et al., in their article referring to ICU intensivists, say, "No longer should we receive referrals that state that a family or patient 'want everything,' when neither clinician, relative nor patient grasps what 'everything' really means and where it might lead, and when expectations are often unrealistic."¹¹⁹⁰

6.A.iii.b. Specific Issues

The following subsection discusses specific issues that plague humanity regarding technology in healthcare. These issues are noted as specific as they do not apply everywhere in the world and are more focused on countries and areas with low socioeconomic groups and poor technological infrastructure. One issue mentioned is the concept of the digital divide, or simply put, the separation of those with access to modern technologies/the Internet/ ICTs and those without. This transitions to the next topic of how this digital divide affects the physician-patient relationship and what barriers this divide imposes on communication, satisfaction, and access to care.

Digital Divide

The digital divide emerges as a problem that requires attention when evaluating the possibilities of information technology.¹¹⁹¹ The term refers to "the gap in computer and Internet access between population groups segmented by income, age, education level, or other parameters."¹¹⁹² In order to reduce this gap, a variety of projects have been started, with a particular emphasis on offering access to technology, online databases, and education. Despite the fact that lower socioeconomic groups are rapidly getting access to the Internet, as new technology become available, it is believed that the digital gap will still remain.¹¹⁹³ For instance, "as sophisticated multimedia services become an integral part of Internet-based applications, broadband access may become as important for accessing healthcare sites as narrowband access
is today for obtaining web-based health information."¹¹⁹⁴ In such instance, just because of various access protocols, there may be a digital gap between two groups having PC hardware and Internet connectivity. Additionally, the digital divide has many other dimensions, with infrastructure access being only one of them. Other crucial elements include proper web content and health literacy. These issues are crucial to solving if informatics applications are to be adopted in palliative care and hospice facilities, particularly if they have the ability to be used in the homes of patients and families.¹¹⁹⁵

When all else is equal, impoverished Americans are less likely than wealthy Americans to have access to the information communication technology (ICT) that enables that access.¹¹⁹⁶ "Similar gaps exist between the affluent developed and impoverished developing world. Although Internet access is increasing across the world, it is still the case that a comparatively small percentage of the developing world's poor has Internet access."¹¹⁹⁷ According to a Statisca analysis from 2022, 63 % of people worldwide have access to the Internet, however "internet penetration in the most economically developed countries is already well above 90 %, there are still approximately 2.5 billion people around the world without access."¹¹⁹⁸

There is a causal link between these disparities in income and access to information and communication technologies (ICTs).¹¹⁹⁹ Most often, those with poor socioeconomic level who are struggling to pay their expenses cannot afford Internet connection or any related training. Unfortunately, by preventing access to the opportunities, information, and resources that the Internet may offer, this vicious cycle may prolong poverty.¹²⁰⁰

The argument being stated here is not whether reducing domestic and global poverty is a complex issue that will be exceedingly difficult to solve, but rather that discrepancies between the affluent and the poor are linked to the moral relevance of the digital divide as an issue that

has to be addressed, as well as between extremely wealthy countries and countries that are living in utter poverty.¹²⁰¹ Although it is possible to make the case that these differences are fundamentally wrong, it appears doubtful given that economic inequalities are seen as harmful more for the suffering they create than for some deontological egalitarian ideal that requires total equality of justice.¹²⁰²

Professional-Patient Relationship

If databases, networks, and computers can improve communication between doctors and patients or between nurses and patients, we will have succeeded.¹²⁰³ "If reliance on computers impedes the abilities of health professionals to establish trust and to communicate compassionately, however, or further contributes to the dehumanization of patients, then we may have paid too dearly for our use of these machines."¹²⁰⁴ Let us imagine that a healthcare professional uses a decision-support tool, which is a procedure that gives physicians, personnel, clients, or other people – in-depth information that is intelligently selected or given at precise times, for the enhancement of health¹²⁰⁵ to create differential diagnoses or test a diagnostic hypothesis, and that the results of that tool determine whether a certain test or therapy should be ordered.¹²⁰⁶ A doctor must be able to explain the extent to which computers aid in their decision-making on whether or not to intervene in a certain situation. This is essential to avoid alienating patients who could otherwise get bored, angry, or confused with the use of technology in decision-making. It is possible that a doctor would prefer to keep this information private in order to maintain patient confidence.¹²⁰⁷

The procedures that make up human decision-making are not completely unknown to patients. When their physicians and nurses utilize machinery to aid sensitive cognitive activities, though, they may be deceived and lose what little understanding they do have. It is still unclear

whether patients should be informed of the automated statistics' accuracy rate before they are given information on prior real-world patients' data.¹²⁰⁸ The question remains if this information would help or hinder the informed consent exchange, or would it just cause confusion and delay decision-making?¹²⁰⁹

These kinds of questions help to ensure that the future of computers and statistics in the clinical world will be used responsibly. Unfortunately, despite numerous attempts to study whether computer use will alienate patients, the data is inconclusive.¹²¹⁰ For instance, the statistics cannot predict whether all patient groups would favorably react to all the various sorts of emails that patients may get from their physicians. A healthcare provider should not, as previously noted, communicate a patient a new diagnostic status through email because of common courtesy and moral obligation. Attempting to answer the question now only would help anticipate future problems caused by technology within the patient-physician relationship. The process of obtaining informed consent must always be kept humanized, intimate, sensitive, and sympathetic. Healthcare professionals must maintain their personability during these close patient meetings and avoid becoming overwhelmed by the vast information that health informatics may supply.¹²¹¹

6.A. Conclusion

Technology has broadly impacted the implementation and future of hospice and palliative care. While these fields have been implemented to help bring back the art of dying that has been lost in Western culture, technology can support and challenge this effort. While the location of death and dying has shifted away from home and into the hospital or similar institutions, the movement for hospice and palliative care has contributed to reintroducing the social and intimate element into end-of-life care in the home. This is supplemented by a new ethical framework for

dying conditioned by a few observations and contemporary challenges that must be addressed. The chapter discussed the opportunities modern technology can provide to hospice and palliative care fields, such as improved communication with telemedicine and medical interventions. Electronic learning has opened infinite possibilities for both patients and medical experts to investigate, educate themselves about, and get assistance for the illnesses that impact them. The Internet provides a base for an interconnected web of participation in an ongoing learning process about health and healthcare. It will continue to expand such knowledge as time and technology progress.

The chapter continues with a highlight of some of the ethical challenges that are posed by such technology, as there are two sides to every coin. Fundamental issues are described, such as technological or medical futility with life-sustaining technology and the debate about the technological imperative to 'do everything possible' to keep people alive at all costs vs. the dehumanizing of the patient with all the machinery. The section then concludes with detail of some of the more specific issues that technology has caused in the broader healthcare context, such as the digital divide that explains the socioeconomic gap that is further spread between the poor and rich by the continual upgrading and use of technology in society and the health literacy that goes along with it. Also, technology can strain trust in the professional-patient relationship, in which sometimes clinical decision support systems can lead a professional to withhold information deemed futile, which puts a strain on trust in the relationship when trust is of the utmost importance in end-of-life care conversations. Overall, technology significantly impacts how healthcare is run and maintained, including in hospice and palliative care. Technology provides both opportunity and challenge, and the ethics of all of it should be assessed, so humanity does not blindly put all the faith into it and lose the sense of humanity and personal

caregiving that was once so prevalent in the art of dying well. The next section of the chapter tackles this issue in the form of care robots that are increasingly used to care for the growing aging population. Care ethics provides a framework to design these robots so that we don't lose that sense of humanity and preserve the art of dying well for years to come.

6.B. Ethical Considerations of Using Care Robots in Aging Populations

6.B. Introduction

With a growing number of elderly people in need of assistance worldwide, accompanied by a decrease in available caregivers – there is a shortage of epidemic proportions in elderly care. However, recent advances in robotics and assistive technologies provide an opportunity to address the deficit with the use of robots in caring for the elderly and aging populations. This mechanization of care work has already been going on for several years in many countries, but are ethics being considered in the design and implementation of these 'carebots?' This section investigates the question: what are the ethical considerations in designing robots used in caring for elderly and aging populations?

In answering this question, we start with the importance of the design of the robots using the foundational concepts of care ethics – specifically those related to elder care, such as relational autonomy and dignity-enhancing care. The section then moves to a more philosophical discussion about which virtues and values are essential to instill and whether it is possible to have an artificial morality within machines. If so, what would a framework look like for an ethical evolution of these care robots? Finally, ethical challenges are posed when using a carecentered framework, such as implementing human capabilities in care robots, giving the robots context sensitivity, and the whole issue of deception in the care provided by these robots. Care robots and other assistive technologies have enormous potential to enhance the lives of the

elderly by helping to open up human caregiver availability, reducing dependence on others, and much more – so long as they are designed carefully and ethically.

6.B.i. The Mechanization of Care Work

6.B.i.a. Scope of the Aging Population

The World Health Organization reports a fast-growing disparity between the number of elderly people who require care and the shrinking pool of caretakers.¹²¹² For instance, in the UK, it was anticipated that between 2015 and 2020, the general population's population of persons over 65 would rise by 12% (1.1 million), that of those over 85 by 18% (300,000), and that of people over 100 by 40%. (7,000).¹²¹³ By 2020, it was anticipated that there would be more people 65 and older than children under five worldwide.¹²¹⁴ A long-term trend that started in Europe a few decades ago is population aging. As the number of persons aged 65 and older rises throughout the EU, people are living longer and having fewer children per couple.¹²¹⁵

Only 50% of people over 65 said their health was "very good" or "good," compared to 88% of the rest of the population, according to the 2011 UK census.¹²¹⁶ According to the same census, 56%, or 5.2 million people aged 65 and older, were married, up from 52% (4.3 million) in 2001.¹²¹⁷ The percentage of married couples climbed from 51 to 54%, while the number of couples who are cohabiting nearly doubled from 1.6 to 2.8%.¹²¹⁸ The statistics relating to marriage rates are of particular note because these couples are less likely to need elderly care as many couples will take care of and support each other in old age. The percentage of people 65 and older who were living alone fell from 34% in 2001 to almost a third (31%) in 2011.¹²¹⁹ As a result, welfare states must pay escalating maintenance and social assistance expenses for senior citizens who are unable to live independently.

The senior population in the United States will increase significantly between 2012 and 2050. The number of people 65 and older is expected to increase to 83.7 million by 2050, nearly double the forecast number of 43.1 million in 2012.¹²²⁰ Since they started reaching 65 in 2011, the baby boomers are mostly to account for this rise in the senior population. The remaining baby boomers will be older than 85 by 2050.¹²²¹ The population is anticipated to age significantly, as this section's goal describes. According to projections, by 2030, more than 20% of Americans would be 65 or older, up from 13% in 2010 and 9.8% in 1970.¹²²²

We shall see a global shortage of care professionals due to the aging population that is rapidly expanding. There is currently a shortage of senior care in several nations, notably Japan. Japan's population is aging quickly as a result of its low birth rate and high life expectancy, with 22% of people over 65 in 2009 and a forecast increase to 34% by 2035.¹²²³ Numerous institutions including elderly care facilities, the Japanese government, and Japanese families are working together to find a solution to the lack of care specialists that have the knowledge and skill to assist in the continuous care of the elderly.¹²²⁴ Jennifer Parks notes, "Japan has responded to this problem by developing technologies to assist with some of the care demands, such as machines to bathe nursing home residents, machines and robots that use surveillance technology to monitor the vital signs and condition of elderly persons, or even robots that can act as companions for the many demented, lonely, and isolated elderly citizens."¹²²⁵ According to Motoki Korenaga, a representative of the Japanese Ministry of Trade and Industry, in March 2009: "Japan wants to become an advanced country in the area of addressing the aging society with the use of robots."1226 His remarks also mentioned intentions for the quick preparation of safety regulations for robot nurses who would likely take care of senior patients over the next five years.¹²²⁷ In preparation for the impending age boom, Germany has also created robots for

caregiving. According to the German federal statistics office, 85 out of every 100 Germans between the ages of 20 and 60 in the next 40 years will be elderly people.¹²²⁸

6.B.i.b. Elderly Care Needs

The concept of assistive technologies casts a wide net to define these as "user-oriented technologies aiming to provide or assist users with public or private welfare services."¹²²⁹ These "welfare technologies" assist citizen participation in society.¹²³⁰ This can include assistance in daily activities, security, self-reliance, and other various forms of participation.¹²³¹ Assistive technologies are grouped into four different ways of providing support: "1) safety and security, e.g., the safety alarm service, 2) compensation and wellness, e.g., memory support, walker, light/heating control, 3) social contact, e.g., video communication, and 4) care and treatment, e.g., blood glucose meter, blood pressure monitor."¹²³² The technologies aim to enhance a person's weakened abilities (like glasses and arm-grasping forceps) or replace lost capabilities (such as a wheelchair or hearing aid).¹²³³

When the patient is elderly and cannot fully be independent, healthcare services can enter their home and bring tools and technologies that assist in enabling a more independent lifestyle. These various technologies take over the tasks traditionally completed by humans, such as replacing the manual medication dispenser with a digital one that can remind its user and their provider to take daily medications. Bratteteig and Wagner have noted that society requires more "work to make home care work," in other words, more effort is needed to progress the tools and technology that enable independent living for elderly adults such that these technologies still require fine-tuning from both the care receiving and providing ends.¹²³⁴ Woll and Bratteteig also emphasize that since these technologies change the caretaking process, it affects the relationships of all those involved: the caregivers and receivers, family, friends, and providers.¹²³⁵ This take

supports their view that desires to put elderly care into a larger context that uses "an elderlycentered way of organizing elderly care."¹²³⁶

At this point, it is helpful to make a distinction between active and passive uses of technology. The distinguishing factor is that levels of automation can differ in the design of the technologies to support users with a wide spectrum of disabilities.¹²³⁷ Woll and Bratteteig argue in another article that "fully automated technology can give users full autonomy, and there is no one-to-one relationship between the level of automation and the user's autonomy."¹²³⁸ The active use of technology requires conscious interaction with technology for a specific purpose, such as using a lift chair or taking pills from a digital medication dispenser. Such actions would need a necessary understanding of how the technology works and the ability to perform the actions to operate it accurately. This active use of technology is different from passive such that passive, as the name suggests, requires no conscious interaction of the user. The most prominent example of this is a fall alarm automatically triggering activation when the user has an accident and alerts others to give assistance to the user and help them up. These circumstances have the user engaging with the technology without understanding the exact method of operation. Mao et al. give a few more examples of passive-use technologies, such as "shower nozzles/water faucets that provide a constant temperature in the bathing/shower water, infrared water tap flow control, or automatic refrigerator door closer, to mention a few."¹²³⁹

The passive use of technology creates a unique space for elderly users who may not be able to understand modern-day operations and interactions of technology that may seem abstract or foreign.¹²⁴⁰ Nursing homes may be one of the first places to benefit from such technologies as dementia prevalence will be highest there, and passive technology could significantly support the users in their daily activities there.¹²⁴¹ People who have been affected by severe cognitive

impairments, like dementia, are highly susceptible to struggling in the active use of technology. The example Woll and Bratteteig give describes a situation where a cognitively impaired person falls and does not know or remember that they have the ability to call for help by pushing the alarm button located on a device that they wear around their neck or wrist.¹²⁴² In addition, the fallen user may not want to be a burden to ask for help from other human caregivers, inconveniencing someone else. Hence, these users need an alternative to active-use technologies that can accommodate their increased needs in the healthcare landscape. This is where care robots have become significant in elder care.

6.B.i.c. Rise of Care Robots

There is a quickly rising view that care robots are the key technological invention to answer the growing elderly caregiver/receiver gap. Care robots can be programmed to be partly or fully self-governed. These robots have the ability to support both caregivers and care-receivers with various physical needs. Bedaf, Geldernlom, & de Witte give examples of "My Spoon Robot can aid someone with eating problems, and the Sanyo Bath Robot provides hygienic care to older adults."¹²⁴³ While some other robots act as more of a social support to the user, such as the seallike robot Paro or the dog-like robot AIBO.¹²⁴⁴ Yet, other care robots exist that utilize both assistive and social functionalities. They assist in social interaction, such as Robovie and Pearl robots.¹²⁴⁵ "Pearl is an autonomous robot able to follow patients around, communicate via a touch screen display, and serve as a telepresence device."¹²⁴⁶ Robovie was created to interact with people and is outfitted with a variety of tactile and vision sensors, microphones, and motors that "facilitate interactive meaningful behavior" via robotic arms, eyes, and head.¹²⁴⁷

According to Vallor, "carebots are robots designed for use in home, hospital, or other settings to assist in, support, or provide care for sick, disabled, young, elderly, or otherwise

vulnerable persons."¹²⁴⁸ Amanda and Noel Sharkey provide a helpful description to differentiate between the functionality of carebots: "performing or providing assistance in caregiving tasks; monitoring the health or behavioral status of those receiving care or the provision of care by caregivers; and providing companionship to those under care."¹²⁴⁹

Aimee van Wynsberghe mentions that no specific capability can be attributed to all care robots. However, they do have a wide variety of abilities, such as motion and types of recognition, including face, voice, and even emotion.¹²⁵⁰ In addition, these care robots have various levels of autonomy, beginning with entirely human-controlled surgical robots (i.e., the da Vinci machine), advancing to TUG (a hospital delivery robot) and RIBA (a robot that lifts patients without human input).¹²⁵¹ Therefore, a care robot "relies on the idea of *interpretive flexibility*, that a robot is defined by its context, users, and task for use."¹²⁵² In other words, depending on its current functionality, the same robot could go by two different names – each specific to its current task. Wynsberghe gives the example of HAL, or the Hybrid Assistive Limb, to demonstrate how the limb can be used for rehabilitation and care purposes depending on whether it's worn by the patient or used by the nurse to help lift a patient.¹²⁵³

An example of research embedded in the broader robotics research program is the "Acceptable robotiCs COMPanions for AgiNg Years" acronym ACCOMPANY.

ACCOMPANY aims to develop:

a robotic companion as part of an intelligent environment, providing services to elderly users in a motivating and socially acceptable manner to facilitate independent living at home... providing physical, cognitive and social assistance in the everyday home tasks, and ... contributing to the reenablement of the user, i.e., assist the user in being able to carry out certain tasks on his/her own.¹²⁵⁴

Amirabdollahian et al. explain that the ACCOMPANY system can operate independently in a smart home while "co-learning" with the user in addition to being mobile and having an

operating arm.¹²⁵⁵ They go on to say that the target demographic for the ACCOMPANY system are elderly individuals living by themselves at home whose physical and mental abilities are declining – limiting their independence at home.¹²⁵⁶ ACCOMPANY is just one response of many to addressing the rising number of older persons who need help with daily tasks. This system simultaneously attempts to offer an economically reasonable solution to provide adequate care to this in-need population. ¹²⁵⁷

Before the general public has access to really reasonable, secure, and useful care robots for institutional or general consumer usage, still more work has to be done. The advancement of care robots will certainly continue considering the serious imbalance of caregivers and care recipients in addition to the capitalist mentality to continue creating new technologies that become available for commercial retail.¹²⁵⁸ Carebot designers and developers may easily inflate the future role of these technologies to help promote their products. The direction the designers are heading in is clear – as stated on the RIBA's temporary marketing website(the robot that lifts patients without human input): "There might come a time when robots will replace human caregivers. An impossible vision it may seem, but do not be too surprised about human caregivers being assisted by robots in the near future."¹²⁵⁹

6.B.ii. Designing Robots Using Care Ethics

6.B.ii.a. Defining Care Ethics & The Value of Care

Care ethics offers a unique and different approach to bioethical issues from many other well-known methods such as principalism, virtue ethics, deontology, and consequentialism, to name a few. The normative questions these various approaches attempt to answer are all relatively the same: What criteria should we use to judge people's moral behavior? How do we

proceed? Which behaviors are morally required? Care ethics tries to answer these questions while considering care – in all its aspects – to be of significant ethical value.

In a Different Voice, written by Carol Gilligan and initially released in 1982, helped the foundations of an ethics of care take off. As a developmental psychologist, Gilligan described psychological and empirical findings that focused on girls' development in their morality as opposed to boys.¹²⁶⁰ Gilligan deduced that there is a significant difference or a "different voice" in how males and females view moral issues. Specifically, females are more focused on interpersonal relationships and are more swayed in their decisions by emotions of love or hate, whereas males value independence and abstract rules of behavior.¹²⁶¹ Gilligan asserted that "although only some women studied adopted this different voice, almost no men did."¹²⁶² In a subsequent piece, she said, "this meant that if women were eliminated from the research sample, care focus in moral reasoning would virtually disappear."¹²⁶³

In her book *Moral Boundaries*, Joan Tronto outlines the four components of care, which serve as an excellent place to start for describing certain aspects of a care ethic. The four elements of care are: "caring about, noticing the need to care in the first place; taking care of, assuming responsibility for care; caregiving, the actual work of care that needs to be done; and care-receiving, the response of that which is cared for to the care."¹²⁶⁴ Four ethical principles of care "attentiveness, responsibility, competence, and responsiveness" come from these four components of care.¹²⁶⁵

A comprehensive study of moral action is provided by the ethical perspective known as care ethics. It examines this behavior in the context of particular care partnerships. Care ethics are marked by this "contextual and relational sensitivity."¹²⁶⁶ As a result, care ethics continue to be tightly tied to real caregiving activities and carefully consider how caregiving obligation

manifests: "the contextual elements, the typical character of a situation, the way people collaborate, and the emotions involved."¹²⁶⁷ Care ethics runs the actual risk of emphasizing process-related components at the expense of content that is value-loaded by concentrating on specific contextual and relational features. The duty lacks a normative reference if care ethics really restricts itself to the "pure reconstruction of an actual care situation," such as who assumes total accountability for the care and all its aspects, without consulting any standards or values to judge if this treatment is suitable.¹²⁶⁸

6.B.ii.b. Relational Autonomy in Elder Care

Over the past few decades, autonomy has become a more important value as healthcare has shifted away from the paternalistic patient-physician relationship.¹²⁶⁹ This is seen as a general improvement in healthcare. However, regarding older adults, especially those with cognitive impairments like dementia, "the privileging of autonomy has caused tremendous difficulties because of competency issues."¹²⁷⁰ Too often, the mistake is made that we assume elderly individuals lack decision-making capacity. Hence, family members or healthcare professionals use their personal preferences to make decisions for the elderly individual.¹²⁷¹ We must stop imposing our own beliefs, values, and preferences on others and allow them to have their wishes expressed in the care process. Absent of any bad intentions, decision makers often make their choices with the patient's best interest at heart. Regardless, several points must be mentioned in this situation.

Despite the best intentions, oftentimes denying a person their autonomy is generally harmful.¹²⁷² Robert Stewart notes that family members or healthcare professionals often override the elder's autonomy too quickly because they believe the elder is deemed to lack decision-making capacity even before a proper evaluation has been performed, which ultimately

dehumanizes the elder.¹²⁷³ Also, there are numerous ways to conceptualize autonomy – certain ways are better than others in managing dementia patients. The essential nature of autonomy is the capacity to self-legislate or act independently of others utilizing reason as a compass.¹²⁷⁴ The "independence" and "rationality" elements of autonomy provide challenges for the elderly, particularly those with dementia.¹²⁷⁵ However, these qualities are less important in other conceptions of autonomy. Particularly, relational autonomy has been supported by care ethics.¹²⁷⁶

Care ethicists have proposed that the popular conception of autonomy is predominately male-centered, based on some empirical research first conducted by Carol Gilligan.¹²⁷⁷ Males and females differ in their values when it comes to relationships and emotional ties, where males place more significance on independence and abstract concepts.¹²⁷⁸ Because our relationships with others play a role in defining who we are as people, Gilligan and others have claimed that these relationships and the emotions that surround them must be taken into account while making decisions.¹²⁷⁹ Therefore, we should consider autonomy in regards of our interactions with others.¹²⁸⁰

The idea of autonomy that underlies the traditional healthcare ethic, which is based on an individualistic view of man, is rejected by the ethics of care.¹²⁸¹ This idea of autonomy disregards each person's ties to and reliance on their social environment. It ignores the social contexts in which these characteristics are important and instead defines autonomy as "rationality, individuality, and self-determination."¹²⁸² According to Virginia Held,

The ethics of care usually works with a conception of persons as relational, rather than as the selfsufficient independent individuals of the dominant moral theories. The dominant theories can be interpreted as importing into moral theory a concept of the person developed primarily for liberal, political and economic theory, seeing the person as a rational, autonomous agent, or a selfinterested individual.¹²⁸³ In order to care for persons with chronic diseases, we need a relational definition of autonomy based on a social perspective of man, one that pays attention to the connections in which the individual sees himself or herself. Only in a relationship that enables them to establish their identity and learn to manage their sickness can people with a chronic illness who have been reliant on outside care experience their independence.¹²⁸⁴ A conceptualization of care that excludes all individual values and attitudes is incompatible with this process.¹²⁸⁵

Responding to the person's vulnerability by family caregivers and professional caregivers is a requirement for integrating dependency and autonomy in elderly dementia care.¹²⁸⁶ Sensitivity and empathy are crucial qualities in care ethics, however, because caring also includes a cognitive component, this does not limit ethics to only an emotional reaction.¹²⁸⁷ Professional understanding of care ethics, which include competence, responsiveness, attention, and accountability in addition to being devoted, involved, and active in the caring process, is necessary to provide effective care for others.¹²⁸⁸ The act of caring requires more than just making the proper choice at a given time; it is a dynamic and continuing process. It necessitates constant participation and decision-making.¹²⁸⁹

When we think of people as relational, it does not follow that we are unable to oppose the many social ties we were raised in or find ourselves in, or to modify whatever relationships we do have.¹²⁹⁰ However, it frequently calls on us to do so. According to the ethics of care, we should take these decisions into account as they relate to the relationships that shape who we are.¹²⁹¹ We constantly change our relationships with others – some are maintained, some are ended, and some are created. Yet, as noted in the contractual model, we don't view these as choices of people acting as though social ties did not exist before making them.¹²⁹² Interestingly, moral agents are inspired by the ethics of care, and while having relationships with other

individuals, these moral agents can still be free.¹²⁹³ When necessary, the ethics of care seeks to encourage the cared-for person's responsible autonomy. Care ethics needs us to focus on autonomy's physical, mental, and emotional requirements.¹²⁹⁴ Because autonomy is exerted within relationships rather than by abstract, autonomous, free, and equal persons, those lacking certain resources cannot be autonomous in a way that is effective.¹²⁹⁵ In a related sense, human dignity and vulnerability of the elderly can provide a normative foundation for care and care ethics as described in the following subsection.

6.B.ii.c. Human Dignity and Vulnerability of the Elderly

According to Vanlaere and Gastmans, "a person is more than just the sum of rational capacities (like the capacity to think) and powers (such as the power for verbal communication)."¹²⁹⁶ This includes people who lack these capacities and self-awareness. As a result, the body plays a key role in a person being a subject. Janssens notes "that the person's being a subject manifests itself in the most unique, but in a sense, also the most vulnerable aspect of our being, that is to say our corporality."¹²⁹⁷ People may get ill physically and mentally, which is a recurrent theme that can be found precisely in our bodily susceptibility. In the journal article "A personalist approach to care ethics," theologian Paul Valadier mentions the human being's bodily frailty as the foundation of caring by saying, "respect and care for the distorted human nature are aroused by the undignified state in which sick or injured persons find themselves."¹²⁹⁸ Being unwell is at the core, perhaps more so than susceptibility as the potential for disease. In the article, using the Good Samaritan tale as an example, Valadier demonstrates that the robbed traveler's damaged body seems to be more than simply a body throughout their interaction, and he says that "the body reveals itself as worthy of respect."¹²⁹⁹

Care begins where a person is lacking in abilities, physical prowess, and mental capacity. Creating dignity is the goal of the compassionate connection between the sufferer and the caretaker.¹³⁰⁰ Therefore, both the one providing the care and the person receiving it develop dignity. Valadier continues, "Dignity manifests itself in the act that makes us focus on the other so that we consider him or her as a person, even if it assumes an appearance of not being human, or even being inhuman."¹³⁰¹ Individuals are interested in one another, and at the heart of this interest is being a person. Since everyone is aware of how important one another is, this suggests that people depend on one another in order to be who they are.¹³⁰²

It is clear from the concept of personalism that caring commitments between persons cannot only be viewed as "being rational and symmetrical."¹³⁰³ Care must be defined within this critical duty to the other's vulnerability. Care requires another person to be vulnerable in some form (physically, mentally, or emotionally), it is within the recognition of this duty to act in caring for another that human dignity is respected. The creation of dignity occurs through the care of the body. Care is not based on logical reasoning but rather respecting the dignity of others and the fact that one sees the other as a person despite the other's undignified look.¹³⁰⁴ Margaret Urban Walker in her book, *Moral Understandings: A Feminist Study in Ethics*, defines care as "a practice of responsibility in which the different persons involved take responsibility in the process of reacting to vulnerability."¹³⁰⁵ Her definition parallels the view of vulnerability described by Vanlaere and Gastmans in that her view of care also requires a commitment to the other's vulnerability and so by extension respects the dignity of others as well. Care appears to be how others relate to people, and this is especially true for those who have limited talents and powers (such as the mentally impaired and those who suffer from severe dementia).¹³⁰⁶ The

Good Samaritan parable was presented earlier as an example of how this treatment might frequently be portrayed while taking care of the body physically.

The essential concepts of care ethics, relational autonomy, personalism, and human dignity must be core to the design of robots for caring for and assisting the vulnerable elderly population. Lacking the foundational concepts or context of care, robot designers do not have the necessary background to program the value of care into a test care robot. While this section only provides a brief of what these concepts entail, it nonetheless provides a starting point from which designers and engineers can work and continue to research and expand their knowledge and implementation of care ethics within the programming of the carebots.

Virtues and Values of Care Robots

Given that the International Organization for Standardization has not yet provided any guidelines for these kinds of robots, there is now a potential to include ethics into the actual construction processes.¹³⁰⁷ Accordingly, "one must first identify the moral precepts of significance followed by an account as to how to operationalize said precepts" if robot design is to take ethics into consideration.¹³⁰⁸ In their arguments regarding how care robots will ultimately replace human caregivers, several care-ethicists also emphasize the benefits inherent to care practices.¹³⁰⁹ These beneficial outcomes are produced for both parties by the morality of the caregiving process involving human carers and care recipients. Consider the reciprocal nature of virtues like kindness, empathy, openness, dignity, and attentiveness.¹³¹⁰ The question of what will happen to these internal goods emerges because numerous authors deny the idea that care robots have a reciprocal dimension unless they do so fraudulently through deceit (more detail on this later).¹³¹¹

Two moral traits that are fundamental—reciprocity and empathy—may, in some ways, be developed through caring activities.¹³¹² These are by no means the only significant moral virtues fostered in caring procedures. Shannon Vallor argues that they highlight two crucial points: "(1) that there are important goods for *caregivers* that are internal to caregiving practices, and that (2) the potential on these from the widespread use of carebots merits our careful attention."¹³¹³ Following Mark Coeckelbergh's robo-ethics, which focuses on the ethical behavior robots should develop, informs us that we may want to give more critical thought and delay giving some of these traits to carebots.¹³¹⁴

The first of the two moral virtues to be discussed is reciprocity that may be described as a "primitive biological impulse which functions as the seed of human sociality, is the unifying feature of all forms of friendship and, with proper moral and cognitive/perceptual habituation, matures into a social virtue."¹³¹⁵ According to the idea of personalism, people are inherently or naturally inclined to care for other people. Being a decent person is having the capacity to reciprocate appropriately, in the appropriate ways, at the proper intervals, and as suitable for certain situations and individuals.¹³¹⁶ However, reciprocity is a virtue that we must cultivate since it is not only a natural disposition.¹³¹⁷ One of the most fundamental virtues is reciprocity, which is necessary for producing and maintaining the basic necessities for the flourishing of human life.¹³¹⁸ This might also imply that we have a moral obligation to alter both the societal structures that promote reciprocity as a virtue and those that work against it.¹³¹⁹ If this is the case, it is even more important to think about how the introduction of carebots can affect the development of reciprocity in humans.

Vallor defines empathy as "an emotive/perceptual capacity that, like reciprocity, develops in most humans from a basic biological impulse, expresses itself fully in the highest forms of

friendship, and, when properly cultivated and expressed, constitutes a virtue."¹³²⁰ Empathy is the emotional and perceptual ability to share the pleasures and sorrows of another sentient person in a meaningful way.¹³²¹ When developed as a virtue, it takes the form of the capacity to receive these emotions in the right relationships and situations and to respond to the other in a way that is as highly tuned to and suitable to these details.¹³²² We will create problems for individuals with whom we sympathize, especially when it comes to caring relationships. When we practice caring, we put ourselves through the pain of observing, experiencing, and reacting to another's suffering in a manner that is appropriate.¹³²³ Sparrow & Sparrow emphasize the need of empathy in caring as a way to demonstrate how carebots cannot replace human carers, saying that "entities which do not understand the facts about human experience and mortality that make tears appropriate will be unable to fulfill this caring role."¹³²⁴

In order for empathy to persist and flourish rather than wither on the vine, it must be carefully and fully intentionally fostered, much like reciprocity. As a result, while considering the moral consequences of robotic care, these components should always be taken into consideration. If the ease of using robots for care tempts us to give up caring behaviors before we have an opportunity to mature and acquire qualities like empathy and reciprocity (among many others), that might irreparably harm society and its morality.¹³²⁵ However, if carebots offer kinds of limited support that encourage us to engage in caring activities, allowing us to feel and give more without having to worry about crushing pressures, the moral impact on carers' character may be quite favorable.¹³²⁶ Imbuing carebots with these various moral virtues, such as empathy and reciprocity, brings us to the concept of artificial morality, which describes in more detail some of the ethical and philosophical questions that arise in the processes of picking and choosing which virtues and attributes should be programmed into the carebots.

Artificial Morality

Robotics, AI, and assistive technology are all developing fields that deal with artificial morality. Simulating or modeling human cognitive skills is the goal of artificial intelligence. The study of whether and how artificial systems may be given moral capabilities is known as artificial morality.¹³²⁷ Another expanding area for the use of artificial morality is elderly care. The idea is to use morally capable autonomous artificial systems for domestic care, helping to address the previously mentioned demographic care shortfall.¹³²⁸ Examples of situations requiring moral judgment in this setting include: What is the ideal frequency and level of intrusiveness for a care system to remind somebody to nourish themselves or receive treatment? If someone has not been moving for a time, should it be reported to the family or a medical service? If yes, how long should they be left alone? Should the system continuously watch the user, and how should it use the information gathered? In this situation, individuality, confidentiality, physical wellbeing, and family interests are ethically significant.¹³²⁹

Artificial morality "takes place between the triad of artificial systems, designers and users where each factor contributes to its own."¹³³⁰ In other words, the moral dilemmas that these technologies will face will get more complex as they become more intelligent and autonomous. The benefits and drawbacks of artificial morality must both be taken into account. Determining each agent's contribution and outlining the effects of moral agency and responsibility are two of the most significant difficulties when discussing artificial morality. This is vital to determine whether or not we, as people and as a community, want artificial systems to be capable of morality.¹³³¹

One of the key issues in artificial morality is the question of whether artificial systems may be regarded as moral agents. However, two more questions are raised by this point. The first

is about agency in general. Clarification of the conditions that must be satisfied for anything to be regarded as a moral agent is required, as is proof that at minimum a few artificial systems satisfy these conditions.¹³³² The second and third problems, respectively, are what constitutes a moral agent and if artificial systems may be both moral and mere agents. Misselhorn denotes two dimensions required for agency in philosophy "the first dimension, or gradual characteristic of a moral agent, involves the idea that agents are self-originating sources of their doings."¹³³³ Moral agents are always responsible for their actions. Another way of looking at that is by saying external factors may not determine the agent's actions, nor are they under the agent's control. "The second dimension that being an agent involves is the capacity to act for reasons."¹³³⁴ Let us imagine that a robot or other system possesses representational states that perform the same function for it that the equivalent mental states do for humans. In such case, the system may be considered to be functionally equivalent to a human agent. They have the capacity to make and carry out plans. This may be sufficient to state that these systems are operating rationally.¹³³⁵

The second worry is whether or not artificial systems now have the ability to behave morally. If an artificial agent is a "self-originating source of behavior" and its motivations are moral, then it may be said to be moral in the most basic sense.¹³³⁶ There are other information processing systems that force systematic change if it is not working ethically, there must be states showing that it is functionally similar to moral beliefs and pro-attitudes.¹³³⁷ One must be mindful that an artificial system's presentation of a primitive type of moral agency does not equal the full moral agency exercised by humans.¹³³⁸ This is partly because AI systems' moral competence is restricted to certain sectors. Human morality, on the other hand, may be used in any circumstance and has a far larger range of applications. Full moral agency, as granted to

humans, also comprises a range of distinct intentional beliefs, phenomenal awareness, free will, and the capacity to submit moral behaviors to analysis and reasoning.¹³³⁹

Framework for Ethical Evaluation of Robots

A framework for the ethical assessment of care robots must take into account the particular context of usage, the individual requirements of users, the jobs the robot will do, as well as the robot's technological capabilities.¹³⁴⁰ Future care robots will need an ethical design framework as a tool, going further than merely a retrospective review of robots.¹³⁴¹ The rights of the person, the unique requirements of a certain demography of care recipients, or the impact particular to the caregiver are some of the viewpoints from which authors debate ethics and robotics.¹³⁴²

Three dimensions could be described when the ethics of robots are mentioned: "First, the ethical systems built into robots; second, the ethical systems of people who design robots; and third, the ethics of how people treat robots."¹³⁴³ Given that robots are socio-technical systems, might lead one to believe that a framework for treating them ethically should take into consideration all three factors.¹³⁴⁴ The paradigm created in this chapter incorporates care-based ethical analysis into a care robot's construction. The framework intends to foster ethical thought from the standpoint of care ethics tradition and to inspire designers' and engineers' ethical thinking, and to clarify the relationship between a care robot's technical components and the ideals that are finally reflected in a care practice.¹³⁴⁵

These three dimensions are addressed with the value-sensitive design (VSD), which also concentrates on the framework's objective. Value-sensitive design is "a theoretically grounded approach to the design of technology that accounts for human values in a principled and comprehensive manner throughout the design process."¹³⁴⁶ The idea that technologies represent

values serves as the foundation for value-sensitive design, which also provides a viable way for assessing the present state of technology design during the course of the planning and execution process.¹³⁴⁷ The argument that software and computer systems are ethically neutral and rely on the user to gain moral importance is refuted by this viewpoint. It is currently possible to predict whether a computer system would favor specific moral standards and ideals.¹³⁴⁸

VSD is used to develop a specific system or object, however its usage in this context aims to establish a generic framework that designers and ethicists may use to assess any care robot's ethical performance or to include ethics into the any care robot's construction.¹³⁴⁹ Coeckelbergh notes, "The VSD blueprint is used to create a framework that examines the connection between technological capabilities and design in order to build care robots with the particular context of their usage, purpose, and clients in mind."¹³⁵⁰ The framework's versatility to be used to current and upcoming care robot designs is one of its main advantages. Designers can understand how their work will affect the care being provided by using previous designs as a guide. The framework may be used to future projects, which by default integrates ethics into the design process.¹³⁵¹

6.B.iii. Ethical Challenges of a Care-Centered Framework

A few ethical challenges accompany the implementation of a care-centered framework for care robots. In order to assess the ethical implications of carebots, the capabilities approach's initial problem depends on human capacities. The integration of a context-sensitive method into the care robot design is the second obstacle. Each poses a set of moral difficulties surrounding the design and implementation of the carebots – but these questions must be answered if the process is to move forward so that ethics is designed from the start – not as an afterthought.

6.B.iii.a. Human Capabilities

Capabilities approaches, "rooted in the economic and political philosophies presuppose that conditions of justice require attention to more than just the availability and social distribution of various external goods, resources, or utilities."¹³⁵² Instead, considering justice necessitates paying attention to fundamental human capacities and/or functionings, which are seen as benefits achieved in daily life through particular types of actions and routines.¹³⁵³ The needs of justice necessitate that we not forgo these skills' development by people solely to maximize aggregate utility, since their realization is essential to human flourishing. The capacities approach may be used more generally in the framework of ethics to help in conceiving the good life and the ways in which it may be fulfilled, despite the fact that it was established to address shortcomings in other notions of justice (such as utilitarianism and Rawls).¹³⁵⁴

When analyzing the ethical implications of carebots, philosophers employ capabilities approaches. Borenstein and Pearson give an example of how caregiver moral responsibilities could be met with the help of carebots without compromising their emotional well-being which could undermine their capacity for love, fun, or friendship as human beings.¹³⁵⁵ The capabilities method also aids in identifying inappropriate uses for carebots, such as robbing carers or those being cared for of control over their surroundings.¹³⁵⁶ Consider a carebot that restricted residents of nursing homes' movements in order to lessen the risk of a fall, protecting their physical health at the cost of social isolation and loss of autonomy. The capabilities approach is another example Coeckelbergh uses to illustrate how to think ethically about carebots.¹³⁵⁷ Although he does not think the capabilities approach addresses all of the ethical issues that carebots bring up, he does advise using it as one method to determine if a particular kind of caring meets the criteria for good care.¹³⁵⁸

Vallor has suggested that we "employ the capabilities approach in one further respect: to help us to conceptualize the goods internal to caring practices for caregivers, goods that might be lost should a person choose to surrender some or all of such practices to carebot."¹³⁵⁹ The benefits inherent in caring behaviors might be included into each of the ten fundamental human capacities that Nussbaum listed in *Frontiers of Justice*.¹³⁶⁰ However, in this section, we will concentrate on only three of them: affiliation, practical reason, and emotion, each of which indicates improved abilities for carers in the objective of providing care.¹³⁶¹ Nussbaum defines these as "Affiliation: being able to live with and towards others, to recognize and show concern for other human beings, to engage in various forms of social interaction; to be able to imagine the situation of another."¹³⁶² "Practical reason: being able to form a conception of the good and to engage in critical reflection about the planning of one's life."¹³⁶⁴

These human capabilities are to be used as qualification benchmarks to determine and judge the use of robotic assistive technology in the healthcare sector. Human dignity should be enriched and promoted through good and reasonable care and thus necessitates that the aforementioned human capabilities are standards that should be met with or without technology in healthcare.¹³⁶⁵ In this circumstance, the capabilities include care-ethical dimensions that have been discussed and others that positively affect good care. However, there is a possibility that for certain capabilities and care tasks, care robots and assistive technologies cannot provide them as well as human beings. This should be determined by each circumstance separately, as a society should not reject the use of care robots outright. Therefore, in some care situations and capacities, care robots and assistive technologies may replace care given by humans. Still, in other cases and capacities, they may only be able to assist and not replace human care.¹³⁶⁶

6.B.iii.b. Context-Sensitive Process

Lacking the foundational concepts or context of care, robot designers do not have the necessary background to program the ability of context-sensitivity into a test care robot. Fieldwork experience in both nursing homes and hospitals is necessary to fully understand the special significance of these concepts in context. For instance, safety in a dementia unit can include keeping patients from wandering outside or keeping them from injuring both themselves and other people. Safety in a nursing home ward often refers to keeping patients from falling or helping them eat so they do not choke. The setting, individual experiences, and particular practice all influence how value is prioritized.¹³⁶⁷ For instance, by lifting, the value of safety is demonstrated by preventing the care recipient from falling or becoming hurt. Safety is of utmost concern in this situation. Contrarily, bathing safety is decided by the water's temperature (ensuring that the patient will not be burned or scalded) and the patient's right placement to minimize risk. While safety is always a top priority when taking a bath, other considerations come first.¹³⁶⁸ For instance, drawing the curtain to offer privacy, speaking to the care recipient to soothe them, and using delicate strokes to show them respect and understanding. These instances highlight the interconnectedness of care values and the value of therapeutic relationships, where the fundamental principles of healthcare are viewed in the context of the relationship.¹³⁶⁹

The context of the care practice must be determined, as was previously indicated. Consider a particular hospital's ward as opposed to a nursing home or a home environment. Numerous factors contribute to the environment in which the care is provided is important. According to recent research, one's approval of utilizing robots to provide care is correlated with their religious views.¹³⁷⁰ It has been suggested that those who believe in "a god" would not be as open to accepting close human-robot intimacy.¹³⁷¹ Thereby, negating all the time, effort, and

resources used to design, build, and test carebots in various settings. It is necessary to do further study to ascertain if and to what degree religious convictions might be detrimental to the acceptance of care robots. Hopefully, in an effort to not waste any time and energy on their prototypes, designers of the robots would not deceive any elderly person to thinking they are interacting with human when it is really a care robot. The final challenge this subsection will discuss is deception of the elderly when it comes to the use and implementation of care robots.

6.B.iii.c. Deception

There may still be problems with deception in the provision of senior care despite the use of technology, but these issues will likely be at the level of actual care in relationships between caregivers and care recipients rather than being a direct result of using technology.¹³⁷² Philosophically, the issue may be defined as needing to make a decision between a responsibility to respect the autonomy of individuals and a consequentialist obligation to occasionally lie for the benefit of an individual.¹³⁷³ However, should these moral intuitions even be characterized as possibilities or as a dilemma? In actuality, both should be balanced and blended while providing care for the elderly.¹³⁷⁴ On the one hand, carers work to uphold the dignity and autonomy of those they care for. Empirically, not all elderly people are entirely autonomous, though. For instance, they could suffer from cognitive difficulties.

Care procedures must take into account this extraordinary factual reality, be sensitive to the challenges that seniors experience, and assist and care for them.¹³⁷⁵ For instance, if interacting with a robot enhances that person's quality of life, elder care may involve assisting a cognitively challenged person with tasks they are unable to perform on their own. The former, of course, should not be taken for granted. Important here is the word "if" - the caregiver needs to evaluate this, though. Deception may thus be used in routine medical procedures. Coeckelbergh

stresses "The deontological requirement to respect a person's transcendental autonomy and not to deceive is, and needs to be balanced with, the consequentialist aim to improve someone's well-being."¹³⁷⁶ This issue cannot have a definitive answer, at least not one that can be offered by dispassionate logic. While moral philosophy can help to define norms and principles in this situation, it is not enough. Ultimately, people who provide care for the elderly must choose the course of action that is most practical, taking into account the person's situation and skills as well as moral principles and perceptions, such as protecting autonomy and never exploiting a human as a simple instrument. ¹³⁷⁷

It should be noted that even if a person is not totally factually independent, caretakers may nevertheless choose to treat them as if they are, inspired by notions from Kant or other thinkers about the dignity and autonomy of the person.¹³⁷⁸ Caretakers aim to respect their desires and provide them options. This is a sort of deception since it gives the impression that the individual is completely independent while in reality, they are not. However, when actions are taken for the "benefit" and "happiness" of the people, caregivers may choose to use a paternalistic kind of deception, even if it involves lying or distorting the facts.¹³⁷⁹

Blackford and Rodogno concentrate on the connection between sentimentality and deception.¹³⁸⁰ Both believe that sentimentality is the over-emotionalization of certain items—in this example, care robots—resulting in a distortion of such objects. Rodogno refers to "the paradox of emotional fiction, which states that one empathizes with fictional characters while at the same time knowing they are fictional, to show: sentimentalizing care robots does not necessarily lead to misrepresenting them."¹³⁸¹ Even though robots are not sentient, older folks may be aware of this and yet exhibit feelings toward them. Rodogno also contends that rather than sentimentalization itself, the primary focus should be on the motivations behind

sentimentalization.¹³⁸² He contends that the impulse to arouse particular emotions is shown in the desire to sentimentalize things.¹³⁸³ He claims sentimentality "involves the attempt to use something to secure the desired feeling or emotional comfort."¹³⁸⁴ This urge for emotional solace is not always driven by cynical factors, such as social isolation. Therefore, it may not necessarily be an issue to sentimentalize care robots. It may be another example of selfish conduct, which sometimes entails misbehavior.¹³⁸⁵ Nevertheless, it creates a challenge "insofar as it disrespects other persons, values, or duties that the individuals have."¹³⁸⁶

6.B. Conclusion

The central question of this section asked, what are the ethical considerations in designing robots used in caring for aging populations? In the preceding sections, many ethical considerations have been identified; however, many more exist that were not identified in this chapter. The issue of the increasing scope of the aging population was described along with a lack of proportionate caregivers. The many diverse needs of the elderly were described with the subsequent rise to meet those needs with the advent of care robots. The importance of the foundational concepts of care ethics, relational autonomy, and human dignity was described to provide a starting point for engineers and designers to program the care robots. Then philosophical considerations of the virtues and values essential in care robots were told, and the argument of whether or not society can embed moral capacities within artificial systems. A framework for the ethical evolution of robots was highlighted so that the future of care robots can be carefully built and considered with ethics in mind. Finally, some ethical challenges of a care-centered framework were described, such as the implementation of human capabilities in robots and whether or not the carebots can have context-sensitivity in caring for the diverse needs of the elderly population. The notion of deception that the care robots can emit through

their care of the vulnerable elderly was also noted. According to this section, the employment of robots in elder care might result in a setting that is quite dystopian in nature if precautions are not made to address these ethical problems. Carefully developed guidelines and even legal restrictions on care robots' development and use could help prevent this dystopian future.

6. Conclusion

In review, the sixth chapter focused on an overall positive or utopian application of various data analytics-driven technologies within the specific context of the elderly population and end-of-life practices such as palliative care and hospice. The reason for the focus on end-of-life practices was due to the exponentially increasing scope of the aging population, an unmet need has risen, now more than ever, to care for the elderly population around the world. Thanks to data-analytics-driven technology, machines have changed how the world deals with death and allow people to live much longer than they could a century ago. A detailed historical and cultural background of death and dying outlined systematic changes towards death over time. Opportunities for technologies in end-of-life practices were explored, such as telemedicine and electronic learning. Challenges were described as well, such as the digital-divide, professional-patient relationship via technology, and fundamental issues such as the technological imperative vs. humanizing. The chapter continued with ethical considerations of using care robots in aging populations using care ethics. Concepts like artificial morality, relational autonomy, and dignity were described within this applied chapter of data analytics in healthcare.

CHAPTER 7: CONCLUSION

The beginning of the technological revolution has caused profound changes in science, society, culture, and many other facets of daily life. Technology and society are becoming progressively more entwined as time goes on, and this trend influences everyday life and healthcare. Through data analytics, technology such as big data information management, artificial intelligence (AI), machine learning (ML), and care robots are changing the healthcare landscape in ways that even a few years ago were not considered imaginable. These changes are being pushed by organizations and restless people who are not satisfied with the state of

healthcare. Clinical and health executives confront a never-ending list of obstacles, such as everexpanding medical skills, limited resources and personnel shortages, and a more varied mix of patients and customers whose requirements only increase. These technological changes are welcomed by many. This dissertation aimed to explore the ethics of these technological changes in healthcare and how this impacted reform and governance policy relating to data analytics.

Chapter 1 introduced the topic of data analytics and its impact on the healthcare world by outlining the dissertation's argumentative road map and the overall scope of the argument, emphasizing that data must be handled as a strategic asset. Without proper management, data often becomes duplicated, of low quality, and is unable to support the insightful conclusions that come from excellent data. Moreover, improper management and governance of data can cause numerous ethical concerns regarding privacy, confidentiality, and trust. Without proper management and governance, adverse outcomes will ensue, causing patient harm, financial woes, and endless legal paperwork. Therefore, ethical governance of data analytics in healthcare is essential. The thesis of this dissertation aimed to explain the ethics of governance of data analytics in healthcare and the challenges and opportunities that accompany this management. The chapters that followed the dissertation's introduction addressed the thesis by establishing the context through an approach to data ethics that focuses on the key ethical challenges of data analytics from a technological perspective and an applied analysis of data ethics in human enhancement and technological practices in end-of-life care.

After the introductory chapter, Chapters 2 and 3 established the background and context for the dissertation in how data analytics plays a role in the dissemination of healthcare. Blending finance and delivery, data analytics impacts care from an individual level all the way to organizational changes that can fundamentally reform healthcare as it is today. Chapter 2

described how data analytics could be used in a broader system context to initiate healthcare reform within the United States. Invoking its power to collect taxes, regulate, and facilitate commerce, the federal government of the United States has regularly approved the gathering and use of medical data to facilitate its activities since the 1950s.¹³⁸⁷ However, somewhat recently, in 2009, the government created a Federal Coordinating Council for Comparative Effectiveness Research, which established a data steward under health reform.¹³⁸⁸ The enormous volumes of healthcare data submitted to government agencies under Medicare, Medicaid, and countless other federal health services are collected by this data steward. Using the power to gather, combine, manage, and safeguard data while operating under open policy guidelines, this council can make strides in advancing population health while simultaneously exploring and researching a reformed health system.¹³⁸⁹

The analysis started with a background of healthcare reform in America. The first section of the chapter highlighted one of the critical ethical concepts that data analytics assists in exercising – distributive justice. In this section, the foundational theories of distributive justice were explored through an array of literature indicating the basic concerns of need and contribution within allocation problems. Implementation considerations of a new healthcare system were described by listing criteria for a better system, four conditions that must be met, and a universal access to basic care that technology will help achieve. A universal health insurance program may be the answer to the healthcare reform that America needs, but it has its own set of problems that must be considered. Many of these healthcare problems can be mitigated using data analytics, which can reduce costs, provide more personalized care access, lead to more accurate diagnostics, expedite care, and provide numerous other benefits for all. The second half of the chapter detailed data analytics' current and future impacts on the

healthcare landscape. A brief overview of what big data is and why it matters, including ethical challenges of security, privacy, and informed consent, was explored. Social policies of genomic data were described in depth, including genetic discrimination, common heritage, and distributive justice. The autonomy paradigm and cloud computing were used, respectively, to explain the constraints and potential of data analytics. Personalized care using predictive modeling and integrative sequencing were described in their effect on the finance and delivery of healthcare.

More data access to produce information pertaining to how the healthcare system operates will depend on the capacity of politicians to strike a balance between the requirement to protect genuine patient-clinician objectives on the one hand and the significance of health data to the general public on the other. In order to provide proof without jeopardizing these legitimate interests, it is necessary to consider whether data used for research, like that by the Federal Coordinating Council for Comparative Effectiveness Research, can be handled properly and securely. Chapter 3 took the context one step further and described the data battle that data analytics utilizes, creating tension between individual data privacy and public health surveillance for the common good. Concluding the section, the debate between individual data privacy and public health surveillance was at the heart of the argument – with ethical arguments on both sides. The section continued by detailing the imperative to influence policy as technology rapidly changes the norms of everyday life. Changes in policy in the form of ethical governance frameworks that enable regulation and monitoring of technologies that are a double-edged sword are crucial. Since healthcare is such a broad topic in scope, multiple frameworks are crucial to achieving ethical policies, processes, and procedures globally. Ethical governance frameworks are necessary as technological and societal changes continually reshape the global healthcare
landscape. These frameworks can withstand the dynamic nature of technology and society's relationship and thus provide a consistent solution to reprioritizing ethics into data analytics governance. Policy at not only a national but global level was conveyed, using examples such as the European Union and UNESCO. While data analytics can assist in creating biometrics governance frameworks, some concerns regarding the use and misuse of data analytics itself were addressed. The next chapter described some of those concerns in two senses: one practical, within genomics, and another theoretical, within the future of digital technologies themselves.

The prior chapters were primarily concerned with the advantages and possibilities that data analytics governance might provide the healthcare environment in terms of financing and delivery, as discussed in Chapter 2. Data analytics governance support was described at the micro level in Chapter 3, with the rights and autonomy of the person being compared to the governance at the macro level with regard to population health and the Common Good. Chapter 4 examined data analytics from a new angle and focused on the ethical issues and difficulties this technology might provide. Chapter 4 addressed the ethical and practical challenges of data analytics in healthcare within genomics and an applied analysis relating to human enhancement.

The social policy of genomic information was explored, including issues with genetic discrimination, common heritage, and further problems with distributive justice. The problems of data analytics within the autonomy paradigm and cloud computing were discussed in this section. Given that cloud computing decentralizes data storage, this problem proves difficult for ethical data analytics governance to overcome. Possessing centralized, secure data storage that is simple to manage and track back to in the case of a data breach is essential for upholding people's autonomy and privacy. The remainder of the chapter described a philosophical/theoretical foundation by discussing how technology has affected healthcare

275

through the digital revolution and its relationship to society. It also looked at the future of this dynamic relationship and how data analytics could significantly alter this relationship, leading to a dehumanized world via human enhancement. Concepts like technological determinism and the Precautionary Principle were addressed with a cautious mindset. Human enhancement was the subject of an applied examination of these notions, which raised ethical questions and served as the context for the discussion. Overall, Chapter 4 presented a more cautionary approach to digital technologies and gave perspective on how to move forward, keeping ethics a priority in the design and implementation of new data analytics technologies.

Chapter 5 began by looking at a few capabilities of artificial intelligence (AI) and its potential for revolutionizing healthcare. Since AI acts as the interpreter of all the data it receives using machine learning and algorithms, AI helps scientists, researchers, and healthcare data stewards, among many others, to understand and adapt to the clinical outcomes that the data forecasts. Philosophical concepts such as free will, moral disagreement, and moral agency were explored considering AI. The section concluded with an analysis of AI's challenges and opportunities in healthcare delivery. The chapter also described the foundations of health information management and its legal and ethical aspects. An applied analysis was detailed within systematic reviews and risk management. Finally, the chapter concluded with a call for health information technology governance – defined its need and core components, and highlighted concepts such as information stewardship and data quality.

If humankind is not mindful of the verge that human augmentation might lead it down, the conclusion of Chapter 4 looked at an example of how data analytics could be employed in a negative or dystopian manner. On the other hand, Chapter 6 concentrated on a generally positive or utopian use of different data analytics-driven technologies within the context of the older

276

population and end-of-life practices like palliative care and hospice. A thorough understanding of death and dying's historical and cultural context revealed how attitudes toward death have evolved. Opportunities for technology, such as telemedicine and electronic learning, were investigated in end-of-life practices. There were also descriptions of difficulties like the digital divide, the professional-patient interaction via technology, and basic problems like the technical imperative vs. humanizing. The last section of the chapter dealt with the moral ramifications of utilizing care robots with elderly people. This chapter on applied healthcare data analytics discussed concepts such as relational autonomy, artificial morality, and dignity.

This dissertation addressed the challenges of data analytics in this fast-developing environment as governance oversight develops effective policies organizationally, nationally, and globally and will help to create a more ethical, secure, and efficient future of healthcare for all humanity. By reprioritizing ethics at the forefront of governance, instead of having it as merely a component of data analytics governance, all those involved with these technologies will be able to produce and maintain ethical data analytics. Thus, each chapter in the dissertation attempted to look at the past, present, and future issues surrounding the governance of data analytics. Understanding the ethical concerns within these issues assisted in providing a forum for those issues that have yet to be addressed by data experts. Further, by locating the commonalities between these ethical implications, it becomes much more feasible to formulate and recommend solutions. Fortunately, this dissertation identified many ethical concerns and provided several recommendations for answers in the form of frameworks for data governance. While these issues are often significant in scope, globally even, usually a concrete solution will not always work practically in all countries. Therefore, this dissertation proposed frameworks for

277

an ethical data governance system that will help organizations and systems worldwide to build and utilize a more ethical data governance for all its stakeholders.

ENDNOTES

- ¹ Sara Rosenbaum, "Data Governance and Stewardship: Designing Data Stewardship Entities and Advancing Data Access," *Health Serv Res* 45, no. 5 Pt 2 (2010): 1443.
- ² Sara Rosenbaum, "Data Governance and Stewardship: Designing Data Stewardship Entities and Advancing Data Access," 1443.
- ³ Sara Rosenbaum, "Data Governance and Stewardship: Designing Data Stewardship Entities and Advancing Data Access," 1148.
- ⁴ Sara Rosenbaum, "Data Governance and Stewardship: Designing Data Stewardship Entities and Advancing Data Access," 1148.
- ⁵ Sara Rosenbaum, "Data Governance and Stewardship: Designing Data Stewardship Entities and Advancing Data Access," 1149.
- ⁶ Steven Camarota and Karen Zeigler, "Immigrant Population Hits Record 46.2 Million in November 2021," Center for Immigration Studies, https://cis.org/Camarota/Immigrant-Population-Hits-Record-462-Million-November-2021.

⁷ Henk ten Have and Bert Gordijn, *Bioethics in a European Perspective*, International Library of Ethics, Law, and the New Medicine (Dordrecht; Boston, MA: Kluwer Academic Publishers, 2001), 199-217.

- ⁸ Anne-Emanuelle Birn et al., "Struggles for National Health Reform in the United States," *American Journal of Public Health* 93, no. 1 (2003): 86-91.
- ⁹ Kant Patel, *Healthcare Politics and Policy in America* (London: Routledge, 2019), 34.
- ¹⁰ Kant Patel, *Healthcare Politics and Policy in America*, 35.
- ¹¹ Anne-Emanuelle Birn et al., "Struggles for National Health Reform in the United States," 86-91.
- ¹² David Blumenthal and James Morone, "The Lessons of Success—Revisiting the Medicare Story," *New England Journal of Medicine* 359, no. 22 (2008): 2384.
- ¹³ David Blumenthal and James Morone, "The Lessons of Success-Revisiting the Medicare Story," 2385.
- ¹⁴ Alan Derickson, *Health Security for All: Dreams of Universal Healthcare in America* (Baltimore: Johns Hopkins University Press, 2005), 42-78.
- ¹⁵ Alan Derickson, *Health Security for All: Dreams of Universal Healthcare in America*, 42-78.
- ¹⁶ Alan Derickson, *Health Security for All: Dreams of Universal Healthcare in America*, 42-78.
- ¹⁷ Alan Derickson, Health Security for All: Dreams of Universal Healthcare in America, 42-78.
- ¹⁸ Congressional Quarterly, "Health Insurance: Hearings on New Proposals," (Washington, DC: Congressional Quarterly Almanac, 1972), 541-544.
- ¹⁹ Congressional Quarterly, "Health Insurance: Hearings on New Proposals," 541-544.
- ²⁰ Flint J. Wainess, "The Ways and Means of National Health Care Reform, 1974 and Beyond," *Journal of Health Politics, Policy, and Law* 24, no. 2 (1999): 305-333.
- ²¹ Flint J. Wainess, "The Ways and Means of National Health Care Reform, 1974 and Beyond," 305-333.
- ²² Flint J. Wainess, "The Ways and Means of National Health Care Reform, 1974 and Beyond," 305-333.
- ²³ Center for Medicaid and CHIP Services, "May 2022 Medicaid and Chip Enrollment Trends Snapshot," ed. Centers for Medicare & Medicaid Services (2022).
- ²⁴ Flint J. Wainess, "The Ways and Means of National Health Care Reform, 1974 and Beyond," 305-333.
- ²⁵ Beatrix Hoffman, "Health Care Reform and Social Movements in the United States," *American Journal of Public Health* 93, no. 1 (2003): 80.
- ²⁶ Dona Cooper Hamilton and Charles V. Hamilton, "The Dual Agenda of African American Organizations since the New Deal: Social Welfare Policies and Civil Rights," *Political Science Quarterly* 107, no. 3 (1992): 245.
- ²⁷ Dona Cooper Hamilton and Charles V. Hamilton, "The Dual Agenda of African American Organizations since the New Deal: Social Welfare Policies and Civil Rights," 244-254.
- ²⁸ David McBride, From Tb to Aids: Epidemics among Urban Blacks since 1900, Suny Series in Afro-American Studies (Albany, NY: State University of New York Press, 1991), 154.

- ²⁹ Jill Quadagno, "Promoting Civil Rights through the Welfare State: How Medicare Integrated Southern Hospitals," Social Problems 47, no. 1 (2000): 68-89.
- ³⁰ Beatrix Hoffman, "Health Care Reform and Social Movements in the United States," 75.
- ³¹ Beatrix Hoffman, "Health Care Reform and Social Movements in the United States," 75.
- ³² Beatrix Hoffman, "Health Care Reform and Social Movements in the United States," 76.
- ³³ Beatrix Hoffman, "Health Care Reform and Social Movements in the United States," 76.
- ³⁴ Donald Paul Sullins, "The Healing Community: A Catholic Social Justice Critique of Modern Health Care," *The Linacre Quarterly* 81, no. 2 (2014): 172.
- ³⁵ Donald Paul Sullins, "The Healing Community: A Catholic Social Justice Critique of Modern Health Care," 172.
- ³⁶ Donald Paul Sullins, "The Healing Community: A Catholic Social Justice Critique of Modern Health Care," 172.
- ³⁷ Donald Paul Sullins, "The Healing Community: A Catholic Social Justice Critique of Modern Health Care," 172.
- ³⁸ United States Conference of Catholic Bishops, "Letter to Senate on Health Care Reform," news release, 2017.
- ³⁹ United States Conference of Catholic Bishops, "Letter to Senate on Health Care Reform."
- ⁴⁰ United States Conference of Catholic Bishops, "Letter to Senate on Health Care Reform."
- ⁴¹ Pope Benedict XVI, "Message of His Holiness Benedict Xvi," news release, November 15, 2010, https://www.vatican.va/content/benedict-xvi/en/letters/2010/documents/hf_ben-xvi_let_20101115_opsanitari.html.
- ⁴² Pope Benedict XVI, "Message of His Holiness Benedict Xvi."
- ⁴³ United States Conference of Catholic Bishops, *Ethical and Religious Directives for Catholic Health Care Services*, 5th ed. (Washington, DC: USCCB Publishing, 2009), 7.
- ⁴⁴ Scott McConnaha, "Catholic Teaching and Disparities in Care," *Health Progress* 87, no. 1 (2006): 48.
- ⁴⁵ Scott McConnaha, "Catholic Teaching and Disparities in Care," 48.
- ⁴⁶ Scott McConnaha, "Catholic Teaching and Disparities in Care," 49.
- ⁴⁷ Scott McConnaha, "Catholic Teaching and Disparities in Care," 49.
- ⁴⁸ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 7th ed. (New York: Oxford University Press, 2013), 271.
- ⁴⁹ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 271.
- ⁵⁰ Michael L. Morgan, *Classics of Moral and Political Theory*, 5th ed. (Indianapolis: Hackett Publishing Company, 2011), 1008-1009.
- ⁵¹ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 254.
- ⁵² Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 254.
- ⁵³ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics.*, 254.
- ⁵⁴ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 255.
- ⁵⁵ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 255.
- ⁵⁶ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 256.
- ⁵⁷ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 256.
- ⁵⁸ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 257.
- ⁵⁹ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 257.
- ⁶⁰ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 258.
- ⁶¹ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 258.
- ⁶² Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 256.
- ⁶³ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 256.
- ⁶⁴ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 256.
- ⁶⁵ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 259.
- ⁶⁶ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 259.
- ⁶⁷ Tom L. Beauchamp and James F. Childress, *Principles of Biomedical Ethics*, 260.
- ⁶⁸ Madison Powers and Ruth R. Faden, Social Justice: The Moral Foundations of Public Health and Health Policy, Issues in Biomedical Ethics (Oxford, NY: Oxford University Press, 2006), 37-41.

- ⁷² Harold W. Baillie, *Health Care Ethics*, 93.
- ⁷³ Harold W. Baillie, *Health Care Ethics*, 93.
- ⁷⁴ Harold W. Baillie, *Health Care Ethics*, 94.
- ⁷⁵ Harold W. Baillie, *Health Care Ethics*, 94.
- ⁷⁶ Harold W. Baillie, *Health Care Ethics*, 94.
- ⁷⁷ Harold W. Baillie, *Health Care Ethics*, 94.
- ⁷⁸ Harold W. Baillie, *Health Care Ethics*, 94.
- ⁷⁹ Harold W. Baillie, *Health Care Ethics*, 95.
- ⁸⁰ Harold W. Baillie, Health Care Ethics, 95.
- ⁸¹ Harold W. Baillie, Health Care Ethics, 95.
- ⁸² David F. Kelly, Gerard Magill, and Henk ten Have, *Contemporary Catholic Health Care Ethics*, 2nd ed. (Washington, DC: Georgetown University Press, 2013), 352.
- ⁸³ David F. Kelly, Gerard Magill, and Henk ten Have, Contemporary Catholic Health Care Ethics, 352.
- ⁸⁴ Harold W. Baillie, *Health Care Ethics*, 99.
- ⁸⁵ Harold W. Baillie, *Health Care Ethics*, 99.
- ⁸⁶ David F. Kelly, Gerard Magill, and Henk ten Have, Contemporary Catholic Health Care Ethics, 354.
- ⁸⁷ David F. Kelly, Gerard Magill, and Henk ten Have, *Contemporary Catholic Health Care Ethics*, 354.
- ⁸⁸ David F. Kelly, Gerard Magill, and Henk ten Have, Contemporary Catholic Health Care Ethics, 355.
- ⁸⁹ David F. Kelly, Gerard Magill, and Henk ten Have, Contemporary Catholic Health Care Ethics, 355.
- ⁹⁰ David F. Kelly, Gerard Magill, and Henk ten Have, Contemporary Catholic Health Care Ethics, 357.
- ⁹¹ David F. Kelly, Gerard Magill, and Henk ten Have, *Contemporary Catholic Health Care Ethics*, 357.
- ⁹² Centers for Disease Control and Prevention, "Cdc Reports on Uninsured in First Six Months of 2021," American Hospital Association, https://www.aha.org/news/headline/2021-11-17-cdc-reports-uninsured-first-sixmonths-2021.
- ⁹³ American College of Physicians, "Access to Health Care: Executive Summary," Annals of Internal Medicine 112, no. 9 (1990): 641.
- ⁹⁴ American College of Physicians, "Access to Health Care: Executive Summary," 641.
- ⁹⁵ The Henry J. Kaiser Family Foundation, "Key Facts About the Uninsured Population," in *Annual Social and Economic Supplement* (Menlo Park, CA 2017), 5.
- ⁹⁶ The Henry J. Kaiser Family Foundation, "Key Facts About the Uninsured Population," 5.
- ⁹⁷ The Henry J. Kaiser Family Foundation, "Key Facts About the Uninsured Population," 5.
- ⁹⁸ The Henry J. Kaiser Family Foundation, "Key Facts About the Uninsured Population," 5.
- ⁹⁹ The Henry J. Kaiser Family Foundation, "Key Facts About the Uninsured Population," 5.
- ¹⁰⁰ Jessica Pierce and George Randels, *Contemporary Bioethics: A Reader with Cases* (New York, NY: Oxford University Press, 2010), 72.
- ¹⁰¹ Jessica Pierce and George Randels, *Contemporary Bioethics: A Reader with Cases*, 72.
- ¹⁰² American College of Physicians, "Access to Health Care: Executive Summary," 641.
- ¹⁰³ American College of Physicians, "Access to Health Care: Executive Summary," 641.
- ¹⁰⁴ American College of Physicians, "Access to Health Care: Executive Summary," 641.
- ¹⁰⁵ Norton J. Greenberger, Nicholas E. Davies, and Euegene A. Hildreth, "Reforming Health Care in the United States," *BMJ: British Medical Journal (International Edition)* 303, no. 6812 (1991): 1216-1217.
- ¹⁰⁶ Norton J. Greenberger, Nicholas E. Davies, and Euegene A. Hildreth, "Reforming Health Care in the United States," 1216-1217.
- ¹⁰⁷ Norton J. Greenberger, Nicholas E. Davies, and Euegene A. Hildreth, "Reforming Health Care in the United States," 1216-1217.

⁶⁹ Madison Powers and Ruth R. Faden, Social Justice: The Moral Foundations of Public Health and Health Policy, 37-41.

⁷⁰ Harold W. Baillie, *Health Care Ethics*, 6th ed. (Boston: Prentice Hall, 2013), 93.

⁷¹ Harold W. Baillie, *Health Care Ethics*, 93.

¹¹¹ American College of Physicians, "Access to Health Care: Executive Summary," 642. ¹¹² American College of Physicians, "Access to Health Care: Executive Summary," 642. ¹¹³ American College of Physicians, "Access to Health Care: Executive Summary," 642. ¹¹⁴ Norman Daniels and James E. Sabin, Setting Limits Fairly: Learning to Share Resources for Health, 2nd ed. (Oxford ; New York: Oxford University Press, 2008), 10. ¹¹⁵ Norman Daniels and James E. Sabin, Setting Limits Fairly: Learning to Share Resources for Health, 10. ¹¹⁶ Norman Daniels and James E. Sabin, Setting Limits Fairly: Learning to Share Resources for Health, 11. ¹¹⁷ Norman Daniels and James E. Sabin, Setting Limits Fairly: Learning to Share Resources for Health, 11. ¹¹⁸ Norman Daniels and James E. Sabin, Setting Limits Fairly: Learning to Share Resources for Health, 12. ¹¹⁹ Norman Daniels and James E. Sabin, Setting Limits Fairly: Learning to Share Resources for Health, 12. ¹²⁰ Norman Daniels and James E. Sabin, Setting Limits Fairly: Learning to Share Resources for Health, 12. ¹²¹ Norman Daniels and James E. Sabin, Setting Limits Fairly: Learning to Share Resources for Health, 12. ¹²² American College of Physicians, "Access to Health Care: Executive Summary," 656. ¹²³ American College of Physicians, "Access to Health Care: Executive Summary," 656. ¹²⁴ American College of Physicians, "Access to Health Care: Executive Summary," 656. ¹²⁵ American College of Physicians, "Access to Health Care: Executive Summary," 656. ¹²⁶ American College of Physicians, "Access to Health Care: Executive Summary," 656. ¹²⁷ American College of Physicians, "Access to Health Care: Executive Summary," 656. ¹²⁸ American College of Physicians, "Access to Health Care: Executive Summary," 657. ¹²⁹ American College of Physicians, "Access to Health Care: Executive Summary," 657. ¹³⁰ American College of Physicians, "Access to Health Care: Executive Summary," 657. ¹³¹ American College of Physicians, "Access to Health Care: Executive Summary," 658. ¹³² American College of Physicians, "Access to Health Care: Executive Summary," 658. ¹³³ Nishita Mehta and Anil Pandit, "Concurrence of Big Data Analytics and Healthcare: A Systematic Review," International Journal of Medical Informatics 114 (2018): 57. ¹³⁴ Nishita Mehta and Anil Pandit, "Concurrence of Big Data Analytics and Healthcare: A Systematic Review," 57. ¹³⁵ Megan M. Miller, "Data as the New Oil: A Slippery Slope of Trade Secret Implications Greased by the California Consumer Privacy Act," Cybaris: An Intellectual Property Law Review 12, no. 1 (2021): 6. ¹³⁶ Samuel Greengard, "Weighing the Impact of Gdpr: The Eu Data Regulation Will Affect Computer, Internet, and Technology Usage within and Outside the Eu; How It Will Play out Remains to Be Seen," Communications of the ACM 61, no. 11 (2018): 16. ¹³⁷ Nancy Davis Kho, "Gdpr for Content Design, Development, and Deployment: 'Gdpr Is a Really Good Opportunity to Deep Clean under the Sofa and Get Rid of All the Dust Bunnies.'," EContent 41, no. 3 (2018): 22. ¹³⁸ Nancy Davis Kho, "Gdpr for Content Design, Development, and Deployment: 'Gdpr Is a Really Good Opportunity to Deep Clean under the Sofa and Get Rid of All the Dust Bunnies.'," 22. ¹³⁹ Nancy Davis Kho, "Gdpr for Content Design, Development, and Deployment: 'Gdpr Is a Really Good Opportunity to Deep Clean under the Sofa and Get Rid of All the Dust Bunnies.'," 22. ¹⁴⁰ Nancy Davis Kho, "Gdpr for Content Design, Development, and Deployment: 'Gdpr Is a Really Good Opportunity to Deep Clean under the Sofa and Get Rid of All the Dust Bunnies.'," 22. ¹⁴¹ Alan Rodger, "Data Privacy Laws: Cutting the Red Tape," (Ovum, 2015), 1. ¹⁴² Mark Thompson, "How Have Companies Adapted to the Gdpr?," Information Age, 11 March 2020, 1. ¹⁴³ Mark Thompson, "How Have Companies Adapted to the Gdpr?," 1. ¹⁴⁴ Mark Thompson, "How Have Companies Adapted to the Gdpr?," 1. ¹⁴⁵ Nancy Davis Kho, "Gdpr for Content Design, Development, and Deployment: 'Gdpr Is a Really Good Opportunity to Deep Clean under the Sofa and Get Rid of All the Dust Bunnies.'," 22.

¹⁰⁸ American College of Physicians, "Access to Health Care: Executive Summary," 642.
 ¹⁰⁹ American College of Physicians, "Access to Health Care: Executive Summary," 642.
 ¹¹⁰ American College of Physicians, "Access to Health Care: Executive Summary," 642.

- ¹⁴⁶ Dixie B. Baker et al., "Privacy, Fairness, and Respect for Individuals," *eGEMS (Generating Evidence & Methods to Improve Patient Outcomes)* 4, no. 2 (2016): 8.
- ¹⁴⁷ Dixie B. Baker et al., "Privacy, Fairness, and Respect for Individuals," 8.
- ¹⁴⁸ Dixie B. Baker et al., "Privacy, Fairness, and Respect for Individuals," 8.
- ¹⁴⁹ Dixie B. Baker et al., "Privacy, Fairness, and Respect for Individuals," 8.
- ¹⁵⁰ Dixie B. Baker et al., "Privacy, Fairness, and Respect for Individuals," 8.
- ¹⁵¹ Kenneth W. Goodman, *Ethics, Medicine, and Information Technology: Intelligent Machines and the Transformation of Health Care* (Cambridge; New York, NY: Cambridge University Press, 2015), 44.
- ¹⁵² Kenneth W. Goodman, *Ethics, Medicine, and Information Technology: Intelligent Machines and the Transformation of Health Care*, 44.
- ¹⁵³ Kenneth W. Goodman, *Ethics, Medicine, and Information Technology: Intelligent Machines and the Transformation of Health Care*, 44.
- ¹⁵⁴ Kenneth W. Goodman, *Ethics, Medicine, and Information Technology: Intelligent Machines and the Transformation of Health Care*, 44.
- ¹⁵⁵ Katherine O Keefe and Daragh O'Brien, *Ethical Data and Information Management: Concepts, Tools and Methods*, 1 Edition. ed. (New York, NY: Kogan Page Ltd, 2018), 216.
- ¹⁵⁶ Katherine O Keefe and Daragh O'Brien, *Ethical Data and Information Management: Concepts, Tools and Methods*, 216.
- ¹⁵⁷ Katherine O Keefe and Daragh O'Brien, *Ethical Data and Information Management: Concepts, Tools and Methods*, 216.
- ¹⁵⁸ Katherine O Keefe and Daragh O'Brien, *Ethical Data and Information Management: Concepts, Tools and Methods*, 216.
- ¹⁵⁹ Katherine O Keefe and Daragh O'Brien, *Ethical Data and Information Management: Concepts, Tools and Methods*, 216.
- ¹⁶⁰ Katherine O Keefe and Daragh O'Brien, *Ethical Data and Information Management: Concepts, Tools and Methods*, 217.
- ¹⁶¹ Kenneth W. Goodman, *Ethics, Medicine, and Information Technology: Intelligent Machines and the Transformation of Health Care*, 129.
- ¹⁶² Kenneth W. Goodman, *Ethics, Medicine, and Information Technology: Intelligent Machines and the Transformation of Health Care*, 129.
- ¹⁶³ Kenneth W. Goodman, *Ethics, Medicine, and Information Technology: Intelligent Machines and the Transformation of Health Care*, 130.
- ¹⁶⁴ Kenneth W. Goodman, *Ethics, Medicine, and Information Technology: Intelligent Machines and the Transformation of Health Care*, 130.
- ¹⁶⁵ Vivek Navale and Philip E. Bourne, "Cloud Computing Applications for Biomedical Science: A Perspective," *PLoS Computational Biology* 14, no. 6 (2018): 2.
- ¹⁶⁶ Vivek Navale and Philip E. Bourne, "Cloud Computing Applications for Biomedical Science: A Perspective," 1.
- ¹⁶⁷ Peter Mell and Timothy Grance, "The Nist Definition of Cloud Computing," (Gaithersburg, MD: National Institute of Standards and Technology - U.S. Department of Commerce, 2011), 2.
- ¹⁶⁸ Vivek Navale and Philip E. Bourne, "Cloud Computing Applications for Biomedical Science: A Perspective," 2.
- ¹⁶⁹ Vivek Navale and Philip E. Bourne, "Cloud Computing Applications for Biomedical Science: A Perspective," 2.
- ¹⁷⁰ Vivek Navale and Philip E. Bourne, "Cloud Computing Applications for Biomedical Science: A Perspective," 2.
- ¹⁷¹ Vivek Navale and Philip E. Bourne, "Cloud Computing Applications for Biomedical Science: A Perspective," 2.
- ¹⁷² Chaowei Yang, Yan Xu, and Douglas Nebert, "Redefining the Possibility of Digital Earth and Geosciences with Spatial Cloud Computing," *International Journal of Digital Earth* 6, no. 4 (2013): 297.
- ¹⁷³ Chaowei Yang et al., "Utilizing Cloud Computing to Address Big Geospatial Data Challenges," Computers, Environment & Urban Systems 61 (2017): 120.
- ¹⁷⁴ Chaowei Yang et al., "Utilizing Cloud Computing to Address Big Geospatial Data Challenges," 120.

¹⁷⁵ Chun-Wei Tsai et al., "Metaheuristic Algorithms for Healthcare: Open Issues and Challenges," *Computers & Electrical Engineering* 53 (2016): 430.

- ¹⁷⁹ Vivek Navale and Philip E. Bourne, "Cloud Computing Applications for Biomedical Science: A Perspective," 6.
- ¹⁸⁰ Vivek Navale and Philip E. Bourne, "Cloud Computing Applications for Biomedical Science: A Perspective," 6.
- ¹⁸¹ Vivek Navale and Philip E. Bourne, "Cloud Computing Applications for Biomedical Science: A Perspective," 6.

Perspective," 6.

- ¹⁸⁴ Vivek Navale and Philip E. Bourne, "Cloud Computing Applications for Biomedical Science: A Perspective," 6.
- ¹⁸⁵ Ashwin Belle et al., "Big Data Analytics in Healthcare," *BioMed Research International* 2015 (2015): 7.
- ¹⁸⁶ Ashwin Belle et al., "Big Data Analytics in Healthcare," 7.
- ¹⁸⁷ Mary J. Wills, "Decisions through Data: Analytics in Healthcare," *Journal of Healthcare Management* 59, no. 4 (2014): 257.
- ¹⁸⁸ Mary J. Wills, "Decisions through Data: Analytics in Healthcare," 257.
- ¹⁸⁹ Mary J. Wills, "Decisions through Data: Analytics in Healthcare," 258.
- ¹⁹⁰ Mary J. Wills, "Decisions through Data: Analytics in Healthcare," 258.
- ¹⁹¹ Jennifer R. Popovic, "Distributed Data Networks: A Blueprint for Big Data Sharing and Healthcare Analytics," *Annals of the New York Academy of Sciences* 1387, no. 1 (2017): 108.
- ¹⁹² Jennifer R. Popovic, "Distributed Data Networks: A Blueprint for Big Data Sharing and Healthcare Analytics," 108.
- ¹⁹³ Jennifer R. Popovic, "Distributed Data Networks: A Blueprint for Big Data Sharing and Healthcare Analytics," 109.
- ¹⁹⁴ Jennifer R. Popovic, "Distributed Data Networks: A Blueprint for Big Data Sharing and Healthcare Analytics," 109.
- ¹⁹⁵ Jennifer R. Popovic, "Distributed Data Networks: A Blueprint for Big Data Sharing and Healthcare Analytics," 109.
- ¹⁹⁶ Jennifer R. Popovic, "Distributed Data Networks: A Blueprint for Big Data Sharing and Healthcare Analytics," 109.
- ¹⁹⁷ Ivan J. Barrick, *Transforming Health Care Management: Integrating Technology Strategies* (Sudbury, MA: Jones and Bartlett Publishers, LLC, 2009), 151.
- ¹⁹⁸ Ivan J. Barrick, *Transforming Health Care Management: Integrating Technology Strategies*, 152.
- ¹⁹⁹ Ivan J. Barrick, Transforming Health Care Management: Integrating Technology Strategies, 152.
- ²⁰⁰ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective* (Chicago, IL: Health Administration Press, 2013), 94.
- ²⁰¹ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 94.
- ²⁰² Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 94.
- ²⁰³ Mark A Musen, Blackford Middleton, and Robert A Greenes, "Clinical Decision-Support Systems," in *Biomedical Informatics*, ed. Edward H. Shortliffe and James J. Cimino (New York, NY: Springer, 2014), 643.
- ²⁰⁴ Mark A Musen, Blackford Middleton, and Robert A Greenes, "Clinical Decision-Support Systems," 643.
- ²⁰⁵ Mark A Musen, Blackford Middleton, and Robert A Greenes, "Clinical Decision-Support Systems," 643.
- ²⁰⁶ Mark A Musen, Blackford Middleton, and Robert A Greenes, "Clinical Decision-Support Systems," 644.
- ²⁰⁷ Mark A Musen, Blackford Middleton, and Robert A Greenes, "Clinical Decision-Support Systems," 644.
- ²⁰⁸ Ana D. Cleveland and Donald B. Cleveland, *Health Informatics for Medical Librarians* (New York, NY: Neal-Schuman Publishers, 2009), 65.
- ²⁰⁹ Ana D. Cleveland and Donald B. Cleveland, *Health Informatics for Medical Librarians*, 66.

¹⁷⁶ Chun-Wei Tsai et al., "Metaheuristic Algorithms for Healthcare: Open Issues and Challenges," 430.

¹⁷⁷ Chun-Wei Tsai et al., "Metaheuristic Algorithms for Healthcare: Open Issues and Challenges," 430.

¹⁷⁸ Chun-Wei Tsai et al., "Metaheuristic Algorithms for Healthcare: Open Issues and Challenges," 430.

 ¹⁸² Vivek Navale and Philip E. Bourne, "Cloud Computing Applications for Biomedical Science: A Perspective," 6.
 ¹⁸³ Vivek Navale and Philip E. Bourne, "Cloud Computing Applications for Biomedical Science: A

²¹⁰ Ana D. Cleveland and Donald B. Cleveland, *Health Informatics for Medical Librarians*, 66.

- ²¹³ Ana D. Cleveland and Donald B. Cleveland, *Health Informatics for Medical Librarians*, 243.
- ²¹⁴ Ana D. Cleveland and Donald B. Cleveland, *Health Informatics for Medical Librarians*, 243.
- ²¹⁵ Ashwin Belle et al., "Big Data Analytics in Healthcare," 2.
- ²¹⁶ Ashwin Belle et al., "Big Data Analytics in Healthcare," 2.
- ²¹⁷ Ashwin Belle et al., "Big Data Analytics in Healthcare," 2.
- ²¹⁸ Leroy Hood and Stephen H. Friend, "Predictive, Personalized, Preventive, Participatory (P4) Cancer Medicine," *Nature Reviews Clinical Oncology* 8, no. 3 (2011): 184-187.
- ²¹⁹ Ashwin Belle et al., "Big Data Analytics in Healthcare," 2.
- ²²⁰ Rui Chen et al., "Personal Omics Profiling Reveals Dynamic Molecular and Medical Phenotypes," *Cell* 148, no. 6 (2012): 1293-1307.
- ²²¹ Ashwin Belle et al., "Big Data Analytics in Healthcare," 2.
- ²²² Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective* (Chicago, IL: Health Administration Press, 2012), 253.
- ²²³ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 253.
- ²²⁴ Cinnamon S. Bloss, Nicholas J. Schork, and Eric J. Topol, "Effect of Direct-to-Consumer Genomewide Profiling to Assess Disease Risk," *The New England Journal of Medicine* 364, no. 6 (2011): 524-534.
- ²²⁵ Cinnamon S. Bloss, Nicholas J. Schork, and Eric J. Topol, "Effect of Direct-to-Consumer Genomewide Profiling to Assess Disease Risk," 524-534.
- ²²⁶ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 253.
- ²²⁷ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 253.
- ²²⁸ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 253.
- ²²⁹ James G. Anderson and Kenneth W. Goodman, *Ethics and Information Technology: A Case-Based Approach to a Health Care System in Transition*, ed. Kathyrn J. Hannah and Marion J. Ball (New York, NY: Springer, 2002), 119.
- ²³⁰ James G. Anderson and Kenneth W. Goodman, *Ethics and Information Technology: A Case-Based Approach to a Health Care System in Transition*, 119.
- ²³¹ World Medical Association, "Wma to Draw up Health Database Guidelines," https://www.wma.net/news-post/wma-to-draw-up-health-database-guidelines/.
- ²³² Lynn Harold Vogel, "Management of Information in Healthcare Organizations," in *Biomedical Informatics*, ed. Edward H. Shortliffe and James J. Cimino (Verlag London: Springer, 2014), 455.
- ²³³ Lynn Harold Vogel, "Management of Information in Healthcare Organizations," 455.
- ²³⁴ Lynn Harold Vogel, "Management of Information in Healthcare Organizations," 455.
- ²³⁵ Lynn Harold Vogel, "Management of Information in Healthcare Organizations," 455.
- ²³⁶ Lynn Harold Vogel, "Management of Information in Healthcare Organizations," 455.
- ²³⁷ Lynn Harold Vogel, "Management of Information in Healthcare Organizations," 455.
- ²³⁸ Lynn Harold Vogel, "Management of Information in Healthcare Organizations," 456.
- ²³⁹ Lynn Harold Vogel, "Management of Information in Healthcare Organizations," 456.
- ²⁴⁰ Chiranjeev Sanyal et al., "Economic Evaluations of Ehealth Technologies: A Systematic Review," *PLoS ONE* 13, no. 6 (2018): 2.
- ²⁴¹ Chiranjeev Sanyal et al., "Economic Evaluations of Ehealth Technologies: A Systematic Review," 2.
- ²⁴² Chiranjeev Sanyal et al., "Economic Evaluations of Ehealth Technologies: A Systematic Review," 2.
- ²⁴³ Chiranjeev Sanyal et al., "Economic Evaluations of Ehealth Technologies: A Systematic Review," 2.

²¹¹ Ana D. Cleveland and Donald B. Cleveland, *Health Informatics for Medical Librarians*, 242.

²¹² Ana D. Cleveland and Donald B. Cleveland, *Health Informatics for Medical Librarians*, 242.

- ²⁴⁴ Chiranjeev Sanyal et al., "Economic Evaluations of Ehealth Technologies: A Systematic Review," 2.
- ²⁴⁵ Chiranjeev Sanyal et al., "Economic Evaluations of Ehealth Technologies: A Systematic Review," 2.

- ²⁴⁷ Leila Agha, "The Effects of Health Information Technology on the Costs and Quality of Medical Care," *Journal of Health Economics* 34 (2014): 2.
- ²⁴⁸ Leila Agha, "The Effects of Health Information Technology on the Costs and Quality of Medical Care," 2.
- ²⁴⁹ Leila Agha, "The Effects of Health Information Technology on the Costs and Quality of Medical Care," 3.
- ²⁵⁰ Leila Agha, "The Effects of Health Information Technology on the Costs and Quality of Medical Care," 3.
- ²⁵¹ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 321.
- ²⁵² Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 321.
- ²⁵³ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 321.
- ²⁵⁴ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 321.
- ²⁵⁵ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 321.
- ²⁵⁶ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 321.
- ²⁵⁷ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 321.
- ²⁵⁸ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 321.
- ²⁵⁹ Michael J. Quinn, *Ethics for the Information Age*, 6th ed. (Pearson Education Inc., 2015), 462.
- ²⁶⁰ Michael J. Quinn, *Ethics for the Information Age*, 462.
- ²⁶¹ Michael J. Quinn, Ethics for the Information Age, 462.
- ²⁶² Michael J. Quinn, Ethics for the Information Age, 462.
- ²⁶³ Michael J. Quinn, *Ethics for the Information Age*, 462.
- ²⁶⁴ Michael J. Quinn, *Ethics for the Information Age*, 463.
- ²⁶⁵ Michael J. Quinn, Ethics for the Information Age, 463.
- ²⁶⁶ Michael J. Quinn, *Ethics for the Information Age*, 463.
- ²⁶⁷ Michael J. Quinn, *Ethics for the Information Age*, 463.
- ²⁶⁸ DAMA International, Dama Dmbok Data Management Body of Knowledge (New Jersey: Technics Publications, 2017), 381.
- ²⁶⁹ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management: Concepts, Tools and Methods* (2018), 112.
- ²⁷⁰ Katherine O Keefe and Daragh O'Brien, *Ethical Data and Information Management: Concepts, Tools and Methods*, 112.
- ²⁷¹ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management : Concepts, Tools and Methods*, 112.
- ²⁷² DAMA International, Dama Dmbok Data Management Body of Knowledge, 381-385.
- ²⁷³ W. Edwards Deming, *Out of the Crisis* (Cambridge, Massachsetts: The MIT Press, 2018), 72.
- ²⁷⁴ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management : Concepts, Tools and Methods*, 114.
- ²⁷⁵ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management : Concepts, Tools and Methods*, 114.
- ²⁷⁶ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management : Concepts, Tools and Methods*, 114.

²⁴⁶ Chiranjeev Sanyal et al., "Economic Evaluations of Ehealth Technologies: A Systematic Review," 2.

- ²⁷⁷ Katherine O Keefe and Daragh O'Brien, *Ethical Data and Information Management: Concepts, Tools and Methods*, 115.
- ²⁷⁸ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management : Concepts, Tools and Methods*, 115.
- ²⁷⁹ Katherine O Keefe and Daragh O'Brien, *Ethical Data and Information Management: Concepts, Tools and Methods*, 115.
- ²⁸⁰ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management : Concepts, Tools and Methods*, 115.
- ²⁸¹ Sara Rosenbaum, "Data Governance and Stewardship: Designing Data Stewardship Entities and Advancing Data Access," 1443.
- ²⁸² Sara Rosenbaum, "Data Governance and Stewardship: Designing Data Stewardship Entities and Advancing Data Access," 1443.
- ²⁸³ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," *Cambridge Quarterly of Healthcare Ethics* 20, no. 4 (2011): 533.
- ²⁸⁴ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 533.
- ²⁸⁵ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 533.
- ²⁸⁶ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 534.
- ²⁸⁷ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 534.
- ²⁸⁸ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 534.
- ²⁸⁹ David C. Thomasma, "Evolving Bioethics and International Human Rights," in Autonomy and Human Rights in Health Care: An International Perspective, ed. David N. Weisstub and Guillermo Diaz Pintos (Dordrecht: Springer, 2008), 14.
- ²⁹⁰ David C. Thomasma, "Evolving Bioethics and International Human Rights," 14.
- ²⁹¹ David C. Thomasma, "Evolving Bioethics and International Human Rights," 15.
- ²⁹² David C. Thomasma, "Evolving Bioethics and International Human Rights," 15.
- ²⁹³ David C. Thomasma, "Evolving Bioethics and International Human Rights," 15.
- ²⁹⁴ David C. Thomasma, "Evolving Bioethics and International Human Rights," 15.
- ²⁹⁵ David C. Thomasma, "Evolving Bioethics and International Human Rights," 15.
- ²⁹⁶ David C. Thomasma, "Evolving Bioethics and International Human Rights," 15.
- ²⁹⁷ Jacob D. Rendtorff, "The Limitations and Accomplishments of Autonomy as a Basic Principle in Bioethics and Biolaw," in *Autonomy and Human Rights in Health Care: An International Perspective*, ed. David N. Weisstub and Guillermo Diaz Pintos (Dordrecht: Springer, 2008), 80.
- ²⁹⁸ Jacob D. Rendtorff, "The Limitations and Accomplishments of Autonomy as a Basic Principle in Bioethics and Biolaw," 80.
- ²⁹⁹ Jacob D. Rendtorff, "The Limitations and Accomplishments of Autonomy as a Basic Principle in Bioethics and Biolaw," 80.
- ³⁰⁰ Jacob D. Rendtorff, "The Limitations and Accomplishments of Autonomy as a Basic Principle in Bioethics and Biolaw," 81.
- ³⁰¹ Jacob D. Rendtorff, "The Limitations and Accomplishments of Autonomy as a Basic Principle in Bioethics and Biolaw," 81.
- ³⁰² Warren T. Reich, ed. *Encyclopedia of Bioethics* (Washington, DC: Simon & Schuster, 1995), 219.
- ³⁰³ Jacob D. Rendtorff, "The Limitations and Accomplishments of Autonomy as a Basic Principle in Bioethics and Biolaw," 81.
- ³⁰⁴ Henk ten Have, *Global Bioethics: An Introduction* (London; New York, NY: Routledge, Taylor & Francis Group, 2016), 39.
- ³⁰⁵ Renée C. Fox, *The Sociology of Medicine : A Participant Observer's View*, Prentice Hall Foundations of Modern Sociology Series (Englewood Cliffs, N.J.: Prentice Hall, 1989), 72.
- ³⁰⁶ Henk ten Have, *Global Bioethics: An Introduction*, 39.
- ³⁰⁷ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 534.

- ³⁰⁸ Alfred I. Tauber, *Patient Autonomy and the Ethics of Responsibility*, Basic Bioethics (Cambridge, MA: MIT Press, 2005), 68.
- ³⁰⁹ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 534.
- ³¹⁰ Bartha Maria Knoppers and Ruth Chadwick, "Human Genetic Research: Emerging Trends in Ethics," *Nature Reviews Genetics* 6, no. 1 (2005): 75-79.
- ³¹¹ Daniel Callahan, "Individual Good and Common Good: A Communitarian Approach to Bioethics," *Perspectives in Biology and Medicine* 46, no. 4 (2003): 496.
- ³¹² Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 542.
- ³¹³ Tuija Takala, "Setting a Dangerous Precedent: Ethical Issues in Human Genetic Database Research," *Medical Law International* 8, no. 2 (2007): 105-137.
- ³¹⁴ Aristotle, *The Politics*, Cambridge Texts in the History of Political Thought (Cambridge: Cambridge University Press, 1988), 1253.
- ³¹⁵ Louis Dupré, "The Common Good and the Open Society," The Review of Politics 55, no. 4 (1993): 687.
- ³¹⁶ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 535-536.
- ³¹⁷ Thomas W. Smith, "Aristotle on the Conditions for and Limits of the Common Good," *The American Political Science Review* 93, no. 3 (1999): 625-636.
- ³¹⁸ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 536.
- ³¹⁹ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 536.
- ³²⁰ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 536.
- ³²¹ Margit Sutrop, "Changing Ethical Frameworks: From Individual Rights to the Common Good?," 542.
- ³²² Andrej Zwitter, "Big Data Ethics," Big Data & Society 1, no. 2 (2014): 2.
- ³²³ Andrej Zwitter, "Big Data Ethics," 2.
- ³²⁴ Andrej Zwitter, "Big Data Ethics," 2.
- ³²⁵ Andrej Zwitter, "Big Data Ethics," 2.
- ³²⁶ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," *Health Information Science and Systems* 2, no. 3 (2014): 1.
- ³²⁷ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 1.
- ³²⁸ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 3.
- ³²⁹ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 3.
- ³³⁰ J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," *Health Affairs* 33, no. 7 (2014): 1116.
- ³³¹ J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," 1116.
- ³³² J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," 1116.
- ³³³ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 3.
- ³³⁴ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 3.
- ³³⁵ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 3.
- ³³⁶ J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," 1116.
- ³³⁷ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 3.
- ³³⁸ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 3.
- ³³⁹ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 4.
- ³⁴⁰ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 4.
- ³⁴¹ J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," 1116.
- ³⁴² Andrew McAfee and Erik Brynjolfsson, "Big Data: The Management Revolution; Exploiting Vast New Flows of Information Can Radically Improve Your Company's Performance. But First You'll Have to Change Your Decision-Making Culture," *Harvard Business Review* 90, no. 10 (2012): 61-68.

- ³⁴³ J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," 1116.
- ³⁴⁴ J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," 1116.
- ³⁴⁵ J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," 1116.
- ³⁴⁶ J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," 1116.
- ³⁴⁷ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 3.
- ³⁴⁸ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 3.
- ³⁴⁹ Wullianallur Raghupathi and Viju Raghupathi, "Big Data Analytics in Healthcare: Promise and Potential," 3.
- ³⁵⁰ Brent D. Mittelstadt and Luciano Floridi, "The Ethics of Big Data: Current and Foreseeable Issues in Biomedical Contexts," *Science and Engineering Ethics* 22, no. 2 (2016): 312.
- ³⁵¹ Ryan Gallagher, "Software That Tracks People on Social Media Created by Defense Firm," *The Guardian* 2013, 1.
- ³⁵² Ryan Gallagher, "Software That Tracks People on Social Media Created by Defense Firm," 1.
- ³⁵³ Andrej Zwitter, "Big Data Ethics," 4.
- ³⁵⁴ Andrej Zwitter, "Big Data Ethics," 4.
- ³⁵⁵ Andrej Zwitter, "Big Data Ethics," 4.
- ³⁵⁶ Andrej Zwitter, "Big Data Ethics," 4.
- ³⁵⁷ J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," 1119.
- ³⁵⁸ J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," 1119.
- ³⁵⁹ J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," 1119.
- ³⁶⁰ Jeantine E. Lunshof et al., "From Genetic Privacy to Open Consent," Nature Reviews Genetics 9, no. 5 (2008): 406-411.
- ³⁶¹ J. Roski, G. W. Bo-Linn, and T. A. Andrews, "Creating Value in Health Care through Big Data: Opportunities and Policy Implications," 1119.
- ³⁶² Omer Tene and Jules Polonetsky, "Privacy in the Age of Big Data: A Time for Big Decisions," Stanford Law Review Online 64 (2011): 63.
- ³⁶³ Brent D. Mittelstadt and Luciano Floridi, "The Ethics of Big Data: Current and Foreseeable Issues in Biomedical Contexts," 314.
- ³⁶⁴ Brent D. Mittelstadt and Luciano Floridi, "The Ethics of Big Data: Current and Foreseeable Issues in Biomedical Contexts," 315.
- ³⁶⁵ Brent D. Mittelstadt and Luciano Floridi, "The Ethics of Big Data: Current and Foreseeable Issues in Biomedical Contexts," 315.
- ³⁶⁶ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," *American Journal of Public Health* 102, no. 1 (2012): 38.
- ³⁶⁷ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 38.
- ³⁶⁸ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security* (New York, NY: Springer, 2010), 34.
- ³⁶⁹ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*,
 34.
- ³⁷⁰ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*, 34.

- ³⁷¹ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*,
 34.
- ³⁷² Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*, 34.
- ³⁷³ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*, 34.
- ³⁷⁴ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*,
 34.
- ³⁷⁵ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*,
 35.
- ³⁷⁶ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*,
 35.
- ³⁷⁷ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*,
 35.
- ³⁷⁸ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*, 35.
- ³⁷⁹ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*, 35.
- ³⁸⁰ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*, 35.
- ³⁸¹ Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*, 35.
- ³⁸² Charles A. Shoniregun, Kudakwashe Dube, and Fredrick Mtenzi, *Electronic Healthcare Information Security*, 35.
- ³⁸³ Lisa M. Lee and Stephen B. Thacker, "Public Health Surveillance and Knowing About Health in the Context of Growing Sources of Health Data," *American Journal of Preventive Medicine* 41, no. 6 (2011): 636.
- ³⁸⁴ Centers for Disease Control and Prevention, "Distinguishing Public Health Research and Public Health Nonresearch," ed. Department of Health & Human Services (2010), 1.
- ³⁸⁵ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 38.
- ³⁸⁶ Stephen B. Thacker, "Historical Development," in *Principles and Practice of Public Health Surveillance*, ed. Lisa M. Lee, Steven M. Teutsch, and Michael E. St. Louis (New York: Oxford University Press, 2010), 1-17.
- ³⁸⁷ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 41.
- ³⁸⁸ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 41.
- ³⁸⁹ Lisa M. Lee and Lawrence O. Gostin, "Ethical Collection, Storage, and Use of Public Health Data: A Proposal for a National Privacy Protection," *JAMA: The Journal of the American Medical Association* 302, no. 1 (2009): 82-84.
- ³⁹⁰ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 41.
- ³⁹¹ Centers for Disease Control and Prevention, "Distinguishing Public Health Research and Public Health Nonresearch," 1.
- ³⁹² Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 41.
- ³⁹³ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 41.

- ³⁹⁴ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 41.
- ³⁹⁵ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ³⁹⁶ SL Groseclose, RR German, and P. Nsubuga, "Evaluating Public Health Surveillance," in *Principles and Practice of Public Health Surveillance*, ed. Lisa M. Lee, et al. (New York: Oxford University Press, 2010), 166.
- ³⁹⁷ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ³⁹⁸ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ³⁹⁹ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ⁴⁰⁰ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ⁴⁰¹ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ⁴⁰² Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ⁴⁰³ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ⁴⁰⁴ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ⁴⁰⁵ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ⁴⁰⁶ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ⁴⁰⁷ Lisa M. Lee and Lawrence O. Gostin, "Ethical Collection, Storage, and Use of Public Health Data: A Proposal for a National Privacy Protection," 82-84.
- ⁴⁰⁸ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ⁴⁰⁹ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 42.
- ⁴¹⁰ Amy L. Fairchild et al., Searching Eyes : Privacy, the State, and Disease Surveillance in America (Berkeley: University of California Press, 2007), 72.
- ⁴¹¹ William H. Foege, Robert C. Hogan, and Ladene H. Newton, "Surveillance Projects for Selected Diseases," *International Journal of Epidemiology* 5, no. 1 (1976): 29-37.
- ⁴¹² Martha F. Rogers, Mary Glenn Fowler, and Mary Lou Lindegren, "Revised Recommendations for Hiv Screening of Pregnant Women," *Morbidity and Mortality Weekly Report: Recommendations and Reports* 50, no. RR-19 (2001): 59-86.
- ⁴¹³ Martha F. Rogers, Mary Glenn Fowler, and Mary Lou Lindegren, "Revised Recommendations for Hiv Screening of Pregnant Women," 59-86.
- ⁴¹⁴ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 43.
- ⁴¹⁵ Lisa M. Lee, Charles M. Heilig, and Angela White, "Ethical Justification for Conducting Public Health Surveillance without Patient Consent," 43.
- ⁴¹⁶ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," *Review of Policy Research* 29, no. 1 (2012): 21.

- ⁴¹⁷ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 21.
- ⁴¹⁸ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 21.
- ⁴¹⁹ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 21.
- ⁴²⁰ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 21.
- ⁴²¹ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," *Bioethics* 22, no. 9 (2008): 496.
- ⁴²² Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 496.
- ⁴²³ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 496.
- ⁴²⁴ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 496.
- ⁴²⁵ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 496.
- ⁴²⁶ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 496.
- ⁴²⁷ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 496.
- ⁴²⁸ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 22.
- ⁴²⁹ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 22.
- ⁴³⁰ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 22.
- ⁴³¹ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 22.
- ⁴³² Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 22.
- ⁴³³ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 489.
- ⁴³⁴ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 489.
- ⁴³⁵ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 489.
- ⁴³⁶ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 22.
- ⁴³⁷ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 22.
- ⁴³⁸ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 22.
- ⁴³⁹ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 22.
- ⁴⁴⁰ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 22.
- ⁴⁴¹ Frank J. Leavitt, "Democracies Restricting Democratic Rights: Some Classical Sources and Implications for Ethics of Biometrics," *ScientificWorldJournal* 11 (2011): 471.
- ⁴⁴² Andrew P. Rebera, Matteo E. Bonfanti, and Silvia Venier, "Societal and Ethical Implications of Anti-Spoofing Technologies in Biometrics," *Science and Engineering Ethics* 20, no. 1 (2014): 155.
- ⁴⁴³ Atsushi Kogetsu, Soichi Ogishima, and Kazuto Kato, "Authentication of Patients and Participants in Health Information Exchange and Consent for Medical Research: A Key Step for Privacy Protection, Respect for Autonomy, and Trustworthiness," *Frontiers in Genetics* 9, no. 167 (2018): 1.
- ⁴⁴⁴ Atsushi Kogetsu, Soichi Ogishima, and Kazuto Kato, "Authentication of Patients and Participants in Health Information Exchange and Consent for Medical Research: A Key Step for Privacy Protection, Respect for Autonomy, and Trustworthiness," 2.

- ⁴⁴⁵ Atsushi Kogetsu, Soichi Ogishima, and Kazuto Kato, "Authentication of Patients and Participants in Health Information Exchange and Consent for Medical Research: A Key Step for Privacy Protection, Respect for Autonomy, and Trustworthiness," 2.
- ⁴⁴⁶ Samantha A. Adams and Carolyn Petersen, "Precision Medicine: Opportunities, Possibilities, and Challenges for Patients and Providers," *Journal of the American Medical Informatics Association* 23, no. 4 (2016): 787-790.
- ⁴⁴⁷ The Global Alliance for Genomics and Health, "A Federated Ecosystem for Sharing Genomic, Clinical Data," *Science* 352, no. 6291 (2016): 1278-1280.
- ⁴⁴⁸ Atsushi Kogetsu, Soichi Ogishima, and Kazuto Kato, "Authentication of Patients and Participants in Health Information Exchange and Consent for Medical Research: A Key Step for Privacy Protection, Respect for Autonomy, and Trustworthiness," 2.
- ⁴⁴⁹ Harriet J. A. Teare et al., "The Rudy Study: Using Digital Technologies to Enable a Research Partnership," *European Journal of Human Genetics* 25, no. 7 (2017): 816-822.
- ⁴⁵⁰ Timothy P. Hogan et al., "Technology-Assisted Patient Access to Clinical Information: An Evaluation Framework for Blue Button," *JMIR Research Protocols* 3, no. 1 (2014): 18.
- ⁴⁵¹ Atsushi Kogetsu, Soichi Ogishima, and Kazuto Kato, "Authentication of Patients and Participants in Health Information Exchange and Consent for Medical Research: A Key Step for Privacy Protection, Respect for Autonomy, and Trustworthiness," 2.
- ⁴⁵² Brianne E. Kirkpatrick et al., "Genomeconnect: Matchmaking between Patients, Clinical Laboratories, and Researchers to Improve Genomic Knowledge," *Human Mutation* 36, no. 10 (2015): 974-978.
- ⁴⁵³ Atsushi Kogetsu, Soichi Ogishima, and Kazuto Kato, "Authentication of Patients and Participants in Health Information Exchange and Consent for Medical Research: A Key Step for Privacy Protection, Respect for Autonomy, and Trustworthiness," 2.
- ⁴⁵⁴ Lawrence O'Gorman, "Comparing Passwords, Tokens, and Biometrics for User Authentication," *Proceedings of the IEEE* 91, no. 12 (2003): 2019-2020.
- ⁴⁵⁵ Lawrence O'Gorman, "Comparing Passwords, Tokens, and Biometrics for User Authentication," 2019-2020.
- ⁴⁵⁶ Lawrence O'Gorman, "Comparing Passwords, Tokens, and Biometrics for User Authentication," 2019-2020.
- ⁴⁵⁷ Atsushi Kogetsu, Soichi Ogishima, and Kazuto Kato, "Authentication of Patients and Participants in Health Information Exchange and Consent for Medical Research: A Key Step for Privacy Protection, Respect for Autonomy, and Trustworthiness," 3.
- ⁴⁵⁸ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 24.
- ⁴⁵⁹ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 24.
- ⁴⁶⁰ Atsushi Kogetsu, Soichi Ogishima, and Kazuto Kato, "Authentication of Patients and Participants in Health Information Exchange and Consent for Medical Research: A Key Step for Privacy Protection, Respect for Autonomy, and Trustworthiness," 5.
- ⁴⁶¹ Atsushi Kogetsu, Soichi Ogishima, and Kazuto Kato, "Authentication of Patients and Participants in Health Information Exchange and Consent for Medical Research: A Key Step for Privacy Protection, Respect for Autonomy, and Trustworthiness," 5.
- ⁴⁶² Robert Biddle, Sonia Chiasson, and P. C. Van Orschot, "Graphical Passwords: Learning from the First Twelve Years," *ACM Computing Surveys* 44, no. 4 (2012): 19.
- ⁴⁶³ Atsushi Kogetsu, Soichi Ogishima, and Kazuto Kato, "Authentication of Patients and Participants in Health Information Exchange and Consent for Medical Research: A Key Step for Privacy Protection, Respect for Autonomy, and Trustworthiness," 5.
- ⁴⁶⁴ Atsushi Kogetsu, Soichi Ogishima, and Kazuto Kato, "Authentication of Patients and Participants in Health Information Exchange and Consent for Medical Research: A Key Step for Privacy Protection, Respect for Autonomy, and Trustworthiness," 5.

- ⁴⁶⁵ Emilio Mordini, "Biometrics," in *Handbook of Global Bioethics*, ed. Henk ten Have and Bert Gordijn (Dordecht: Springer, 2014), 513.
- ⁴⁶⁶ Emilio Mordini, "Biometrics," 513.
- ⁴⁶⁷ Emilio Mordini, "Biometrics," 513.
- ⁴⁶⁸ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 494.
- ⁴⁶⁹ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 494.
- ⁴⁷⁰ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 494.
- ⁴⁷¹ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 495.
- ⁴⁷² Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 495.
- ⁴⁷³ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 495.
- ⁴⁷⁴ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 495.
- ⁴⁷⁵ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 495.
- ⁴⁷⁶ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 495.
- ⁴⁷⁷ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 495.
- ⁴⁷⁸ Emilio Mordini, "Biometrics," 523.

⁴⁷⁹ Henk ten Have and Michèle Jean, eds., *The Unesco Universal Declaration on Bioethics and Human Rights: Background, Principles and Application*, Ethics Series (Paris: UNESCO Publishing, 2009), 355-366.

- ⁴⁸⁰ Emilio Mordini, "Biometrics," 523.
- ⁴⁸¹ Henk ten Have and Michèle Jean, *The Unesco Universal Declaration on Bioethics and Human Rights:* Background, Principles and Application, 355-366.
- ⁴⁸² Emilio Mordini, "Biometrics," 524.
- ⁴⁸³ Henk ten Have and Michèle Jean, *The Unesco Universal Declaration on Bioethics and Human Rights: Background, Principles and Application*, 355-366.
- ⁴⁸⁴ Henk ten Have and Michèle Jean, *The Unesco Universal Declaration on Bioethics and Human Rights:* Background, Principles and Application, 355-366.
- ⁴⁸⁵ Emilio Mordini, "Biometrics," 524.
- ⁴⁸⁶ Henk ten Have and Michèle Jean, *The Unesco Universal Declaration on Bioethics and Human Rights:* Background, Principles and Application, 355-366.
- ⁴⁸⁷ Emilio Mordini, "Biometrics," 524.
- ⁴⁸⁸ Emilio Mordini, "Biometrics," 524.
- ⁴⁸⁹ Emilio Mordini, "Biometrics," 524.
- ⁴⁹⁰ John R. Vacca, *Biometric Technologies and Verification Systems* (Amsterdam; Boston, MA: Butterworth-Heinemann/Elsevier, 2007), 51.
- ⁴⁹¹ John R. Vacca, Biometric Technologies and Verification Systems, 51.
- ⁴⁹² John R. Vacca, *Biometric Technologies and Verification Systems*, 51.
- ⁴⁹³ John R. Vacca, *Biometric Technologies and Verification Systems*, 51.
- ⁴⁹⁴ John R. Vacca, *Biometric Technologies and Verification Systems*, 51.
- ⁴⁹⁵ Anil K. Jain, Ruud Bolle, and Sharath Pankanti, *Biometrics: Personal Identification in Networked Society*, The Kluwer International Series in Engineering and Computer Science (Boston: Kluwer, 1999), 35.
- ⁴⁹⁶ Anil K. Jain, Ruud Bolle, and Sharath Pankanti, *Biometrics: Personal Identification in Networked Society*, 35.
- ⁴⁹⁷ Anil K. Jain, Ruud Bolle, and Sharath Pankanti, *Biometrics: Personal Identification in Networked Society*, 35.
- ⁴⁹⁸ Anil K. Jain, Ruud Bolle, and Sharath Pankanti, *Biometrics: Personal Identification in Networked Society*, 35.
- ⁴⁹⁹ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 27.
- ⁵⁰⁰ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 27.
- ⁵⁰¹ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 27.

- ⁵⁰² Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 33.
- ⁵⁰³ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 33.
- ⁵⁰⁴ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 33.
- ⁵⁰⁵ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 33.
- ⁵⁰⁶ Emilio Mordini, "Biometrics," 520.
- ⁵⁰⁷ Emilio Mordini, "Biometrics," 520.
- ⁵⁰⁸ Emilio Mordini, "Biometrics," 520.
- ⁵⁰⁹ Emilio Mordini, "Biometrics," 520.
- ⁵¹⁰ Anil K. Jain, Ruud Bolle, and Sharath Pankanti, Biometrics: Personal Identification in Networked Society, 311.
- ⁵¹¹ Anil K. Jain, Ruud Bolle, and Sharath Pankanti, Biometrics: Personal Identification in Networked Society, 311.
- ⁵¹² Anil K. Jain, Ruud Bolle, and Sharath Pankanti, Biometrics: Personal Identification in Networked Society, 312.
- ⁵¹³ Anil K. Jain, Ruud Bolle, and Sharath Pankanti, Biometrics: Personal Identification in Networked Society, 312.
- ⁵¹⁴ Anil K. Jain, Ruud Bolle, and Sharath Pankanti, Biometrics: Personal Identification in Networked Society, 312.
- ⁵¹⁵ Anil K. Jain, Ruud Bolle, and Sharath Pankanti, Biometrics: Personal Identification in Networked Society, 312.
- ⁵¹⁶ Anil K. Jain, Ruud Bolle, and Sharath Pankanti, Biometrics: Personal Identification in Networked Society, 312.
- ⁵¹⁷ J. Deliversky and M. Deliverska, "Ethical and Legal Considerations in Biometric Data Usage-Bulgarian Perspective," *Front Public Health* 6, no. 25 (2018): 3.
- ⁵¹⁸ J. Deliversky and M. Deliverska, "Ethical and Legal Considerations in Biometric Data Usage-Bulgarian Perspective," 3.
- ⁵¹⁹ J. Deliversky and M. Deliverska, "Ethical and Legal Considerations in Biometric Data Usage-Bulgarian Perspective," 3.
- ⁵²⁰ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 26.
- ⁵²¹ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 26.
- ⁵²² Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 26.
- ⁵²³ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 26.
- ⁵²⁴ Margit Sutrop and Katrin Laas-Mikko, "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics," 26.
- ⁵²⁵ Xinghua Shi and Xintao Wu, "An Overview of Human Genetic Privacy," Annals of the New York Academy of Sciences 1387, no. 1 (2017): 64.
- ⁵²⁶ Xinghua Shi and Xintao Wu, "An Overview of Human Genetic Privacy," 64.
- ⁵²⁷ Xinghua Shi and Xintao Wu, "An Overview of Human Genetic Privacy," 64.
- ⁵²⁸ Xinghua Shi and Xintao Wu, "An Overview of Human Genetic Privacy," 65.
- ⁵²⁹ Xinghua Shi and Xintao Wu, "An Overview of Human Genetic Privacy," 65.
- ⁵³⁰ Xinghua Shi and Xintao Wu, "An Overview of Human Genetic Privacy," 65.
- ⁵³¹ Xinghua Shi and Xintao Wu, "An Overview of Human Genetic Privacy," 65.
- ⁵³² Xinghua Shi and Xintao Wu, "An Overview of Human Genetic Privacy," 65.
- 533 Xinghua Shi and Xintao Wu, "An Overview of Human Genetic Privacy," 65.
- ⁵³⁴ Emilio Mordini, "Biometrics," 517.
- ⁵³⁵ Emilio Mordini, "Biometrics," 517.
- ⁵³⁶ Emilio Mordini, "Biometrics," 517.
- ⁵³⁷ Emilio Mordini, "Biometrics," 518.

- ⁵³⁸ Emilio Mordini, "Biometrics," 518.
- ⁵³⁹ Emilio Mordini, "Biometrics," 518.
- ⁵⁴⁰ Emilio Mordini, "Biometrics," 518.
- ⁵⁴¹ Emilio Mordini, "Biometrics," 519.
- ⁵⁴² Emilio Mordini, "Biometrics," 519.
- ⁵⁴³ Emilio Mordini, "Biometrics," 519.
- ⁵⁴⁴ Emilio Mordini, "Biometrics," 519.
- ⁵⁴⁵ Emilio Mordini, "Biometrics," 519.
- ⁵⁴⁶ Andrew P. Rebera, Matteo E. Bonfanti, and Silvia Venier, "Societal and Ethical Implications of Anti-Spoofing Technologies in Biometrics," 160.
- ⁵⁴⁷ Andrew P. Rebera, Matteo E. Bonfanti, and Silvia Venier, "Societal and Ethical Implications of Anti-Spoofing Technologies in Biometrics," 160.
- ⁵⁴⁸ Andrew P. Rebera, Matteo E. Bonfanti, and Silvia Venier, "Societal and Ethical Implications of Anti-Spoofing Technologies in Biometrics," 161.
- ⁵⁴⁹ Andrew P. Rebera, Matteo E. Bonfanti, and Silvia Venier, "Societal and Ethical Implications of Anti-Spoofing Technologies in Biometrics," 161.
- ⁵⁵⁰ Andrew P. Rebera, Matteo E. Bonfanti, and Silvia Venier, "Societal and Ethical Implications of Anti-Spoofing Technologies in Biometrics," 161.
- ⁵⁵¹ J. Deliversky and M. Deliverska, "Ethical and Legal Considerations in Biometric Data Usage-Bulgarian Perspective," 2.
- ⁵⁵² J. Deliversky and M. Deliverska, "Ethical and Legal Considerations in Biometric Data Usage-Bulgarian Perspective," 2.
- ⁵⁵³ J. Deliversky and M. Deliverska, "Ethical and Legal Considerations in Biometric Data Usage-Bulgarian Perspective," 2.
- ⁵⁵⁴ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 492.
- ⁵⁵⁵ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 492.
- ⁵⁵⁶ Emilio Mordini and Sonia Massari, "Body, Biometrics and Identity," 492.
- ⁵⁵⁷ Mary Rundle and Chris Conley, *Ethical Implications of Emerging Technologies: A Survey*, ed. Boyan Radoykov (Paris: UNESCO, 2007), 78.
- ⁵⁵⁸ Mary Rundle and Chris Conley, *Ethical Implications of Emerging Technologies: A Survey*, 78.
- ⁵⁵⁹ Mary Rundle and Chris Conley, *Ethical Implications of Emerging Technologies: A Survey*, 78.
- ⁵⁶⁰ Mary Rundle and Chris Conley, *Ethical Implications of Emerging Technologies: A Survey*, 79.
- ⁵⁶¹ Mary Rundle and Chris Conley, *Ethical Implications of Emerging Technologies: A Survey*, 79.
- ⁵⁶² Mary Rundle and Chris Conley, *Ethical Implications of Emerging Technologies: A Survey*, 79.
- ⁵⁶³ Mary Rundle and Chris Conley, Ethical Implications of Emerging Technologies: A Survey, 79.
- ⁵⁶⁴ Mary Rundle and Chris Conley, *Ethical Implications of Emerging Technologies: A Survey*, 80.
- ⁵⁶⁵ Mary Rundle and Chris Conley, Ethical Implications of Emerging Technologies: A Survey, 80.
- ⁵⁶⁶ Mary Rundle and Chris Conley, *Ethical Implications of Emerging Technologies: A Survey*, 80.
- ⁵⁶⁷ Mary Rundle and Chris Conley, Ethical Implications of Emerging Technologies: A Survey, 80.
- ⁵⁶⁸ Mary Rundle and Chris Conley, Ethical Implications of Emerging Technologies: A Survey, 81.
- ⁵⁶⁹ Mary Rundle and Chris Conley, *Ethical Implications of Emerging Technologies: A Survey*, 81.
- ⁵⁷⁰ Mary-Claire King, Joan H. Marks, and Jessica B. Mandell, "Breast and Ovarian Cancer Risks Due to Inherited Mutations in Brca1 and Brca2," *Science* 302, no. 5645 (2003): 643-646.
- ⁵⁷¹ Ben Eggleston, "Genetic Discrimination in Health Insurance: An Ethical and Economic Analysis," in *The Human Genome Project in College Curriculum: Ethical Issues and Practical Strategies* ed. Aine Donovan and Ronald Michael Green (Hanover, NH: Dartmouth College Press, 2008), 46-47.
- ⁵⁷² Ben Eggleston, "Genetic Discrimination in Health Insurance: An Ethical and Economic Analysis," 46-47.
- ⁵⁷³ Ben Eggleston, "Genetic Discrimination in Health Insurance: An Ethical and Economic Analysis," 46-47.
- ⁵⁷⁴ Ben Eggleston, "Genetic Discrimination in Health Insurance: An Ethical and Economic Analysis," 46-47.

- ⁵⁷⁵ Francis S. Collins and James D. Watson, "Genetic Discrimination: Time to Act," Science 302, no. 5646 (2003): 745.
- ⁵⁷⁶ Allen E. Buchanan et al., *From Chance to Choice: Genetics and Justice* (New York, NY: Cambridge University Press, 2000), 339.
- ⁵⁷⁷ Mark A. Rothstein, "Currents in Contemporary Ethics: Gina, the Ada, and Genetic Discrimination in Employment," *Journal of Law, Medicine & Ethics* 36, no. 4 (2008): 837-840.
- ⁵⁷⁸ Mark A. Rothstein, "Currents in Contemporary Ethics: Gina, the Ada, and Genetic Discrimination in Employment," 837-840.
- ⁵⁷⁹ 42 U.S.C. § 12102(2).
- 580 29 C.F.R. § 1630.2(j).
- ⁵⁸¹ Mark A. Rothstein, "Currents in Contemporary Ethics: Gina, the Ada, and Genetic Discrimination in Employment," 837-840.
- ⁵⁸² Henk ten Have, *Global Bioethics: An Introduction*, 117.
- ⁵⁸³ Henk ten Have, *Global Bioethics: An Introduction*, 117.
- ⁵⁸⁴ Henk ten Have, *Global Bioethics: An Introduction*, 117.
- ⁵⁸⁵ Henk ten Have, *Global Bioethics: An Introduction*, 117.
- ⁵⁸⁶ Henk ten Have, *Global Bioethics: An Introduction*, 117.
- ⁵⁸⁷ UNESCO, "Universal Declaration on the Human Genome and Human Rights," *Journal of Medicine & Philosophy* 23, no. 3 (1998): 334-341.
- ⁵⁸⁸ UNESCO, "Universal Declaration on the Human Genome and Human Rights," 334-341.
- ⁵⁸⁹ UNESCO, "Universal Declaration on the Human Genome and Human Rights," 334-341.
- ⁵⁹⁰ UNESCO, "Universal Declaration on the Human Genome and Human Rights," 334-341.
- ⁵⁹¹ UNESCO, "Universal Declaration on the Human Genome and Human Rights," 334-341.
- ⁵⁹² Henk ten Have, *Global Bioethics: An Introduction*, 118.
- ⁵⁹³ Bartha M. Knoppers and Yann Joly, "Our Social Genome?," Trends in Biotechnology 25, no. 7 (2007): 286-288.
- ⁵⁹⁴ Bartha M. Knoppers and Yann Joly, "Our Social Genome?," 286-288.
- ⁵⁹⁵ Bartha M. Knoppers and Yann Joly, "Our Social Genome?," 286-288.
- ⁵⁹⁶ Bartha M. Knoppers and Yann Joly, "Our Social Genome?," 286-288.
- ⁵⁹⁷ Human Genome Organisation (HUGO), "Statement on Human Genomic Databases," *Journal International de Bioéthique* 14, no. 3 (2003): 207-210.
- ⁵⁹⁸ Human Genome Organisation (HUGO), "Human Genome Organisation (Hugo). Statement on Human Genomic Databases. December 2002," 207-210.
- ⁵⁹⁹ Bartha M. Knoppers and Yann Joly, "Our Social Genome?," 286-288.
- ⁶⁰⁰ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," The Monist 89, no. 4 (2006): 568-569.
- ⁶⁰¹ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 568-569.
- ⁶⁰² Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 568-569.
- ⁶⁰³ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 568-569.
- ⁶⁰⁴ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 568-569.
- 605 Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 568-569.
- ⁶⁰⁶ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 568-569.
- ⁶⁰⁷ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 568-569.
- ⁶⁰⁸ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 568-569.
- ⁶⁰⁹ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 568-569.
- ⁶¹⁰ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 569.
- ⁶¹¹ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 570-571.
- ⁶¹² Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 573-574.
- ⁶¹³ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 577-578.
- ⁶¹⁴ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 577-578.
- ⁶¹⁵ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 586.

- ⁶¹⁶ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 586-587.
- ⁶¹⁷ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 586-587.
- ⁶¹⁸ Martin O'Neill, "Genetic Information, Life Insurance, and Social Justice," 586.
- ⁶¹⁹ Judit Sándor, *Society and Genetic Information: Codes and Laws in the Genetic Era*, English ed., Cps Books (Budapest, Hungary: Central European University Press, 2003), 192-193.
- ⁶²⁰ Judit Sándor, Society and Genetic Information: Codes and Laws in the Genetic Era, 192-193.

⁶²¹ Judit Sándor, Society and Genetic Information: Codes and Laws in the Genetic Era, 192-193.

⁶²² Judit Sándor, Society and Genetic Information: Codes and Laws in the Genetic Era, 192-193.

⁶²³ Judit Sándor, Society and Genetic Information: Codes and Laws in the Genetic Era, 192-193.

⁶²⁴ Judit Sándor, Society and Genetic Information: Codes and Laws in the Genetic Era, 192-193.

⁶²⁵ Judit Sándor, Society and Genetic Information: Codes and Laws in the Genetic Era, 192-193.

⁶²⁶ Judit Sándor, Society and Genetic Information: Codes and Laws in the Genetic Era, 192-193.

⁶²⁷ Judit Sándor, Society and Genetic Information: Codes and Laws in the Genetic Era, 192-193.

- ⁶²⁸ Judit Sándor, Society and Genetic Information: Codes and Laws in the Genetic Era, 192-193.
- ⁶²⁹ Judit Sándor, Society and Genetic Information: Codes and Laws in the Genetic Era, 192-193.
- ⁶³⁰ Judit Sándor, Society and Genetic Information: Codes and Laws in the Genetic Era, 192-193.
- ⁶³¹ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," *Journal* of Law, Medicine & Ethics 43, no. 4 (2015): 675-689.
- ⁶³² Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 675-689.
- ⁶³³ Graeme Laurie et al., "Managing Access to Biobanks: How Can We Reconcile Individual Privacy and Public Interests in Genetic Research," *Medical Law International* 10, no. 4 (2010): 317.
- ⁶³⁴ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 680.
- ⁶³⁵ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 680-681.
- ⁶³⁶ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 680-681.
- ⁶³⁷ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 680-681.
- ⁶³⁸Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 680.

⁶³⁹ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 680.

- ⁶⁴⁰ Graeme Laurie et al., "Managing Access to Biobanks: How Can We Reconcile Individual Privacy and Public Interests in Genetic Research," 318.
- ⁶⁴¹ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 680-681.
- ⁶⁴² Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 680.
- ⁶⁴³ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 680-681.
- ⁶⁴⁴ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 680-681.
- ⁶⁴⁵ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 681-682.
- ⁶⁴⁶ Edward S. Dove, Bartha M. Knoppers, and Ma'n H. Zawati, "Towards an Ethics Safe Harbor for Global Biomedical Research," *Journal of Law and the Biosciences* 1, no. 1 (2014): 16-17.
- ⁶⁴⁷ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 682.
- ⁶⁴⁸ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 682-683.
- ⁶⁴⁹ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 682-683.

- ⁶⁵⁰ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 675-689.
- ⁶⁵¹ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 675-689.
- ⁶⁵² Jane Kaye, "Do We Need a Uniform Regulatory System for Biobanks across Europe?," *European Journal of Human Genetics* 14, no. 2 (2006): 245-248.
- ⁶⁵³ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 675-689.
- ⁶⁵⁴ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 675-689.
- ⁶⁵⁵ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 675-689.
- ⁶⁵⁶ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 684.
- ⁶⁵⁷ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 675-689.
- ⁶⁵⁸ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 675-689.
- ⁶⁵⁹ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 675-689.
- ⁶⁶⁰ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 675-689.
- ⁶⁶¹ Edward S. Dove, "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework," 675-689.
- ⁶⁶² Paula Boddington, *Towards a Code of Ethics for Artificial Intelligence* (New York, NY: Springer Berlin Heidelberg, 2017), 65.
- ⁶⁶³ Nicholas Agar, How to Be Human in the Digital Economy (Cambridge, MA: The MIT Press, 2019), 1.
- ⁶⁶⁴ Hannah Arendt, *The Human Condition*, Charles R. Walgreen Foundation Lectures (University of Chicago Press, 1958), 39-44.
- ⁶⁶⁵ Sally Wyatt, "Technological Determinisim Is Dead; Long Live Technological Determinism," in *The Handbook of Science and Technology Studies*, ed. Edward J. Hackett, et al. (The MIT Press, 2007), 165-180.
- ⁶⁶⁶ Lewis Mumford, "History: Neglected Clue to Technological Change," *Technology and Culture* 2, no. 3 (1961): 230-236.
- ⁶⁶⁷ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," *Philosophy & Technology* 31, no. 4 (2018): 656.
- ⁶⁶⁸ Shuo-Yan Chou, "The Fourth Industrial Revolution: Digital Fusion with Internet of Things," Journal of International Affairs 72, no. 1 (2019): 107.
- ⁶⁶⁹ Shuo-Yan Chou, "The Fourth Industrial Revolution: Digital Fusion with Internet of Things," 107.
- ⁶⁷⁰ Shuo-Yan Chou, "The Fourth Industrial Revolution: Digital Fusion with Internet of Things," 107.
- ⁶⁷¹ Shuo-Yan Chou, "The Fourth Industrial Revolution: Digital Fusion with Internet of Things," 107.
- ⁶⁷² Jaap Bloem et al., "The Fourth Industrial Revolution: Things to Tighten the Link between It and Ot," in VINT Research Report (Sogeti, 2014), 14.
- ⁶⁷³ Shuo-Yan Chou, "The Fourth Industrial Revolution: Digital Fusion with Internet of Things," 108.
- ⁶⁷⁴ Shuo-Yan Chou, "The Fourth Industrial Revolution: Digital Fusion with Internet of Things," 108.
- ⁶⁷⁵ Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies," *Ethical Theory & Moral Practice* 18, no. 5 (2015): 1028.
- ⁶⁷⁶ Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies," 1028.

⁶⁷⁷ Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic

Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies," 1028.

⁶⁷⁸ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 656.

⁶⁷⁹ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 656.

⁶⁸⁰ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 656.

⁶⁸² Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 656.

⁶⁸³ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 656.

- ⁶⁸⁴ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 656.
- ⁶⁸⁵ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 656.
- ⁶⁸⁶ Natascia Boeri, "Technology and Society as Embedded: An Alternative Framework for Information and Communication Technology and Development," *Media, Culture & Society* 38, no. 1 (2016): 108.
- ⁶⁸⁷ Caroline Pade-Khene, Brenda Mallinson, and David Sewry, "Sustainable Rural Ict Project Management Practice for Developing Countries: Investigating the Dwesa and Rumep Projects," *Information Technology for Development* 17, no. 3 (2011): 187-212.

⁶⁸⁸ Natascia Boeri, "Technology and Society as Embedded: An Alternative Framework for Information and Communication Technology and Development," 108.

- ⁶⁸⁹ Judy Wajcman, "Addressing Technological Change: The Challenge to Social Theory," *Current Sociology* 50, no. 3 (2002): 347-363.
- ⁶⁹⁰ Saskia Sassen, "Towards a Sociology of Information Technology," *Current Sociology* 50, no. 3 (2002): 365.

⁶⁹¹ Natascia Boeri, "Technology and Society as Embedded: An Alternative Framework for Information and Communication Technology and Development," 108.

- ⁶⁹² Natascia Boeri, "Technology and Society as Embedded: An Alternative Framework for Information and Communication Technology and Development," 108.
- ⁶⁹³ Natascia Boeri, "Technology and Society as Embedded: An Alternative Framework for Information and Communication Technology and Development," 108.
- ⁶⁹⁴ Nicholas Agar, How to Be Human in the Digital Economy, 9.
- ⁶⁹⁵ Nicholas Agar, How to Be Human in the Digital Economy, 9.
- ⁶⁹⁶ Nicholas Agar, How to Be Human in the Digital Economy, 9.
- ⁶⁹⁷ Nicholas Agar, How to Be Human in the Digital Economy, 10.
- ⁶⁹⁸ Nicholas Agar, How to Be Human in the Digital Economy, 10.
- ⁶⁹⁹ Daniel Wegner and Gray Kurt, *The Mind Club: Who Thinks, What Feels, and Why It Matters* (New York, New York: Penguin Books, 2016), 3.
- ⁷⁰⁰ Nicholas Agar, *How to Be Human in the Digital Economy*, 10.
- ⁷⁰¹ Nicholas Agar, *How to Be Human in the Digital Economy*, 10.
- ⁷⁰² Nicholas Agar, *How to Be Human in the Digital Economy*, 13.
- ⁷⁰³ Nicholas Agar, *How to Be Human in the Digital Economy*, 13.
- ⁷⁰⁴ Nicholas Agar, *How to Be Human in the Digital Economy*, 13.
- ⁷⁰⁵ Nicholas Agar, *How to Be Human in the Digital Economy*, 13.
- ⁷⁰⁶ Nicholas Agar, *How to Be Human in the Digital Economy*, 13.
- ⁷⁰⁷ Nicholas Agar, *How to Be Human in the Digital Economy*, 13.
- ⁷⁰⁸ Robert C. Scharff and Val Dusek, *Philosophy of Technology: The Technological Condition: An Anthology*, 2nd ed. (Wiley-Blackwell, 2014), 458.
- ⁷⁰⁹ Robert C. Scharff and Val Dusek, *Philosophy of Technology: The Technological Condition: An Anthology*, 458.
- ⁷¹⁰ Robert C. Scharff and Val Dusek, *Philosophy of Technology: The Technological Condition: An Anthology*, 458.
- ⁷¹¹ Robert C. Scharff and Val Dusek, *Philosophy of Technology: The Technological Condition: An Anthology*, 458.
- ⁷¹² George W. Bush, "State of the Union Address by the President," ed. Executive Branch (Washington D.C.: United States Capitol, 2006).
- ⁷¹³ Robert C. Scharff and Val Dusek, *Philosophy of Technology: The Technological Condition: An Anthology*, 458.

⁶⁸¹ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 656.

- ⁷¹⁴ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 656-657.
- ⁷¹⁵ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 657.
- ⁷¹⁶ Hans Radder, "Science, Technology, and the Science-Technology Relationship," in *Handbook of the Philosophy of Science: Philosophy of Technology and Engineering Sciences*, ed. Anthonie Meijers (North Holland: Elsevier, 2009), 65-92.
- ⁷¹⁷ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 657.
- ⁷¹⁸ Norbert Wiener, *Cybernetics: Or, Control and Communication in the Animal and the Machine*, 2nd ed., The M.I.T. Paperback Series (Cambridge, MA: M.I.T. Press, 1965), 39-44.
- ⁷¹⁹ Heather A. Love, "Newsreels, Novels, and Cybernetics: Reading the Random Patterns of John Dos Passos's U.S.A," *Journal of Modern Literature* 40, no. 2 (2017): 114.
- ⁷²⁰ Heather A. Love, "Newsreels, Novels, and Cybernetics: Reading the Random Patterns of John Dos Passos's U.S.A," 114.
- ⁷²¹ Norbert Wiener, *Cybernetics: Or, Control and Communication in the Animal and the Machine*, 39-44.
- ⁷²² Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 657.
- ⁷²³ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 657.
- ⁷²⁴ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 657.
- ⁷²⁵ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 657.
- ⁷²⁶ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 657.
- ⁷²⁷ Heather Douglas, "Pure Science and the Problem of Progress," *Studies in History and Philosophy of Science* 46 (2014): 55-63.
- ⁷²⁸ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 658.
- ⁷²⁹ Federica Russo, "Digital Technologies, Ethical Questions, and the Need of an Informational Framework," 658.
- ⁷³⁰ Jonathan Gabe, Mike Bury, and Mary Ann Elston, *Key Concepts in Medical Sociology* (Sage Publications Ltd, 2004), 147.
- ⁷³¹ Stefan Timmermans and Marc Berg, "The Practice of Medical Technology," *Sociology of Health & Illness* 25, no. 3 (2003): 99.
- ⁷³² Jonathan Gabe, Mike Bury, and Mary Ann Elston, Key Concepts in Medical Sociology, 147.
- ⁷³³ Jonathan Gabe, Mike Bury, and Mary Ann Elston, *Key Concepts in Medical Sociology*, 147.
- ⁷³⁴ Jonathan Gabe, Mike Bury, and Mary Ann Elston, Key Concepts in Medical Sociology, 150-151.
- ⁷³⁵ Henk ten Have, "Genetics and Culture: The Geneticization Thesis," *Medicine, Health Care and Philosophy* 4, no. 3 (2001)., 1.
- ⁷³⁶ Jonathan Gabe, Mike Bury, and Mary Ann Elston, Key Concepts in Medical Sociology., 147.
- ⁷³⁷ Stefan Timmermans and Marc Berg, "The Practice of Medical Technology," 101.
- ⁷³⁸ Anselm Strauss et al., Social Organization of Medical Work (New Brunswick, N.J.: Transaction Publishers, 1997), 44.
- ⁷³⁹ Jonathan Gabe, Mike Bury, and Mary Ann Elston, Key Concepts in Medical Sociology., 148.
- ⁷⁴⁰ Jonathan Gabe, Mike Bury, and Mary Ann Elston, Key Concepts in Medical Sociology., 148.
- ⁷⁴¹ Bruno Latour, Science in Action: How to Follow Scientists and Engineers through Society (London: Open University Press, 1987), 68-69.
- ⁷⁴² Jonathan Gabe, Mike Bury, and Mary Ann Elston, Key Concepts in Medical Sociology., 149.
- ⁷⁴³ Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies," 1034.
- ⁷⁴⁴ "The Wingspread Consensus Statement on the Precautionary Principle", (paper presented at the Wingspread Conference, Racine, WI, 1998).
- ⁷⁴⁵ Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies," 1034.
- ⁷⁴⁶ Jonathan Hughes, "How Not to Criticize the Precautionary Principle," *Journal of Medicine & Philosophy* 31, no. 5 (2006): 447-464.

- ⁷⁴⁷ Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies.", 1034.
- ⁷⁴⁸ Stephen M. Gardiner, "A Core Precautionary Principle," *Journal of Political Philosophy* 14, no. 1 (2006): 33-60.

⁷⁴⁹ Stephen M. Gardiner, "A Core Precautionary Principle," 33-60.

- ⁷⁵⁰ Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies," 1034.
- ⁷⁵¹ Philip Brey, "Anticipating Ethical Issues in Emerging It," *Ethics & Information Technology* 14, no. 4 (2012): 305-317.
- ⁷⁵² Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies.", 1035.
- ⁷⁵³ Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies.", 1035.
- ⁷⁵⁴ Thien A. Tran and Tugrul Daim, "A Taxonomic Review of Methods and Tools Applied in Technology Assessment," *Technological Forecasting & Social Change* 75, no. 9 (2008): 1396-1405.
- ⁷⁵⁵ Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies."
- ⁷⁵⁶ Philip Brey, "Anticipating Ethical Issues in Emerging It," 305-317.
- ⁷⁵⁷ Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies.", 1035.
- ⁷⁵⁸ Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies."
- ⁷⁵⁹ Armin Grunwald, "Technology Assessment: Concepts and Methods," in *Philosophy of Technology and Engineering Sciences* (Elsevier, 2009), 9.
- ⁷⁶⁰ Brent D. Mittelstadt, Bernd Stahl, and N. Ben Fairweather, "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies.", 1035.
- ⁷⁶¹ Arnold R. Eiser, *The Ethos of Medicine in Postmodern America: Philosophical, Cultural, and Social Considerations* (Lanham: Lexington Books, 2014), 171.
- ⁷⁶² Thomas E. McCollough, *The Moral Imagination and Public Life: Raising the Ethical Question*, Chatham House Studies in Political Thinking (Chatham, NJ: Chatham Books, 1992), 72.
- ⁷⁶³ Arnold R. Eiser, The Ethos of Medicine in Postmodern America: Philosophical, Cultural, and Social Considerations, 172.
- ⁷⁶⁴ Henry A. Giroux, Stealing Innocence: Youth, Corporate Power and the Politics of Culture (New York, NY: Palgrave Macmillan, 2000), 172.
- ⁷⁶⁵ Roy M. Poses, "A Cautionary Tale: The Dysfunction of American Health Care," *European Journal of Internal Medicine* 14, no. 2 (2003): 123-130.
- ⁷⁶⁶ Arnold R. Eiser, *The Ethos of Medicine in Postmodern America: Philosophical, Cultural, and Social Considerations*, 172.
- ⁷⁶⁷ Arnold R. Eiser, *The Ethos of Medicine in Postmodern America: Philosophical, Cultural, and Social Considerations*, 172.
- ⁷⁶⁸ Albert Borgmann, *Technology and the Character of Contemporary Life: A Philosophical Inquiry* (Chicago, IL: University of Chicago Press, 1984), 41.
- ⁷⁶⁹ Albert Borgmann, *Technology and the Character of Contemporary Life: A Philosophical Inquiry*, 41.
- ⁷⁷⁰ Arnold R. Eiser, *The Ethos of Medicine in Postmodern America: Philosophical, Cultural, and Social Considerations*, 175.
- ⁷⁷¹ Arnold R. Eiser, *The Ethos of Medicine in Postmodern America: Philosophical, Cultural, and Social Considerations*, 175.
- ⁷⁷² Arnold R. Eiser, The Ethos of Medicine in Postmodern America: Philosophical, Cultural, and Social Considerations, 175.

⁷⁷³ Robert J. Sternberg, "Wisdom : Its Nature, Origins, and Development," ed. Robert J. Sternberg (New York, NY: Cambridge University Press, 2000), 170-175.

- ⁷⁷⁵ Arnold R. Eiser, *The Ethos of Medicine in Postmodern America: Philosophical, Cultural, and Social Considerations*, 173.
- ⁷⁷⁶ Ellen M. McGee, "Bioelectronics and Implanted Devices," in *Medical Enhancement and Posthumanity*, International Library of Ethics, Law and Technology (Dordrecht: Springer, 2008), 207.
- ⁷⁷⁷ Ellen M. McGee, "Bioelectronics and Implanted Devices," 207.
- ⁷⁷⁸ Ellen M. McGee, "Bioelectronics and Implanted Devices," 207.
- ⁷⁷⁹ Ellen M. McGee, "Bioelectronics and Implanted Devices," 207.
- ⁷⁸⁰ Ellen M. McGee, "Bioelectronics and Implanted Devices," 207.
- ⁷⁸¹ Ellen M. McGee, "Bioelectronics and Implanted Devices," 208.
- ⁷⁸² Ellen M. McGee, "Bioelectronics and Implanted Devices," 208.
- ⁷⁸³ Ellen M. McGee, "Bioelectronics and Implanted Devices," 208.
- ⁷⁸⁴ Ellen M. McGee, "Bioelectronics and Implanted Devices," 208.
- ⁷⁸⁵ Ellen M. McGee, "Bioelectronics and Implanted Devices," 208.
- ⁷⁸⁶ Ellen M. McGee, "Bioelectronics and Implanted Devices," 208.
- ⁷⁸⁷ Ellen M. McGee, "Bioelectronics and Implanted Devices," 208.
- ⁷⁸⁸ Ellen M. McGee, "Bioelectronics and Implanted Devices," 208.
- ⁷⁸⁹ Ellen M. McGee, "Bioelectronics and Implanted Devices," 208.
- ⁷⁹⁰ Nicholas Agar, *Truly Human Enhancement: A Philosophical Defense of Limits*, Basic Bioethics (Cambridge, MA: MIT Press, 2014), 2.
- ⁷⁹¹ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 2.
- ⁷⁹² Nicholas Agar, *Truly Human Enhancement: A Philosophical Defense of Limits*, Basic Bioethics (Cambridge, MA: MIT Press, 2014), 2.
- ⁷⁹³ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 2.
- ⁷⁹⁴ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 2.
- ⁷⁹⁵ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 3.
- ⁷⁹⁶ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 3.
- ⁷⁹⁷ Nicholas Agar, *Truly Human Enhancement: A Philosophical Defense of Limits*, 3.
- ⁷⁹⁸ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 3.
- ⁷⁹⁹ Nicholas Agar, *Truly Human Enhancement: A Philosophical Defense of Limits*, 3.
- ⁸⁰⁰ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 3.
- ⁸⁰¹ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 3.
- ⁸⁰² Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 3.
- ⁸⁰³ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 4.
- ⁸⁰⁴ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 4.
- ⁸⁰⁵ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 4.
- ⁸⁰⁶ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 4.
- ⁸⁰⁷ Ellen M. McGee, "Bioelectronics and Implanted Devices," 213.
- ⁸⁰⁸ Ellen M. McGee, "Bioelectronics and Implanted Devices," 213.
- ⁸⁰⁹ Ellen M. McGee, "Bioelectronics and Implanted Devices," 213.
- ⁸¹⁰ Ellen M. McGee, "Bioelectronics and Implanted Devices," 214.
- ⁸¹¹ Ellen M. McGee, "Bioelectronics and Implanted Devices," 214.
- ⁸¹² Ellen M. McGee, "Bioelectronics and Implanted Devices," 214.
- ⁸¹³ Ellen M. McGee, "Bioelectronics and Implanted Devices," 214.
- ⁸¹⁴ Ellen M. McGee, "Bioelectronics and Implanted Devices," 214.
- ⁸¹⁵ Ellen M. McGee, "Bioelectronics and Implanted Devices," 214.
- ⁸¹⁶ Ellen M. McGee, "Bioelectronics and Implanted Devices," 214.

⁷⁷⁴ Robert J. Sternberg, "Wisdom : Its Nature, Origins, and Development," 170-175.

- ⁸¹⁷ Ellen M. McGee, "Bioelectronics and Implanted Devices," 215.
- ⁸¹⁸ Ellen M. McGee, "Bioelectronics and Implanted Devices," 215.
- ⁸¹⁹ Ellen M. McGee, "Bioelectronics and Implanted Devices," 215.
- ⁸²⁰ Francis Fukuyama, Our Posthuman Future: Consequences of the
- Biotechnology Revolution (New York, NY: Farrar, Straus and Giroux, 2002), 70-71.
- ⁸²¹ Leon R. Kass, "Ageless Bodies, Happy Souls: Biotechnology and the Pursuit of Perfection," *The New Atlantis*, no. 1 (2003): 9-28.
- ⁸²² Ellen M. McGee, "Bioelectronics and Implanted Devices," 215.
- ⁸²³ Ellen M. McGee, "Bioelectronics and Implanted Devices," 215.
- ⁸²⁴ Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 5.
- 825 Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 5.
- 826 Nicholas Agar, Truly Human Enhancement: A Philosophical Defense of Limits, 5.
- ⁸²⁷ Nick Bostrom and Julian Savulescu, "Human Enhancement Ethics: The State of the Debate," in *Human Enhancement*, ed. Julian Savulescu and Nick Bostrom (Oxford: Oxford University Press, 2009), 19-20.
- ⁸²⁸ Nick Bostrom and Julian Savulescu, "Human Enhancement Ethics: The State of the Debate," 19-20.
- ⁸²⁹ Nick Bostrom and Julian Savulescu, "Human Enhancement Ethics: The State of the Debate," 19-20.
- ⁸³⁰ Nick Bostrom and Julian Savulescu, "Human Enhancement Ethics: The State of the Debate," 19-20.
- ⁸³¹ Lawry Tom, *Ai in Health: A Leader's Guide to Winning in the New Age of Intelligent Health Systems* (Boca Raton, FL: HIMSS Publishing, 2020), 147.
- ⁸³² Wolfgang Ertel, *Introduction to Artificial Intelligence*, Undergraduate Topics in Computer Science (London; New York: Springer, 2011), 1.
- ⁸³³ Hector J. Levesque, *Common Sense, the Turing Test, and the Quest for Real Ai* (Cambridge, Massachusetts: MIT Press, 2017), 1.
- ⁸³⁴ Hector J. Levesque, Common Sense, the Turing Test, and the Quest for Real Ai, 1.
- ⁸³⁵ Hector J. Levesque, Common Sense, the Turing Test, and the Quest for Real Ai, 2.
- ⁸³⁶ Hector J. Levesque, Common Sense, the Turing Test, and the Quest for Real Ai, 2.
- ⁸³⁷ Hector J. Levesque, Common Sense, the Turing Test, and the Quest for Real Ai, 2.
- ⁸³⁸ Hector J. Levesque, Common Sense, the Turing Test, and the Quest for Real Ai, 2.
- ⁸³⁹ Hector J. Levesque, Common Sense, the Turing Test, and the Quest for Real Ai, 3.
- ⁸⁴⁰ Hector J. Levesque, Common Sense, the Turing Test, and the Quest for Real Ai, 3.
- ⁸⁴¹ Hector J. Levesque, Common Sense, the Turing Test, and the Quest for Real Ai, 4.
- ⁸⁴² Terrence J. Sejnowski, *The Deep Learning Revolution* (Cambridge, Massachusetts: The MIT Press, 2018), 1.
- ⁸⁴³ Terrence J. Sejnowski, The Deep Learning Revolution, 1.
- ⁸⁴⁴ Terrence J. Sejnowski, *The Deep Learning Revolution*, 1.
- ⁸⁴⁵ Terrence J. Sejnowski, *The Deep Learning Revolution*, 1.
- ⁸⁴⁶ Ethem Alpaydin, *Machine Learning: The New Ai*, Mit Press Essential Knowledge Series (Cambridge, MA: MIT Press, 2016), 17.
- ⁸⁴⁷ Ethem Alpaydin, Machine Learning: The New Ai, 17-18.
- 848 Ethem Alpaydin, Machine Learning: The New Ai, 18.
- 849 Ethem Alpaydin, Machine Learning: The New Ai, 19.
- ⁸⁵⁰ Ethem Alpaydin, Machine Learning: The New Ai, 19.
- ⁸⁵¹ Ethem Alpaydin, Machine Learning: The New Ai, 19.
- ⁸⁵² Ethem Alpaydin, Machine Learning: The New Ai, 20.
- 853 A. M. Turing, "Computing Machinery and Intelligence," Mind 59, no. 236 (1950): 433-460.
- ⁸⁵⁴ A. M. Turing, "Computing Machinery and Intelligence," 442.
- ⁸⁵⁵ Murray Shanahan, *The Technological Singularity*, The Mit Press Essential Knowledge Series (Cambridge, Massachusetts: The MIT Press, 2015), 1.
- ⁸⁵⁶ Murray Shanahan, The Technological Singularity, 2.
- 857 A. M. Turing, "Computing Machinery and Intelligence," 436.

- ⁸⁵⁸ Murray Shanahan, The Technological Singularity, 2.
- ⁸⁵⁹ Murray Shanahan, The Technological Singularity, 2.
- ⁸⁶⁰ Murray Shanahan, The Technological Singularity, 2.
- ⁸⁶¹ Murray Shanahan, *The Technological Singularity*, 3.
- ⁸⁶² Murray Shanahan, The Technological Singularity, 3.
- ⁸⁶³ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," in *The Cambridge Handbook of Artificial Intelligence*, ed. Keith Frankish and William M. Ramsey (Cambridge, UK: Cambridge University Press, 2014), 316.
- ⁸⁶⁴ Roberto Simanowski, Michel Brodmer, and Jefferson Chase, "On the Ethics of Algorithmic Intelligence," Social Research: An International Quarterly 86, no. 2 (2019): 423.
- ⁸⁶⁵ Roberto Simanowski, Michel Brodmer, and Jefferson Chase, "On the Ethics of Algorithmic Intelligence," 424.
- ⁸⁶⁶ Theresa May, "Prime Minister's Address to the World Economic Forum," (Davos, Switzerland: Prime Minister's Office, 2018), 1.
- ⁸⁶⁷ Roberto Simanowski, Michel Brodmer, and Jefferson Chase, "On the Ethics of Algorithmic Intelligence," 425.
- ⁸⁶⁸ Roberto Simanowski, Michel Brodmer, and Jefferson Chase, "On the Ethics of Algorithmic Intelligence," 425.
- ⁸⁶⁹ Luciano Floridi et al., "Ai4people—an Ethical Framework for a Good Ai Society: Opportunities, Risks, Principles, and Recommendations," *Minds and Machines: Journal for Artificial Intelligence, Philosophy* and Cognitive Science 28, no. 4 (2018): 689-707.
- ⁸⁷⁰ Roberto Simanowski, Michel Brodmer, and Jefferson Chase, "On the Ethics of Algorithmic Intelligence," 425.
- ⁸⁷¹ Roberto Simanowski, Michel Brodmer, and Jefferson Chase, "On the Ethics of Algorithmic Intelligence," 425.
- ⁸⁷² Roberto Simanowski, Michel Brodmer, and Jefferson Chase, "On the Ethics of Algorithmic Intelligence," 425.
- ⁸⁷³ Ethics Commission, "Ethics Commission Automated and Connected Driving," (Federal Ministry of Transport and Digital Infrastructure (Germany), 2017), 20.
- ⁸⁷⁴ Roberto Simanowski, Michel Brodmer, and Jefferson Chase, "On the Ethics of Algorithmic Intelligence," 426.
- ⁸⁷⁵ Roberto Simanowski, Michel Brodmer, and Jefferson Chase, "On the Ethics of Algorithmic Intelligence," 426.
- ⁸⁷⁶ Roberto Simanowski, Michel Brodmer, and Jefferson Chase, "On the Ethics of Algorithmic Intelligence," 427.
- ⁸⁷⁷ Kyle Bogosian, "Implementation of Moral Uncertainty in Intelligent Machines," *Minds & Machines* 27, no. 4 (2017): 592.
- ⁸⁷⁸ Kyle Bogosian, "Implementation of Moral Uncertainty in Intelligent Machines," 592.
- ⁸⁷⁹ Kyle Bogosian, "Implementation of Moral Uncertainty in Intelligent Machines," 592.
- ⁸⁸⁰ Carl Shulman, Nick Tarleton, and Henrik Jonsson, "Which Consequentialism? Machine Ethics and Moral Divergence," in AP-CAP 2009: The Fifth Asia-Pacific Computing and Philosophy Conference, ed. Carson Reynolds and Alvaro Cassinelli (University of Tokyo, Japan2009), 23-25.
- ⁸⁸¹ Kyle Bogosian, "Implementation of Moral Uncertainty in Intelligent Machines," 592.
- ⁸⁸² Kyle Bogosian, "Implementation of Moral Uncertainty in Intelligent Machines," 593.
- ⁸⁸³ Kyle Bogosian, "Implementation of Moral Uncertainty in Intelligent Machines," 593.
- ⁸⁸⁴ Kyle Bogosian, "Implementation of Moral Uncertainty in Intelligent Machines," 593.
- ⁸⁸⁵ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 318.
- ⁸⁸⁶ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 318.
- ⁸⁸⁷ Ben Goertzel and Cassio Pennachin, eds., Artificial General Intelligence (Berlin: Springer, 2007), 1-27.
- ⁸⁸⁸ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 318.
- ⁸⁸⁹ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 318.
- ⁸⁹⁰ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 318.
- ⁸⁹¹ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 319.
- ⁸⁹² Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 319.
- ⁸⁹³ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 319.
- ⁸⁹⁴ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 319.
- ⁸⁹⁵ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 319.
- ⁸⁹⁶ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 319.

898 Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 320. ⁸⁹⁹ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 320. ⁹⁰⁰ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 320. ⁹⁰¹ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 320. ⁹⁰² Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 320. ⁹⁰³ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 320. ⁹⁰⁴ Paula Boddington, *Towards a Code of Ethics for Artificial Intelligence*, 89. ⁹⁰⁵ Paula Boddington, *Towards a Code of Ethics for Artificial Intelligence*, 89. ⁹⁰⁶ Paula Boddington, *Towards a Code of Ethics for Artificial Intelligence*, 89. ⁹⁰⁷ Paula Boddington, *Towards a Code of Ethics for Artificial Intelligence*, 89. ⁹⁰⁸ Paula Boddington, *Towards a Code of Ethics for Artificial Intelligence*, 89. ⁹⁰⁹ Paula Boddington, *Towards a Code of Ethics for Artificial Intelligence*, 89. ⁹¹⁰ Paula Boddington, Towards a Code of Ethics for Artificial Intelligence, 89. ⁹¹¹ Paula Boddington, *Towards a Code of Ethics for Artificial Intelligence*, 89. ⁹¹² Paula Boddington, *Towards a Code of Ethics for Artificial Intelligence*, 89. ⁹¹³ Paula Boddington, *Towards a Code of Ethics for Artificial Intelligence*, 90. ⁹¹⁴ Paula Boddington, Towards a Code of Ethics for Artificial Intelligence, 90. ⁹¹⁵ Paula Boddington, Towards a Code of Ethics for Artificial Intelligence, 90. ⁹¹⁶ Paula Boddington, *Towards a Code of Ethics for Artificial Intelligence*, 90. ⁹¹⁷ Paula Boddington, Towards a Code of Ethics for Artificial Intelligence, 90. ⁹¹⁸ Paula Boddington, *Towards a Code of Ethics for Artificial Intelligence*, 90. ⁹¹⁹ Paula Boddington, Towards a Code of Ethics for Artificial Intelligence, 90. ⁹²⁰ Ritu Agarwal et al., "Research Commentary: The Digital Transformation of Healthcare: Current Status and the Road Ahead," Information Systems Research 21, no. 4 (2010): 796-809. 921 Deborah Lupton, "The Commodification of Patient Opinion: The Digital Patient Experience Economy in the Age of Big Data," Sociology of Health & Illness 36, no. 6 (2013): 856-869. ⁹²² Fei Jiang et al., "Artificial Intelligence in Healthcare: Past, Present and Future," Stroke and vascular neurology 2, no. 4 (2017): 230-243. 923 Mike Joyce et al., Digital Therapeutics: Preparing for Takeoff, Mckinsey Insights (New York: McKinsev & Company, 2018), 1-16. ⁹²⁴ Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital Health Research Supported by Artificial Intelligence," BMC Medicine 17, no. 137 (2019): 2. 925 Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital Health Research Supported by Artificial Intelligence," 2. 926 E. H. Kluge, "Health Information, the Fair Information Principles and Ethics," Methods of information in medicine 33, no. 4 (1994): 336-345. ⁹²⁷ E. H. Kluge, "Health Information, the Fair Information Principles and Ethics," 336-345. 928 Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital Health Research Supported by Artificial Intelligence," 2. 929 Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital

⁸⁹⁷ Nick Bostrom and Eliezer Yudkowsky, "The Ethics of Artificial Intelligence," 320.

- Health Research Supported by Artificial Intelligence," 3. ⁹³⁰ Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital
- Health Research Supported by Artificial Intelligence," 3.
- ⁹³¹ Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital Health Research Supported by Artificial Intelligence," 3.
- ⁹³² Camille Nebeker, Rebecca J. Bartlett Ellis, and John Torous, "Development of a Decision-Making Checklist Tool to Support Technology Selection in Digital Health Research," *Translational Behavioral Medicine* 10, no. 4 (2020): 1004-1015.

- ⁹³³ Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital Health Research Supported by Artificial Intelligence," 3.
- ⁹³⁴ Mohammad Adibuzzaman et al., "Big Data in Healthcare the Promises, Challenges and Opportunities from a Research Perspective: A Case Study with a Model Database," *AMIA Annual Symposium proceedings* 2017 (2018): 384-392.
- ⁹³⁵ Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital Health Research Supported by Artificial Intelligence," 4.
- ⁹³⁶ Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital Health Research Supported by Artificial Intelligence," 5.
- ⁹³⁷ Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital Health Research Supported by Artificial Intelligence," 5.
- ⁹³⁸ Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital Health Research Supported by Artificial Intelligence," 5.
- ⁹³⁹ Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital Health Research Supported by Artificial Intelligence," 5.
- ⁹⁴⁰ Camille Nebeker, John Torous, and Rebecca J. Bartlett Ellis, "Building the Case for Actionable Ethics in Digital Health Research Supported by Artificial Intelligence," 5.
- ⁹⁴¹ Jim Warren, "How Will Ai Change Health Care Delivery?," New Zealand Medical Student Journal, no. 28 (2019): 10.
- ⁹⁴² Jim Warren, "How Will Ai Change Health Care Delivery?," 10.
- ⁹⁴³ Sue Wells et al., "Cohort Profile: The Predict Cardiovascular Disease Cohort in New Zealand Primary Care (Predict-Cvd 19)," *International Journal of Epidemiology* 46, no. 1 (2017): 22.
- ⁹⁴⁴ Sue Wells et al., "Cohort Profile: The Predict Cardiovascular Disease Cohort in New Zealand Primary Care (Predict-Cvd 19)," 22.
- 945 Jim Warren, "How Will Ai Change Health Care Delivery?," 10.
- ⁹⁴⁶ Jim Warren, "How Will Ai Change Health Care Delivery?," 10.
- ⁹⁴⁷ Jim Warren, "How Will Ai Change Health Care Delivery?," 11.
- ⁹⁴⁸ Jim Warren, "How Will Ai Change Health Care Delivery?," 11.
- ⁹⁴⁹ Jim Warren, "How Will Ai Change Health Care Delivery?," 11.
- ⁹⁵⁰ Jeana Frost et al., "Patient-Reported Outcomes as a Source of Evidence in Off-Label Prescribing: Analysis of Data from Patientslikeme," *Journal of Medical Internet Research* 13, no. 1 (2011): 6.
- ⁹⁵¹ Eric J. Topol, The Patient Will See You Now : The Future of Medicine Is in Your Hands (New York: Basic Books, 2015), 16.
- ⁹⁵² Henri-Corto Stoeklé et al., "23andme: A New Two-Sided Data-Banking Market Model," *BMC Medical Ethics* 17 (2016): 19.
- ⁹⁵³ Caroline Free et al., "Smoking Cessation Support Delivered Via Mobile Phone Text Messaging (Txt2stop): A Single-Blind, Randomised Trial," *The Lancet* 378, no. 9785 (2011): 49-55.
- ⁹⁵⁴ Jim Warren, "How Will Ai Change Health Care Delivery?," 11.
- 955 Jim Warren, "How Will Ai Change Health Care Delivery?," 11.
- 956 Jim Warren, "How Will Ai Change Health Care Delivery?," 11.
- ⁹⁵⁷ Jim Warren, "How Will Ai Change Health Care Delivery?," 11.
- ⁹⁵⁸ Roman V. Yampolskiy, *Artificial Superintelligence: A Futuristic Approach* (Boca Raton: Taylor & Francis, CRC Press, 2016), 136.
- ⁹⁵⁹ Robin Hanson to Prefer Law to Values, 2009, https://www.overcomingbias.com/2009/10/prefer-law-to-values.html/trackback. 1.
- ⁹⁶⁰ Roman V. Yampolskiy, Artificial Superintelligence: A Futuristic Approach, 136.
- ⁹⁶¹ Eric Drexler, *Engines of Creation*, 1st ed. (Anchor Press/Doubleday, 1986), 10.

- ⁹⁶² Stuart Armstrong, Anders Sandberg, and Nick Bostrom, "Thinking inside the Box: Controlling and Using an Oracle Ai," *Minds and Machines: Journal for Artificial Intelligence, Philosophy and Cognitive Science* 22, no. 4 (2012): 299-324.
- ⁹⁶³ David J. Chalmers, "The Singularity: A Philosophical Analysis," *Journal of Consciousness Studies* 17, no. 9-10 (2010): 7-65.
- ⁹⁶⁴ Roman V. Yampolskiy, Artificial Superintelligence: A Futuristic Approach, 136.
- ⁹⁶⁵ Roman V. Yampolskiy, Artificial Superintelligence: A Futuristic Approach, 137.
- ⁹⁶⁶ Roman V. Yampolskiy, Artificial Superintelligence: A Futuristic Approach, 137.
- ⁹⁶⁷ Roman V. Yampolskiy, Artificial Superintelligence: A Futuristic Approach, 137.
- ⁹⁶⁸ Roman V. Yampolskiy, Artificial Superintelligence: A Futuristic Approach, 137.
- ⁹⁶⁹ Roman V. Yampolskiy, Artificial Superintelligence: A Futuristic Approach, 137.
- ⁹⁷⁰ Roman V. Yampolskiy, Artificial Superintelligence: A Futuristic Approach, 137.
- ⁹⁷¹ Roman V. Yampolskiy, Artificial Superintelligence: A Futuristic Approach, 138.
- ⁹⁷² Roman V. Yampolskiy, Artificial Superintelligence: A Futuristic Approach, 138.
- ⁹⁷³ Roman V. Yampolskiy, Artificial Superintelligence: A Futuristic Approach, 138.
- ⁹⁷⁴ Roman V. Yampolskiy, Artificial Superintelligence: A Futuristic Approach, 138.
- ⁹⁷⁵ Edward H. Shortliffe and G. Octo Barnett, "Biomedical Data: Their Acquisition, Storage, and Use," in *Biomedical Informatics: Computer Applications in Health Care and Biomedicine*, ed. Edward H. Shortliffe and James J. Cimino (London: Springer, 2014), 39.
- ⁹⁷⁶ Edward H. Shortliffe and G. Octo Barnett, "Biomedical Data: Their Acquisition, Storage, and Use," 39.
- ⁹⁷⁷ Edward H. Shortliffe and G. Octo Barnett, "Biomedical Data: Their Acquisition, Storage, and Use," 39.
- ⁹⁷⁸ Edward H. Shortliffe and G. Octo Barnett, "Biomedical Data: Their Acquisition, Storage, and Use," 40.
- ⁹⁷⁹ Ramona Nelson and Nancy Staggers, *Health Informatics-E-Book: An Interprofessional Approach* (Elsevier Health Sciences, 2016), 594.
- ⁹⁸⁰ Ramona Nelson and Nancy Staggers, Health Informatics-E-Book: An Interprofessional Approach, 594.
- ⁹⁸¹ Ramona Nelson and Nancy Staggers, Health Informatics-E-Book: An Interprofessional Approach, 594.
- ⁹⁸² Ramona Nelson and Nancy Staggers, *Health Informatics-E-Book: An Interprofessional Approach*, 596.
- ⁹⁸³ Ramona Nelson and Nancy Staggers, Health Informatics-E-Book: An Interprofessional Approach, 596.
- ⁹⁸⁴ Ramona Nelson and Nancy Staggers, Health Informatics-E-Book: An Interprofessional Approach, 597.
- ⁹⁸⁵ Ramona Nelson and Nancy Staggers, Health Informatics-E-Book: An Interprofessional Approach, 597.
- ⁹⁸⁶ Ramona Nelson and Nancy Staggers, *Health Informatics-E-Book: An Interprofessional Approach*, 597.
- ⁹⁸⁷ Ramona Nelson and Nancy Staggers, Health Informatics-E-Book: An Interprofessional Approach, 597.
- 988 James Lee Sr., "Knowledge Management: The Intellectual Revolution," IIE Solutions 32, no. 10 (2000): 34-37.
- ⁹⁸⁹ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 153.
- ⁹⁹⁰ William A Yasnoff, *Health Information Infrastructure*, ed. Edward H. Shortliffe and James J. Cimino, 4th ed., Biomedical Informatics: Computer Applications in Health Care and Biomedicine (Springer, 2014), 423.
- ⁹⁹¹ William A Yasnoff, Health Information Infrastructure, 423.
- ⁹⁹² William A Yasnoff, *Health Information Infrastructure*, 423.
- 993 William A Yasnoff, Health Information Infrastructure, 423.
- ⁹⁹⁴ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 283.
- ⁹⁹⁵ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 283.
- ⁹⁹⁶ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 283.
- ⁹⁹⁷ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 284.

- ⁹⁹⁸ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 284.
- ⁹⁹⁹ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 284.
- ¹⁰⁰⁰ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 285.
- ¹⁰⁰¹ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 285.
- ¹⁰⁰² Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 285.
- ¹⁰⁰³ Gordon D. Brown, Timothy B. Patrick, and Kalyan S. Pasupathy, *Health Informatics: A Systems Perspective*, 285.
- ¹⁰⁰⁴ Mervat Abdelhak, Sara Grostick, and Mary Alice Hanken, *Health Information: Management of a Strategic Resource* (Elsevier Health Sciences, 2014), 58.
- ¹⁰⁰⁵ Mervat Abdelhak, Sara Grostick, and Mary Alice Hanken, *Health Information: Management of a Strategic Resource*, 58.
- ¹⁰⁰⁶ Mervat Abdelhak, Sara Grostick, and Mary Alice Hanken, *Health Information: Management of a Strategic Resource*, 58.
- ¹⁰⁰⁷ Mervat Abdelhak, Sara Grostick, and Mary Alice Hanken, *Health Information: Management of a Strategic Resource*, 58.
- ¹⁰⁰⁸ Kenneth W. Goodman, Reid Cushman, and Randolph A. Miller, "Ethics in Biomedical and Health Informatics: Users, Standards, and Outcomes," in *Biomedical Informatics*, ed. Edward H. Shortliffe and James J. Cimino (Verlag-London: Springer, 2014), 330.
- ¹⁰⁰⁹ Kenneth W. Goodman, Reid Cushman, and Randolph A. Miller, "Ethics in Biomedical and Health Informatics: Users, Standards, and Outcomes," 330.
- ¹⁰¹⁰ Kenneth W. Goodman, Reid Cushman, and Randolph A. Miller, "Ethics in Biomedical and Health Informatics: Users, Standards, and Outcomes," 330.
- ¹⁰¹¹ Dana C. McWay, Legal and Ethical Aspects of Health Information Management (Cengage Learning, 2015), 131.
- ¹⁰¹² Dana C. McWay, Legal and Ethical Aspects of Health Information Management, 131.
- ¹⁰¹³ Dana C. McWay, Legal and Ethical Aspects of Health Information Management, 131.
- ¹⁰¹⁴ Dana C. McWay, Legal and Ethical Aspects of Health Information Management, 131.
- ¹⁰¹⁵ Dana C. McWay, Legal and Ethical Aspects of Health Information Management, 131.
- ¹⁰¹⁶ Dana C. McWay, Legal and Ethical Aspects of Health Information Management, 131.
- ¹⁰¹⁷ Eileen E. Morrison, *Ethics in Health Administration: A Practical Approach for Decision Makers*, 3rd ed. (Jones & Bartlett Publishers, 2016), 169.
- ¹⁰¹⁸ Eileen E. Morrison, *Ethics in Health Administration: A Practical Approach for Decision Makers*, 169.
- ¹⁰¹⁹ Eileen E. Morrison, *Ethics in Health Administration: A Practical Approach for Decision Makers*, 169.
- ¹⁰²⁰ Eileen E. Morrison, *Ethics in Health Administration: A Practical Approach for Decision Makers*, 169.
- ¹⁰²¹ Eileen E. Morrison, *Ethics in Health Administration: A Practical Approach for Decision Makers*, 169.
- ¹⁰²² Eileen E. Morrison, *Ethics in Health Administration: A Practical Approach for Decision Makers*, 170.
- ¹⁰²³ Eileen E. Morrison, *Ethics in Health Administration: A Practical Approach for Decision Makers*, 170.
- ¹⁰²⁴ Eileen E. Morrison, *Ethics in Health Administration: A Practical Approach for Decision Makers*, 170.
- ¹⁰²⁵ Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," *Medicine, Health Care and Philosophy* 19, no. 4 (2016): 540.
- ¹⁰²⁶ Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," 540.
- ¹⁰²⁷ Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," 540.

- ¹⁰²⁸ Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," 540.
- ¹⁰²⁹ Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," 540.

¹⁰³⁰ Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," 534.

- ¹⁰³¹ Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," 540.
- ¹⁰³² Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," 540.
- ¹⁰³³ Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," 540.
- ¹⁰³⁴ Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," 540.
- ¹⁰³⁵ Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," 540.
- ¹⁰³⁶ Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," 540.
- ¹⁰³⁷ Corine Mouton Dorey, "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'," 540.
- ¹⁰³⁸ Dana C. McWay, Legal and Ethical Aspects of Health Information Management, 135.
- ¹⁰³⁹ Dana C. McWay, Legal and Ethical Aspects of Health Information Management, 135.
- ¹⁰⁴⁰ Dana C. McWay, Legal and Ethical Aspects of Health Information Management, 135.
- ¹⁰⁴¹ Dana C. McWay, Legal and Ethical Aspects of Health Information Management, 135.
- ¹⁰⁴² Dana C. McWay, Legal and Ethical Aspects of Health Information Management, 135.
- ¹⁰⁴³ Dana C. McWay, Legal and Ethical Aspects of Health Information Management, 135.
- ¹⁰⁴⁴ Ramona Nelson and Nancy Staggers, Health Informatics-E-Book: An Interprofessional Approach, 493.
- ¹⁰⁴⁵ Ramona Nelson and Nancy Staggers, *Health Informatics-E-Book: An Interprofessional Approach*, 493.
- ¹⁰⁴⁶ Ramona Nelson and Nancy Staggers, Health Informatics-E-Book: An Interprofessional Approach, 493.
- ¹⁰⁴⁷ Ramona Nelson and Nancy Staggers, Health Informatics-E-Book: An Interprofessional Approach, 493.
- ¹⁰⁴⁸ Ramona Nelson and Nancy Staggers, *Health Informatics-E-Book: An Interprofessional Approach*, 493.
- ¹⁰⁴⁹ Ramona Nelson and Nancy Staggers, Health Informatics-E-Book: An Interprofessional Approach, 493.
- ¹⁰⁵⁰ Mervat Abdelhak, Sara Grostick, and Mary Alice Hanken, *Health Information: Management of a Strategic Resource*, 128.
- ¹⁰⁵¹ Mervat Abdelhak, Sara Grostick, and Mary Alice Hanken, *Health Information: Management of a Strategic Resource*, 129.
- ¹⁰⁵² Mervat Abdelhak, Sara Grostick, and Mary Alice Hanken, *Health Information: Management of a Strategic Resource*, 129.
- ¹⁰⁵³ Mervat Abdelhak, Sara Grostick, and Mary Alice Hanken, *Health Information: Management of a Strategic Resource*, 129.
- ¹⁰⁵⁴ Mervat Abdelhak, Sara Grostick, and Mary Alice Hanken, *Health Information: Management of a Strategic Resource*, 129.
- ¹⁰⁵⁵ Eileen E. Morrison, Ethics in Health Administration: A Practical Approach for Decision Makers, 247.
- ¹⁰⁵⁶ Eileen E. Morrison, Ethics in Health Administration: A Practical Approach for Decision Makers, 247.
- ¹⁰⁵⁷ Eileen E. Morrison, Ethics in Health Administration: A Practical Approach for Decision Makers, 248.
- ¹⁰⁵⁸ Eileen E. Morrison, *Ethics in Health Administration: A Practical Approach for Decision Makers*, 248.
- ¹⁰⁵⁹ Eileen E. Morrison, *Ethics in Health Administration: A Practical Approach for Decision Makers*, 248.
- ¹⁰⁶⁰ Eileen E. Morrison, *Ethics in Health Administration: A Practical Approach for Decision Makers*, 248.
- ¹⁰⁶¹ Eileen E. Morrison, *Ethics in Health Administration: A Practical Approach for Decision Makers*, 248.
- ¹⁰⁶² Eileen E. Morrison, Ethics in Health Administration: A Practical Approach for Decision Makers, 248.
- ¹⁰⁶³ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management : Concepts, Tools and Methods*, 236.
- ¹⁰⁶⁴ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management : Concepts, Tools and Methods*, 236.
- ¹⁰⁶⁵ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management : Concepts, Tools and Methods*, 237.
- ¹⁰⁶⁶ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management : Concepts, Tools and Methods*, 237.
- ¹⁰⁶⁷ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management : Concepts, Tools and Methods*, 237.
- ¹⁰⁶⁸ Katherine O'Keefe and Daragh O'Brien, *Ethical Data and Information Management : Concepts, Tools and Methods*, 237.
- ¹⁰⁶⁹ Sarah Coakley, "Introductory Remarks: The Theology and Ethics of Contemporary Dying," *Studies in Christian Ethics* 29, no. 3 (2016): 246.
- ¹⁰⁷⁰ Sarah Coakley, "Introductory Remarks: The Theology and Ethics of Contemporary Dying," 246.
- ¹⁰⁷¹ Philippe Ariès, Western Attitudes toward Death: From the Middle Ages to the Present (JHU Press, 1974), 85.
- ¹⁰⁷² Philippe Ariès, Western Attitudes toward Death: From the Middle Ages to the Present, 87.
- ¹⁰⁷³ Philippe Ariès, Western Attitudes toward Death: From the Middle Ages to the Present, 87-88.
- ¹⁰⁷⁴ Philippe Ariès, Western Attitudes toward Death: From the Middle Ages to the Present, 88.
- ¹⁰⁷⁵ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well* (Cambridge, Massachusetts: The MIT Press, 2015), 43.
- ¹⁰⁷⁶ Pedro Gozalo et al., "End-of-Life Transitions among Nursing Home Residents with Cognitive Issues," *The New England Journal of Medicine* 365, no. 13 (2011): 1212-1221.
- ¹⁰⁷⁷ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 43.
- ¹⁰⁷⁸ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 46.
- ¹⁰⁷⁹ Brandy Schillace, *Death's Summer Coat: What the History of Death and Dying Teaches Us About Life and Living* (New York, NY: Pegasus Books, 2015), 72.
- ¹⁰⁸⁰ Philippe Ariès, At the Hour of Our Death, 1st American ed. (New York, NY: Knopf, 1981), 5.
- ¹⁰⁸¹ Philippe Ariès, At the Hour of Our Death, 5.
- ¹⁰⁸² Brandy Schillace, *Death's Summer Coat: What the History of Death and Dying Teaches Us About Life and Living*, 73.
- ¹⁰⁸³ Brandy Schillace, Death's Summer Coat: What the History of Death and Dying Teaches Us About Life and Living, 73.
- ¹⁰⁸⁴ Paul Ramsey, "The Indignity of 'Death with Dignity'," The Hastings Center Studies 2, no. 2 (1974): 54.
- ¹⁰⁸⁵ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 48.
- ¹⁰⁸⁶ Ronald M. Davis and Myron Genel, "Good Care of the Dying Patient," *Journal of the American Medical Association* 275, no. 6 (1996): 474-478.
- ¹⁰⁸⁷ Ronald M. Davis and Myron Genel, "Good Care of the Dying Patient," 475.
- ¹⁰⁸⁸ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 49.
- ¹⁰⁸⁹ Ronald M. Davis and Myron Genel, "Good Care of the Dying Patient," 474-478.
- ¹⁰⁹⁰ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 49.
- ¹⁰⁹¹ Jeffrey P. Bishop, *The Anticipatory Corpse: Medicine, Power, and the Care of the Dying* (Notre Dame, IN: University of Notre Dame Press, 2011), 60.

- ¹⁰⁹² Atul Gawande, *Being Mortal: Medicine and What Matters in the End*, 1st ed. (New York: Metropolitan Books: Henry Holt & Company, 2014), 79.
- ¹⁰⁹³ Jeffrey P. Bishop, *The Anticipatory Corpse: Medicine, Power, and the Care of the Dying*, 69.

- ¹⁰⁹⁵ Jeffrey P. Bishop, The Anticipatory Corpse: Medicine, Power, and the Care of the Dying, 69.
- ¹⁰⁹⁶ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 80.
- ¹⁰⁹⁷ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 80.
- ¹⁰⁹⁸ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 80.
- ¹⁰⁹⁹ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 80.
- ¹¹⁰⁰ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 80.
- ¹¹⁰¹ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 80.
- ¹¹⁰² Lydia S. Dugdale, Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well, 80.
- ¹¹⁰³ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 80.
- ¹¹⁰⁴ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 81.
- ¹¹⁰⁵ Daniel Callahan, *The Troubled Dream of Life: In Search of a Peaceful Death* (Georgetown University Press, 2000), 54.
- ¹¹⁰⁶ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 81.
- ¹¹⁰⁷ Lydia S. Dugdale, *Dying in the Twenty-First Century: Toward a New Ethical Framework for the Art of Dying Well*, 81.
- ¹¹⁰⁸ Sarah Coakley, "Introductory Remarks: The Theology and Ethics of Contemporary Dying," 246.
- ¹¹⁰⁹ Sarah Coakley, "Introductory Remarks: The Theology and Ethics of Contemporary Dying," 246-247.
- ¹¹¹⁰ Sarah Coakley, "Introductory Remarks: The Theology and Ethics of Contemporary Dying," 247.
- ¹¹¹¹ Sarah Coakley, "Introductory Remarks: The Theology and Ethics of Contemporary Dying," 247.
- ¹¹¹² Sarah Coakley, "Introductory Remarks: The Theology and Ethics of Contemporary Dying," 247.
- ¹¹¹³ Sarah Coakley, "Introductory Remarks: The Theology and Ethics of Contemporary Dying," 247.
- ¹¹¹⁴ Sarah Coakley, "Introductory Remarks: The Theology and Ethics of Contemporary Dying," 247.
- ¹¹¹⁵ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review," *Journal of Medical Systems* 31, no. 2 (2007): 97-101.
- ¹¹¹⁶ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review,"98.
- ¹¹¹⁷ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review,"98.
- ¹¹¹⁸ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review," 98.
- ¹¹¹⁹ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review,"98.
- ¹¹²⁰ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review,"98.

¹⁰⁹⁴ Sarah Coakley, "Introductory Remarks: The Theology and Ethics of Contemporary Dying," 246.

- ¹¹²¹ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review," 98-99.
- ¹¹²² Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review,"99.
- ¹¹²³ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review,"99.
- ¹¹²⁴ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review,"99.
- ¹¹²⁵ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review,"99.
- ¹¹²⁶ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review," 100.
- ¹¹²⁷ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review," 100.
- ¹¹²⁸ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review," 100.
- ¹¹²⁹ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review," 100.
- ¹¹³⁰ Lia Willis, George Demiris, and Debra P. Oliver, "Internet Use by Hospice Families and Providers: A Review," 100.
- ¹¹³¹ Karen E. Steinhauser et al., "Factors Considered Important at the End of Life by Patients, Family, Physicians, and Other Care Providers," *JAMA: The Journal of the American Medical Association* 284, no. 19 (2000): 2476-2482.
- ¹¹³² Eduardo Bruera and Sriram Yennurajalingam, "Palliative Care in Advanced Cancer Patients: How and When?," Oncologist 17, no. 2 (2012): 267-273.
- ¹¹³³ Jelle van Gurp et al., "How Outpatient Palliative Care Teleconsultation Facilitates Empathic Patient-Professional Relationships: A Qualitative Study," *PLoS ONE* 10, no. 4 (2015): 2.
- ¹¹³⁴ Timothy E. Quill and Amy P. Abernethy, "Generalist Plus Specialist Palliative Care Creating a More Sustainable Model," *The New England Journal of Medicine* 368, no. 13 (2013): 1173-1175.
- ¹¹³⁵ Emily Collis and R. Al-Qurainy, "Care of the Dying Patient in the Community," *BMJ: British Medical Journal* 347, no. 7915 (2013): 28.
- ¹¹³⁶ Jelle van Gurp et al., "How Outpatient Palliative Care Teleconsultation Facilitates Empathic Patient-Professional Relationships: A Qualitative Study," 2.
- ¹¹³⁷ Jelle van Gurp et al., "How Outpatient Palliative Care Teleconsultation Facilitates Empathic Patient-Professional Relationships: A Qualitative Study," 2.
- ¹¹³⁸ K. Kroenke et al., "Effect of Telecare Management on Pain and Depression in Patients with Cancer: A Randomized Trial," *JAMA: Journal of the American Medical Association* 304, no. 2 (2010): 163-171.
- ¹¹³⁹ Pamela Whitten, Gary Doolittle, and Michael Mackert, "Telehospice in Michigan: Use and Patient Acceptance," *American Journal of Hospice & Palliative Medicine* 21, no. 3 (2004): 191-195.
- ¹¹⁴⁰ Jeannette Pols, "Wonderful Webcams: About Active Gazes and Invisible Technologies," Science, Technology, & Human Values 36, no. 4 (2011): 451-473.
- ¹¹⁴¹ Jelle van Gurp et al., "How Outpatient Palliative Care Teleconsultation Facilitates Empathic Patient-Professional Relationships: A Qualitative Study," 2.
- ¹¹⁴² Jelle van Gurp et al., "How Outpatient Palliative Care Teleconsultation Facilitates Empathic Patient-Professional Relationships: A Qualitative Study," 2.
- ¹¹⁴³ Jose Pereira et al., "Palliative Cancer Patients and Their Families on the Internet: Motivation and Impact," *Journal of Palliative Care* 16, no. 4 (2000): 15.
- ¹¹⁴⁴ Jose Pereira et al., "Palliative Cancer Patients and Their Families on the Internet: Motivation and Impact," 15.

- ¹¹⁴⁵ Raisa B. Deber, Nancy Kraetschmer, and Jane Irvine, "What Role Do Patients Wish to Play in Treatment Decision Making?," Archives of Internal Medicine 156, no. 13 (1996): 1414-1420.
- ¹¹⁴⁶ Lawrence E. Widman and David A. Tong, "Requests for Medical Advice from Patients and Families to Health Care Providers Who Publish on the World Wide Web," *Archives of Internal Medicine* 157, no. 2 (1997): 209-212.
- ¹¹⁴⁷ Jose Pereira et al., "Palliative Cancer Patients and Their Families on the Internet: Motivation and Impact," 16.
- ¹¹⁴⁸ Hilary Bower, "Internet Sees Growth of Unverified Health Claims," *British Medical Journal* 313, no. 7054 (1996): 381.
- ¹¹⁴⁹ Jose Pereira et al., "Palliative Cancer Patients and Their Families on the Internet: Motivation and Impact," 16.
- ¹¹⁵⁰ Jose Pereira et al., "Palliative Cancer Patients and Their Families on the Internet: Motivation and Impact," 16.
- ¹¹⁵¹ Thomas N. Robinson, "Community Health Behavior Change through Computer Network Health Promotion: Preliminary Findings from Stanford Health-Net," *Computer Methods and Programs in Biomedicine* 30, no. 2 (1989): 137-144.
- ¹¹⁵² David H. Gustafson et al., "Computer Support for Elderly Women with Breast Cancer," JAMA: The Journal of the American Medical Association 280, no. 15 (1998): 1305.
- ¹¹⁵³ Samantha J. Scolamiero, "Support Groups in Cyberspace," M.D. Computing 14, no. 1 (1997): 12-14.
- ¹¹⁵⁴ Paula Klemm, Karla Reppert, and Lori Visich, "A Nontraditional Cancer Support Group: The Internet," *Computers in Nursing* 16, no. 1 (1998): 33-36.
- ¹¹⁵⁵ Jose Pereira et al., "Palliative Cancer Patients and Their Families on the Internet: Motivation and Impact," 16.
- ¹¹⁵⁶ Henk ten Have, *Vulnerability: Challenging Bioethics* (London ; New York: Routledge, Taylor & Francis Group, 2016), 211.
- ¹¹⁵⁷ Henk ten Have, *Vulnerability: Challenging Bioethics*, 211.
- ¹¹⁵⁸ Henk ten Have, *Vulnerability: Challenging Bioethics*, 211.
- ¹¹⁵⁹ Henk ten Have, *Vulnerability: Challenging Bioethics*, 211.
- ¹¹⁶⁰ Henk ten Have, *Vulnerability: Challenging Bioethics*, 212.
- ¹¹⁶¹ Henk ten Have, *Vulnerability: Challenging Bioethics*, 212.
- ¹¹⁶² Henk ten Have, *Vulnerability: Challenging Bioethics*, 212.
- ¹¹⁶³ Henk ten Have, *Vulnerability: Challenging Bioethics*, 212.
- ¹¹⁶⁴ Henk ten Have, *Vulnerability: Challenging Bioethics*, 212.
- ¹¹⁶⁵ Shan Mohammed and Elizabeth Peter, "Rituals, Death and the Moral Practice of Medical Futility," *Nursing Ethics* 16, no. 3 (2009): 292-301.
- ¹¹⁶⁶ Alberto Coustasse, Theresa Quiroz, and Sue G. Lurie, "To the Bitter End: Disparities in End-of-Life Care," *Journal of Hospital Marketing & Public Relations* 18, no. 2 (2008): 167-185.
- ¹¹⁶⁷ Lawrence J. Schneiderman, Nancy S. Jecker, and Albert R. Jonsen, "Medical Futility: Its Meaning and Ethical Implications," *Annals of Internal Medicine* 112, no. 12 (1990): 951.
- ¹¹⁶⁸ Lawrence J. Schneiderman, Nancy S. Jecker, and Albert R. Jonsen, "Medical Futility: Its Meaning and Ethical Implications," 949-954.
- ¹¹⁶⁹ Alberto Coustasse, Theresa Quiroz, and Sue G. Lurie, "To the Bitter End: Disparities in End-of-Life Care," 167-185.
- ¹¹⁷⁰ Institute of Medicine, *Dying in America: Improving Quality and Honoring Individual Preferences near the End* of Life (Washington, DC: The National Academies Press, 2015), 16.
- ¹¹⁷¹ H. Montgomery, M. Grocott, and M. Mythen, "Critical Care at the End of Life: Balancing Technology with Compassion and Agreeing When to Stop," *British Journal of Anaesthesia* 119, no. S1 (2017): 87.
- ¹¹⁷² H. Montgomery, M. Grocott, and M. Mythen, "Critical Care at the End of Life: Balancing Technology with Compassion and Agreeing When to Stop," 87.
- ¹¹⁷³ H. Montgomery, M. Grocott, and M. Mythen, "Critical Care at the End of Life: Balancing Technology with Compassion and Agreeing When to Stop," 88.
- ¹¹⁷⁴ Terhikki T. Miettinen and Reijo S. Tilvis, "Medical Futility as a Cause of Suffering of Dying Patients—the Family Members' Perspective," *Journal of Palliative Care* 15, no. 2 (1999): 26-29.

- ¹¹⁷⁵ Terhikki T. Miettinen and Reijo S. Tilvis, "Medical Futility as a Cause of Suffering of Dying Patients—the Family Members' Perspective," 26-29.
- ¹¹⁷⁶ Terhikki T. Miettinen and Reijo S. Tilvis, "Medical Futility as a Cause of Suffering of Dying Patients—the Family Members' Perspective," 26-29.
- ¹¹⁷⁷ Henk ten Have and David Clark, *The Ethics of Palliative Care : European Perspectives*, Facing Death (Buckingham England ; Phildelphia, PA: Open University Press, 2002), 216.
- ¹¹⁷⁸ Henk ten Have and David Clark, *The Ethics of Palliative Care : European Perspectives*, 216.
- ¹¹⁷⁹ Janet Sopcheck, "Social, Economic, and Political Issues Affecting End-of-Life Care," *Policy, Politics & Nursing Practice* 17, no. 1 (2016): 33.
- ¹¹⁸⁰ Janet Sopcheck, "Social, Economic, and Political Issues Affecting End-of-Life Care," 33.
- ¹¹⁸¹ Carol O. Long et al., "Home Health and Hospice Use of Computers and the Internet," *Home Healthcare Nurse* 18, no. 10 (2000): 45.
- ¹¹⁸² Carol O. Long et al., "Home Health and Hospice Use of Computers and the Internet," 45.
- ¹¹⁸³ Carol O. Long et al., "Home Health and Hospice Use of Computers and the Internet," 46.
- ¹¹⁸⁴ Carol O. Long et al., "Home Health and Hospice Use of Computers and the Internet," 46.
- ¹¹⁸⁵ Carol O. Long et al., "Home Health and Hospice Use of Computers and the Internet," 46.
- ¹¹⁸⁶ H. Montgomery, M. Grocott, and M. Mythen, "Critical Care at the End of Life: Balancing Technology with Compassion and Agreeing When to Stop," 87.
- ¹¹⁸⁷ H. Montgomery, M. Grocott, and M. Mythen, "Critical Care at the End of Life: Balancing Technology with Compassion and Agreeing When to Stop," 87.
- ¹¹⁸⁸ H. Montgomery, M. Grocott, and M. Mythen, "Critical Care at the End of Life: Balancing Technology with Compassion and Agreeing When to Stop," 86.
- ¹¹⁸⁹ H. Montgomery, M. Grocott, and M. Mythen, "Critical Care at the End of Life: Balancing Technology with Compassion and Agreeing When to Stop," 87.
- ¹¹⁹⁰ H. Montgomery, M. Grocott, and M. Mythen, "Critical Care at the End of Life: Balancing Technology with Compassion and Agreeing When to Stop," 87.
- ¹¹⁹¹ George Demiris, Debra Parker Oliver, and Elaine Wittenberg-Lyles, "Technologies to Support End of Life Care," *Seminars in Oncology Nursing* 27, no. 3 (2011): 3.
- ¹¹⁹² George Demiris, Debra Parker Oliver, and Elaine Wittenberg-Lyles, "Technologies to Support End of Life Care," 3.
- ¹¹⁹³ George Demiris, Debra Parker Oliver, and Elaine Wittenberg-Lyles, "Technologies to Support End of Life Care," 3.
- ¹¹⁹⁴ George Demiris, Debra Parker Oliver, and Elaine Wittenberg-Lyles, "Technologies to Support End of Life Care," 4.
- ¹¹⁹⁵ George Demiris, Debra Parker Oliver, and Elaine Wittenberg-Lyles, "Technologies to Support End of Life Care," 3.
- ¹¹⁹⁶ Maria Canellopoulou-Bottis and Kenneth E. Himma, "The Digital Divide: A Perspective for the Future," in *The Handbook of Information and Computer Ethics*, ed. Kenneth Einar Himma and Herman T. Tavani (Hoboken, N.J.: Wiley, 2008), 623.
- ¹¹⁹⁷ Maria Canellopoulou-Bottis and Kenneth E. Himma, "The Digital Divide: A Perspective for the Future," 622.
- ¹¹⁹⁸ Statista, "Global Internet Penetration Rate," (Statista Research Department, 2022).
- ¹¹⁹⁹ Dan K. Morhaim and Keshia M. Pollack, "End-of-Life Care Issues: A Personal, Economic, Public Policy, and Public Health Crisis," *American Journal of Public Health* 103, no. 6 (2013): 112.
- ¹²⁰⁰ Dan K. Morhaim and Keshia M. Pollack, "End-of-Life Care Issues: A Personal, Economic, Public Policy, and Public Health Crisis," 21.
- ¹²⁰¹ Dan K. Morhaim and Keshia M. Pollack, "End-of-Life Care Issues: A Personal, Economic, Public Policy, and Public Health Crisis," 21.
- ¹²⁰² Maria Canellopoulou-Bottis and Kenneth E. Himma, "The Digital Divide: A Perspective for the Future," 622.

- ¹²⁰³ Dan K. Morhaim and Keshia M. Pollack, "End-of-Life Care Issues: A Personal, Economic, Public Policy, and Public Health Crisis," 21.
- ¹²⁰⁴ Kenneth W. Goodman, Reid Cushman, and Randolph A. Miller, "Ethics in Biomedical and Health Informatics: Users, Standards, and Outcomes," 344.
- ¹²⁰⁵ Kenneth W. Goodman, Reid Cushman, and Randolph A. Miller, "Ethics in Biomedical and Health Informatics: Users, Standards, and Outcomes," 344.
- ¹²⁰⁶ Jackie Crossen-Sills, Irene Toomey, and Mary E. Doherty, "Technology and Home Care: Implementing Systems to Enhance Aging in Place," *Nursing Clinics of North America* 44, no. 2 (2009): 12.
- ¹²⁰⁷ Kenneth W. Goodman, Reid Cushman, and Randolph A. Miller, "Ethics in Biomedical and Health Informatics: Users, Standards, and Outcomes," 344.
- ¹²⁰⁸ Jackie Crossen-Sills, Irene Toomey, and Mary E. Doherty, "Technology and Home Care: Implementing Systems to Enhance Aging in Place," 21.
- ¹²⁰⁹ Kenneth W. Goodman, Reid Cushman, and Randolph A. Miller, "Ethics in Biomedical and Health Informatics: Users, Standards, and Outcomes," 345.
- ¹²¹⁰ Jackie Crossen-Sills, Irene Toomey, and Mary E. Doherty, "Technology and Home Care: Implementing Systems to Enhance Aging in Place," 12.
- ¹²¹¹ Kenneth W. Goodman, Reid Cushman, and Randolph A. Miller, "Ethics in Biomedical and Health Informatics: Users, Standards, and Outcomes," 345.
- ¹²¹² World Health Organization, "World Report on Ageing and Health," (Geneva, Switzerland2015), 3.
- ¹²¹³ Heather Draper and Tom Sorell, "Ethical Values and Social Care Robots for Older People: An International Qualitative Study," *Ethics and Information Technology* 19, no. 1 (2017): 49.
- ¹²¹⁴ Richard Suzman et al., "Health in an Ageing World—What Do We Know?," *The Lancet* 385, no. 9967 (2015): 484-486.
- ¹²¹⁵ European Commission et al., "Short Analytical Web Note : Demography Report. 3/2015," (Publications Office, 2015), 4.
- ¹²¹⁶ European Commission et al., "Short Analytical Web Note : Demography Report. 3/2015," 4.
- ¹²¹⁷ European Commission et al., "Short Analytical Web Note : Demography Report. 3/2015," 5.
- ¹²¹⁸ European Commission et al., "Short Analytical Web Note : Demography Report. 3/2015," 5.
- ¹²¹⁹ Heather Draper and Tom Sorell, "Ethical Values and Social Care Robots for Older People: An International Qualitative Study," 50.
- ¹²²⁰ Jennifer A. Parks, "Lifting the Burden of Women's Care Work: Should Robots Replace the 'Human Touch'?," *Hypatia* 25, no. 1 (2010): 100.
- ¹²²¹ Jennifer A. Parks, "Lifting the Burden of Women's Care Work: Should Robots Replace the 'Human Touch'?," 100.
- ¹²²² Jennifer A. Parks, "Lifting the Burden of Women's Care Work: Should Robots Replace the 'Human Touch'?," 100.
- ¹²²³ Amanda Sharkey and Noel Sharkey, "Granny and the Robots: Ethical Issues in Robot Care for the Elderly," *Ethics and Information Technology* 14, no. 1 (2012): 28.
- ¹²²⁴ Jennifer A. Parks, "Lifting the Burden of Women's Care Work: Should Robots Replace the 'Human Touch'?," 101.
- ¹²²⁵ Jennifer A. Parks, "Lifting the Burden of Women's Care Work: Should Robots Replace the 'Human Touch'?," 101.
- ¹²²⁶ Amanda Sharkey and Noel Sharkey, "Granny and the Robots: Ethical Issues in Robot Care for the Elderly," 28.
- ¹²²⁷ Amanda Sharkey and Noel Sharkey, "Granny and the Robots: Ethical Issues in Robot Care for the Elderly," 28.
- ¹²²⁸ Jennifer A. Parks, "Lifting the Burden of Women's Care Work: Should Robots Replace the 'Human Touch'?," 101.
- ¹²²⁹ Anita Woll and Tone Bratteteig, "A Trajectory for Technology-Supported Elderly Care Work," Computer Supported Cooperative Work: The Journal of Collaborative Computing 28, no. 1/2 (2019): 131.
- ¹²³⁰ Anita Woll and Tone Bratteteig, "A Trajectory for Technology-Supported Elderly Care Work," 131.

- ¹²³¹ Anita Woll and Tone Bratteteig, "A Trajectory for Technology-Supported Elderly Care Work," 131.
- ¹²³² Anita Woll and Tone Bratteteig, "A Trajectory for Technology-Supported Elderly Care Work," 131.
- ¹²³³ Anita Woll and Tone Bratteteig, "A Trajectory for Technology-Supported Elderly Care Work," 131.
- ¹²³⁴ Tone Bratteteig and Ina Wagner, "Moving Healthcare to the Home: The Work to Make Homecare Work" (paper presented at the ECSCW 2013: Proceedings of the 13th European Conference on Computer Supported Cooperative Work, Paphos, Cyprus, September 2013), 145.
- ¹²³⁵ Anita Woll and Tone Bratteteig, "Activity Theory as a Framework to Analyze Technology-Mediated Elderly Care," *Mind, Culture, and Activity* 25, no. 1 (2018): 6-21.
- ¹²³⁶ Anita Woll and Tone Bratteteig, "A Trajectory for Technology-Supported Elderly Care Work," 132.
- ¹²³⁷ Anita Woll and Tone Bratteteig, "A Trajectory for Technology-Supported Elderly Care Work," 132.
- ¹²³⁸ Anita Woll and Tone Bratteteig, "Activity Theory as a Framework to Analyze Technology-Mediated Elderly Care," 6-21.
- ¹²³⁹ Hui-Fen Mao et al., "Indicators of Perceived Useful Dementia Care Assistive Technology: Caregivers' Perspectives," *Geriatrics and Gerontology International* 15, no. 8 (2015): 1049-1057.
- ¹²⁴⁰ Anita Woll and Tone Bratteteig, "Activity Theory as a Framework to Analyze Technology-Mediated Elderly Care," 6-21.
- ¹²⁴¹ Anita Woll and Tone Bratteteig, "A Trajectory for Technology-Supported Elderly Care Work," 132.
- ¹²⁴² Anita Woll and Tone Bratteteig, "A Trajectory for Technology-Supported Elderly Care Work," 132.
- ¹²⁴³ Sandra Bedaf, Gert Jan Gelderblom, and Luc De Witte, "Overview and Categorization of Robots Supporting Independent Living of Elderly People: What Activities Do They Support and How Far Have They Developed," Assistive Technology 27, no. 2 (2015): 88-100.
- ¹²⁴⁴ Roger Bemelmans et al., "Socially Assistive Robots in Elderly Care: A Systematic Review into Effects and Effectiveness," *Journal of the American Medical Directors Association* 13, no. 2 (2012): 114-120.
- ¹²⁴⁵ Reza Kachouie et al., "Socially Assistive Robots in Elderly Care: A Mixed-Method Systematic Literature Review," *International Journal of Human-Computer Interaction* 30, no. 5 (2014): 369-393.
- ¹²⁴⁶ Reza Kachouie et al., "Socially Assistive Robots in Elderly Care: A Mixed-Method Systematic Literature Review," 385.
- ¹²⁴⁷ Reza Kachouie et al., "Socially Assistive Robots in Elderly Care: A Mixed-Method Systematic Literature Review," 385.
- ¹²⁴⁸ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," *Philosophy & Technology* 24, no. 3 (2011): 252.
- ¹²⁴⁹ Amanda Sharkey and Noel Sharkey, "Granny and the Robots: Ethical Issues in Robot Care for the Elderly," 27.
- ¹²⁵⁰ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," Science & Engineering Ethics 19, no. 2 (2013): 409.
- ¹²⁵¹ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 409.
- ¹²⁵² D. Howcroft, N. Mitev, and M. Wilson, "What We May Learn from the Social Shaping of Technology Approach," in *Social Theory and Philosophy for Information Systems*, ed. John Mingers and Leslie Willcocks (Chichester, West Sussex, England: John Wiley and Sons, Inc, 2004), 329-371.
- ¹²⁵³ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 409.
- ¹²⁵⁴ AGE Platform Europe, "Accompany," European Commussion, https://www.age-platform.eu/goodpractice/accompany.
- ¹²⁵⁵ Farshid Amirabdollahian et al., "Assistive Technology Design and Development for Acceptable Robotics Companions for Ageing Years," *Journal of Behavioral Robotics* 4, no. 2 (2013): 94-112.
- ¹²⁵⁶ Farshid Amirabdollahian et al., "Assistive Technology Design and Development for Acceptable Robotics Companions for Ageing Years," 94-112.
- ¹²⁵⁷ Heather Draper and Tom Sorell, "Ethical Values and Social Care Robots for Older People: An International Qualitative Study," 50.
- ¹²⁵⁸ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 253.

- ¹²⁵⁹ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 253.
- ¹²⁶⁰ Carol Gilligan, In a Different Voice: Psychological Theory and Women's Development (Cambridge, Massachusetts: Harvard University Press, 1993), 8.
- ¹²⁶¹ Carol Gilligan, In a Different Voice: Psychological Theory and Women's Development, 8.
- ¹²⁶² Carol Gilligan, In a Different Voice: Psychological Theory and Women's Development, 10.
- ¹²⁶³ Carol Gilligan, "Moral Orientation and Moral Development," in *Women and Moral Theory*, ed. Eva Kittay and Diana Meyers (Totowa, NJ: Rowman & Littlefield, 1987), 25.
- ¹²⁶⁴ Joan C. Tronto, Moral Boundaries: A Political Argument for an Ethic of Care (New York: Routledge, 1993), 127.
- ¹²⁶⁵ Joan C. Tronto, Moral Boundaries, 127.
- ¹²⁶⁶ Joan C. Tronto, Moral Boundaries, 127.
- ¹²⁶⁷ Linus Vanlaere and Chris Gastmans, "A Personalist Approach to Care Ethics," *Nursing Ethics* 18, no. 2 (2011): 162.
- ¹²⁶⁸ Linus Vanlaere and Chris Gastmans, "A Personalist Approach to Care Ethics," 162.
- ¹²⁶⁹ Robert Scott Stewart, "Some Ethical Issues in Treating and Caring for People with Dementia," *Online Journal of Health Ethics* 14, no. 1 (2018): 4.
- ¹²⁷⁰ Robert Scott Stewart, "Some Ethical Issues in Treating and Caring for People with Dementia," 4.
- ¹²⁷¹ Robert Scott Stewart, "Some Ethical Issues in Treating and Caring for People with Dementia," 4.
- ¹²⁷² Robert Scott Stewart, "Some Ethical Issues in Treating and Caring for People with Dementia," 4.
- ¹²⁷³ Robert Scott Stewart, "Some Ethical Issues in Treating and Caring for People with Dementia," 4.
- ¹²⁷⁴ Robert Scott Stewart, "Some Ethical Issues in Treating and Caring for People with Dementia," 4.
- ¹²⁷⁵ Robert Scott Stewart, "Some Ethical Issues in Treating and Caring for People with Dementia," 5.
- ¹²⁷⁶ Robert Scott Stewart, "Some Ethical Issues in Treating and Caring for People with Dementia," 5.
- ¹²⁷⁷ Carol Gilligan, In a Different Voice: Psychological Theory and Women's Development, 8-9.
- ¹²⁷⁸ Carol Gilligan, In a Different Voice: Psychological Theory and Women's Development, 8-9.
- ¹²⁷⁹ Carol Gilligan, In a Different Voice: Psychological Theory and Women's Development, 8-9.
- ¹²⁸⁰ Robert Scott Stewart, "Some Ethical Issues in Treating and Caring for People with Dementia," 5.
- ¹²⁸¹ Ruud ter Meulen, "Ethics of Care," in *The Sage Handbook of Health Care Ethics: Core and Emerging Issues*, ed. Henk ten Have Ruth Chadwick, and Eric M. Meslin (London: SAGE Publications Ltd, 2011), 45.
- ¹²⁸² Ruud ter Meulen, "The Sage Handbook of Health Care Ethics," 45.
- ¹²⁸³ Virginia Held, The Ethics of Care: Personal, Political, and Global (Oxford: Oxford University Press, 2006), 13.
- ¹²⁸⁴ Ruud ter Meulen, "The Sage Handbook of Health Care Ethics," 45.
- ¹²⁸⁵ Ruud ter Meulen, "The Sage Handbook of Health Care Ethics," 46.
- ¹²⁸⁶ Chris Gastmans, "Dignity-Enhancing Nursing Care: A Foundational Ethical Framework," *Nursing Ethics* 20, no. 2 (2013): 142-149.
- ¹²⁸⁷ Kari Lislerud Smebye et al., "Ethical Dilemmas Concerning Autonomy When Persons with Dementia Wish to Live at Home: A Qualitative, Hermeneutic Study," *BMC Health Services Research* 16 (2016): 9.
- ¹²⁸⁸ Joan C. Tronto, Moral Boundaries, 3.
- ¹²⁸⁹ Kari Lislerud Smebye et al., "Ethical Dilemmas Concerning Autonomy When Persons with Dementia Wish to Live at Home: A Qualitative, Hermeneutic Study," 9.
- ¹²⁹⁰ Virginia Held, The Ethics of Care, 84.
- ¹²⁹¹ Virginia Held, The Ethics of Care, 84.
- ¹²⁹² Virginia Held, The Ethics of Care, 84.
- ¹²⁹³ Virginia Held, The Ethics of Care, 84.
- ¹²⁹⁴ Virginia Held, The Ethics of Care, 84.
- ¹²⁹⁵ Virginia Held, The Ethics of Care, 84.
- ¹²⁹⁶ Linus Vanlaere and Chris Gastmans, "A Personalist Approach to Care Ethics," 169.
- ¹²⁹⁷ Louis Janssens, "Artificial Insemination: Ethical Considerations," *Louvain Studies* 8, no. 1 (1980): 9.

- ¹²⁹⁸ Linus Vanlaere and Chris Gastmans, "A Personalist Approach to Care Ethics," 169.
- ¹²⁹⁹ Linus Vanlaere and Chris Gastmans, "A Personalist Approach to Care Ethics," 169.
- ¹³⁰⁰ Linus Vanlaere and Chris Gastmans, "A Personalist Approach to Care Ethics," 170.
- ¹³⁰¹ Linus Vanlaere and Chris Gastmans, "A Personalist Approach to Care Ethics," 170.
- ¹³⁰² Linus Vanlaere and Chris Gastmans, "A Personalist Approach to Care Ethics," 170.
- ¹³⁰³ Linus Vanlaere and Chris Gastmans, "A Personalist Approach to Care Ethics," 170.
- ¹³⁰⁴ Linus Vanlaere and Chris Gastmans, "A Personalist Approach to Care Ethics," 170.
- ¹³⁰⁵ Margaret Urban Walker, *Moral Understandings: A Feminist Study in Ethics* (New York: Oxford University Press, 2008), 72.
- ¹³⁰⁶ Linus Vanlaere and Chris Gastmans, "A Personalist Approach to Care Ethics," 170.
- ¹³⁰⁷ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 408.
- ¹³⁰⁸ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 408.
- ¹³⁰⁹ Tijs Vandemeulebroucke, Bernadette Dierckx de Casterlé, and Chris Gastmans, "The Use of Care Robots in Aged Care: A Systematic Review of Argument-Based Ethics Literature," *Archives of Gerontology & Geriatrics* 74 (2018): 22.
- ¹³¹⁰ Tijs Vandemeulebroucke, Bernadette Dierckx de Casterlé, and Chris Gastmans, "The Use of Care Robots in Aged Care: A Systematic Review of Argument-Based Ethics Literature," 22.
- ¹³¹¹ Tijs Vandemeulebroucke, Bernadette Dierckx de Casterlé, and Chris Gastmans, "The Use of Care Robots in Aged Care: A Systematic Review of Argument-Based Ethics Literature," 22.
- ¹³¹² Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 257.
- ¹³¹³ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 257.
- ¹³¹⁴ Mark Coeckelbergh, "Care Robots and the Future of Ict-Mediated Elderly Care: A Response to Doom Scenarios," AI & Society 31, no. 4 (2016): 220.
- ¹³¹⁵ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 257.
- ¹³¹⁶ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 257.
- ¹³¹⁷ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 257.
- ¹³¹⁸ Lawrence C. Becker, *Reciprocity* (London: Routledge, 2014), 86.
- ¹³¹⁹ Lawrence C. Becker, *Reciprocity*, 86.
- ¹³²⁰ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 258.
- ¹³²¹ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 258.
- ¹³²² Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 259.
- ¹³²³ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 259.
- ¹³²⁴ Robert Sparrow and Linda Sparrow, "In the Hands of Machines? The Future of Aged Care," *Minds & Machines* 16, no. 2 (2006): 154.
- ¹³²⁵ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 260.
- ¹³²⁶ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 260.
- ¹³²⁷ Catrin Misselhorn, "Artificial Morality. Concepts, Issues and Challenges," Society 55, no. 2 (2018): 161.
- ¹³²⁸ Catrin Misselhorn, "Artificial Morality. Concepts, Issues and Challenges," 161.

¹³²⁹ Catrin Misselhorn,	"Artificial Morality	. Concepts, Issue	es and Challenges," 1	62.
,	2			

- ¹³³⁰ Catrin Misselhorn, "Artificial Morality. Concepts, Issues and Challenges," 162.
- ¹³³¹ Catrin Misselhorn, "Artificial Morality. Concepts, Issues and Challenges," 162.
- ¹³³² Catrin Misselhorn, "Artificial Morality. Concepts, Issues and Challenges," 163.
- ¹³³³ Catrin Misselhorn, "Artificial Morality. Concepts, Issues and Challenges," 163.
- ¹³³⁴ Catrin Misselhorn, "Artificial Morality. Concepts, Issues and Challenges," 163.
- ¹³³⁵ Catrin Misselhorn, "Artificial Morality. Concepts, Issues and Challenges," 163.
- ¹³³⁶ Catrin Misselhorn, "Artificial Morality. Concepts, Issues and Challenges," 164.
- ¹³³⁷ Catrin Misselhorn, "Artificial Morality. Concepts, Issues and Challenges," 164.
- ¹³³⁸ Catrin Misselhorn, "Artificial Morality. Concepts, Issues and Challenges," 164.
- ¹³³⁹ Catrin Misselhorn, "Artificial Morality. Concepts, Issues and Challenges," 164.
- ¹³⁴⁰ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 409.
- ¹³⁴¹ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 409.
- ¹³⁴² Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 409.
- ¹³⁴³ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 410.
- ¹³⁴⁴ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 410.
- ¹³⁴⁵ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 410.
- ¹³⁴⁶ B. Friedman and P. H. Kahn, Jr., "Human Values, Ethics, and Design," in *The Human-Computer Interaction Handbook: Fundamentals, Evolving Technologies and Emerging Applications* (Mahwah, N.J.: Lawrence Erlbaum Associates, Inc., 2003), 1178.
- ¹³⁴⁷ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 411.
- ¹³⁴⁸ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 411.
- ¹³⁴⁹ Mark Coeckelbergh, "Artificial Agents, Good Care, and Modernity," *Theoretical Medicine and Bioethics* 36, no. 4 (2015): 269.
- ¹³⁵⁰ Mark Coeckelbergh, "Artificial Agents, Good Care, and Modernity," 269.
- ¹³⁵¹ Aimee Wynsberghe, "Designing Robots for Care: Care Centered Value-Sensitive Design," 411.
- ¹³⁵² Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 262.
- ¹³⁵³ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 262.
- ¹³⁵⁴ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 262.
- ¹³⁵⁵ Jason Borenstein and Yvette Pearson, "Robot Caregivers: Harbingers of Expanded Freedom for All?," *Ethics and Information Technology* 12, no. 3 (2010): 278.
- ¹³⁵⁶ Jason Borenstein and Yvette Pearson, "Robot Caregivers: Harbingers of Expanded Freedom for All?," 279.
- ¹³⁵⁷ Mark Coeckelbergh, "Health Care, Capabilities, and Ai Assistive Technologies," *Ethical Theory & Moral Practice* 13, no. 2 (2010): 186.
- ¹³⁵⁸ Mark Coeckelbergh, "Health Care, Capabilities, and Ai Assistive Technologies," 185.
- ¹³⁵⁹ Shannon Vallor, "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century," 262.
- ¹³⁶⁰ Martha C. Nussbaum, *Frontiers of Justice: Disability, Nationality, Species Membership*, 1st Harvard University Press ed. (London: Belknap, 2007), 76.
- ¹³⁶¹ Martha C. Nussbaum, Frontiers of Justice: Disability, Nationality, Species Membership, 76.
- ¹³⁶² Martha C. Nussbaum, Frontiers of Justice: Disability, Nationality, Species Membership, 77.
- ¹³⁶³ Martha C. Nussbaum, Frontiers of Justice: Disability, Nationality, Species Membership, 77.
- ¹³⁶⁴ Martha C. Nussbaum, Frontiers of Justice: Disability, Nationality, Species Membership, 77.
- ¹³⁶⁵ Mark Coeckelbergh, "Health Care, Capabilities, and Ai Assistive Technologies," 185.
- ¹³⁶⁶ Mark Coeckelbergh, "Health Care, Capabilities, and Ai Assistive Technologies," 185.

- ¹³⁶⁷ Jim Torresen, "A Review of Future and Ethical Perspectives of Robotics and Ai," *Frontiers in Robotics and AI* 4, no. 75 (2018): 6.
- ¹³⁶⁸ Jim Torresen, "A Review of Future and Ethical Perspectives of Robotics and Ai," 6.
- ¹³⁶⁹ Jim Torresen, "A Review of Future and Ethical Perspectives of Robotics and Ai," 6.
- ¹³⁷⁰ T. Metzler and L. Lewis, "Ethical View, Religious Views, and Acceptance of Robotic Applications: A Pilot Study," http://www.aaai.org.
- ¹³⁷¹ T. Metzler and L. Lewis, "Ethical View, Religious Views, and Acceptance of Robotic Applications: A Pilot Study".
- ¹³⁷² Mark Coeckelbergh, "Care Robots and the Future of Ict-Mediated Elderly Care: A Response to Doom Scenarios," 460.
- ¹³⁷³ Mark Coeckelbergh, "Care Robots and the Future of Ict-Mediated Elderly Care: A Response to Doom Scenarios," 460.
- ¹³⁷⁴ Mark Coeckelbergh, "Care Robots and the Future of Ict-Mediated Elderly Care: A Response to Doom Scenarios," 460.
- ¹³⁷⁵ Mark Coeckelbergh, "Care Robots and the Future of Ict-Mediated Elderly Care: A Response to Doom Scenarios," 460.
- ¹³⁷⁶ Mark Coeckelbergh, "Care Robots and the Future of Ict-Mediated Elderly Care: A Response to Doom Scenarios," 460.
- ¹³⁷⁷ Mark Coeckelbergh, "Care Robots and the Future of Ict-Mediated Elderly Care: A Response to Doom Scenarios," 460.
- ¹³⁷⁸ Mark Coeckelbergh, "Care Robots and the Future of Ict-Mediated Elderly Care: A Response to Doom Scenarios," 460.
- ¹³⁷⁹ Mark Coeckelbergh, "Care Robots and the Future of Ict-Mediated Elderly Care: A Response to Doom Scenarios," 460.
- ¹³⁸⁰ Russell Blackford, "Robots and Reality: A Reply to Robert Sparrow," *Ethics and Information Technology* 14, no. 1 (2012): 41-51.
- ¹³⁸¹ Raffaele Rodogno, "Social Robots, Fiction, and Sentimentality," *Ethics and Information Technology* 18, no. 4 (2016): 7.
- ¹³⁸² Raffaele Rodogno, "Social Robots, Fiction, and Sentimentality," 7.
- ¹³⁸³ Raffaele Rodogno, "Social Robots, Fiction, and Sentimentality," 8.
- ¹³⁸⁴ Raffaele Rodogno, "Social Robots, Fiction, and Sentimentality," 8.
- ¹³⁸⁵ Raffaele Rodogno, "Social Robots, Fiction, and Sentimentality," 9.
- ¹³⁸⁶ Raffaele Rodogno, "Social Robots, Fiction, and Sentimentality," 9.
- ¹³⁸⁷ Sara Rosenbaum, "Data Governance and Stewardship: Designing Data Stewardship Entities and Advancing Data Access," 1443.
- ¹³⁸⁸ Sara Rosenbaum, "Data Governance and Stewardship: Designing Data Stewardship Entities and Advancing Data Access," 1443.
- ¹³⁸⁹ Sara Rosenbaum, "Data Governance and Stewardship: Designing Data Stewardship Entities and Advancing Data Access," 1443.

BIBLIOGRAPHY

- (HUGO), Human Genome Organisation. "Statement on Human Genomic Databases." [In Fr]. Journal International de Bioéthique 14, no. 3 (2003): 207-210.
- Abdelhak, Mervat, Sara Grostick, and Mary Alice Hanken. *Health Information: Management of a Strategic Resource*. Elsevier Health Sciences, 2014.
- Adams, Samantha A., and Carolyn Petersen. "Precision Medicine: Opportunities, Possibilities, and Challenges for Patients and Providers." *Journal of the American Medical Informatics Association* 23, no. 4 (07// 2016): 787-790.
- Adibuzzaman, Mohammad, Poching DeLaurentis, Jennifer Hill, and Brian D. Benneyworth. "Big Data in Healthcare - the Promises, Challenges and Opportunities from a Research Perspective: A Case Study with a Model Database." *AMIA Annual Symposium proceedings* 2017 (2018): 384-392.
- Agar, Nicholas. How to Be Human in the Digital Economy. Cambridge, MA: The MIT Press, 2019.

——. *Truly Human Enhancement: A Philosophical Defense of Limits.* Basic Bioethics. Cambridge, MA: MIT Press, 2014.

———. *Truly Human Enhancement: A Philosophical Defense of Limits.* Basic Bioethics. Cambridge, MA: MIT Press, 2014.

- Agarwal, Ritu, Guodong Gao, Catherine DesRoches, and Ashish K. Jha. "Research Commentary: The Digital Transformation of Healthcare: Current Status and the Road Ahead." *Information Systems Research* 21, no. 4 (2010): 796-809.
- Agha, Leila. "The Effects of Health Information Technology on the Costs and Quality of Medical Care." *Journal of Health Economics* 34 (Mar 2014): 19-30.
- Alpaydin, Ethem. *Machine Learning: The New Ai.* Mit Press Essential Knowledge Series. Cambridge, MA: MIT Press, 2016.
- Amirabdollahian, Farshid, den Akker Rieks op, Sandra Bedaf, Richard Bormann, Heather Draper, Vanessa Evers, Gallego Pérez Jorge, *et al.* "Assistive Technology Design and Development for Acceptable Robotics Companions for Ageing Years." *Journal of Behavioral Robotics* 4, no. 2 (2013): 94-112.
- Anderson, James G., and Kenneth W. Goodman. *Ethics and Information Technology: A Case-Based Approach to a Health Care System in Transition*. Edited by Kathyrn J. Hannah and Marion J. Ball. New York, NY: Springer, 2002.

- Arendt, Hannah. *The Human Condition*. Charles R. Walgreen Foundation Lectures. University of Chicago Press, 1958.
- Ariès, Philippe. At the Hour of Our Death. 1st American ed. New York, NY: Knopf, 1981.
 - ——. Western Attitudes toward Death: From the Middle Ages to the Present. JHU Press, 1974.
- Aristotle. *The Politics*. Cambridge Texts in the History of Political Thought. Cambridge: Cambridge University Press, 1988.
- Armstrong, Stuart, Anders Sandberg, and Nick Bostrom. "Thinking inside the Box: Controlling and Using an Oracle Ai." *Minds and Machines: Journal for Artificial Intelligence, Philosophy and Cognitive Science* 22, no. 4 (2012): 299-324.
- Association, World Medical. "Wma to Draw up Health Database Guidelines." <u>https://www.wma.net/news-post/wma-to-draw-up-health-database-guidelines/</u>.
- Baillie, Harold W. Health Care Ethics. 6th ed. Boston: Prentice Hall, 2013.
- Baker, Dixie B., Jane Kaye, Grad Dip Leg, and Sharon F. Terry. "Privacy, Fairness, and Respect for Individuals." *eGEMS (Generating Evidence & Methods to Improve Patient Outcomes)* 4, no. 2 (2016): 1-17.
- Barrick, Ivan J. Transforming Health Care Management: Integrating Technology Strategies. Sudbury, MA: Jones and Bartlett Publishers, LLC, 2009.
- Beauchamp, Tom L., and James F. Childress. *Principles of Biomedical Ethics*. 7th ed. New York: Oxford University Press, 2013.
- Becker, Lawrence C. Reciprocity. London: Routledge, 2014.
- Bedaf, Sandra, Gert Jan Gelderblom, and Luc De Witte. "Overview and Categorization of Robots Supporting Independent Living of Elderly People: What Activities Do They Support and How Far Have They Developed." *Assistive Technology* 27, no. 2 (2015): 88-100.
- Belle, Ashwin, Raghuram Thiagarajan, S. M. Reza Soroushmehr, Fatemeh Navidi, Daniel A. Beard, and Kayvan Najarian. "Big Data Analytics in Healthcare." *BioMed Research International* 2015 (2015): 1-16.
- Bemelmans, Roger, Gert Jan Gelderblom, Pieter Jonker, and Luc de Witte. "Socially Assistive Robots in Elderly Care: A Systematic Review into Effects and Effectiveness." *Journal of the American Medical Directors Association* 13, no. 2 (2012): 114-120.
- Biddle, Robert, Sonia Chiasson, and P. C. Van Orschot. "Graphical Passwords: Learning from the First Twelve Years." *ACM Computing Surveys* 44, no. 4 (2012): 19:11-19:41.

- Birn, Anne-Emanuelle, Theodore M. Brown, Elizabeth Fee, and Walter J. Lear. "Struggles for National Health Reform in the United States." *American Journal of Public Health* 93, no. 1 (Jan 2003): 86-91.
- Bishop, Jeffrey P. *The Anticipatory Corpse: Medicine, Power, and the Care of the Dying.* Notre Dame, IN: University of Notre Dame Press, 2011.
- Bishops, United States Conference of Catholic. *Ethical and Religious Directives for Catholic Health Care Services.* 5th ed. Washington, DC: USCCB Publishing, 2009.

. "Letter to Senate on Health Care Reform." news release, 2017.

- Blackford, Russell. "Robots and Reality: A Reply to Robert Sparrow." *Ethics and Information Technology* 14, no. 1 (2012): 41-51.
- Bloem, Jaap, Menno van Doorn, Sander Duivestein, David Excoffier, René Maas, and Erik van Ommeren. "The Fourth Industrial Revolution: Things to Tighten the Link between It and Ot." In *VINT Research Report*: Sogeti, 2014.
- Bloss, Cinnamon S., Nicholas J. Schork, and Eric J. Topol. "Effect of Direct-to-Consumer Genomewide Profiling to Assess Disease Risk." *The New England Journal of Medicine* 364, no. 6 (2011): 524.
- Blumenthal, David, and James Morone. "The Lessons of Success—Revisiting the Medicare Story." *New England Journal of Medicine* 359, no. 22 (Nov 27 2008): 2384-2389.
- Boddington, Paula. *Towards a Code of Ethics for Artificial Intelligence*. New York, NY: Springer Berlin Heidelberg, 2017.
- Boeri, Natascia. "Technology and Society as Embedded: An Alternative Framework for Information and Communication Technology and Development." *Media, Culture & Society* 38, no. 1 (2016): 107-118.
- Bogosian, Kyle. "Implementation of Moral Uncertainty in Intelligent Machines." Minds & Machines 27, no. 4 (2017): 591-608.
- Borenstein, Jason, and Yvette Pearson. "Robot Caregivers: Harbingers of Expanded Freedom for All?". *Ethics and Information Technology* 12, no. 3 (2010): 277-288.
- Borgmann, Albert. *Technology and the Character of Contemporary Life: A Philosophical Inquiry.* Chicago, IL: University of Chicago Press, 1984.
- Bostrom, Nick, and Julian Savulescu. "Human Enhancement Ethics: The State of the Debate." In *Human Enhancement*, edited by Julian Savulescu and Nick Bostrom, Oxford: Oxford University Press, 2009, 1-22.

- Bostrom, Nick, and Eliezer Yudkowsky. "The Ethics of Artificial Intelligence." In *The Cambridge Handbook of Artificial Intelligence*, edited by Keith Frankish and William M. Ramsey, Cambridge, UK: Cambridge University Press, 2014, xi, 354 pages.
- Bower, Hilary. "Internet Sees Growth of Unverified Health Claims." *British Medical Journal* 313, no. 7054 (1996): 381-381.
- Bratteteig, Tone, and Ina Wagner. "Moving Healthcare to the Home: The Work to Make Homecare Work." Paper presented at the ECSCW 2013: Proceedings of the 13th European Conference on Computer Supported Cooperative Work, Paphos, Cyprus, September 2013.
- Brey, Philip. "Anticipating Ethical Issues in Emerging It." *Ethics & Information Technology* 14, no. 4 (2012): 305-317.
- Brown, Gordon D., Timothy B. Patrick, and Kalyan S. Pasupathy. *Health Informatics: A Systems Perspective*. Chicago, IL: Health Administration Press, 2013.

- Bruera, Eduardo, and Sriram Yennurajalingam. "Palliative Care in Advanced Cancer Patients: How and When?". *Oncologist* 17, no. 2 (2012): 267-774.
- Buchanan, Allen E., Dan W. Brock, Norman Daniels, and Daniel Wikler. *From Chance to Choice: Genetics and Justice*. New York, NY: Cambridge University Press, 2000.
- Bush, George W. "State of the Union Address by the President." edited by Executive Branch. Washington D.C.: United States Capitol, 2006.
- Callahan, Daniel. "Individual Good and Common Good: A Communitarian Approach to Bioethics." *Perspectives in Biology and Medicine* 46, no. 4 (2003): 496.
- ------. The Troubled Dream of Life: In Search of a Peaceful Death. Georgetown University Press, 2000.
- Camarota, Steven, and Karen Zeigler. "Immigrant Population Hits Record 46.2 Million in November 2021." Center for Immigration Studies, <u>https://cis.org/Camarota/Immigrant-Population-Hits-Record-462-Million-November-2021</u>.
- Canellopoulou-Bottis, Maria, and Kenneth E. Himma. "The Digital Divide: A Perspective for the Future." Chap. 26 In *The Handbook of Information and Computer Ethics*, edited by Kenneth Einar Himma and Herman T. Tavani, Hoboken, N.J.: Wiley, 2008, 621-637.
- Chalmers, David J. "The Singularity: A Philosophical Analysis." *Journal of Consciousness Studies* 17, no. 9-10 (2010): 7-65.

- Chen, Rui, George I Mias, Jennifer Li-Pook-Than, Lihua Jiang, Hugo Y K. Lam, Rong Chen, Elana Miriami, *et al.* "Personal Omics Profiling Reveals Dynamic Molecular and Medical Phenotypes." *Cell* 148, no. 6 (2012): 1293-1307.
- Chou, Shuo-Yan. "The Fourth Industrial Revolution: Digital Fusion with Internet of Things." Journal of International Affairs 72, no. 1 (2019): 107-120.
- Cleveland, Ana D., and Donald B. Cleveland. *Health Informatics for Medical Librarians*. New York, NY: Neal-Schuman Publishers, 2009.
- Coakley, Sarah. "Introductory Remarks: The Theology and Ethics of Contemporary Dying." *Studies in Christian Ethics* 29, no. 3 (2016): 245-248.
- Coeckelbergh, Mark. "Artificial Agents, Good Care, and Modernity." *Theoretical Medicine and Bioethics* 36, no. 4 (August 01 2015): 265-277.

——. "Care Robots and the Future of Ict-Mediated Elderly Care: A Response to Doom Scenarios." *AI & Society* 31, no. 4 (2016): 455-462.

- ———. "Health Care, Capabilities, and Ai Assistive Technologies." *Ethical Theory & Moral Practice* 13, no. 2 (2010): 181-190.
- Collins, Francis S., and James D. Watson. "Genetic Discrimination: Time to Act." *Science* 302, no. 5646 (2003): 745.
- Collis, Emily, and R. Al-Qurainy. "Care of the Dying Patient in the Community." *BMJ: British Medical Journal* 347, no. 7915 (2013): 27-30.
- Commission, Ethics. "Ethics Commission Automated and Connected Driving." Federal Ministry of Transport and Digital Infrastructure (Germany), 2017.
- Commission, European, Social Affairs Directorate-General for Employment, Inclusion, and Eurostat. "Short Analytical Web Note : Demography Report. 3/2015." Publications Office, 2015.
- Coustasse, Alberto, Theresa Quiroz, and Sue G. Lurie. "To the Bitter End: Disparities in End-of-Life Care." *Journal of Hospital Marketing & Public Relations* 18, no. 2 (2008): 167-185.
- Crossen-Sills, Jackie, Irene Toomey, and Mary E. Doherty. "Technology and Home Care: Implementing Systems to Enhance Aging in Place." *Nursing Clinics of North America* 44, no. 2 (2009/06/01/ 2009): 239-246.
- Daniels, Norman, and James E. Sabin. Setting Limits Fairly: Learning to Share Resources for Health. 2nd ed. Oxford ; New York: Oxford University Press, 2008.
- Davis, Ronald M., and Myron Genel. "Good Care of the Dying Patient." *Journal of the American Medical Association* 275, no. 6 (1996): 474-478.

- Deber, Raisa B., Nancy Kraetschmer, and Jane Irvine. "What Role Do Patients Wish to Play in Treatment Decision Making?". *Archives of Internal Medicine* 156, no. 13 (1996): 1414-1420.
- Deliversky, J., and M. Deliverska. "Ethical and Legal Considerations in Biometric Data Usage-Bulgarian Perspective." *Front Public Health* 6, no. 25 (2018).
- Deming, W. Edwards. Out of the Crisis. Cambridge, Massachsetts: The MIT Press, 2018.
- Demiris, George, Debra Parker Oliver, and Elaine Wittenberg-Lyles. "Technologies to Support End of Life Care." *Seminars in Oncology Nursing* 27, no. 3 (2011): 211-217.
- Derickson, Alan. *Health Security for All: Dreams of Universal Healthcare in America*. Baltimore: Johns Hopkins University Press, 2005.
- Dorey, Corine Mouton. "Rethinking the Ethical Approach to Health Information Management through Narration: Pertinence of Ricœur's 'Little Ethics'." *Medicine, Health Care and Philosophy* 19, no. 4 (2016): 531-543.
- Douglas, Heather. "Pure Science and the Problem of Progress." *Studies in History and Philosophy* of Science 46 (2014): 55-63.
- Dove, Edward S. "Biobanks, Data Sharing, and the Drive for a Global Privacy Governance Framework." *Journal of Law, Medicine & Ethics* 43, no. 4 (Winter 2015): 675-689.
- Dove, Edward S., Bartha M. Knoppers, and Ma'n H. Zawati. "Towards an Ethics Safe Harbor for Global Biomedical Research." *Journal of Law and the Biosciences* 1, no. 1 (2014): 3-51.
- Draper, Heather, and Tom Sorell. "Ethical Values and Social Care Robots for Older People: An International Qualitative Study." *Ethics and Information Technology* 19, no. 1 (2017): 49-68.
- Drexler, Eric. Engines of Creation. 1st ed.: Anchor Press/Doubleday, 1986.
- Dugdale, Lydia S. Dying in the Twenty-First Century: Toward a New Ethical Framework for the *Art of Dying Well*. Cambridge, Massachusetts: The MIT Press, 2015.
- Dupré, Louis. "The Common Good and the Open Society." *The Review of Politics* 55, no. 4 (1993): 687-712.
- Eggleston, Ben. "Genetic Discrimination in Health Insurance: An Ethical and Economic Analysis." In *The Human Genome Project in College Curriculum: Ethical Issues and Practical Strategies* edited by Aine Donovan and Ronald Michael Green, Hanover, NH: Dartmouth College Press, 2008.
- Eiser, Arnold R. The Ethos of Medicine in Postmodern America: Philosophical, Cultural, and Social Considerations. Lanham: Lexington Books, 2014.

- Ertel, Wolfgang. Introduction to Artificial Intelligence. Undergraduate Topics in Computer Science. London; New York: Springer, 2011.
- Europe, AGE Platform. "Accompany." European Commussion, <u>https://www.age-platform.eu/good-practice/accompany</u>.
- Fairchild, Amy L., Ronald Bayer, James Keith Colgrove, and Daniel Wolfe. *Searching Eyes : Privacy, the State, and Disease Surveillance in America.* Berkeley: University of California Press, 2007.
- Floridi, Luciano, Josh Cowls, Monica Beltrametti, Raja Chatila, Patrice Chazerand, Virginia Dignum, Christoph Luetge, et al. "Ai4people—an Ethical Framework for a Good Ai Society: Opportunities, Risks, Principles, and Recommendations." Minds and Machines: Journal for Artificial Intelligence, Philosophy and Cognitive Science 28, no. 4 (2018): 689-707.
- Foege, William H., Robert C. Hogan, and Ladene H. Newton. "Surveillance Projects for Selected Diseases." *International Journal of Epidemiology* 5, no. 1 (01// 1976): 29-37.
- Foundation, The Henry J. Kaiser Family. "Key Facts About the Uninsured Population." In *Annual Social and Economic Supplement*. Menlo Park, CA 2017.
- Fox, Renée C. *The Sociology of Medicine : A Participant Observer's View.* Prentice Hall Foundations of Modern Sociology Series. Englewood Cliffs, N.J.: Prentice Hall, 1989.
- Free, Caroline, Rosemary Knight, Steven Robertson, Robyn Whittaker, Phil Edwards, Weiwei Zhou, Anthony Rodgers, et al. "Smoking Cessation Support Delivered Via Mobile Phone Text Messaging (Txt2stop): A Single-Blind, Randomised Trial." *The Lancet* 378, no. 9785 (2011): 49-55.
- Friedman, B., and P. H. Kahn, Jr. "Human Values, Ethics, and Design." In *The Human-Computer Interaction Handbook: Fundamentals, Evolving Technologies and Emerging Applications*, Mahwah, N.J.: Lawrence Erlbaum Associates, Inc., 2003, 1177-1201.
- Frost, Jeana, Sally Okun, Timothy Vaughan, James Heywood, and Paul Wicks. "Patient-Reported Outcomes as a Source of Evidence in Off-Label Prescribing: Analysis of Data from Patientslikeme." *Journal of Medical Internet Research* 13, no. 1 (2011): e6-e6.
- Fukuyama, Francis. Our Posthuman Future: Consequences of the
- Biotechnology Revolution. New York, NY: Farrar, Straus and Giroux, 2002.
- Gabe, Jonathan, Mike Bury, and Mary Ann Elston. Key Concepts in Medical Sociology. Sage Publications Ltd, 2004.
- Gallagher, Ryan. "Software That Tracks People on Social Media Created by Defense Firm." *The Guardian*, 2013.

- Gardiner, Stephen M. "A Core Precautionary Principle." *Journal of Political Philosophy* 14, no. 1 (2006): 33-60.
- Gastmans, Chris. "Dignity-Enhancing Nursing Care: A Foundational Ethical Framework." *Nursing Ethics* 20, no. 2 (2013): 142-149.
- Gawande, Atul. Being Mortal: Medicine and What Matters in the End. 1st ed. New York: Metropolitan Books: Henry Holt & Company, 2014.
- Gilligan, Carol. In a Different Voice: Psychological Theory and Women's Development. Cambridge, Massachusetts: Harvard University Press, 1993.

———. "Moral Orientation and Moral Development." In *Women and Moral Theory*, edited by Eva Kittay and Diana Meyers, Totowa, NJ: Rowman & Littlefield, 1987, 19-33.

- Giroux, Henry A. Stealing Innocence: Youth, Corporate Power and the Politics of Culture. [in English] New York, NY: Palgrave Macmillan, 2000.
- Goertzel, Ben, and Cassio Pennachin, eds. Artificial General Intelligence. Berlin: Springer, 2007.
- Goodman, Kenneth W. Ethics, Medicine, and Information Technology: Intelligent Machines and the Transformation of Health Care. Cambridge; New York, NY: Cambridge University Press, 2015.
- Goodman, Kenneth W., Reid Cushman, and Randolph A. Miller. "Ethics in Biomedical and Health Informatics: Users, Standards, and Outcomes." Chap. 10 In *Biomedical Informatics*, edited by Edward H. Shortliffe and James J. Cimino, Verlag-London: Springer, 2014.
- Gozalo, Pedro, Joan M. Teno, Susan L. Mitchell, Jon Skinner, Julie Bynum, Denise Tyler, and Vincent Mor. "End-of-Life Transitions among Nursing Home Residents with Cognitive Issues." *The New England Journal of Medicine* 365, no. 13 (2011): 1212-1221.
- Greenberger, Norton J., Nicholas E. Davies, and Euegene A. Hildreth. "Reforming Health Care in the United States." *BMJ: British Medical Journal (International Edition)* 303, no. 6812 (1991): 1216.
- Greengard, Samuel. "Weighing the Impact of Gdpr: The Eu Data Regulation Will Affect Computer, Internet, and Technology Usage within and Outside the Eu; How It Will Play out Remains to Be Seen." *Communications of the ACM* 61, no. 11 (2018): 16-18.
- Groseclose, SL, RR German, and P. Nsubuga. "Evaluating Public Health Surveillance." In *Principles and Practice of Public Health Surveillance*, edited by Lisa M. Lee, SM Teutsch, Stephen B. Thacker and Michael E. St. Louis, New York: Oxford University Press, 2010, 166-197.
- Grunwald, Armin. "Technology Assessment: Concepts and Methods." In *Philosophy of Technology and Engineering Sciences*: Elsevier, 2009, 1103-1146.

- Gustafson, David H., Fiona McTavish, Robert Hawkins, Suzanne Pingree, Neeraj Arora, John Mendenhall, and Greg E. Simmons. "Computer Support for Elderly Women with Breast Cancer." *JAMA: The Journal of the American Medical Association* 280, no. 15 (1998): 1305-1305.
- Hamilton, Dona Cooper, and Charles V. Hamilton. "The Dual Agenda of African American Organizations since the New Deal: Social Welfare Policies and Civil Rights." *Political Science Quarterly* 107, no. 3 (1992): 435-452.

Hanson, Robin. "Prefer Law to Values." In Prefer Law to Values: Overcoming Bias, 2009.

- Have, Henk ten. "Genetics and Culture: The Geneticization Thesis." *Medicine, Health Care and Philosophy* 4, no. 3 (2001/10/01 2001): 295-304.
- -------. *Global Bioethics: An Introduction*. London; New York, NY: Routledge, Taylor & Francis Group, 2016.
- ------. *Vulnerability: Challenging Bioethics*. London ; New York: Routledge, Taylor & Francis Group, 2016.
- Have, Henk ten, and David Clark. *The Ethics of Palliative Care : European Perspectives*. Facing Death. Buckingham England ; Phildelphia, PA: Open University Press, 2002.
- Have, Henk ten, and Bert Gordijn. *Bioethics in a European Perspective*. International Library of Ethics, Law, and the New Medicine. Dordrecht; Boston, MA: Kluwer Academic Publishers, 2001.
- Have, Henk ten, and Michèle Jean, eds. *The Unesco Universal Declaration on Bioethics and Human Rights: Background, Principles and Application*, Ethics Series. Paris: UNESCO Publishing, 2009.
- Health, The Global Alliance for Genomics and. "A Federated Ecosystem for Sharing Genomic, Clinical Data." *Science* 352, no. 6291 (2016): 1278-1280.
- Held, Virginia. *The Ethics of Care: Personal, Political, and Global.* Oxford: Oxford University Press, 2006.
- Hoffman, Beatrix. "Health Care Reform and Social Movements in the United States." *American Journal of Public Health* 93, no. 1 (Jan 2003): 75-85.
- Hogan, Timothy P., Kim M. Nazi, Tana M. Luger, Daniel J. Amante, Bridget M. Smith, Anna Barker, Stephanie L. Shimada, *et al.* "Technology-Assisted Patient Access to Clinical Information: An Evaluation Framework for Blue Button." *JMIR Research Protocols* 3, no. 1 (2014): e18-e18.
- Hood, Leroy, and Stephen H. Friend. "Predictive, Personalized, Preventive, Participatory (P4) Cancer Medicine." *Nature Reviews Clinical Oncology* 8, no. 3 (2011): 184.

- Howcroft, D., N. Mitev, and M. Wilson. "What We May Learn from the Social Shaping of Technology Approach." In Social Theory and Philosophy for Information Systems, edited by John Mingers and Leslie Willcocks, Chichester, West Sussex, England: John Wiley and Sons, Inc, 2004, 329-371.
- Hughes, Jonathan. "How Not to Criticize the Precautionary Principle." *Journal of Medicine & Philosophy* 31, no. 5 (2006): 447-464.
- International, DAMA. *Dama Dmbok Data Management Body of Knowledge*. New Jersey: Technics Publications, 2017.
- Jain, Anil K., Ruud Bolle, and Sharath Pankanti. *Biometrics: Personal Identification in Networked Society.* The Kluwer International Series in Engineering and Computer Science. Boston: Kluwer, 1999.
- Janssens, Louis. "Artificial Insemination: Ethical Considerations." *Louvain Studies* 8, no. 1 (1980): 3-29.
- Jiang, Fei, Yong Jiang, Hui Zhi, Yi Dong, Hao Li, Sufeng Ma, Yilong Wang, et al. "Artificial Intelligence in Healthcare: Past, Present and Future." Stroke and vascular neurology 2, no. 4 (2017): 230-243.
- Joyce, Mike, Olivier Leclerc, Kirsten Westhues, and Hui Xue. *Digital Therapeutics: Preparing for Takeoff.* Mckinsey Insights. New York: McKinsey & Company, 2018.
- Kachouie, Reza, Sima Sedighadeli, Rajiv Khosla, and Mei-Tai Chu. "Socially Assistive Robots in Elderly Care: A Mixed-Method Systematic Literature Review." *International Journal of Human-Computer Interaction* 30, no. 5 (2014): 369-393.
- Kass, Leon R. "Ageless Bodies, Happy Souls: Biotechnology and the Pursuit of Perfection." *The New Atlantis*, no. 1 (2003): 9-28.
- Kaye, Jane. "Do We Need a Uniform Regulatory System for Biobanks across Europe?". *European Journal of Human Genetics* 14, no. 2 (2006): 245-248.
- Kelly, David F., Gerard Magill, and Henk ten Have. *Contemporary Catholic Health Care Ethics*. 2nd ed. Washington, DC: Georgetown University Press, 2013.
- Kho, Nancy Davis. "Gdpr for Content Design, Development, and Deployment: 'Gdpr Is a Really Good Opportunity to Deep Clean under the Sofa and Get Rid of All the Dust Bunnies.'." *EContent* 41, no. 3 (2018): 20-24.
- King, Mary-Claire, Joan H. Marks, and Jessica B. Mandell. "Breast and Ovarian Cancer Risks Due to Inherited Mutations in Brca1 and Brca2." *Science* 302, no. 5645 (2003): 643-646.
- Kirkpatrick, Brianne E., Erin Rooney Riggs, Danielle R. Azzariti, Vanessa Rangel Miller, David H. Ledbetter, David T. Miller, Heidi Rehm, Christa Lese Martin, and Andrew W. Faucett.

"Genomeconnect: Matchmaking between Patients, Clinical Laboratories, and Researchers to Improve Genomic Knowledge." *Human Mutation* 36, no. 10 (2015): 974-978.

- Klemm, Paula, Karla Reppert, and Lori Visich. "A Nontraditional Cancer Support Group: The Internet." *Computers in Nursing* 16, no. 1 (1998): 31-36.
- Kluge, E. H. "Health Information, the Fair Information Principles and Ethics." *Methods of information in medicine* 33, no. 4 (1994): 336-345.
- Knoppers, Bartha M., and Yann Joly. "Our Social Genome?". *Trends in Biotechnology* 25, no. 7 (Jul 2007): 284-288.
- Knoppers, Bartha Maria, and Ruth Chadwick. "Human Genetic Research: Emerging Trends in Ethics." *Nature Reviews Genetics* 6, no. 1 (2005): 75-79.
- Kogetsu, Atsushi, Soichi Ogishima, and Kazuto Kato. "Authentication of Patients and Participants in Health Information Exchange and Consent for Medical Research: A Key Step for Privacy Protection, Respect for Autonomy, and Trustworthiness." *Frontiers in Genetics* 9, no. 167 (2018): 167.
- Kroenke, K., D. Theobald, J. Wu, K. Norton, G. Morrison, J. Carpenter, W. Tu, *et al.* "Effect of Telecare Management on Pain and Depression in Patients with Cancer: A Randomized Trial." *JAMA: Journal of the American Medical Association* 304, no. 2 (2010): 163-171.
- Latour, Bruno. Science in Action: How to Follow Scientists and Engineers through Society. London: Open University Press, 1987.
- Laurie, Graeme, Pierre Mallia, David A. Frenkel, Atina Krajewska, Helena Moniz, Salvor Nordal, Claudia Pitz, and Judit Sandor. "Managing Access to Biobanks: How Can We Reconcile Individual Privacy and Public Interests in Genetic Research." *Medical Law International* 10, no. 4 (2010): 315-338.
- Leavitt, Frank J. "Democracies Restricting Democratic Rights: Some Classical Sources and Implications for Ethics of Biometrics." *ScientificWorldJournal* 11 (Mar 1 2011): 463-473.
- Lee, Lisa M., and Lawrence O. Gostin. "Ethical Collection, Storage, and Use of Public Health Data: A Proposal for a National Privacy Protection." *JAMA: The Journal of the American Medical Association* 302, no. 1 (07/01/2009): 82-84.
- Lee, Lisa M., Charles M. Heilig, and Angela White. "Ethical Justification for Conducting Public Health Surveillance without Patient Consent." *American Journal of Public Health* 102, no. 1 (Jan 2012): 38-44.
- Lee, Lisa M., and Stephen B. Thacker. "Public Health Surveillance and Knowing About Health in the Context of Growing Sources of Health Data." *American Journal of Preventive Medicine* 41, no. 6 (2011): 636-640.

- Lee Sr., James. "Knowledge Management: The Intellectual Revolution." *IIE Solutions* 32, no. 10 (2000): 34.
- Levesque, Hector J. Common Sense, the Turing Test, and the Quest for Real Ai. Cambridge, Massachusetts: MIT Press, 2017.
- Lislerud Smebye, Kari, Marit Kirkevold, Knut Engedal, and Kari Lislerud Smebye. "Ethical Dilemmas Concerning Autonomy When Persons with Dementia Wish to Live at Home: A Qualitative, Hermeneutic Study." *BMC Health Services Research* 16 (2016): 1-12.
- Long, Carol O., Edward A. Greenberg, Robert L. Ismeurt, and Gary Smith. "Home Health and Hospice Use of Computers and the Internet." *Home Healthcare Nurse* 18, no. 10 (Nov-Dec 2000): 666-671.
- Love, Heather A. "Newsreels, Novels, and Cybernetics: Reading the Random Patterns of John Dos Passos's U.S.A." *Journal of Modern Literature* 40, no. 2 (2017): 112.
- Lunshof, Jeantine E., Ruth Chadwick, Daniel B. Vorhaus, and George M. Church. "From Genetic Privacy to Open Consent." *Nature Reviews Genetics* 9, no. 5 (05/01/2008): 406.
- Lupton, Deborah. "The Commodification of Patient Opinion: The Digital Patient Experience Economy in the Age of Big Data." *Sociology of Health & Illness* 36, no. 6 (2013): 856.
- Mao, Hui-Fen, Ling-Hui Chang, Grace Yao, Wan-Yin Chen, and Wen-Ni Wennie Huang. "Indicators of Perceived Useful Dementia Care Assistive Technology: Caregivers' Perspectives." *Geriatrics and Gerontology International* 15, no. 8 (2015): 1049-1057.
- May, Theresa. "Prime Minister's Address to the World Economic Forum." Davos, Switzerland: Prime Minister's Office, 2018.
- McAfee, Andrew, and Erik Brynjolfsson. "Big Data: The Management Revolution; Exploiting Vast New Flows of Information Can Radically Improve Your Company's Performance. But First You'll Have to Change Your Decision-Making Culture." *Harvard Business Review* 90, no. 10 (2012): 60.
- McBride, David. From Tb to Aids: Epidemics among Urban Blacks since 1900. Suny Series in Afro-American Studies. Albany, NY: State University of New York Press, 1991.
- McCollough, Thomas E. *The Moral Imagination and Public Life: Raising the Ethical Question*. Chatham House Studies in Political Thinking. Chatham, NJ: Chatham Books, 1992.
- McConnaha, Scott. "Catholic Teaching and Disparities in Care." *Health Progress* 87, no. 1 (2006): 46-50.
- McGee, Ellen M. "Bioelectronics and Implanted Devices." In *Medical Enhancement and Posthumanity*. International Library of Ethics, Law and Technology, Dordrecht: Springer, 2008, vi, 240 pages.

- McWay, Dana C. Legal and Ethical Aspects of Health Information Management. Cengage Learning, 2015.
- Medicine, Institute of. *Dying in America: Improving Quality and Honoring Individual Preferences near the End of Life.* Washington, DC: The National Academies Press, 2015.
- Mehta, Nishita, and Anil Pandit. "Concurrence of Big Data Analytics and Healthcare: A Systematic Review." *International Journal of Medical Informatics* 114 (2018): 57-65.
- Mell, Peter, and Timothy Grance. "The Nist Definition of Cloud Computing." Gaithersburg, MD: National Institute of Standards and Technology U.S. Department of Commerce, 2011.
- Metzler, T., and L. Lewis. "Ethical View, Religious Views, and Acceptance of Robotic Applications: A Pilot Study." <u>http://www.aaai.org</u>.
- Miettinen, Terhikki T., and Reijo S. Tilvis. "Medical Futility as a Cause of Suffering of Dying Patients—the Family Members' Perspective." *Journal of Palliative Care* 15, no. 2 (1999): 26-29.
- Miller, Megan M. "Data as the New Oil: A Slippery Slope of Trade Secret Implications Greased by the California Consumer Privacy Act." *Cybaris: An Intellectual Property Law Review* 12, no. 1 (2021): 1-34.
- Misselhorn, Catrin. "Artificial Morality. Concepts, Issues and Challenges." Society 55, no. 2 (2018): 161-169.
- Mittelstadt, Brent D., and Luciano Floridi. "The Ethics of Big Data: Current and Foreseeable Issues in Biomedical Contexts." *Science and Engineering Ethics* 22, no. 2 (Apr 2016): 303-341.
- Mittelstadt, Brent D., Bernd Stahl, and N. Ben Fairweather. "How to Shape a Better Future? Epistemic Difficulties for Ethical Assessment and Anticipatory Governance of Emerging Technologies." *Ethical Theory & Moral Practice* 18, no. 5 (2015): 1027-1047.
- Mohammed, Shan, and Elizabeth Peter. "Rituals, Death and the Moral Practice of Medical Futility." *Nursing Ethics* 16, no. 3 (2009): 292-302.
- Montgomery, H., M. Grocott, and M. Mythen. "Critical Care at the End of Life: Balancing Technology with Compassion and Agreeing When to Stop." *British Journal of Anaesthesia* 119, no. S1 (2017): i85-i89.
- Mordini, Emilio. "Biometrics." Chap. 31 In *Handbook of Global Bioethics*, edited by Henk ten Have and Bert Gordijn, Dordecht: Springer, 2014, 505-526.
- Mordini, Emilio, and Sonia Massari. "Body, Biometrics and Identity." *Bioethics* 22, no. 9 (2008): 488-498.
- Morgan, Michael L. *Classics of Moral and Political Theory*. 5th ed. Indianapolis: Hackett Publishing Company, 2011.

- Morhaim, Dan K., and Keshia M. Pollack. "End-of-Life Care Issues: A Personal, Economic, Public Policy, and Public Health Crisis." *American Journal of Public Health* 103, no. 6 (2013): e8-e10.
- Morrison, Eileen E. *Ethics in Health Administration: A Practical Approach for Decision Makers.* 3rd ed.: Jones & Bartlett Publishers, 2016.
- Mumford, Lewis. "History: Neglected Clue to Technological Change." *Technology and Culture* 2, no. 3 (1961): 230-236.
- Musen, Mark A, Blackford Middleton, and Robert A Greenes. "Clinical Decision-Support Systems." In *Biomedical Informatics*, edited by Edward H. Shortliffe and James J. Cimino, New York, NY: Springer, 2014, 643-674.
- Navale, Vivek, and Philip E. Bourne. "Cloud Computing Applications for Biomedical Science: A Perspective." *PLoS Computational Biology* 14, no. 6 (2018): 1-14.
- Nebeker, Camille, Rebecca J. Bartlett Ellis, and John Torous. "Development of a Decision-Making Checklist Tool to Support Technology Selection in Digital Health Research." *Translational Behavioral Medicine* 10, no. 4 (08/01/ 2020): 1004.
- Nebeker, Camille, John Torous, and Rebecca J. Bartlett Ellis. "Building the Case for Actionable Ethics in Digital Health Research Supported by Artificial Intelligence." *BMC Medicine* 17, no. 137 (2019).
- Nelson, Ramona, and Nancy Staggers. *Health Informatics-E-Book: An Interprofessional Approach*. Elsevier Health Sciences, 2016.
- Nussbaum, Martha C. Frontiers of Justice: Disability, Nationality, Species Membership. 1st Harvard University Press ed. London: Belknap, 2007.
- O Keefe, Katherine, and Daragh O'Brien. *Ethical Data and Information Management: Concepts, Tools and Methods.* 1 Edition. ed. New York, NY: Kogan Page Ltd, 2018.
- O'Gorman, Lawrence. "Comparing Passwords, Tokens, and Biometrics for User Authentication." *Proceedings of the IEEE* 91, no. 12 (2003): 2021-2040.
- O'Keefe, Katherine, and Daragh O'Brien. *Ethical Data and Information Management: Concepts, Tools and Methods.* 2018.
- O'Neill, Martin. "Genetic Information, Life Insurance, and Social Justice." *The Monist* 89, no. 4 (2006): 567-592.
- Organization, World Health. "World Report on Ageing and Health." 267. Geneva, Switzerland, 2015.

- Pade-Khene, Caroline, Brenda Mallinson, and David Sewry. "Sustainable Rural Ict Project Management Practice for Developing Countries: Investigating the Dwesa and Rumep Projects." *Information Technology for Development* 17, no. 3 (2011): 187-212.
- Parks, Jennifer A. "Lifting the Burden of Women's Care Work: Should Robots Replace the 'Human Touch'?". *Hypatia* 25, no. 1 (Winter2010 2010): 100-120.
- Patel, Kant. Healthcare Politics and Policy in America. London: Routledge, 2019.
- Pereira, Jose, Eduardo Bruera, Karen Macmillan, and Sandra Kavanagh. "Palliative Cancer Patients and Their Families on the Internet: Motivation and Impact." [In eng]. *Journal of Palliative Care* 16, no. 4 (2000): 13-19.
- Physicians, American College of. "Access to Health Care: Executive Summary." [In eng]. Annals of Internal Medicine 112, no. 9 (May 1 1990): 641-661.
- Pierce, Jessica, and George Randels. *Contemporary Bioethics: A Reader with Cases*. New York, NY: Oxford University Press, 2010.
- Pols, Jeannette. "Wonderful Webcams: About Active Gazes and Invisible Technologies." *Science, Technology, & Human Values* 36, no. 4 (2011): 451-473.
- Popovic, Jennifer R. "Distributed Data Networks: A Blueprint for Big Data Sharing and Healthcare Analytics." *Annals of the New York Academy of Sciences* 1387, no. 1 (2017): 105-111.
- Poses, Roy M. "A Cautionary Tale: The Dysfunction of American Health Care." *European Journal* of Internal Medicine 14, no. 2 (2003): 123-130.
- Powers, Madison, and Ruth R. Faden. Social Justice: The Moral Foundations of Public Health and Health Policy. Issues in Biomedical Ethics. Oxford, NY: Oxford University Press, 2006.
- Prevention, Centers for Disease Control and. "Cdc Reports on Uninsured in First Six Months of 2021." American Hospital Association, <u>https://www.aha.org/news/headline/2021-11-17-cdc-reports-uninsured-first-six-months-2021</u>.

—. "Distinguishing Public Health Research and Public Health Nonresearch." edited by Department of Health & Human Services, 2010.

- Quadagno, Jill. "Promoting Civil Rights through the Welfare State: How Medicare Integrated Southern Hospitals." *Social Problems* 47, no. 1 (2000): 68-89.
- Quarterly, Congressional. "Health Insurance: Hearings on New Proposals." 541-544. Washington, DC: Congressional Quarterly Almanac, 1972.
- Quill, Timothy E., and Amy P. Abernethy. "Generalist Plus Specialist Palliative Care Creating a More Sustainable Model." *The New England Journal of Medicine* 368, no. 13 (2013): 1173-1175.

Quinn, Michael J. Ethics for the Information Age. 6th ed.: Pearson Education Inc., 2015.

- Radder, Hans. "Science, Technology, and the Science-Technology Relationship." Chap. 4 In *Handbook of the Philosophy of Science: Philosophy of Technology and Engineering Sciences*, edited by Anthonie Meijers, North Holland: Elsevier, 2009, 65-92.
- Raghupathi, Wullianallur, and Viju Raghupathi. "Big Data Analytics in Healthcare: Promise and Potential." *Health Information Science and Systems* 2, no. 3 (2014): 1-10.
- Ramsey, Paul. "The Indignity of 'Death with Dignity'." *The Hastings Center Studies* 2, no. 2 (1974): 47-62.
- Rebera, Andrew P., Matteo E. Bonfanti, and Silvia Venier. "Societal and Ethical Implications of Anti-Spoofing Technologies in Biometrics." *Science and Engineering Ethics* 20, no. 1 (Mar 2014): 155-169.
- Reich, Warren T., ed. Encyclopedia of Bioethics. Washington, DC: Simon & Schuster, 1995.
- Rendtorff, Jacob D. "The Limitations and Accomplishments of Autonomy as a Basic Principle in Bioethics and Biolaw." Chap. 6 In *Autonomy and Human Rights in Health Care: An International Perspective*, edited by David N. Weisstub and Guillermo Diaz Pintos, Dordrecht: Springer, 2008.
- Robinson, Thomas N. "Community Health Behavior Change through Computer Network Health Promotion: Preliminary Findings from Stanford Health-Net." *Computer Methods and Programs in Biomedicine* 30, no. 2 (1989): 137-144.
- Rodger, Alan. "Data Privacy Laws: Cutting the Red Tape." Ovum, 2015.
- Rodogno, Raffaele. "Social Robots, Fiction, and Sentimentality." *Ethics and Information Technology* 18, no. 4 (2016): 257-268.
- Rogers, Martha F., Mary Glenn Fowler, and Mary Lou Lindegren. "Revised Recommendations for Hiv Screening of Pregnant Women." *Morbidity and Mortality Weekly Report: Recommendations and Reports* 50, no. RR-19 (2001): 59-85.
- Rosenbaum, Sara. "Data Governance and Stewardship: Designing Data Stewardship Entities and Advancing Data Access." *Health Serv Res* 45, no. 5 Pt 2 (2010): 1442-1455.
- Roski, J., G. W. Bo-Linn, and T. A. Andrews. "Creating Value in Health Care through Big Data: Opportunities and Policy Implications." *Health Affairs* 33, no. 7 (Jul 2014): 1115-1122.
- Rothstein, Mark A. "Currents in Contemporary Ethics: Gina, the Ada, and Genetic Discrimination in Employment." *Journal of Law, Medicine & Ethics* 36, no. 4 (Winter 2008): 837-840.
- Rundle, Mary, and Chris Conley. *Ethical Implications of Emerging Technologies: A Survey*. Edited by Boyan Radoykov. Paris: UNESCO, 2007.

- Russo, Federica. "Digital Technologies, Ethical Questions, and the Need of an Informational Framework." *Philosophy & Technology* 31, no. 4 (2018): 655-667.
- Sándor, Judit. Society and Genetic Information: Codes and Laws in the Genetic Era. Cps Books. English ed. Budapest, Hungary: Central European University Press, 2003.
- Sanyal, Chiranjeev, Paul Stolee, Don Juzwishin, and Don Husereau. "Economic Evaluations of Ehealth Technologies: A Systematic Review." *PLoS ONE* 13, no. 6 (2018): 1-11.
- Sassen, Saskia. "Towards a Sociology of Information Technology." *Current Sociology* 50, no. 3 (2002): 365.
- Scharff, Robert C., and Val Dusek. *Philosophy of Technology: The Technological Condition: An Anthology.* 2nd ed.: Wiley-Blackwell, 2014.
- Schillace, Brandy. *Death's Summer Coat: What the History of Death and Dying Teaches Us About Life and Living.* New York, NY: Pegasus Books, 2015.
- Schneiderman, Lawrence J., Nancy S. Jecker, and Albert R. Jonsen. "Medical Futility: Its Meaning and Ethical Implications." *Annals of Internal Medicine* 112, no. 12 (1990): 949-954.
- Scolamiero, Samantha J. "Support Groups in Cyberspace." M.D. Computing 14, no. 1 (1997): 12.
- Sejnowski, Terrence J. The Deep Learning Revolution. Cambridge, Massachusetts: The MIT Press, 2018.
- Services, Center for Medicaid and CHIP. "May 2022 Medicaid and Chip Enrollment Trends Snapshot." edited by Centers for Medicare & Medicaid Services, 2022.
- Shanahan, Murray. *The Technological Singularity*. The Mit Press Essential Knowledge Series. Cambridge, Massachusetts: The MIT Press, 2015.
- Sharkey, Amanda, and Noel Sharkey. "Granny and the Robots: Ethical Issues in Robot Care for the Elderly." *Ethics and Information Technology* 14, no. 1 (2012): 27-40.
- Shi, Xinghua, and Xintao Wu. "An Overview of Human Genetic Privacy." Annals of the New York Academy of Sciences 1387, no. 1 (Jan 2017): 61-72.
- Shoniregun, Charles A., Kudakwashe Dube, and Fredrick Mtenzi. *Electronic Healthcare Information Security*. New York, NY: Springer, 2010.
- Shortliffe, Edward H., and G. Octo Barnett. "Biomedical Data: Their Acquisition, Storage, and Use." Chap. 2 In *Biomedical Informatics: Computer Applications in Health Care and Biomedicine*, edited by Edward H. Shortliffe and James J. Cimino, London: Springer, 2014.
- Shulman, Carl, Nick Tarleton, and Henrik Jonsson. "Which Consequentialism? Machine Ethics and Moral Divergence." In AP-CAP 2009: The Fifth Asia-Pacific Computing and

Philosophy Conference, edited by Carson Reynolds and Alvaro Cassinelli, 23-25. University of Tokyo, Japan, 2009.

- Simanowski, Roberto, Michel Brodmer, and Jefferson Chase. "On the Ethics of Algorithmic Intelligence." *Social Research: An International Quarterly* 86, no. 2 (2019): 423-447.
- Smith, Thomas W. "Aristotle on the Conditions for and Limits of the Common Good." *The American Political Science Review* 93, no. 3 (1999): 625-636.
- Sopcheck, Janet. "Social, Economic, and Political Issues Affecting End-of-Life Care." *Policy, Politics & Nursing Practice* 17, no. 1 (2016): 32-42.
- Sparrow, Robert, and Linda Sparrow. "In the Hands of Machines? The Future of Aged Care." *Minds & Machines* 16, no. 2 (2006): 141-161.
- Statista. "Global Internet Penetration Rate." Statista Research Department, 2022.
- Steinhauser, Karen E., Nicholas A. Christakis, Elizabeth C. Clipp, Maya McNeilly, Lauren McIntyre, and James A. Tulsky. "Factors Considered Important at the End of Life by Patients, Family, Physicians, and Other Care Providers." JAMA: The Journal of the American Medical Association 284, no. 19 (2000): 2476-2482.
- Sternberg, Robert J. "Wisdom : Its Nature, Origins, and Development." edited by Robert J. Sternberg, New York, NY: Cambridge University Press, 2000.
- Stewart, Robert Scott. "Some Ethical Issues in Treating and Caring for People with Dementia." Online Journal of Health Ethics 14, no. 1 (2018): 29-39.
- Stoeklé, Henri-Corto, Marie-France Mamzer-Bruneel, Guillaume Vogt, and Christian Hervé. "23andme: A New Two-Sided Data-Banking Market Model." *BMC Medical Ethics* 17 (2016).
- Strauss, Anselm, Shizuko Fagerhaugh, Barbara Suczek, and Carolyn Wiener. *Social Organization of Medical Work*. New Brunswick, N.J.: Transaction Publishers, 1997.
- Sullins, Donald Paul. "The Healing Community: A Catholic Social Justice Critique of Modern Health Care." *The Linacre Quarterly* 81, no. 2 (2014): 172-181.
- Sutrop, Margit. "Changing Ethical Frameworks: From Individual Rights to the Common Good?". *Cambridge Quarterly of Healthcare Ethics* 20, no. 4 (2011): 533-545.
- Sutrop, Margit, and Katrin Laas-Mikko. "From Identity Verification to Behavior Prediction: Ethical Implications of Second Generation Biometrics." *Review of Policy Research* 29, no. 1 (2012): 21-36.
- Suzman, Richard, John R. Beard, Ties Boerma, and Somnath Chatterji. "Health in an Ageing World—What Do We Know?". *The Lancet* 385, no. 9967 (2015): 484-486.

- Takala, Tuija. "Setting a Dangerous Precedent: Ethical Issues in Human Genetic Database Research." *Medical Law International* 8, no. 2 (2007): 105-138.
- Tauber, Alfred I. *Patient Autonomy and the Ethics of Responsibility*. Basic Bioethics. Cambridge, MA: MIT Press, 2005.
- Teare, Harriet J. A., Joanna Hogg, Jane Kaye, Raashid Luqmani, Elaine Rush, Alison Turner, Laura Watts, Melanie Williams, and M. Kassim Javaid. "The Rudy Study: Using Digital Technologies to Enable a Research Partnership." *European Journal of Human Genetics* 25, no. 7 (2017): 816-822.
- Tene, Omer, and Jules Polonetsky. "Privacy in the Age of Big Data: A Time for Big Decisions." *Stanford Law Review Online* 64 (2011): 63-69.
- ter Meulen, Ruud. "Ethics of Care." In *The Sage Handbook of Health Care Ethics: Core and Emerging Issues*, edited by Henk ten Have Ruth Chadwick, and Eric M. Meslin, London: SAGE Publications Ltd, 2011, 39-49.
- Thacker, Stephen B. "Historical Development." In *Principles and Practice of Public Health Surveillance*, edited by Lisa M. Lee, Steven M. Teutsch and Michael E. St. Louis, New York: Oxford University Press, 2010, 1-17.
- Thomasma, David C. "Evolving Bioethics and International Human Rights." Chap. 1 In *Autonomy* and Human Rights in Health Care: An International Perspective, edited by David N. Weisstub and Guillermo Diaz Pintos, Dordrecht: Springer, 2008.
- Thompson, Mark. "How Have Companies Adapted to the Gdpr?" Information Age, 11 March 2020.
- Timmermans, Stefan, and Marc Berg. "The Practice of Medical Technology." *Sociology of Health & Illness* 25, no. 3 (2003): 97-114.
- Tom, Lawry. *Ai in Health: A Leader's Guide to Winning in the New Age of Intelligent Health Systems.* [in English] Boca Raton, FL: HIMSS Publishing, 2020.
- Topol, Eric J. *The Patient Will See You Now : The Future of Medicine Is in Your Hands.* New York: Basic Books, 2015.
- Torresen, Jim. "A Review of Future and Ethical Perspectives of Robotics and Ai." *Frontiers in Robotics and AI* 4, no. 75 (2018-January-15 2018).
- Tran, Thien A., and Tugrul Daim. "A Taxonomic Review of Methods and Tools Applied in Technology Assessment." *Technological Forecasting & Social Change* 75, no. 9 (2008): 1396-1405.
- Tronto, Joan C. Moral Boundaries: A Political Argument for an Ethic of Care. New York: Routledge, 1993.

- Tsai, Chun-Wei, Ming-Chao Chiang, Adlen Ksentini, and Min Chen. "Metaheuristic Algorithms for Healthcare: Open Issues and Challenges." *Computers & Electrical Engineering* 53 (2016): 421-434.
- Turing, A. M. "Computing Machinery and Intelligence." Mind 59, no. 236 (1950): 433-460.
- UNESCO. "Universal Declaration on the Human Genome and Human Rights." Journal of Medicine & Philosophy 23, no. 3 (1998): 334-341.
- Vacca, John R. *Biometric Technologies and Verification Systems*. Amsterdam; Boston, MA: Butterworth-Heinemann/Elsevier, 2007.
- Vallor, Shannon. "Carebots and Caregivers: Sustaining the Ethical Ideal of Care in the Twenty-First Century." *Philosophy & Technology* 24, no. 3 (2011): 251-268.
- van Gurp, Jelle, Martine van Selm, Kris Vissers, Evert van Leeuwen, and Jeroen Hasselaar. "How Outpatient Palliative Care Teleconsultation Facilitates Empathic Patient-Professional Relationships: A Qualitative Study." *PLoS ONE* 10, no. 4 (2015): 1-13.
- Vandemeulebroucke, Tijs, Bernadette Dierckx de Casterlé, and Chris Gastmans. "The Use of Care Robots in Aged Care: A Systematic Review of Argument-Based Ethics Literature." *Archives of Gerontology & Geriatrics* 74 (2018): 15-25.
- Vanlaere, Linus, and Chris Gastmans. "A Personalist Approach to Care Ethics." *Nursing Ethics* 18, no. 2 (2011): 161-173.
- Vogel, Lynn Harold. "Management of Information in Healthcare Organizations." Chap. 14 In *Biomedical Informatics*, edited by Edward H. Shortliffe and James J. Cimino, Verlag London: Springer, 2014, 443-474.
- Wainess, Flint J. "The Ways and Means of National Health Care Reform, 1974 and Beyond." Journal of Health Politics, Policy, and Law 24, no. 2 (Apr 1999): 305-333.
- Wajcman, Judy. "Addressing Technological Change: The Challenge to Social Theory." *Current Sociology* 50, no. 3 (2002): 347.
- Walker, Margaret Urban. *Moral Understandings: A Feminist Study in Ethics*. New York: Oxford University Press, 2008.
- Warren, Jim. "How Will Ai Change Health Care Delivery?". New Zealand Medical Student Journal, no. 28 (2019): 10-12.
- Wegner, Daniel, and Gray Kurt. *The Mind Club: Who Thinks, What Feels, and Why It Matters.* [in English] New York, New York: Penguin Books, 2016.
- Wells, Sue, Tania Riddell, Andrew Kerr, Romana Pylypchuk, Carol Chelimo, Roger Marshall, Daniel J. Exeter, *et al.* "Cohort Profile: The Predict Cardiovascular Disease Cohort in New

Zealand Primary Care (Predict-Cvd 19)." *International Journal of Epidemiology* 46, no. 1 (2017): 22-22.

- Whitten, Pamela, Gary Doolittle, and Michael Mackert. "Telehospice in Michigan: Use and Patient Acceptance." *American Journal of Hospice & Palliative Medicine* 21, no. 3 (2004): 191-195.
- Widman, Lawrence E., and David A. Tong. "Requests for Medical Advice from Patients and Families to Health Care Providers Who Publish on the World Wide Web." *Archives of Internal Medicine* 157, no. 2 (1997): 209-212.
- Wiener, Norbert. *Cybernetics: Or, Control and Communication in the Animal and the Machine.* The M.I.T. Paperback Series. 2nd ed. Cambridge, MA: M.I.T. Press, 1965.
- Willis, Lia, George Demiris, and Debra P. Oliver. "Internet Use by Hospice Families and Providers: A Review." *Journal of Medical Systems* 31, no. 2 (2007): 97-101.
- Wills, Mary J. "Decisions through Data: Analytics in Healthcare." Journal of Healthcare Management 59, no. 4 (2014): 254-262.
- "The Wingspread Consensus Statement on the Precautionary Principle." Paper presented at the Wingspread Conference, Racine, WI, 1998.
- Woll, Anita, and Tone Bratteteig. "Activity Theory as a Framework to Analyze Technology-Mediated Elderly Care." *Mind, Culture, and Activity* 25, no. 1 (2018): 6-21.
 - ——. "A Trajectory for Technology-Supported Elderly Care Work." *Computer Supported Cooperative Work: The Journal of Collaborative Computing* 28, no. 1/2 (2019): 127-168.
- Wyatt, Sally. "Technological Determinisim Is Dead; Long Live Technological Determinism." In *The Handbook of Science and Technology Studies*, edited by Edward J. Hackett, Olga Amsterdamska, Michael Lynch and Judy Wajcman: The MIT Press, 2007, 165-180.
- Wynsberghe, Aimee. "Designing Robots for Care: Care Centered Value-Sensitive Design." Science & Engineering Ethics 19, no. 2 (2013): 407-433.
- XVI, Pope Benedict. "Message of His Holiness Benedict Xvi." news release, November 15, 2010, <u>https://www.vatican.va/content/benedict-xvi/en/letters/2010/documents/hf_ben-</u> xvi_let_20101115_op-sanitari.html.
- Yampolskiy, Roman V. Artificial Superintelligence: A Futuristic Approach. Boca Raton: Taylor & Francis, CRC Press, 2016.
- Yang, Chaowei, Yan Xu, and Douglas Nebert. "Redefining the Possibility of Digital Earth and Geosciences with Spatial Cloud Computing." *International Journal of Digital Earth* 6, no. 4 (2013): 297-312.

- Yang, Chaowei, Manzhu Yu, Fei Hu, Yongyao Jiang, and Yun Li. "Utilizing Cloud Computing to Address Big Geospatial Data Challenges." *Computers, Environment & Urban Systems* 61 (2017): 120-128.
- Yasnoff, William A. *Health Information Infrastructure*. Biomedical Informatics: Computer Applications in Health Care and Biomedicine. Edited by Edward H. Shortliffe and James J. Cimino. 4th ed.: Springer, 2014.

Zwitter, Andrej. "Big Data Ethics." Big Data & Society 1, no. 2 (2014): 1-6.